

SECURITIES & EXCHANGE COMMISSION EDGAR FILING

National American University Holdings, Inc.

Form: S-1/A

Date Filed: 2007-11-27

Corporate Issuer CIK: 1399855

Symbol: NAUH

SIC Code: 8200

Fiscal Year End: 05/31

© Copyright 2014, Issuer Direct Corporation. All Right Reserved. Distribution of this document is strictly prohibited, subject to the terms of use.

File No. 333-143098

-

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

AMENDMENT NO. 6 TO
FORM S-1
REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

CAMDEN LEARNING CORPORATION

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

(Exact Name of Registrant as Specified in Its Charter) 6770

(Primary Standard Industrial Classification Code Number)

83-0479936

(I.R.S. Employer Identification Number)

500 East Pratt Street Suite 1200 Baltimore, MD 21202 (410) 878-6800

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant's Principal Executive Offices)

David L. Warnock
President and Chief Executive Officer
500 East Pratt Street
Suite 1200
Baltimore, MD 21202
(410) 878-6800

(Name, Address, Including Zip Code, and Telephone Number, Including Area Code, of Agent for Service)

Copies to:

Douglas S. Ellenoff, Esq. Stuart Neuhauser, Esq. Adam Mimeles, Esq. Ellenoff Grossman & Schole LLP 370 Lexington Avenue, 19th Floor New York, New York 10017 (212) 370-1300 Joel L. Rubinstein, Esq. Morgan Fox Walbridge, Esq. McDermott Will & Emery LLP 340 Madison Avenue New York, New York 10173 (212) 547-5400

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this registration statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933 check the following box. ☑

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

TABLE OF CONTENTS

CALCULATION OF REGISTRATION FEE CHART

Title of Each Class of Security to Be Registered	Amount to Be Registered	M C P	roposed laximum Offering trice Per Unit ⁽¹⁾	Proposed Maximum Aggregate Offering Price ⁽¹⁾		Maximum I Aggregate			Amount of legistration Fee
Units, each consisting of one share of Common Stock, \$.0001 par value, and one Warrant ⁽²⁾	7,187,500	\$	8.00	\$	57,500,000	\$1	,765.25		
Shares of Common Stock included as part of the Units ⁽²⁾	7,187,500		_		_		(3)		
Warrants included as part of the Units ⁽²⁾	7,187,500		_		_		(3)		
Shares of Common Stock underlying the Warrants included in the Units ⁽²⁾⁽⁴⁾	7,187,500	\$	5.50	\$	39,531,250	\$1	,213.60		
Representative's Unit Purchase Option	1	\$	100	\$	100	\$	0		
Units underlying the Representative's Unit Purchase Option ("Representative's Units") ⁽⁴⁾	625,000	\$	9.60	\$	6,000,000	\$	184.20		
Shares of Common Stock included as part of the Representative's Units ⁽⁴⁾	625,000		_		_		(3)		
Warrants included as part of the Representative's Units ⁽⁴⁾	625,000		_		_		(3)		
Shares of Common Stock underlying the Warrants included in the Representative's Units ⁽⁴⁾	625,000	\$	6.71	\$	4,193,750	\$	128.75		
Total				\$	105,968,850	\$3	3,291.80 ⁽⁵⁾		

⁽¹⁾ Estimated solely for the purpose of calculating the registration fee.

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the registration statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

TABLE OF CONTENTS

The information in this prospectus is not complete and may be changed. We may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

⁽²⁾ Includes 937,500 Units, 937,500 shares of Common Stock, 937,500 Warrants underlying such Units and 937,500 shares of Common Stock underlying the Warrants included in such Units, which may be issued upon exercise of a 45-day option granted to the Underwriters to cover over-allotments, if any.

⁽³⁾ No fee pursuant to Rule 457(g).

⁽⁴⁾ Pursuant to Rule 416, there are also being registered such indeterminable additional securities as may be issued to prevent dilution as a result of stock splits, stock dividends or similar transactions.

^{(5) \$2,428.68} previously paid.

\$50,000,000

CAMDEN LEARNING CORPORATION

6,250,000 units

Camden Learning Corporation is a blank check company recently incorporated for the purpose of merging with, engaging in a capital stock exchange with, purchasing all or substantially all of the assets of, or engaging in any other similar business combination with one or more operating businesses in the education industry focusing on early childcare, K-12 or post-secondary education or corporate training and related businesses. We do not have any specific business combination under consideration or contemplation and we have not, nor has anyone on our behalf, contacted any potential target business or had any discussions, formal or otherwise, with respect to such a transaction.

This is an initial public offering of our securities. Each unit is being sold at a purchase price of \$8.00 per unit and consists of:

- · one share of our common stock; and
- one warrant.

Each warrant entitles the holder to purchase one share of our common stock at a price of \$5.50. Each warrant will become exercisable on the later of our completion of a business combination or ______, 2008 [one year from the date of this prospectus], and will expire on ______, 2011 [four years from the date of this prospectus], or earlier upon redemption.

Our sponsor, Camden Learning, LLC, a limited liability company indirectly controlled and partially owned by certain of our officers and directors, owns 1,458,334 shares of our common stock, and will purchase an aggregate of 2,800,000 warrants, or insider warrants, from us at a price of \$1.00 per warrant in a private placement to be completed immediately prior to this offering. All of the proceeds received from the sale of the insider warrants (an aggregate of \$2,800,000) will be placed in the trust account described below. The insider warrants will be identical to those sold in this offering but (i) will not be subject to redemption, (ii) may be exercised on a "cashless" basis, in each case if held by our sponsor or its permitted assigns and (iii) may not be sold, assigned or transferred prior to the 90th day following consummation of a business combination. The holders of insider warrants will not have any right to any liquidation distributions with respect to the shares underlying such insider warrants in the event we fail to consummate a business combination, in which event the insider warrants will expire worthless.

We have granted Morgan Joseph & Co. Inc., the representative of the underwriters, a 45-day option to purchase up to 937,500 additional units (over and above the 6,250,000 units referred to above) solely to cover over-allotments, if any. We have also agreed to sell to Morgan Joseph & Co. Inc. for \$100, as additional compensation, an option to purchase up to 625,000 units at a per unit price of \$9.60. The units issuable upon exercise of this option are identical to those offered by this prospectus except that the exercise price for the warrants included in these units is \$6.71 per share. The purchase option and its underlying securities have been registered under the registration statement of which this prospectus forms a part.

There is presently no public market for our units, common stock or warrants. We anticipate that our units will be quoted on the OTC Bulletin Board under the symbol [""] on or promptly after the date of this prospectus. Each of the common stock and warrants shall trade separately on the 90th day after the date of this prospectus, unless Morgan Joseph & Co. Inc. determines an earlier date is acceptable. Once the securities comprising the units begin separate trading, we expect that the common stock and warrants will be quoted on the OTC Bulletin Board under the symbols [""] and ["""], respectively. We cannot assure you, however, that our securities will continue to be quoted on the OTC Bulletin Board in the future.

Investing in our securities involves a high degree of risk. See "Risk Factors" beginning on page 7 of this prospectus for a discussion of information that should be considered in connection with an investment in our securities.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

	Public Offering Price				Proceeds, Before Expenses, to Us	
Per unit	\$	8.00	\$	0.56	\$	7.44
Total	\$50,00	0,000	\$	3,500,000	\$	46,500,000

⁽¹⁾ Includes deferred underwriting discount and commissions in the amount of \$1,500,000, or \$0.24 per unit, payable to the underwriters only upon consummation of a business combination and then only with respect to those units as to which the component shares have not been redeemed for cash by those stockholders who voted against the business combination and exercised their redemption rights.

Of the proceeds we receive from this offering and the private placement to be made prior to the effective date of this offering to our sponsor, \$49,500,000 (\$7.92 per unit) will be deposited into a trust account at J.P. Morgan Chase Bank, N.A. maintained by Continental Stock Transfer & Trust Company acting as trustee. This amount includes deferred underwriting discount and commissions in the amount of \$1,500,000, or \$0.24 per unit, payable to the underwriters only upon consummation of a business combination and then only with respect to those units as to which the component

shares have not been redeemed for cash by those stockholders who voted against the business combination and exercised their redemption rights.

We are offering the units for sale on a firm-commitment basis. Morgan Joseph & Co. Inc., acting as representative of the underwriters, expects to deliver our securities to investors in the offering on or about 1, 2007.

MORGAN JOSEPH

Ferris, Baker Watts
Incorporated

Legend Merchant Group

The date of this prospectus is ______, 2007

TABLE OF CONTENTS

PROSPECTUS SUMMARY

This summary highlights certain information appearing elsewhere in this prospectus. For a more complete understanding of this offering, you should read the entire prospectus carefully, including the risk factors and the financial statements.

Unless otherwise stated in this prospectus:

- references to "we," "us" or "our company" are to Camden Learning Corporation;
- references to the "Camden III Funds" are to Camden Partners Strategic Fund III, L.P. and Camden Partners Strategic Fund III-A, L.P., collectively;
- references to a "business combination" are to a merger, capital stock exchange, asset acquisition or other similar business combination between us and one or more operating businesses in the education industry;
- references to "existing stockholders" are to all of our stockholders before this offering;
- references to "private placement" are to the sale of 2,800,000 warrants to our sponsor at a price of \$1.00 per warrant, for an aggregate purchase price of \$2,800,000, in a private placement that will occur immediately prior to the closing of this offering;
- references to "public stockholders" are to the holders of common stock sold as part of the units in this offering or acquired in the aftermarket, including any existing stockholders to the extent they acquire such shares (and solely with respect to such shares);
- references to our "sponsor" are to Camden Learning, LLC, a limited liability company owned by Camden Partners Strategic Fund III, L.P. (96.01%) and Camden Partners Strategic Fund III-A, L.P. (3.99%), each of which are indirectly controlled and partially owned by David L. Warnock, our Chairman, President and Chief Executive Officer, and Donald W. Hughes, our Chief Financial Officer and Secretary. The general partner of each limited partnership is Camden Partners Strategic III, LLC. Messrs. Warnock and Hughes are two of the four managing members of Camden Partners Strategic III, LLC;
- references to a "target business" are to one or more operating businesses which, after completion of this offering, we may target for a potential business combination; and
- the information in this prospectus assumes that the representative of the underwriters will not exercise its overallotment option and that no stockholder exercises its right of redemption as described elsewhere in this prospectus.
- unless expressly stated to the contrary, the information in this prospectus gives retroactive effect to a forward stock split in the form of a dividend of 0.3888888 shares of common stock for each outstanding share of common stock declared and paid on November 20, 2007.

The Company

We are a blank check company organized under the laws of the State of Delaware on April 10, 2007. We were formed for the purpose of merging with, engaging in a capital stock exchange with, purchasing all or substantially all of the assets of, or engaging in any other similar business combination with one or more operating businesses in the education industry focusing on early childcare, K-12 or post-secondary education or corporate training and related businesses. As used throughout this prospectus, the term "education industry" refers broadly to the operation or management of learning facilities, the provision of educational instruction or training or the provision of education-related services. To date, our efforts have been limited to organizational activities and activities relating to this offering. We do not have any specific business combination under consideration or contemplation and we have not, nor has anyone on our behalf, contacted any potential target business or had any discussions, formal or otherwise, with respect to such a transaction.

Education Industry

The U.S. education industry has continued to show substantial growth in the past decade, due to what we believe to be the importance of developing a skilled workforce. A skilled workforce is increasingly reliant on

TABLE OF CONTENTS

intellectual capital as the U.S. economy continues its shift to become focused on services rather than manufacturing. While post-secondary graduates constitute approximately 30% of the U.S. population, more than 85% of the U.S. population has completed its K-12 (kindergarten through twelfth grade) education according to the National Center for Education Statistics report entitled "Digest of Education Statistics: 2005". International competition, especially in math and science, has driven education legislation, requiring minimum performance levels and allocating funding for supplemental services in underperforming schools. In addition to state and government spending, the U.S. has the second highest level of education funding from private sources in the world at 28%, led only by Korea, according to the Organisation for Economic Co-Operation and Development's report entitled "Education at a Glance 2006". These factors have contributed to the overall increase in education spending with total expenditures for education expected to amount to 7.5 percent of the gross domestic product in 2003-04, which is approximately 0.6 percentage points higher than in 1993-94 according to the National Center for Education Statistics report entitled "Digest of Education Statistics: 2005". Post-secondary education is broadly defined as a formal instructional program whose curriculum is designed for students who have completed the requirements for a high school diploma or its equivalent. This includes programs whose purpose is academic, vocational and continuing professional education and excludes adult basic education programs.

Our Management

Our management and board of directors have established an extensive network of relationships from which to identify and generate acquisition opportunities within the education industry. David L. Warnock, our President. Chairman and Chief Executive Officer, has over 24 years of investment experience in the education and business and financial services industries. Mr. Warnock serves on the boards of directors of American Public Education, Inc., New Horizons Worldwide, Inc., Nobel Learning Communities, Inc., Primo Water Corporation and Questar Assessment, Inc., formerly Touchstone Applied Science Associates. Mr. Hughes also serves on the boards of directors of New Horizons Worldwide, Inc. and Questar Assessment, Inc. Dr. Therese Crane, our director, was previously President of Jostens Learning Corporation and its successor, Compass Learning and previously was Vice President of Information and Education Products at America Online. Ronald Tomalis, our director, was previously counselor to the US Secretary of Education and Acting Assistant Secretary of Elementary and Secondary Education. William Jews, our director, is a former governor of the Federal Reserve Bank and was the President and Chief Executive Officer of CareFirst Inc./CareFirst Blue Cross Blue Shield from 1993 through 2006, an organization with more than \$5 billion in annual revenues. Mr. Jews has previously been a director of MBNA, MuniMae Inc., Nations Bank, Ecolab, Inc. and Crown Central Petroleum, and currently serves on the boards of directors of The Ryland Group, a national home builder and mortgage provider, Choice Hotels International, a worldwide lodging franchisor and Fortress International Group, Inc., the parent company of Total Site Solutions, which supplies industry and government with secure data centers and other facilities designed to survive terrorist attacks, natural disasters and blackouts. In addition, we believe the experience of our officers and directors in private equity and investment banking investments will be beneficial in structuring and consummating a business combination.

To date, none of our officers or directors has approached their contacts to identify potential target businesses, and no such contacts have presented or identified potential target businesses to any of our officers or directors. We expect, from time to time, after the offering is completed, that these contacts or sources will advise either our management team or directors of the existence of one or more potential acquisition candidates or that potential acquisition candidates will become known to our management team or directors through their other business activities. Our management will evaluate these leads and determine whether to pursue discussions with any of these candidates.

Business Combination

We have until ______, 2009 [24 months from the date of this prospectus] to consummate a business combination. If we are unable to consummate a business combination by such date, our corporate existence will cease, except for the purpose of winding up our affairs and liquidating. We will not pursue a business combination with any company that is a portfolio company of, or otherwise affiliated with, or has received financial investment from, any of the private equity firms with which our existing stockholders, executive officers or directors are affiliated. Our initial business combination must be with a target business or businesses whose aggregate fair market value is at least equal to 80% of the amount in our trust account (less the

2

TABLE OF CONTENTS

deferred underwriting discount and commissions and taxes payable) at the time of such transaction. Consequently, it is likely we will have the ability to effect only a single business combination, although this may entail the simultaneous acquisitions of several assets or closely related operating businesses at the same time. Should we elect to pursue more than one acquisition of target businesses simultaneously, we could encounter difficulties in consummating all or a portion of such acquisitions due to a lack of adequate resources, including the inability of management to devote sufficient time to the due diligence, negotiation and documentation of each acquisition. Even if we complete the acquisition of more than one target business at the same time, there can be no assurance we will be able to integrate

the operations of such target businesses. In no instance will we acquire less than majority voting control of a target business. However, in the case of a reverse merger or other similar transaction in which we issue a substantial number of new shares, our stockholders immediately prior to such transaction may own less than a majority of our shares subsequent to such transaction. Even if we acquire less than 100% of one or more target businesses in our initial business combination, the aggregate fair market value of the interests we acquire must equal at least 80% of the amount held in the trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction(s), with such interests being evaluated based upon generally accepted financial standards.

In seeking a business combination, we intend to utilize cash derived from the proceeds of this offering and the private placement, as well as our capital stock, debt, or a combination of cash, capital stock and debt, and there is no limit on the issuance of capital stock or incurrence of debt we may undertake in effecting a business combination. This may allow us to acquire a target business or businesses with an aggregate fair market value in excess of 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of the transaction. If we were to seek such additional funds, any such arrangement would only be consummated simultaneously with our consummation of a business combination. As of the date of this prospectus, we have not engaged or retained, had any discussions with, or entered into any agreements with, any third party regarding any such potential financing transactions. In the event a business combination is consummated, all sums remaining in the trust account will be released to us immediately thereafter, and there will be no restriction on our use of such funds.

Our officers and directors will not receive any compensation in this offering or for services rendered to us prior to, or in connection with, the consummation of a business combination. Our officers and directors will be entitled to reimbursement for out-of-pocket expenses incurred by them or their affiliates on our behalf.

We maintain executive offices at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 and our telephone number is (410) 878-6800.

Private Placement

Prior to the closing of this offering, our sponsor will purchase an aggregate of 2,800,000 warrants, which we refer to as the insider warrants, from us at a price of \$1.00 per warrant in a private placement pursuant to Regulation D of the Securities Act of 1933, as amended. The insider warrants will be identical to those sold in this offering but (i) will not be subject to redemption, (ii) may be exercised on a "cashless" basis, in each case if held by our sponsor or its permitted assigns and (iii) may not be sold, assigned or transferred prior to the 90th day following consummation of a business combination. The holders of insider warrants will not have any right to any liquidation distributions with respect to the shares underlying such insider warrants in the event we fail to consummate a business combination, in which event the insider warrants will expire worthless. No commissions, fees or other compensation will be payable in connection with such private placement.

All of the gross proceeds from the sale of the 2,800,000 warrants in the private placement, or \$2,800,000, will be deposited into the trust account. Until the 90th day following consummation of a business combination, the insider warrants may only be transferred in certain limited circumstances, and the transferees receiving such insider warrants will be subject to the same sale restrictions imposed on the initial purchaser and its member transferees. If the holders of the insider warrants acquire warrants for their own account in the open market, any such warrants will be redeemable. If our other outstanding warrants are redeemed (including the warrants subject to the underwriters' unit purchase option) and the market price of a share of our common stock rises following such redemption, holders of the insider warrants could potentially realize a larger gain on exercise or sale of those warrants than is available to other warrant holders, although we do not know if the price of our common stock would increase following a warrant redemption. If our share price declines in

3

TABLE OF CONTENTS

periods subsequent to a warrant redemption and the insiders continue to hold the insider warrants, the value of those warrants still held by such persons may also decline. The insider warrants will be differentiated from warrants, if any, purchased in or following this offering by any holder of insider warrants through the legends contained on the certificates representing the insider warrants indicating the restrictions and rights specifically applicable to such warrants as are described in this prospectus.

Additional Purchases by Our Sponsor

Our sponsor has informed us it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Our sponsor has agreed to vote any such shares of common stock in favor of our initial business combination, representing a possible maximum aggregate of 3.4% of the public shares entitled to vote on the business combination. Our sponsor has also entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Such open market purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price per share of not more than the per share amount held in the trust account (less taxes payable) as reported in such 8-K and will be made by a broker-dealer mutually agreed upon by our sponsor and the representative of the underwriters in such amounts and at such times as such broker-dealer may determine, in its sole discretion, so long as the purchase price does not exceed the above-referenced per share purchase price. Our sponsor has agreed to vote any such shares of common stock purchased in

the open market in favor of our initial business combination, representing a possible maximum aggregate of 8% of the public shares entitled to vote on the business combination. Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell such shares, provided it will be entitled to participate in any liquidating distributions with respect to the shares purchased in the open market. In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our sponsor has agreed to purchase from us in a private placement, which we may refer to as the co-investment, a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this co-investment. This co-investment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders.

Our sponsor, whose sole owners are the Camden III Funds, has agreed to such purchases because the managing members of the general partner of the Camden III Funds, including David L. Warnock, our Chairman, President and Chief Executive Officer and Donald W. Hughes, our Chief Financial Officer and Secretary, want the Camden III Funds to have a substantial cash investment in us, including any target business we may acquire.

4

TABLE OF CONTENTS

THE OFFERING

Securities offered:

6,250,000 units, at \$8.00 per unit, each unit consisting of:

- one share of common stock; and
- one warrant

The units will begin trading on or promptly after the date of this prospectus. Each of the common stock and warrants shall trade separately on the 90th day after the date of this prospectus unless Morgan Joseph & Co. Inc. determines that an earlier date is acceptable, based on its assessment of the relative strengths of the securities markets and small capitalization companies in general, and the trading pattern of, and demand for, our securities in particular. However, Morgan Joseph & Co. Inc. may decide to allow continued trading of the units following such separation, in which case holders of units will be required to have their brokers contact our transfer agent in order to separate the units into common stock and warrants. In no event will Morgan Joseph & Co. Inc. allow separate trading of the common stock and warrants until (i) we file an audited balance sheet reflecting our receipt of the gross proceeds of this offering and the private placement, including any proceeds we receive from the exercise of the over-allotment option, if such option is exercised on the date of this prospectus, (ii) we file a Current Report on Form 8-K and issue a press release announcing when such separate trading will begin and (iii) the expiration of the underwriters over-allotment option or its exercise in full. We will file a Current Report on Form 8-K, including an audited balance sheet, upon the consummation of this offering, which is anticipated to take place three business days from the date of this prospectus. The audited balance sheet will include proceeds we receive from the exercise of the over-allotment option if the over-allotment option is exercised on the date of this prospectus. If the over-allotment option is exercised following the date of this prospectus, an additional Current Report on Form 8-K will be filed to disclose the exercise and closing of the over-allotment option.

If you are not an institutional investor, you may purchase securities in this offering only if you reside within the states in which we have applied to have the securities registered. We have registered the securities in: Colorado, Delaware, the District of Columbia, Florida, Georgia, Hawaii, Illinois, Louisiana, Maryland, New York and Rhode Island.

Additional Purchases by our Sponsor:

Our sponsor will purchase an aggregate of 2,800,000 warrants, or insider warrants, from us at a price of \$1.00 per warrant in a private placement to be completed immediately prior to this offering, All of the proceeds received from the sale of the insider warrants (an aggregate of \$2,800,000) will be placed in the trust account described below. The sale of the warrants is not expected to result in compensation expense because they are being sold at or above fair market value. The insider warrants will be identical to those sold in this offering but (i) will not be subject to redemption, (ii) may be exercised on a "cashless" basis, in each case if held by our sponsor or its permitted assigns and (iii) may not be sold.

5

TABLE OF CONTENTS

assigned or transferred prior to the 90th day following the consummation of a business combination. The holders of the insider warrants will not have any rights to any liquidation distributions with respect to the shares underlying such insider warrants in the event we fail to consummate a business combination, in which event the insider warrants will expire worthless.

Our sponsor has informed us it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Our sponsor has agreed to vote any such shares of common stock in favor of our initial business combination, representing a possible maximum aggregate of 3.4% of the public shares entitled to vote on the business combination. In addition, our sponsor has entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Such open market purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price per share of not more than the per share amount held in the trust account (less taxes payable) as reported in such 8-K and will be made by a brokerdealer mutually agreed upon by our sponsor and the representative of the underwriters in such amounts and at such times as such brokerdealer may determine, in its sole discretion, so long as the purchase price does not exceed the above-referenced per share purchase price. Our sponsor has agreed to vote any such shares of common stock purchased in the open market in favor of our initial business combination, representing a possible maximum aggregate of 8% of the public shares entitled to vote on the business combination. Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell such shares, provided it will be entitled to participate in any liquidating distributions with respect to the shares purchased in the open market. In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our sponsor has agreed to purchase from us in a private placement a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this coinvestment. This co-investment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders. The private placement units may not be sold, assigned or transferred unless and until our stockholders approve a business

No commissions, fees or other compensation will be payable in connection with the private placement or the co-investment.

6

combination.

TABLE OF CONTENTS

Common stock:

Number outstanding before this offering:

offering: 1,562,500 shares

Number to be outstanding after this offering:

Onom

7,812,500 shares

Warrants:

Number outstanding before this offering and private placement:

Number to be outstanding after this offering and private placement:

None

9.050.000 warrants

Exercisability:	Each warrant is exercisable for one share of common stock.
Exercise price:	\$5.50 per share
Exercise period:	The warrants sold in the offering will become exercisable on the later of:
	 the completion of a business combination, or
	• , 2008 [one year from the date of this

prospectus]

The warrants held by public stockholders will only be exercisable if a registration statement covering the common stock issuable upon exercise of the warrants is effective and current. The insider warrants will be identical to those sold in this offering but (i) will not be subject to redemption, (ii) may be exercised on a "cashless" basis, in each case if held by our sponsor or its permitted assigns and (iii) may not be sold, assigned or transferred prior to the 90th day following consummation of a business combination. The warrants included in the units sold in this offering, as well as the insider warrants, will expire at 5:00 p.m., New York City time, on _______, 2011 [four years from the date of this prospectus] or earlier upon redemption.

We may redeem the outstanding warrants included in the units sold in this offering and the warrants issued upon exercise of the representative's unit purchase option:

- in whole and not in part,
- at a price of \$0.01 per warrant at any time while the warrants are exercisable,
- upon a minimum of 30 days' prior written notice of redemption, and
- if, and only if, the last closing sales price of our common stock equals or exceeds \$11.50 per share for any 20 trading days within a 30 trading day period ending three business days before we send the notice of redemption.

We have established this last criterion to provide warrant holders with a premium to the initial warrant exercise price as well as a

7

TABLE OF CONTENTS

degree of liquidity to cushion the market reaction, if any, to our redemption call. If the foregoing conditions are satisfied and we call the warrants for redemption, each warrant holder shall then be entitled to exercise his or her warrant prior to the date scheduled for redemption. However, there can be no assurance that the price of our common stock will exceed the call trigger price (\$11.50) or the warrant exercise price after the redemption call is made.

None of the warrants issued in the private placement are redeemable while held by the initial purchasers or their permitted assigns.

There will be no fees, reimbursements or cash payments made to our existing stockholders and/or officers and directors other than:

- repayment of the \$200,000 promissory note bearing interest at a rate of 4.9% per annum made by our sponsor, to fund organizational and offering expenses, prior to the closing date of this offering;
- reimbursement of any expenses incident to finding a suitable business combination; and

Redemption:

Limited Payments to Insiders:

 payment to Camden Partners Holdings, LLC of \$7,500 per month for certain general and administrative services, including but not limited to receptionist, secretarial and general office services.

Certificate of Incorporation:

As discussed below, there are specific provisions in our amended and restated certificate of incorporation that may not be amended prior to our consummation of a business combination without the prior consent of holders of 95% of the shares purchased in this offering, including our requirements to seek stockholder approval of such a business combination and to allow our stockholders to seek redemption of their shares if they do not approve of such a business combination. While we have been advised that such provisions limiting our ability to amend our certificate of incorporation may not be enforceable under Delaware law, we view these provisions, which are contained in Article Third and Sixth of our amended and restated certificate of incorporation, as obligations to our stockholders, and our officers and directors have agreed they will not propose, or vote in favor of, any amendment to these provisions.

Our amended and restated certificate of incorporation also provides we will continue in existence only until _______, 2009 [twenty four months from the date of this prospectus]. If we have not completed a business combination by such date, our corporate existence will cease except for the purposes of winding up our affairs and liquidating, pursuant to Section 278 of the Delaware General Corporation Law. This has the same effect as if our board of directors and stockholders had formally voted to approve our dissolution pursuant to Section 275 of the Delaware General Corporation Law. Accordingly, limiting our corporate existence to a specified date as permitted by Section 102(b)(5) of the Delaware General Corporation Law removes the necessity to comply with the formal procedures set forth in Section 275 (which would have required our board of directors and stockholders to formally vote to

8

TABLE OF CONTENTS

approve our dissolution and liquidation and to have filed a certificate of dissolution with the Delaware Secretary of State). In connection with any proposed business combination we submit to our stockholders for approval, we will also submit to our stockholders a proposal to amend our amended and restated certificate of incorporation to provide for our perpetual existence, thereby removing this limitation on our corporate life. We will only consummate a business combination if stockholders vote both in favor of such business combination and our amendment to provide for our perpetual existence. The approval of the proposal to amend our amended and restated certificate of incorporation to provide for our perpetual existence would require the affirmative vote of a majority of our outstanding shares of common stock. We view this _, 2009 provision terminating our corporate life by [twenty four months from the date of this prospectus] as an obligation to our stockholders and will not take any action to amend or waive this provision to allow us to survive for a longer period of time except in connection with the consummation of a business combination.

Proposed OTC Bulletin Board symbols for our:

Units: "[]"

Common Stock: "[]"

Warrants: "[]"

Offering and private placement proceeds to be held in trust:

\$49,500,000 of the proceeds from this offering and the private placement will be placed in a trust account at J.P. Morgan Chase Bank, N.A. maintained by Continental Stock Transfer & Trust Company, pursuant to an investment management trust agreement to be signed on the date of this prospectus. Of this amount, up to \$48,000,000 may be used by us for the purpose of effecting a business combination and up to \$1,500,000 will be paid to Morgan Joseph & Co. Inc. if a business combination is consummated (less \$0.24 for each share redeemed for cash in connection with our business combination), but will be forfeited by Morgan Joseph & Co. Inc. if a business combination is not consummated. These funds will not be released until the earlier of the completion of a business combination or our liquidation; provided, however, there can be released to us from the interest income, after taxes, accrued on the trust account prior to, or upon the consummation of, a business combination or our liquidation, amounts for payments of taxes on interest earned and up to \$600,000 to fund our expenses relating to investigating and selecting a target business, other working capital requirements and expenses incurred in connection with our dissolution if we fail to consummate a business combination. Other than as described above, the funds held in the trust account will not be available for our use for any expenses related to this offering or expenses which we may incur related to the investigation and selection of a target business and the negotiation of an agreement to acquire a target business, unless and until a business combination is consummated. The \$1,500,000 of the funds

9

TABLE OF CONTENTS

attributable to Morgan Joseph & Co. Inc.'s deferred underwriting discount and commissions in connection with this offering will be released to Morgan Joseph & Co. Inc., less \$0.24 per share for any public stockholders exercising their redemption rights, upon completion of a business combination on the terms described in this prospectus, or to our public stockholders upon liquidation of the trust account as part of our plan of dissolution and liquidation, but will in no event be available for use by us in a business combination. Expenses we may incur prior to consummation of a business combination may only be paid from the net proceeds of this offering and the private placement not held in the trust account, and any interest earned and released to us as provided above.

In the event a business combination is consummated, all sums remaining in the trust account will be released to us and there will be no restriction on our use of such funds, which shall be available for working capital to pay officer and director salaries, make change of control payments, pay fees to affiliates or for any other uses as we may determine.

Stockholders must approve business combination:

We will seek stockholder approval before we effect any business combination, even if the nature of the acquisition would not ordinarily require stockholder approval under applicable state law. In connection with the vote required for any business combination, all of our existing stockholders, including all of our officers and directors, have agreed to vote the shares of common stock owned by them immediately before this offering in accordance with the majority of the shares of common stock voted by the public stockholders. Any shares acquired in this offering or in the aftermarket by existing stockholders and their designees will be voted in favor of the business combination.

Accordingly, our existing stockholders will not be able to exercise redemption rights with respect to a potential business combination.

Assuming the 212,500 units are purchased in the offering and the \$4,000,000 of common stock are purchased by our sponsor in the open market, our existing stockholders, collectively, will beneficially own 29.18% of the then issued and outstanding shares of common stock, with only 9.18% of such shares automatically voting in favor of the business combination and the remaining 20% voting in accordance with the majority of the shares voted by the public stockholders. None of our other existing stockholders, officers and directors has indicated to us they intend to purchase units in this offering, or units or warrants on the open market following the offering. For a more complete discussion, please see the section of this prospectus entitled "Principal Stockholders."

Except as expressly set forth herein, we are not aware of any present intention on the part of our sponsor, officers or directors to make any purchases in this offering or in the aftermarket. Although we do not know for certain the factors that would cause our sponsor, officers or directors to purchase our securities in addition to those purchases expressly set forth herein, we believe some of the factors they would consider are: (i) the trading price of our securities, (ii) their aggregate investment in our securities, (iii) whether it

10

TABLE OF CONTENTS

appears that a substantial number of public stockholders are voting against a proposed business combination, and (iv) their interest in the target business once the target business has been identified.

Although our sponsor and directors have no current intentions to make such purchases, the sponsor and directors are not prohibited from purchasing additional shares of our common stock in the open market and/or in privately negotiated transactions. Any such privately negotiated transaction with a stockholder would include a contractual acknowledgement that such stockholder, although still a holder of record of our common stock, is no longer the beneficial owner thereof and therefore agrees to vote such shares of common stock as directed by the sponsor, directors or officers, as the case may be. In the event the sponsor or directors purchase shares in privately negotiated transactions from stockholders who have already submitted proxies to vote against the proposed combination and requested redemption of their shares, such selling stockholders would be required to revoke their prior proxies to vote against the proposed acquisition and to revoke their prior elections to redeem their shares and to cast new votes in favor of the proposed acquisition. The revocation of prior proxies to vote against the business combination and substitution therefor of votes in favor of the proposed acquisition would have the effect of reducing redemptions and increasing votes in favor of the proposed acquisition, thereby making it more likely that a proposed business combination would be approved. Our officers are prohibited by the terms of their agreements with the Camden III Funds from investing in portfolio companies of the Camden III Funds.

We will proceed with a business combination only if a majority of the shares of our common stock cast at the meeting are voted in favor of the business combination and public stockholders owning up to one share less than 30% of the shares sold in this offering vote against the business combination and exercise their redemption rights described below. Our threshold for redemption rights has been established at 30% in order for this offering to be competitive with other offerings by blank check companies currently in the market. However, a 20% threshold is more typical in offerings of this type. We have selected the higher threshold to reduce the risk of a small group of stockholders exercising undue influence on the stockholder approval process. We will not propose a business combination that is conditioned on one share less than 30% of the public stockholders exercising their redemption rights. However, the 30% threshold entails certain risks described under the heading "Risk Factors — The fact we will proceed with the business combination if public stockholders holding one share less than 30% of shares sold in this offering exercise their redemption rights, rather than the 20% threshold of most other blank check companies, may hinder our ability to consummate a business combination in the most efficient manner to optimize our capital structure."

Redemption rights for stockholders voting to reject a business combination:

Public stockholders voting against a business combination and electing redemption of their shares will be entitled to redeem their

common stock for a pro rata share of the trust account including \$0.24 per share being held in the trust account attributable to the deferred underwriting discount and commissions and any interest earned on the portion of the trust account, excluding up to \$600,000 of interest income (after taxes) previously released to us to fund our working capital requirements and amounts released to us for payment of taxes on interest earned. Public stockholders that redeem their stock for their pro rata share of the trust account will continue to have the right to exercise any warrants they may hold. Stockholders may be requested to tender their shares of common stock before a business combination is consummated. Redeeming stockholders will be sent instructions on how to tender their shares of common stock and when they should expect to receive the redemption amount. In order to ensure accuracy in determining whether or not the redemption threshold has been met, each redeeming stockholder must continue to hold their shares of common stock until the consummation of the business combination. We will not charge redeeming stockholders any fees in connection with the tender of shares for redemption. Because our existing stockholders have agreed to vote any shares acquired by them in this offering or the aftermarket in favor of a business combination negotiated by our executive officers, they are not entitled to redemption rights with respect to any such shares if the business combination is approved and completed.

Liquidation if no business combination:

As described above, if we have not consummated a business combination by , 2009 [twenty four months from the date of this prospectus], our corporate existence will cease by operation of law and we will promptly distribute only to our public stockholders the amount in our trust account (including any accrued interest, after taxes payable on such interest) plus any remaining net assets. At that time, pursuant to Section 281 of the Delaware General Corporation Law, we will adopt a plan that will provide for our payment, based on facts known to us at such time, of (i) all existing claims, (ii) all pending claims and (iii) all claims that may be potentially brought against us within the subsequent 10 years. Accordingly, we would be required to provide for any creditors known to us at that time as well as provide for any claims that we believe could potentially be brought against us within the subsequent 10 years prior to distributing the funds held in the trust to our public stockholders. We cannot assure you that we will properly assess all claims that may be potentially brought against us or that distributions from the trust account, if we liquidate, will not be reduced by such claims. As such, our stockholders could potentially be liable for any claims of creditors to the extent of distributions received by them (but no more). Furthermore, we will use our reasonable best efforts to have all vendors and service providers (which would include any third parties we engaged to assist us in any way in connection with our search for a target business) and prospective target businesses execute agreements with us waiving any right, title, interest or claim of any kind they may have in or to any monies held in the trust account. We may elect to forego obtaining waivers only if we receive the approval of our Chief

Executive Officer and the approving vote or written consent of at least a majority of our board of directors. However, there is no guarantee these third parties will execute such waivers, or even if they execute such waivers, that they will not seek recourse against the trust account or that a court would not conclude that such waivers are not legally enforceable. In order to protect the amounts held in the trust account. our sponsor has agreed to indemnify us for claims of creditors, vendors, service providers and target businesses that have not executed a valid and binding waiver of their right to seek payment of amounts due to them out of the trust account. As further assurance our sponsor will have the necessary funds required to meet these indemnification obligations, (i) the sole owners and members of our sponsor, Camden Partners Strategic Fund III, L.P. (96.01% ownership of the sponsor) and Camden Partners Strategic Fund III-A, L.P. (3.99%), have agreed, under our sponsor's limited liability company agreement, to make capital contributions to our sponsor as and when required in order for the sponsor to fulfill its indemnification obligations and (ii) our sponsor has agreed to take all such action reasonably necessary to request its members make such capital contributions. Additionally, in the event either of the Camden III Funds undertakes a liquidating distribution while the indemnification obligations of the sponsor are outstanding, they have agreed, in our sponsor's limited liability company agreement, to use reasonable efforts to set aside from such distribution, adequate reserves to cover the reasonably anticipated liabilities which may be incurred by our sponsor. We and the representative of the underwriters are named as express third party beneficiaries in and with respect to the provisions of our sponsor's limited liability company agreement which require the Camden III Funds to make such capital contributions and establish such reserves. Although we have a fiduciary obligation to pursue the sponsor to enforce its indemnification obligations, and intend to pursue such actions as and when we deem appropriate, there can be no assurance it or the Camden III Funds will be able to satisfy those obligations, if required to do so. The Camden III Funds are exempt from registration under the Investment Company Act of 1940 pursuant to either Section 3(c)(1) or Section 3(c)(7) of such Act and have informed us they expect to continue to qualify for such exemption following this offering.

Our existing stockholders have agreed to waive their respective rights to participate in any liquidation as part of our plan of dissolution and liquidation with respect to those shares of common stock acquired by them prior to this offering. They will participate in any liquidation distribution with respect to any shares of common stock acquired as part of this offering or in the aftermarket. In addition, Morgan Joseph & Co. Inc. has agreed to waive its right to the \$1,500,000 (\$1,725,000 if the underwriters' over-allotment option is exercised in full) of deferred compensation deposited in the trust account if a business combination is not consummated. We anticipate the distribution of the funds in the trust account to our public stockholders will occur within 10 business days from the date our corporate existence ceases.

13

TABLE OF CONTENTS

Accordingly, in the event we liquidate the trust account, our public stockholders will receive \$7.92 per share (plus a portion of the interest on the trust account, but net of: (i) taxes payable on the interest income earned and (ii) up to \$600,000 of interest income released to us to fund our working capital requirements) subject to reduction by claims of creditors as described above.

Escrow of existing stockholders' securities:

On the date of this prospectus, all of our existing stockholders, including all of our officers and directors, will place the shares they owned before this offering into an escrow account maintained by Continental Stock Transfer & Trust Company acting as escrow agent. Other than transfers to family members, to trusts for estate planning purposes or upon death (upon any of which such transfer they shall remain subject to the escrow agreement), these shares will not be transferable and will not be released from escrow until one year after consummation of a business combination, unless we were to consummate a transaction after the consummation of the initial business combination that results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property. If we are forced to dissolve and liquidate, these shares will be cancelled. On the date of this prospectus, the insider warrants will be placed into the escrow account maintained by Continental Stock Transfer & Trust Company, acting as escrow agent. Other than transfers to family members, to trusts for estate planning purposes or upon death (upon any of which such transfer they shall remain subject to the escrow agreement), the insider warrants will not be transferable and will not be released from escrow until the 90th day after the completion of our business combination. Additionally, the membership interests of Camden Learning, LLC, our sponsor, are not transferable until one year after consummation of a business combination.

If holders of more than 20% of the shares sold in this offering vote against a proposed business combination and seek to exercise their redemption rights and such business combination is consummated, our existing stockholders have agreed to forfeit, on a pro rata basis, and return to us for cancellation, a number of the initial 1,562,500 shares of our common stock purchased, up to a maximum of 195,312 shares, so that the existing stockholders will collectively own no more than 23.81% (without regard to any purchase of units in this offering, any open market purchases or private purchases of units by the sponsor directly from us, as set forth elsewhere herein) of our outstanding common stock immediately prior to the consummation of such business combination.

14

TABLE OF CONTENTS

Risks

We are a newly formed company and until we complete a business combination, we will have no operations and will generate no operating revenues. In making your decision on whether to invest in our securities, you should take into account not only the backgrounds of our management team, but also the special risks we face as a blank check company, as well as the fact that this offering is not being conducted in compliance with Rule 419 promulgated under the Securities Act of 1933, as amended, and, therefore, you will not be entitled to protections normally afforded to investors in Rule 419 blank check offerings. Additionally, our existing stockholders' initial equity investment is below that which is required under the guidelines of the North American Securities Administrators' Association, Inc. and we do not satisfy such association's policy regarding unsound financial condition. You should carefully consider these and the other risks set forth in the section entitled "Risk Factors" beginning on page 17 of this prospectus.

15

TABLE OF CONTENTS

SUMMARY FINANCIAL DATA

The following table summarizes the relevant financial data for our business and should be read with our financial statements, which are included in this prospectus. We have not had any significant operations to date, so only balance

sheet data is presented.

	September 30, 2007				
alance Sheet Datext-align:		Actual		As Adjusted ⁽¹⁾	
Working capital (deficit)	\$	(198,356)	\$	919,787	
Total assets		265,081		50,419,787	
Total liabilities		234,582		1,500,000	
Value of common stock which may be redeemed for an interest in		_		14,849,992	
the trust account					
Stockholders' equity		30,499		34.069.795	

(1) The "as adjusted" information gives effect to the sale of the units in this offering and the sale of warrants in the private placement, including the application of the related gross proceeds and the payment of the estimated remaining costs from such transactions.

The "actual" information for working capital (deficit) excludes \$228,855 of costs related to this offering and the private placement which were incurred or paid prior to September 30, 2007. These deferred offering costs have been recorded as a long-term asset and are reclassified against stockholders' equity in the "as adjusted" information. The "actual" information for working capital (deficit) also excludes the \$10,712 excess of the face amount of a \$200,000 note payable to an affiliate over the recorded liability of \$189,288. The "as adjusted" information gives effect to the payment of this note at its full face amount.

The total assets amount, as adjusted, includes the \$49,500,000 (\$7.92 per share) to be held in the trust account, \$48,000,000 of which will be available to us to consummate a business combination within the time period described in this prospectus, with the balance of \$1,500,000 (\$0.24 per share) to be used to pay the deferred underwriting discount and commissions payable to Morgan Joseph & Co. Inc., less amounts payable to redeeming stockholders, upon consummation of a business combination. If a business combination is not so consummated, we will be dissolved and the proceeds held in the trust account (including \$1,500,000 of deferred compensation to be held for the benefit of Morgan Joseph & Co. Inc.) will be distributed solely to our public stockholders (subject to our obligations under Delaware law to provide for claims of creditors).

We will not proceed with a business combination if public stockholders owning more than 29.99% of the shares sold in this offering vote against the business combination and exercise their redemption rights. Accordingly, we may effect a business combination only if stockholders owning one share less than 30% of the shares sold in this offering exercise their redemption rights. If this occurred, we would be required to redeem for cash up to approximately one share less than 30% of the 6,250,000 shares of common stock sold in this offering, or 1,874,999 shares of common stock, at a pershare redemption price of \$7.92 (plus a portion of the interest earned on the trust account, but net of (i) taxes payable on interest earned and (ii) up to \$600,000 of interest income released to us to fund our working capital), which includes \$0.24 per share of deferred underwriting discount and commissions which Morgan Joseph & Co. Inc. has agreed to forfeit to pay redeeming stockholders. The actual per-share redemption price will be equal to:

- the initial amount in the trust account (\$7.92 per share) which includes the amount attributable to deferred underwriting discounts and commissions and including all accrued interest (after taxes payable and up to \$600,000 of interest income released to us to fund our working capital), as of two business days prior to the proposed consummation of the business combination, *divided by*
- · the number of shares of common stock sold in the offering.

In the event holders of more than 20% of the shares sold in this offering elect to redeem their shares, our existing stockholders have agreed to forfeit a number of the initial 1,562,500 shares of our common stock purchased, up to a maximum of 195,312 shares, so that the existing stockholders will collectively own no more than 23.81% (without regard to any purchase of units in this offering, any open market purchases or private purchases of units by the sponsor directly from us, as set forth elsewhere herein) of our outstanding common stock immediately prior to the consummation of such business combination after giving effect to the redemption.

16

TABLE OF CONTENTS

RISK FACTORS

An investment in our securities involves a high degree of risk. You should consider carefully all of the material risks described below, together with the other information contained in this prospectus before making a decision to invest in our securities. If any of the following events occur, our business, financial condition and results of operations may be materially adversely affected. In such event, the trading price of our securities could decline and you could lose all or part of your investment.

Risks Associated with Our Business

We are a recently formed development stage company with no operating history and no revenues and, accordingly, you will not have any basis on which to evaluate our ability to achieve our business objective.

We are a recently formed development stage company with no operating results to date. Therefore, our ability to begin operations is dependent upon obtaining financing through the public offering of our securities. Since we do not have any operations or an operating history, you will have no basis upon which to evaluate our ability to achieve our business objective, which is to acquire, merge with, engage in a capital stock exchange with, purchase all or substantially all of the assets of, or engage in any other similar business combination with, a single operating entity, or one or more related or unrelated operating entities with a business in the education industry. We do not have any specific merger, capital stock exchange, asset acquisition or other business combination under consideration or contemplation and we have not, nor has anyone on our behalf, contacted any potential target business or had any discussions, formal or otherwise, with respect to such a transaction. Moreover, we have not engaged or retained any agent or other representative to identify or locate any suitable acquisition candidate for us. We cannot assure you as to when or if a business combination will occur. We have no present revenue and will not generate any revenues until, at the earliest, after the completion of a business combination. The report of our independent registered public accountants on our financial statements includes an explanatory paragraph stating that our ability to continue as a going concern is dependent on the consummation of this offering. The financial statements do not include any adjustments that might result from our inability to consummate this offering or our ability to continue as a going concern.

If we are unable to complete a business combination and are forced to liquidate, our public stockholders will receive less than \$8.00 per share upon distribution of the trust account and our warrants will expire worthless.

If we are unable to complete a business combination within 24 months from the date of this prospectus and are forced to liquidate our assets, the per share liquidation proceeds will be less than \$8.00 because of the expenses incurred in connection with this offering, our general and administrative expenses and the anticipated costs of seeking a business combination. The per share liquidation value will be \$7.92 per share, plus interest earned thereon (after taxes payable and up to \$600,000 of interest income released to us), which includes the net proceeds of this offering and the private placement of the insider warrants and \$1,500,000 (\$0.24 per share) of deferred underwriting discount and commissions. While we will pay, or reserve for payment, from funds not held in trust, our liabilities and obligations, and our sponsor has agreed to indemnify us under certain circumstances for such liabilities and obligations, we cannot assure you, where it is subsequently determined that the reserve for liabilities is insufficient, that stockholders will not be liable for such amounts to creditors. Furthermore, there will be no distribution with respect to our outstanding warrants and, accordingly, the warrants will expire worthless if we liquidate before the completion of a business combination. For a more complete discussion of the effects on our stockholders if we are unable to complete a business combination, see the section below entitled "Effecting a business combination — Liquidation if no business combination."

If we are unable to consummate a business combination, our public stockholders will be forced to wait the full 24 months before receiving liquidation distributions.

We have 24 months in which to complete a business combination. We have no obligation to return funds to investors prior to such date unless we consummate a business combination prior thereto and only then in cases where investors have sought redemption of their shares. Only after the expiration of this full time period will public stockholders be entitled to liquidation distributions if we are unable to complete a business combination. Accordingly, investors' funds may be unavailable to them until such date.

17

TABLE OF CONTENTS

You will not be entitled to protections normally afforded to investors of blank check companies, including the ability to receive all interest earned on the amount held in trust.

Since the net proceeds of this offering are intended to be used to complete a business combination with a target business which has not been identified, we may be deemed to be a "blank check" company under the United States securities laws. However, since we will have net tangible assets in excess of \$5,000,000 upon the consummation of this offering and will file a Current Report on Form 8-K with the Securities and Exchange Commission, or the SEC, upon consummation of this offering, including an audited balance sheet demonstrating this fact, we are exempt from rules promulgated by the SEC to protect investors of blank check companies such as Rule 419. Accordingly, investors will not be afforded the benefits or protections of those rules, such as entitlement to all the interest earned on the funds deposited into the trust account. Because we are not subject to Rule 419, a portion of the interest earned on the funds deposited in the trust account will be released to us to fund our working capital and will not be available at all to those public stockholders redeeming in connection with a business combination. For a more detailed comparison of our offering to offerings under Rule 419, see the section entitled "Comparison to Offerings of Blank Check Companies" below

Because there are numerous companies with a business plan similar to ours seeking to effectuate a business combination, it may be more difficult for us to complete a business combination.

Based upon publicly available information, we have identified approximately 137 similarly structured blank check companies which have completed initial public offerings since August 2003 and 47 others with registration statements currently pending before the SEC. Of the blank check companies which have completed their public offerings, only 40 companies have consummated a business combination, 24 other companies have announced they have entered into a definitive agreement for a business combination but have not consummated such business combination and eight have liquidated or will be liquidating. Accordingly, there are approximately 65 blank check companies with approximately \$9.3 billion in trust seeking to carry out a business plan similar to our business plan. While many of these companies are targeted towards specific industries in which they must complete a business combination, certain of these

companies may consummate a business combination in any industry they choose. As a result, there may be significant demand for the types of privately-held companies we target, which demand may limit the number of potential acquisition targets for us.

Further, because only 64 of such companies have either consummated a business combination or entered into a definitive agreement for a business combination, it may indicate there are fewer attractive target businesses available to such entities or that many privately-held target businesses are not inclined to enter into these types of transactions with publicly-held blank check companies like ours. We cannot assure you we will be able to successfully compete for an attractive business combination. Additionally, because of this competition, we cannot assure you we will be able to effectuate a business combination within the prescribed time period. If we are unable to consummate a business combination within the prescribed time period, we will be forced to liquidate.

The terms on which we may effect a business combination can be expected to become less favorable as we approach our twenty four month deadline.

Pursuant to our amended and restated certificate of incorporation, we must adopt a plan of dissolution and liquidation and initiate procedures for our dissolution and liquidation and the distribution of our assets, including the funds held in the trust account, if we do not effect a business combination within 24 months after the completion of this offering. We have agreed with the trustee to promptly adopt a plan of dissolution and liquidation and initiate procedures for our dissolution and liquidation and the distribution of our assets, including the funds held in the trust account, upon expiration of the time periods set forth above unless we obtain the written consent of holders of 95% of the shares purchased in this offering.

Any entity with which we negotiate, or attempt to negotiate, a business combination, will be aware of this time limitation and can be expected to negotiate accordingly. In such event, we may not be able to reach an agreement with any proposed target prior to such period and any agreement that is reached may be on terms less favorable to us than if we did not have the time period restrictions set forth above. Additionally, as the 24 month time period draws closer, we may not have the desired amount of negotiating leverage in the

18

TABLE OF CONTENTS

event any new information comes to light after entering into definitive agreements with any proposed target but prior to consummation of a business transaction.

The fact we will proceed with the business combination if public stockholders holding one share less than 30% of the shares sold in this offering exercise their redemption rights, rather than the 20% threshold of most other blank check companies, may hinder our ability to consummate a business combination in the most efficient manner or to optimize our capital structure.

Unlike most other blank check offerings which have a 20% redemption threshold, we will proceed with the business combination if public stockholders holding one share less than 30% of the shares sold in this offering exercise their redemption rights. As a result of our higher redemption threshold, we may have less cash available to complete a business combination. Because we will not know how many stockholders may exercise such redemption rights, we will need to structure a business combination meeting the 80% of our net assets test that requires less cash, or we may need to arrange third party financing to help fund the transaction in case a larger percentage of stockholders exercise their redemption rights than we expect. Alternatively, to compensate for the potential shortfall in cash, we may be required to structure the business combination, in whole or in part, using the issuance of our stock as consideration. Accordingly, this increase in redemption threshold to 30% may hinder our ability to consummate a business combination in the most efficient manner or to optimize our capital structure. At the same time, ownership by the sponsor of the 212,500 units purchased in the offering and \$4,000,000 worth of common stock purchased on the open market would represent 9.18% of the voting securities, assuming our sponsor purchases 212,500 units in the offering and the maximum of \$4,000,000 of open market purchases, all of which the sponsor has indicated it will vote in favor of the business combination, further increases the likelihood the business combination will be approved.

Exercise of redemption rights must be effected pursuant to a specific process which may take time to complete and may result in the expenditure of funds by stockholders seeking redemption.

A stockholder requesting redemption of his, her or its common stock into cash may be required to do so at any time after the mailing to our stockholders of the proxy statement and prior to the vote taken with respect to a proposed business combination. A stockholder would have from the time we send out our proxy statement through the vote on the business combination to tender (either electronically or through the delivery of physical stock certificates) his shares of common stock if he, she or it wishes to seek to exercise his, her or its redemption rights, a period which is expected to be not less than 10 nor more than 60 days. There is a nominal cost associated with the above-referenced tendering process and the act of certificating the shares or delivering them through the DWAC (Deposit/Withdrawal At Custodian) system. The transfer agent will typically charge the tendering broker \$35 and it would be up to the broker whether or not to pass this cost on to the redeeming holder. There may be additional mailing and other nominal charges depending on the particular process used to tender common stock. Although we believe the time period, costs and other potential burdens associated with the tendering process are not onerous for an average investor, this process may result in additional burdens for stockholders, including mis-delivery or any other defect in the tendering process.

Additionally, if a vote on our initial business combination is held and the business combination is not approved, or is not completed for any other reason, we may continue to try to consummate a business combination with a different

target until twenty four months from the date of this prospectus. Public stockholders voting against our initial business combination who exercised their redemption rights would not be entitled to convert their shares of common stock into a pro rata share of the aggregate amount then on deposit in the trust account in respect of the unconsummated initial business combination. In such case, if we have required public stockholders to tender their certificates prior to the meeting, we will promptly return such certificates to the tendering public stockholder, who would have to comply with the tendering process again for any vote against a subsequent business combination.

If third parties bring claims against us, the proceeds held in trust could be reduced and the per-share liquidation price received by stockholders from the trust account will be less than \$7.92 per share.

Our placing of funds in trust may not protect those funds from third party claims against us. Although we will use our reasonable best efforts to have all vendors, prospective target businesses or other entities with which we execute agreements waive any right, title, interest or claim of any kind in or to any monies held in

19

TABLE OF CONTENTS

the trust account for the benefit of our public stockholders, there is no guarantee they will execute such waivers, or even if they execute such waivers that they would be prevented from bringing claims against the trust account including but not limited to fraudulent inducement, breach of fiduciary responsibility and other similar claims, as well as claims challenging the enforceability of the waiver, in each case in order to gain an advantage with a claim against our assets, including the funds held in the trust account. If any third party refused to execute an agreement waiving such claims to the monies held in the trust account, we would perform an analysis of the alternatives available to us if we chose not to engage such third party and evaluate if such engagement would be in the best interest of our stockholders if such third party refused to waive such claims. We may elect to forego obtaining waivers only if we receive the approval of our Chief Executive Officer and the approving vote or written consent of at least a majority of our board of directors. Examples of possible instances where we may engage a third party that refused to execute a waiver include the engagement of a third party consultant whose particular expertise or skills are believed by management to be significantly superior to those of other consultants that would agree to execute a waiver or in cases where management is unable to find a provider of required services willing to provide the waiver. In addition, there is no guarantee such entities will agree to waive any claims they may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with us and not seek recourse against the trust account for any reason.

Accordingly, the proceeds held in trust could be subject to claims that could take priority over the claims of our public stockholders and the per-share liquidation price could be less than the \$7.92 per share held in the trust account, plus interest (net of any taxes due on such interest, which taxes, if any, shall be paid from the trust account and net of any amounts released to us as working capital), due to claims of such creditors. If we are unable to complete a business combination and are forced to dissolve and liquidate, our sponsor will be liable to ensure the proceeds in the trust account are not reduced by the claims of various vendors, prospective target businesses or other entities owed money by us for services rendered or products sold to us, to the extent necessary to ensure such claims do not reduce the amount in the trust account. In order to protect the amounts held in the trust account, our sponsor has agreed to indemnify us for claims of any vendors, service providers, prospective target businesses or creditors that have not executed a valid and binding waiver of amounts due to them out of the trust account. As further assurance our sponsor will have the necessary funds required to meet these indemnification obligations, (i) the Camden III Funds have agreed, under our sponsor's limited liability company agreement, to make capital contributions to our sponsor as and when required in order for the sponsor to fulfill its indemnification obligations and (ii) our sponsor has agreed to take all such action reasonably necessary to request its members make such capital contributions. Additionally, in the event either of the Camden III Funds undertakes a liquidating distribution while the indemnification obligations of the sponsor are outstanding, they have each agreed, in our sponsor's limited liability company agreement, to use reasonable efforts to set aside from such distribution adequate reserves to cover the reasonably anticipated liabilities which may be incurred by our sponsor. We and the representative of the underwriters are named as express third party beneficiaries in and with respect to the provisions of our sponsor's limited liability company agreement which require the Camden III Funds to make such capital contributions and establish such reserves. Although we have a fiduciary obligation to pursue the sponsor to enforce its indemnification obligations, and intend to pursue such actions as and when we deem appropriate, there can be no assurance it or the Camden III Funds will be able to satisfy those obligations, if required to do so.

Our officers and directors are currently, and may in the future become, affiliated with entities engaged in business activities similar to those intended to be conducted by us, and, accordingly, may have conflicts of interest in determining which entity a particular business opportunity should be presented to.

None of our directors or officers has been a principal of, or affiliated with, a "blank check" company that executed a business plan similar to our business plan and none of such individuals is currently affiliated with any such entity. However, our officers and directors are currently, and may in the future become, affiliated with entities, including other "blank check" companies, engaged in business activities similar to those intended to be conducted by us. While our officers and directors have agreed to present business opportunities first to the company, subject to any pre-existing duty they may have, they may become aware of business opportunities which may be appropriate for presentation to us, as well as the other entities to which they owe fiduciary duties. Accordingly, they may have conflicts of interest in determining to which entity a particular business opportunity should be presented.

TABLE OF CONTENTS

Each of our officers and directors has pre-existing fiduciary obligations to other businesses of which they are officers or directors. To the extent they identify business opportunities which may be suitable for the entities to which they owe a pre-existing fiduciary obligation, our officers and directors will honor those fiduciary obligations, subject to the "right of first refusal" described below. Accordingly, they may not present opportunities to us that otherwise may be attractive to us unless the entities to which they owe a pre-existing fiduciary obligation (and any successors to such entities) have declined to accept such opportunities.

David L. Warnock serves on the boards of directors of American Public Education, Inc. and Nobel Learning Communities, Inc., and both he and Mr. Hughes serve on the boards of directors of New Horizons Worldwide, Inc. and Questar Assessment, Inc., formerly Touchstone Applied Science Associates, all of which are portfolio companies of one or both of the Camden III Funds in the education industry. Both Messrs. Warnock and Hughes have a pre-existing fiduciary duty to each of these companies and may not present opportunities to us that otherwise may be attractive to us unless these entities have declined to accept such opportunities.

Messrs. Warnock and Hughes also have fiduciary obligations to the Camden III Funds. The Camden III Funds are private equity funds focused on investing in micro-cap public and, to a lesser extent, late stage private companies, in the business and financial services, healthcare and education industries. In order to minimize potential conflicts, or the appearance of conflicts, which may arise from the affiliations that Messrs. Warnock and Hughes have with the Camden III Funds, the Camden III Funds have granted us a "right of first refusal" with respect to an acquisition of voting control of any company or business in the education industry whose aggregate fair market value is at least equal to 80% of the balance of the trust account (less the deferred underwriting discounts and commissions and taxes payable), which is the minimum size of a target business for our initial business combination. Pursuant to this right of first refusal, each of the Camden III Funds has agreed to present any investment or purchase opportunity in a company meeting these criteria to a committee of our independent directors for our review and that it will not enter into any agreement to purchase or invest in such company until our committee of independent directors has had a reasonable period of time to determine whether or not to pursue such opportunity. This right of first refusal will expire upon the earlier of (i) our consummation of an initial business combination or (ii) 24 months after the consummation of this offering. Furthermore, we have agreed with the representative of the underwriters that any target company with respect to which either of the Camden III Funds has initiated any contacts or entered into any negotiations or discussions, formal or informal, will not be a potential acquisition target for us.

Therese Kreig Crane, Ed.D, currently serves as Chairman of the Board of Directors of Nobel Learning Communities Inc. and as a director of Questia Inc. and Tutor.com. Ronald Tomalis is a director and owner of The Chartwell Educational Group. While we do not know if any of these entities will be competitive with us, each of these directors has a pre-existing fiduciary duty to each of these companies and may not present opportunities to us that otherwise may be attractive to us unless these entities have declined to accept such opportunities.

For a more complete discussion of our management's affiliations and the potential conflicts of interest that you should be aware of, see the sections below entitled "Management — Directors and Executive Officers" and "Management — Conflicts of Interest." We cannot assure you these conflicts will be resolved in our favor.

Our stockholders may be held liable for claims by third parties against us to the extent of distributions received by them.

We will dissolve and liquidate if we do not complete a business combination within 24 months after the consummation of this offering. In the event of a dissolution, stockholders may be held liable under the Delaware General Corporation Law for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. If the corporation complies with certain procedures set forth in Section 280 of the Delaware General Corporation Law intended to ensure it makes reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder's pro rata share of

21

TABLE OF CONTENTS

the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution. However, it is our intention to make liquidating distributions to our stockholders within 10 business days after the 24 month period and, therefore, we do not intend to comply with these procedures. Because we will not be complying with Section 280, we will comply with Section 281(b) of the Delaware General Corporation Law, requiring us to adopt a plan of dissolution that will provide for our payment, based on facts known to us at such time, of (i) all existing claims, (ii) all pending claims and (iii) all claims that may be potentially brought against us within the subsequent 10 years. However, because we are a blank check company, rather than an operating company, and our operations will be limited to searching for prospective target businesses to acquire, the only likely claims to arise would be from our vendors (such as accountants, lawyers, investment bankers, etc.) or potential target businesses. As described above, we will use our reasonable best efforts to have all vendors and prospective target businesses execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account. As

a result, the claims that could be made against us are significantly limited and the likelihood any claim would result in any liability extending to the trust is minimal. However, because we will not be complying with Section 280, our public stockholders could potentially be liable for any claims to the extent of distributions received by them in a dissolution, and any such liability of our stockholders will likely extend beyond the third anniversary of such dissolution. Accordingly, we cannot assure you that third parties will not seek to recover from our public stockholders amounts owed to them by us.

If we are forced to file a bankruptcy case or an involuntary bankruptcy case is filed against us which is not dismissed, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a "preferential transfer" or a "fraudulent conveyance". As a result, a bankruptcy court could seek to recover all amounts received by our stockholders in our dissolution. Furthermore, because we intend to distribute the proceeds held in the trust account to our public stockholders promptly after [], 2009, this may be viewed or interpreted as giving preference to our public stockholders over any potential creditors with respect to access to or distributions from our assets. Additionally, our board may be viewed as having breached their fiduciary duties to our creditors and/or may have acted in bad faith, thereby exposing itself and our company to claims of punitive damages, by paying public stockholders from the trust account prior to addressing the claims of creditors. We cannot assure you that claims will not be brought against us for these reasons.

We may choose to redeem our outstanding warrants at a time that is disadvantageous to our warrant holders.

Subject to there being a current prospectus under the Securities Act of 1933 with respect to the shares of common stock issuable upon exercise of the warrants, we may redeem the warrants issued as a part of our units at any time while the warrants are exercisable in whole and not in part, at a price of \$0.01 per warrant, upon a minimum of 30 days' prior written notice of redemption, if and only if, the last sales price of our common stock equals or exceeds \$11.50 per share for any 20 trading days within a 30 trading day period ending three business days before we send the notice of redemption. Redemption of the warrants could force the warrant holders (i) to exercise the warrants and pay the exercise price thereafter at a time when it may be disadvantageous for the holders to do so, (ii) to sell the warrants at the then current market price when they might otherwise wish to hold the warrants, or (iii) to accept the nominal redemption price which, at the time the warrants are called for redemption, is likely to be substantially less than the market value of the warrants. The foregoing does not apply to the warrants included as part of the 2,800,000 insider warrants purchased prior to this offering, as such warrants are not subject to redemption while held by the initial holder or any permitted transferee of such initial holder.

Although we are required to use our best efforts to have an effective registration statement covering the issuance of the shares underlying the warrants at the time that our warrant holders exercise their warrants, we cannot guarantee a registration statement will be effective, in which case our warrant holders may not be able to exercise our warrants.

Holders of our warrants will be able to exercise the warrant only if (i) a current registration statement under the Securities Act of 1933 relating to the shares of our common stock underlying the warrants is then effective and (ii) such shares are qualified for sale or exempt from qualification under the applicable securities

22

TABLE OF CONTENTS

laws of the states in which the various holders of warrants reside. Although we have undertaken in the warrant agreement, and therefore have a contractual obligation, to use our best efforts to maintain a current registration statement covering the shares underlying the warrants following completion of this offering to the extent required by federal securities law, and we intend to comply with such undertaking, we cannot assure you we will be able to do so. At the time the warrants become exercisable (following our completion of a business combination), we expect to be listed on a national securities exchange, which would provide an exemption from registration in every state, but we cannot assure you this will be the case. Accordingly, we believe holders in every state will be able to exercise their warrants as long as our prospectus relating to the common stock issuable upon exercise of the warrants is current. However, we cannot assure you of this fact. The value of the warrants may be greatly reduced if a registration statement covering the shares issuable upon the exercise of the warrants is not kept current or if the securities are not qualified, or exempt from qualification, in the states in which the holders of warrants reside. For example, some states may not permit us to register the shares issuable on exercise of our warrants for sale. We are not obligated to pay cash or other consideration to the holders of the warrants in such situations. Holders of warrants who reside in jurisdictions in which the shares underlying the warrants are not qualified and in which there is no exemption will be unable to exercise their warrants and would either have to sell their warrants in the open market or allow them to expire worthless. In the event the warrants expire worthless or we choose to redeem the warrants at a time when the holders of such warrants are unable to exercise them, the purchasers of units will have effectively paid the full purchase price of the units solely for the common stock underlying such units. If and when the warrants become redeemable by us, we may exercise our redemption right even if we are unable to qualify the underlying securities for sale under all applicable state securities laws.

Because the warrants sold in the private placement were originally issued pursuant to an exemption from the registration requirements under the federal securities laws, holders of such warrants will be able to exercise their warrants even if, at the time of exercise, a prospectus relating to the common stock issuable upon exercise of such warrants is not current.

We may issue shares of our capital stock or debt securities to complete a business combination, which would reduce the equity interest of our stockholders and likely cause a change in control of our ownership.

Our amended and restated certificate of incorporation authorizes the issuance of up to 20,000,000 shares of common stock, par value \$.0001 per share, and 1,000,000 shares of preferred stock, par value \$.0001 per share. Immediately after this offering, there will be 1,887,500 authorized but unissued shares of our common stock available for issuance (after appropriate reservation for the issuance of shares upon full exercise of our outstanding warrants and underwriter's unit purchase option) and all of the 1,000,000 shares of preferred stock available for issuance. Although we have no commitments as of the date of this offering to issue our securities, we may issue a substantial number of additional shares of our common stock or preferred stock, or a combination of common and preferred stock, to complete a business combination. The issuance of additional shares of our common stock or any number of shares of our preferred stock:

- · may significantly reduce the equity interest of investors in this offering;
- will likely cause a change in control if a substantial number of our shares of common stock are issued, which
 may affect, among other things, our ability to use our net operating loss carry forwards, if any, and most likely
 also result in the resignation or removal of our present officers and directors;
- may adversely affect prevailing market prices for our common stock; and
- may subordinate the rights of holders of our common stock if preferred stock is issued with rights senior to those afforded to the common stock.

If we finance the purchase of assets or operations through the issuance of debt securities, it could result in:

 default and foreclosure on our assets if our operating cash flow after a business combination were insufficient to pay our debt obligations;

23

TABLE OF CONTENTS

- acceleration of our obligations to repay the indebtedness even if we have made all principal and interest
 payments when due, if the debt security contained covenants that required the maintenance of certain financial
 ratios or reserves and any such covenant were breached without a waiver or renegotiation of that covenant;
- · our immediate payment of all principal and accrued interest, if any, if the debt security was payable on demand;
- · covenants that limit our ability to acquire capital assets or make additional acquisitions; and
- our inability to obtain additional financing, if necessary, if the debt security contained covenants restricting our ability to obtain additional financing while such security was outstanding.

We are not currently able to assess the likelihood we will need to finance a business combination through the issuance of debt securities. However, since we will proceed with the business combination if public stockholders holding less than 30% of the shares sold in this offering exercise their redemption rights, unlike the 20% threshold adopted by many companies similar to ours, we may have less cash available to complete a business combination. Because we will not know how many stockholders may exercise such redemption rights, we will need to structure a business combination meeting the 80% of our net assets test that requires less cash, or we may need to arrange third party financing to help fund the transaction in case a larger percentage of stockholders exercise their redemption rights than we expect. For a more complete discussion of the possible structure of a business combination, see the section below entitled "Effecting a business combination — Selection of a target business and structuring of a business combination."

Substantial resources could be expended in researching business combinations that are not consummated, which could materially adversely affect subsequent attempts to locate and consummate a business combination.

We anticipate the investigation of each specific target business and the negotiation, drafting, and execution of relevant agreements, disclosure documents and other instruments will require substantial management time and attention and substantial costs for accountants, attorneys and other third party fees and expenses. If we decide not to enter into an agreement with respect to a specific proposed business combination we have investigated, the costs incurred up to that point for the proposed transaction likely would not be recoverable. Furthermore, even if an agreement is reached relating to a specific target business, we may fail to consummate the business combination for any number of reasons including those beyond our control, such as public stockholders holding 30% or more of the shares sold in this offering voting against the business combination and opting to have us redeem their stock into a pro rata share of the trust account even if a majority of our stockholders approve the business combination. Any such event will result in a loss to us of the related costs incurred which could materially adversely affect subsequent attempts to locate and consummate a business combination.

We may have insufficient resources to cover our operating expenses and the expenses of consummating a business combination.

We have reserved approximately \$900,000 from the proceeds of this offering and the private placement held outside the trust account and up to \$600,000 of interest, after taxes, we may earn on funds in the trust account to cover our operating expenses for the next 24 months, to cover the expenses incurred in connection with a business combination and to cover expenses in connection with our dissolution if we do not complete a business combination in the allowed time. This amount is based on management's estimates of the costs needed to fund our operations for the next 24 months, to consummate a business combination or dissolve. Those estimates may prove inaccurate, especially if a

portion of the available proceeds is used to make a down payment or pay exclusivity or similar fees in connection with a business combination, or if we expend a significant portion of the available proceeds in pursuit of a business combination that is not consummated. In addition, the amounts available from interest earned on the proceeds held in the trust account will be dependent on the length of time since our initial public offering and prevailing interest rates. If we do not have sufficient proceeds available to fund our expenses, we may be forced to obtain additional financing, either from our management or the existing stockholders or from third parties. We may not be able to obtain additional financing and our existing stockholders and management are not obligated to provide any additional

24

TABLE OF CONTENTS

financing. If we do not have sufficient proceeds and cannot find additional financing, we may be forced to dissolve and liquidate prior to consummating a business combination.

Our ability to effect a business combination and to execute any potential business plan afterwards will be dependent upon the efforts of our key personnel, some of whom may join us following a business combination and whom we would have only a limited ability to evaluate.

Our ability to effect a business combination will be dependent upon the efforts of our key personnel. The future role of our key personnel following a business combination cannot presently be fully ascertained. Although members of our management and other key personnel may remain associated with us following a business combination, including Messrs. Warnock, Hughes, Tomalis and Dr. Crane, we may employ other personnel following the business combination. While we intend to closely scrutinize any additional individuals we engage after a business combination, we cannot assure you our assessment of these individuals will prove to be correct. Moreover, our current management will only be able to remain with the combined company after the consummation of a business combination if they are able to negotiate terms with the combined company as part of any such combination. If we acquired a target business in an all-cash transaction, it would be more likely that current members of management would remain with us if they chose to do so. If a business combination were structured as a merger whereby the stockholders of the target company were to control the combined company following a business combination, it may be less likely management would remain with the combined company unless it was negotiated as part of the transaction via the acquisition agreement. an employment or consulting agreement or other arrangement. The determination to remain as officers of the resulting business will be determined prior to the completion of the transaction and will depend upon the appropriateness or necessity of current management to remain. In making the determination as to whether current management should remain with us following the business combination, management will analyze the experience and skill set of the target business' management, and negotiate as part of the business combination that certain members of current management remain if it is believed to be in the best interests of the combined company post-business combination. If management negotiates to be retained post-business combination as a condition to any potential business combination, such negotiations may result in a conflict of interest.

None of our officers or directors has ever been associated with a blank check company, which could adversely affect our ability to consummate a business combination.

None of our officers or directors has ever been associated with a blank check company. Accordingly, you may not have sufficient information with which to evaluate the ability of our management team to identify and complete a business combination using the proceeds of this offering and the private placement. Our management's lack of experience in operating a blank check company could adversely affect our ability to consummate a business combination and force us to dissolve and liquidate the trust account to our public stockholders as part of our stockholder-approved plan of dissolution and liquidation.

Our officers and directors may allocate their time to other businesses thereby causing conflicts of interest in their determination as to how much time to devote to our affairs. This could have a negative impact on our ability to consummate a business combination.

Our officers and directors are not required to commit their full time to our affairs, which may result in a conflict of interest in allocating their time between our operations and other businesses. We do not intend to have any full time employees prior to the consummation of a business combination. Each of our officers are engaged in several other business endeavors and are not obligated to contribute any specific number of hours per week to our affairs. If our officers' other business affairs require them to devote more substantial amounts of time to such affairs, it could limit their ability to devote time to our affairs and could have a negative impact on our ability to consummate a business combination. For a discussion of potential conflicts of interest that you should be aware of, see the section below entitled "Management — Conflicts of Interest." We cannot assure you these conflicts will be resolved in our favor.

25

TABLE OF CONTENTS

warrants purchased in the private placement, which will not participate in liquidation distributions and therefore our management may have a conflict of interest in determining whether a particular target business is appropriate for a business combination.

Certain directors and officers of ours, as well as our sponsor, own shares of our common stock, and our sponsor will own the warrants purchased in the private placement. Our sponsor is owned 96.01% by Camden Partners Strategic Fund III, L.P. and 3.99% by Camden Partners Strategic Fund III-A, L.P., each of which are indirectly controlled and partially owned by David L. Warnock, our Chairman, President and Chief Executive Officer, and Donald W. Hughes, our Chief Financial Officer and Secretary. The general partner of each limited partnership is Camden Partners Strategic III, LLC, an entity of which Messrs. Warnock and Hughes are two of the four managing members. Our sponsor, as well as those officers and directors that own our common stock, have each waived their right to the liquidation of the trust account if we are unable to complete a business combination. The shares and warrants owned by our sponsor and officers and directors will be worthless if we do not consummate a business combination. The personal and financial interests of our sponsor and officers and directors may influence their motivation in identifying and selecting a target business and completing a business combination in a timely manner. Consequently, our officers' and directors' discretion in identifying and selecting a suitable target business may result in a conflict of interest when determining whether the terms, conditions and timing of a particular business combination are appropriate and in our stockholders' best interest.

Since our existing stockholders will lose their entire investment in us if a business combination is not consummated and may be required to pay costs associated with our liquidation, our existing stockholders may purchase shares of our common stock from stockholders who would otherwise choose to vote against a proposed business combination or exercise their redemption rights in connection with such business combination.

Our existing stockholders own shares of our common stock (which were purchased for an aggregate of \$25,000) that will be worthless if we do not consummate a business combination. In addition, our sponsor purchased warrants exercisable for our common stock (for an aggregate of \$2,800,000), which will also be worthless if we do not consummate a business combination, and has entered into an agreement to place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Given the interest that our existing stockholders have in a business combination being consummated, it is possible our existing stockholders will acquire additional securities from public stockholders (in the open market and/or in privately negotiated transactions) who have elected to redeem their shares of our common stock in order to change their vote and insure the business combination will be approved (which could result in a business combination being approved even if, after the announcement of the business combination, 30.0% or more of our public stockholders would have elected to exercise their redemption rights, or more than 50.0% of our public stockholders would have voted against the business combination, but for the purchases made by our existing stockholder). At this time we have no indication any of our existing stockholders intend to do so. Although we do not know for certain the factors that would cause our existing stockholders to purchase such additional securities, we believe some of the factors they would consider are: (i) the trading price of our securities, (ii) their aggregate investment in our securities, (iii) whether it appears that a substantial number of public stockholders are voting against a proposed business combination, and (iv) their interest in the target business once the target business has been identified.

Any privately negotiated transaction with a stockholder would include a contractual acknowledgement that such stockholder, although still the record holder of our common stock, is no longer the beneficial owner thereof and therefore agrees to vote such shares of common stock as directed by our existing stockholder. In the event our existing stockholder purchases shares in privately negotiated transactions from public stockholders who have already submitted proxies to vote against a proposed business combination and requested redemption of their shares, such selling stockholders would be required to revoke their prior proxies to vote

26

TABLE OF CONTENTS

against the proposed acquisition and to revoke their prior elections to redeem their shares and to cast new votes in favor of the proposed acquisition. The revocation of prior proxies to vote against the business combination and substitution therefor of votes in favor of the proposed acquisition would have the effect of reducing redemptions and increasing votes in favor of the proposed acquisition, thereby making it more likely that a proposed business combination would be approved.

Our existing stockholders will not receive reimbursement for any out-of-pocket expenses incurred by them to the extent such expenses exceed the amount available outside the trust account (plus a portion of the interest earned on the amounts in the trust account) unless the business combination is consummated, and therefore they may have a conflict of interest.

Our existing stockholders will not receive reimbursement for any out-of-pocket expenses incurred by them to the extent such expenses exceed the amount available outside the trust account (plus a portion of the interest earned on the amounts in the trust account), unless the business combination is consummated. The amount of available proceeds is based on management estimates of the capital needed to fund our operations for the next 24 months and to consummate a business combination. Those estimates may prove to be inaccurate, especially if a portion of the available proceeds is used to make a down payment or pay exclusivity or similar fees in connection with a business

combination, or if we expend a significant portion in pursuit of an acquisition which is not consummated. The financial interests of such persons could influence their motivation in selecting a target business and thus, there may be a conflict of interest when determining whether a particular business combination is in the stockholders' best interest.

If our common stock becomes subject to the SEC's penny stock rules, broker-dealers may experience difficulty in completing customer transactions and trading activity in our securities may be adversely affected.

If at any time we have net tangible assets of less than \$5,000,000 and our common stock has a market price per share of less than \$5.00, transactions in our common stock may be subject to the "penny stock" rules promulgated under the Securities Exchange Act of 1934, as amended. Under these rules, broker-dealers who recommend such securities to persons other than institutional accredited investors must:

- make a special written suitability determination for the purchaser;
- receive the purchaser's written agreement to a transaction prior to sale;
- provide the purchaser with risk disclosure documents which identify certain risks associated with investing in "penny stocks" and which describe the market for these "penny stocks" as well as a purchaser's legal remedies; and
- obtain a signed and dated acknowledgment from the purchaser demonstrating the purchaser has actually received the required risk disclosure document before a transaction in a "penny stock" can be completed.

If our common stock becomes subject to these rules, broker-dealers may find it difficult to effectuate customer transactions and trading activity in our securities may be adversely affected. As a result, the market price of our securities may be depressed and you may find it more difficult to sell our securities.

It is probable our initial business combination will be with a single target business, which may cause us to be solely dependent on a single business and a limited number of products or services. Additionally, we may face obstacles to completing simultaneous acquisitions.

Our initial business combination must be with a business or businesses with an aggregate fair market value of at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction, which amount is required as a condition to the consummation of our initial business combination. We may not be able to acquire more than one target business because of various factors, including the amount of funds available to consummate a business combination, possible complex accounting issues, which would include generating pro forma financial statements reflecting the operations of several target businesses as if they had been combined, and numerous logistical issues, which could include attempting to coordinate the timing of negotiations, proxy statement

27

TABLE OF CONTENTS

disclosure and closings with multiple target businesses. In addition, we may not have sufficient management, financial or other resources to effectively investigate the business and affairs of multiple acquisition candidates simultaneously or to negotiate the terms of multiple acquisition agreements at the same time, which could result in a failure to properly evaluate multiple acquisitions. Further, we would also be exposed to the risk that conditions to closings with respect to the acquisition of one or more of the target businesses would not be satisfied, bringing the aggregate fair market value of the initial business combination below the required fair market value of 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable). Accordingly, while it is possible we may attempt to effect our initial business combination with more than one target business, we are more likely to choose a single target business if deciding between one target business meeting such 80% threshold and comparable multiple target business candidates collectively meeting the 80% threshold. Consequently, it is probable that, unless the purchase price consists substantially of our equity, we will have the ability to complete only a single initial business combination with the proceeds of this offering and the private placement. Accordingly, the prospects for our success may be:

- solely dependent upon the performance of a single business; or
- dependent upon the development or market acceptance of a single or limited number of products or services.

In this case, we will not be able to diversify our operations or benefit from the possible spreading of risks or offsetting of losses, unlike other entities which may have the resources to complete several business combinations in different industries or different areas of a single industry.

The ability of our stockholders to exercise their redemption rights may not allow us to effectuate the most desirable business combination or optimize our capital structure.

At the time we seek stockholder approval of any business combination, we will offer each public stockholder the right to have such stockholder's shares of common stock redeemed for cash if the stockholder votes against the business combination and the business combination is approved and completed. Accordingly, if our business combination requires us to use substantially all of our cash to pay the purchase price, because we will not know how many stockholders may exercise such redemption rights, we may either need to reserve part of the trust account for possible payment upon such redemption, or we may need to arrange third party financing to help fund our business combination in case a larger percentage of stockholders exercise their redemption rights than we expected. Therefore, we may not be able to consummate a business combination that requires us to use all of the funds held in the trust

account as part of the purchase price, or we may end up having a leverage ratio not optimal for our business combination. This may limit our ability to effectuate the most attractive business combination available to us.

We will not be required to obtain an opinion from an investment banking firm as to the fair market value of a proposed business combination if our board of directors independently determines the target business has sufficient fair market value.

The initial target business or businesses we acquire must have an aggregate fair market value equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction. There is no limitation on our ability to raise funds privately or through loans that would allow us to acquire a target business or businesses with an aggregate fair market value in an amount considerably greater than 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction. We have not had any preliminary discussions, or made any agreements or arrangements, with respect to financing arrangements with any third party. The fair market value of such business will be determined by our board of directors based upon standards generally accepted by the financial community, such as actual and potential sales, earnings and cash flow and book value, and the price for which comparable businesses have recently been sold. If our board is not able to independently determine whether the target business has a sufficient fair market value, we will obtain an opinion from an unaffiliated, independent investment banking firm which is a member of the Financial Industry Regulatory Authority, or FINRA, with respect to the satisfaction of such criteria.

28

TABLE OF CONTENTS

We may be unable to obtain additional financing, if required, to complete a business combination or to fund the operations and growth of the target business, which could compel us to restructure the transaction or abandon a particular business combination.

Although we believe the net proceeds of this offering and the private placement will be sufficient to allow us to consummate a business combination, in as much as we have not yet identified any prospective target business, we cannot ascertain the capital requirements for any particular transaction. If the net proceeds of this offering and the private placement prove to be insufficient, either because of the size of the business combination or the depletion of the available net proceeds in search of a target business, or because we become obligated to redeem for cash a significant number of shares from dissenting stockholders, we will be required to seek additional financing. We cannot assure you such financing would be available on acceptable terms, if at all. To the extent additional financing proves to be unavailable when needed to consummate a particular business combination, we would be compelled to restructure the transaction or abandon that particular business combination and seek an alternative target business candidate. In addition, if we consummate a business combination, we may require additional financing to fund the operations or growth of the target business. The failure to secure additional financing could have a material adverse effect on the continued development or growth of the target business. None of our officers, directors or stockholders is required to provide any financing to us in connection with or after a business combination.

Our existing stockholders, including our officers and directors, control a substantial interest in us and thus may influence certain actions requiring stockholder vote including a business combination.

Upon consummation of this offering, our existing stockholders (including all of our officers and directors) will collectively own approximately 20% of our issued and outstanding shares of common stock (not including the purchase of 2,800,000 warrants in the private placement by our sponsor, and assuming no additional purchases by our existing stockholders in the offering). These shares will be voted in accordance with the majority of the shares with respect to the approval of a proposed business combination. Our sponsor has informed us it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Our sponsor has agreed to vote all such shares of common stock in favor of our initial business combination, representing a possible maximum aggregate of 3.4% of the public shares entitled to vote on the business combination. Assuming only these securities are purchased, immediately after this offering, our existing stockholders, collectively, will beneficially own 23.4% of the then issued and outstanding shares of common stock. As a result of this ownership block, our current stockholders may be able to effectively influence the outcome of any other matters requiring approval by our stockholders, including the election of directors and approval of significant corporate transactions.

In addition, our sponsor has entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Assuming the 212,500 units are purchased in the offering and the \$4,000,000 of common stock are purchased by our sponsor in the open market, and further assuming such open market purchases of stock occur at the initial trust amount per share of \$7.92 per unit, our existing stockholders, collectively, will beneficially own 29.18% of the then issued and outstanding shares of common stock, with only 9.18% of such shares automatically voting in favor of the business combination and the remaining 20% voting in accordance with the majority of the shares. None of our other existing stockholders, officers and directors has indicated to us they intend to purchase units in this offering, or units or warrants on the open market following the offering. For a more complete discussion, please see the section of this prospectus entitled "Principal Stockholders."

Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell any shares purchased in the open market, provided it will be entitled to participate in any liquidating distributions with respect to

such shares. Our sponsor has agreed to vote all shares of common stock purchased in the open market in favor of our initial business combination. Accordingly, these purchases will have the effect of increasing the percentage of shares owned by our sponsor and make it more likely the stockholder vote to approve the business combination or amend or waive any provision of our amended certificate of incorporation, or any other matter for which stockholder approval is sought, will be successful.

29

TABLE OF CONTENTS

In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our sponsor has agreed to purchase from us in a private placement a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this co-investment. This co-investment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders. Accordingly, any such private placement purchases will have no influence on the vote on any proposed business combination.

Our board of directors is divided into two classes, each of which will generally serve for a term of two years, with only one class of directors being elected in each year. It is unlikely there will be an annual meeting of stockholders to elect new directors prior to the consummation of a business combination, in which case all of the current directors will continue in office at least until the consummation of the business combination. If there is an annual meeting, as a consequence of our "staggered" board of directors, initially only a minority of the board of directors will be considered for election and our existing stockholders, because of their ownership position, will have considerable influence regarding the outcome.

Accordingly, our existing stockholders will continue to exert control at least until the consummation of a business combination. In addition, our existing stockholders and their affiliates and relatives are not prohibited from purchasing additional units in this offering or in the open market. If they do, we cannot assure you our existing stockholders will not have considerable influence upon the vote in connection with a business combination.

Our existing stockholders paid an aggregate of \$25,000, or approximately \$.016 per share, for their shares and, accordingly, you will experience immediate and substantial dilution from the purchase of our common stock.

The difference between the public offering price per share of our common stock and the pro forma net tangible book value per share of our common stock after this offering and the private placement constitutes the dilution to you and the other investors in this offering. The fact our existing stockholders acquired their shares of common stock at a nominal price has significantly contributed to this dilution. Assuming the offering and the private placement are completed, you and the other new investors will incur an immediate and substantial dilution of approximately 28.25% or \$2.26 per share (the difference between the pro forma net tangible book value per share of \$5.74, and the initial offering price of \$8.00 per unit).

Our outstanding warrants may have an adverse effect on the market price of common stock and make it more difficult to effect a business combination.

In connection with this offering as part of the units, and the private placement, we will be issuing warrants to purchase up to 9,050,000 shares (9,987,500 if the underwriters' over-allotment option is exercised in full) of our common stock. To the extent we issue shares of common stock to effect a business combination, the potential for the issuance of substantial numbers of additional shares upon exercise of these warrants could make us a less attractive acquisition vehicle in the eyes of a target business as such securities, when exercised, will increase the number of issued and outstanding shares of our common stock and reduce the value of the shares issued to complete the business combination. Accordingly, our warrants may make it more difficult to effectuate a business combination or increase the cost of the target business. Additionally, the sale, or even the possibility of sale, of the shares underlying the warrants could have an adverse effect on the market price for our securities or on our ability to obtain future public financing. If and to the extent these warrants are exercised, you may experience dilution to your holdings.

If our existing stockholders exercise their registration rights, it may have an adverse effect on the market price of our common stock and the existence of these rights may make it more difficult to effect a business combination.

Our existing stockholders, including our sponsor, which purchased warrants in the private placement, are entitled to require us to register the resale of their shares of common stock and warrants (as well as the shares of common stock issuable upon exercise of warrants included as part of the insider warrants) at any time after

30

TABLE OF CONTENTS

the date on which their securities are released from escrow. If such existing security holders exercise their registration rights with respect to all of their securities, there will be an additional 1,562,500 shares of common stock and 2,800,000

warrants (as well as 2,800,000 shares of common stock issuable upon exercise of the warrants) eligible for trading in the public market and we will bear the costs of registering such securities. The presence of this additional number of shares of common stock eligible for trading in the public market may have an adverse effect on the market price of our common stock. In addition, the existence of these rights may make it more difficult to effectuate a business combination or increase the cost of the target business, as the stockholders of the target business may be discouraged from entering into a business combination with us or will request a higher price for their securities as a result of these registration rights and the potential future effect their exercise may have on the trading market for our common stock.

There is currently no market for our securities and a market for our securities may not develop, which could adversely affect the liquidity and price of our securities.

As of the date of this prospectus there is no market for our securities. Therefore, stockholders should be aware they cannot benefit from information about prior market history as to their decisions to invest, which means they are at further risk if they invest. In addition, the price of the securities, after the offering, can vary due to general economic conditions and forecasts, our general business condition and the release of our financial reports.

If you are not an institutional investor, you may purchase securities in this offering only if you reside within the states in which we will apply to have the securities registered. Although the states are preempted from regulating the resales of our securities, state securities regulators who view blank check offerings unfavorably could use or threaten to use their investigative or enforcement powers to hinder resales in their states.

We have applied to register our securities, or have obtained or will seek to obtain an exemption from registration, in Colorado, Delaware, the District of Columbia, Florida, Georgia, Hawaii, Illinois, Louisiana, Maryland, New York and Rhode Island. If you are not an "institutional investor," you must be a resident of these jurisdictions to purchase our securities in the offering. The definition of an "institutional investor" varies from state to state but generally includes financial institutions, broker-dealers, banks, insurance companies and other qualified entities. Institutional investors in every state except in Idaho may purchase the units in this offering pursuant to exemptions provided to such entities under the Blue Sky laws of various states. Under the National Securities Market Improvement Act of 1996, the states are pre-empted from regulating transactions in covered securities. We will file periodic and annual reports under the Securities Exchange Act of 1934, as amended, and our securities will be considered covered securities. Therefore, the states will be pre-empted from regulating the resales of the units, from and after the effective date, and the common stock and warrants comprising the units, once they become separately transferable. However, the states retain the jurisdiction to investigate and bring enforcement actions with respect to fraud or deceit, or unlawful conduct by a broker or dealer, in connection with the sale of securities. Although we are not aware of a state having used these powers to prohibit or restrict resales of securities issued by blank check companies generally, certain state securities commissioners view blank check companies unfavorably and might use these powers, or threaten to use these powers, to hinder the resale of securities of blank check companies in their states. For a more complete discussion of the state securities laws and registrations affecting this offering, please see "Underwriting — State Blue Sky Information" below.

We intend to have our securities quoted on the OTC Bulletin Board, which will limit the liquidity and price of our securities more than if our securities were quoted or listed on the Nasdaq Stock Market or a national exchange.

We anticipate our securities will be traded in the over-the-counter market. It is anticipated they will be quoted on the OTC Bulletin Board, a FINRA-sponsored and operated inter-dealer automated quotation system for equity securities not included in the Nasdaq Stock Market. Quotation of our securities on the OTC Bulletin Board will limit the liquidity and price of our securities more than if our securities were quoted or listed on The Nasdaq Stock Market or a national exchange. Lack of liquidity will limit the price at which you may be able to sell our securities or your ability to sell our securities at all.

31

TABLE OF CONTENTS

If we are deemed to be an investment company, we may be required to institute burdensome compliance requirements and our activities may be restricted, which may make it difficult for us to complete a business combination.

We do not plan to operate as an investment fund or investment company, or to be engaged in the business of investing, reinvesting or trading in securities. Our plan is to acquire, hold, operate and grow for the long term one or more operating businesses in the education industry. We do not plan to operate as a passive investor or as a merchant bank seeking dividends or gains from purchases and sales of securities.

Companies that fall within the definition of an "investment company" set forth in Section 3 of the Investment Company Act of 1940, as amended, and the regulations thereunder, which we refer to as the 1940 Act, are subject to registration and substantive regulation under the 1940 Act. Companies that are subject to the 1940 Act that do not become registered are normally required to liquidate and are precluded from entering into transactions or enforceable contracts other than as an incident to liquidation. The basic definition of an "investment company" in the 1940 Act and related SEC rules and interpretations includes a company (1) that is, proposes to be, or holds itself out as being engaged primarily in investing, reinvesting or trading in securities; or (2) that has more than 40% of its assets (exclusive of U.S. government securities and cash items) in "investment securities," or (3) that is a "special situation investment company" (such as a merchant bank or private equity fund).

For example, if we were deemed to be an investment company under the 1940 Act, we would be required to

become registered under the 1940 Act (or liquidate) and our activities would be subject to a number of restrictions, including, among others:

- corporate governance requirements and requirements regarding mergers and share exchanges;
- restrictions on the nature of our investments;
- restrictions on our capital structure and use of multiple classes of securities; and
- · restrictions on our use of leverage and collateral;

each of which may make it difficult for us to consummate an initial business combination.

In addition, we may have imposed upon us burdensome requirements, including:

- registration as an investment company;
- adoption of a specific form of corporate structure; and
- reporting, record keeping, voting, proxy, and disclosure requirements, and other rules and regulations;

compliance with which would reduce the funds we have available outside the trust account to consummate an initial business combination.

In order not to be regulated as an investment company under the 1940 Act, unless we can qualify for an exclusion, we must ensure that we are engaged primarily in an initial business other than investing, reinvesting or trading of securities and that our activities do not include investing, reinvesting, owning, holding or trading "investment securities." Our business will be to identify and consummate a business combination and thereafter to operate the acquired business or businesses for the long term. We do not plan to buy companies with a view to resale or profit from sale of the businesses. We do not plan to buy unrelated businesses or to be a passive investor. We do not believe that our anticipated principal activities will subject us to the 1940 Act. To this end, the proceeds held in the trust account may only be invested by the trustee in U.S. government securities and in assets that are considered "cash items" for purposes of Section 3(a)(2) of the 1940 Act. Pursuant to the trust agreement, the trustee is not permitted to invest in securities or assets that are considered "investment securities" within the meaning of Section 3(a) of the 1940 Act. By restricting the investment of the proceeds to these instruments, and by having a business plan targeted at acquiring, growing and operating a business or businesses for the long term in the education industry (rather than on buying and selling companies in the manner of a merchant bank or private equity fund) we intend to avoid being deemed an "investment company" within the meaning of the 1940 Act. This offering is not intended for persons who are

32

TABLE OF CONTENTS

seeking a return on investments in government securities or investment securities. The trust account and the purchase of government securities for the trust account is intended as a holding place for funds pending the earlier to occur of either: (i) the consummation of our primary business objective, which is a business combination, or (ii) absent a business combination, our dissolution and return of the funds held in the trust account to our public stockholders as part of our plan of dissolution and liquidation. If we do not invest the proceeds as discussed above, we may be deemed to be subject to the 1940 Act. If we were deemed to be subject to the 1940 Act, compliance with these additional regulatory burdens would require additional expense for which we have not accounted.

Our directors may not be considered "independent" under the policies of the North American Securities Administrators Association, Inc. and, therefore, may take actions or incur expenses not deemed to be independently approved or independently determined to be in our best interest.

Under the policies of the North American Securities Administrators Association, Inc., an international organization devoted to investor protection, because each of our directors own shares of our common stock and may receive reimbursement for out-of-pocket expenses incurred by them in connection with activities on our behalf, such as identifying potential target businesses and performing due diligence on suitable business combinations, state securities administrators could take the position such individual is not "independent." If this were the case, they would take the position that we would not have the benefit of independent directors examining the propriety of expenses incurred on our behalf and subject to reimbursement. Additionally, there is no limit on the amount of out-of-pocket expenses that could be incurred and there will be no review of the reasonableness of the expenses by anyone other than our board of directors, which would include persons who may seek reimbursement, or a court of competent jurisdiction if such reimbursement is challenged. To the extent such out-of-pocket expenses exceed the available proceeds not deposited in the trust account, such out-of-pocket expenses would not be reimbursed by us unless we consummate a business combination, in which event this reimbursement obligation would in all likelihood be negotiated with the owners of a target business. Although we believe all actions taken by our directors on our behalf will be in our best interests. whether or not they are deemed to be "independent" under the policies of the North American Securities Administrator Association, we cannot assure you this will actually be the case. If actions are taken, or expenses are incurred that are actually not in our best interests, it could have a material adverse effect on our business and operations and the price of our stock held by the public stockholders.

Because our existing stockholders' initial equity investment was only \$25,000, our offering may be disallowed by state administrators that follow the North American Securities Administrators Association, Inc. Statement of Policy on promotional or development stage companies.

Pursuant to the Statement of Policy Regarding Promoter's Equity Investment promulgated by The North American

Securities Administrators Association, Inc., an international organization devoted to investor protection, any state administrator may disallow an offering of a promotional or development stage company if the initial equity investment by a company's promoters does not equal a certain percentage of the aggregate public offering price. Our existing stockholders' initial investment of \$25,000 is less than the required minimum amount pursuant to this policy (assuming non-exercise of the overallotment option). Accordingly, a state administrator would have the discretion to disallow our offering if it wanted to, in which case we would be prohibited from conducting this offering in that state. We cannot assure you our offering would not be disallowed pursuant to this policy.

Because of our current financial condition, our offering may be disallowed by state administrators following the North American Securities Administrators Association, Inc. Statement of Policy Regarding Unsound Financial Condition.

Pursuant to the Statement of Policy Regarding Unsound Financial Condition promulgated by North American Securities Administrators Association, Inc., any state administrator may disallow an offering if the financial statements of a company contain a footnote or the independent auditor's report contains an explanatory paragraph regarding that company's ability to continue as a going concern and that company has, among other things, an accumulated deficit and no revenues from operations. The report of Eisner LLP, our independent registered accounting firm, contains a going concern explanatory paragraph and we have no revenues

33

TABLE OF CONTENTS

from our operations and an accumulated deficit. Accordingly, a state administrator would have the discretion to disallow our offering if it wanted to, in which case we would be prohibited from conducting this offering in that state. We cannot assure you our offering would not be disallowed pursuant to this policy.

Since we have not currently selected a prospective target business with which to complete a business combination, investors in this offering are unable to currently ascertain the merits or risks of the target business' operations.

Since we have not yet identified a prospective target, investors in this offering have no current basis to evaluate the possible merits or risks of the target business' operations. Accordingly, we may be affected by numerous risks inherent in the business operations of those entities. Although our management will endeavor to evaluate the risks inherent in a particular target business, we cannot assure you we will properly ascertain or assess all of the significant risk factors. We also cannot assure you an investment in our units will not ultimately prove to be less favorable to investors in this offering than a direct investment, if an opportunity were available, in a target business. For a more complete discussion of our selection of a target business, see the section below entitled "Effecting a business combination — We have not identified a target business."

Your only opportunity to evaluate and affect the investment decision regarding a potential business combination will be limited to voting for or against the business combination submitted to our stockholders for approval.

At the time of your investment in us, you will not be provided with an opportunity to evaluate the specific merits or risks of one or more target businesses. Accordingly, your only opportunity to evaluate and affect the investment decision regarding a potential business combination will be limited to voting for or against the business combination submitted to our stockholders for approval. In addition, a proposal you vote against could still be approved if a sufficient number of public stockholders vote for the proposed business combination. Alternatively, a proposal you vote for could still be rejected if a sufficient number of public stockholders vote against the proposed business combination.

Our determination of the offering price of our units and of the aggregate amount of proceeds we are raising in this offering was more arbitrary than would typically be the case if we were an operating company rather than a blank check company.

There was no public market for any of our securities prior to this offering. The public offering price of the units, the terms of the warrants, the aggregate proceeds we are raising and the amount to be placed in trust were the products of a negotiation between the underwriters and us. Factors that were considered in making these determinations included: management's perception of the number of potential competitors that exist to acquire businesses we may find attractive, the financial resources of those potential competitors and, therefore, the potential target size of the businesses they may seek to acquire, as well as management's belief as to the capital required to facilitate a combination with one or more viable target businesses with sufficient scale to operate as a stand-alone public entity. However, although these factors were considered, the determination of our per unit offering price and the aggregate proceeds we are raising in this offering is more arbitrary than the pricing of securities for an operating company in a particular industry since we have no historical operations or financial results that could provide a basis for such determination. There can be no assurance management's perceptions accurately reflect the actual facts with respect to any of the factors considered.

We do not intend to establish an audit or compensation committee until consummation of a business combination.

We currently do not have a compensation or audit committee and do not intend to establish either one prior to consummation of a business combination. Additionally, there will not be a separate committee comprised of some members of our board of directors with specialized accounting and financial knowledge to meet, analyze and discuss

solely financial matters concerning potential target businesses. This may have the result of a less comprehensive discussion among board members with accounting expertise about accounting and financial matters. The lack of a compensation or audit committee may prove to have a material impact on our analysis of potential target businesses which may harm our future operating prospects.

34

TABLE OF CONTENTS

Risks Associated with the Our Acquisition of a Target Business in the Education Industry

We intend to seek a business combination with a company in the education industry. The following risk factors address issues that may arise in connection with the purchase of such a company. Because we have not yet identified a target business, there are likely to be additional risks applicable to any particular target business, and some of the following risks may be inapplicable. Therefore, you should bear in mind that the following risks are illustrative only.

Failure of any acquired schools to comply with the extensive regulatory requirements for school operations could result in financial penalties, restrictions on operations and loss of external financial aid funding, which could affect revenues and impose significant operating restrictions on any business we acquire.

Whether the school has a central location where all students attend classes or whether the curriculum is provided via the Internet, telephone or other remote access, any school we acquire can be expected to be subject to extensive regulation by federal and state governmental agencies and by accrediting commissions. In particular, the Higher Education Act of 1965, as amended, and the regulations promulgated thereunder by the Department of Education, or DOE, set forth numerous standards schools must satisfy to participate in various federal student financial assistance programs under Title IV Programs. To participate in Title IV Programs, schools must receive and maintain authorization by the applicable education agencies in the state in which each school is physically located, be accredited by an accrediting commission recognized by the DOE and be certified as an eligible institution by the DOE. These regulatory requirements can be expected to cover the vast majority of operations of any business we acquire in the education services market, including educational programs, facilities, instructional and administrative staff, administrative procedures, marketing, recruiting, financial operations and financial condition. These regulatory requirements may also affect our ability to acquire or open additional schools, add new educational programs, expand existing educational programs, and change our corporate structure and ownership.

If any acquired schools fails to comply with applicable regulatory requirements, the school and its related main campus and/or additional locations, if any, could be subject to the loss of state licensure or accreditation, the loss of eligibility to participate in and receive funds under the Title IV Programs, the loss of the ability to grant degrees, diplomas and certificates, provisional certification, or the imposition of liabilities or monetary penalties, each of which could adversely affect our revenues and impose significant operating restrictions upon us. The various regulatory agencies periodically revise their requirements and modify their interpretations of existing requirements and restrictions. We cannot predict with certainty how any of these regulatory requirements will be applied or whether each of our schools will be able to comply with these requirements or any additional requirement instituted in the future. Additionally, Congress may change the law or reduce funding for Title IV Programs, which could reduce student population, revenues or profit margin of any business we acquire.

Our success depends in part on the ability of any business we acquire to update and expand the content of existing programs and develop new programs in a cost-effective manner and on a timely basis.

Prospective employers demand that employees possess appropriate technological skills. These skills are becoming more sophisticated in line with technological advancements across all industries. Accordingly, educational programs must keep pace with those technological advancements. The expansion of our existing programs and the development of new programs may not be accepted by our students, prospective employers or the education market. Even if we are able to develop acceptable new programs, we may not be able to introduce these new programs as quickly as our competitors or as quickly as employers demand. If we are unable to adequately respond to changes in market requirements due to financial constraints, unusually rapid technological changes or other factors, our ability to attract and retain students could be impaired, our placement rates could suffer and our revenues could be adversely affected.

We may not be able to retain our key personnel or hire and retain the personnel needed to sustain and grow our business.

The success of any business we acquire will depend largely on the skills, efforts and motivation of our executive officers, including Messrs. Warnock, Hughes, Tomalis and Dr. Crane, who generally have significant experience within the education industry. Our success also depends in large part upon our ability to attract and

35

TABLE OF CONTENTS

sets we seek. In addition, key personnel may leave and subsequently compete against us. The loss of the services of any of our key personnel, or our failure to attract and retain other qualified and experienced personnel on acceptable terms, could have an adverse effect on our ability to operate our business efficiently and execute our growth strategy.

Competition could decrease the market share of any business we acquire, causing it to lower tuition rates and negatively impact our results.

The education market is highly competitive, as we expect any business we acquire to compete for students and faculty with colleges, universities and providers of specialized vocational training and continuing professional education, many of which have greater financial and other resources than we expect to have, which may, among other things, allow our competitors to secure strategic relationships with some or all of our existing strategic partners or develop other high profile strategic relationships or devote more resources to expanding their programs and their school network, all of which could affect the success of our marketing programs. If any business we acquire is unable to compete effectively for students, our student enrollments and revenues will be adversely affected.

Any business we acquire may be required to reduce tuition or increase spending in response to competition in order to retain or attract students or pursue new market opportunities. As a result, its market share, revenues and operating margin may be decreased. We cannot be sure any business we acquire will be able to compete successfully against current or future competitors or that the competitive pressures of such target business will not adversely affect our revenues and profitability.

An increase in interest rates could adversely affect the ability of any business we acquire to attract and retain students.

Interest rates have reached historic lows in recent years, creating a favorable borrowing environment for students. Much of the financing students receive is tied to floating interest rates. However, interest rates have increased recently, resulting in a corresponding increase in the cost to students of financing their education. Higher interest rates can also contribute to higher default rates with respect to student repayment of education loans. Higher default rates may in turn adversely impact eligibility for Title IV Program participation or the willingness of private lenders to make private loan programs available to students who attend any schools we may acquire, which could result in a reduction in student population.

36

TABLE OF CONTENTS

FORWARD-LOOKING STATEMENTS

This prospectus includes forward-looking statements regarding, among other things, our plans, strategies and prospects, both business and financial. All statements other than statements of current or historical fact contained in this prospectus are forward-looking statements. The words "believe," "expect," "anticipate," "should," "would," "could," "plan," "will," "may," "intend," "estimate," "potential," "continue" or similar expressions or the negative of these terms are intended to identify forward-looking statements. Forward-looking statements in this prospectus may include, for example, statements about our:

- ability to complete a business combination;
- success in retaining or recruiting, or changes required in, our officers, key employees or directors following the consummation of a business combination;
- officers and directors allocating their time to other businesses and potentially having conflicts of interest with our business or in approving a business combination as a result of which they would then receive expense reimbursements;
- potential ability to obtain additional financing to complete a business combination:
- pool of prospective target businesses;
- the ability of our officers and directors to generate a number of potential investment opportunities;
- potential change in control of we acquire one or more target businesses for stock;
- · our public securities' potential liquidity and trading;
- use of proceeds not held in the trust account or available to us from interest income on the trust account balance; or
- financial performance following this offering.

We have based these forward-looking statements largely on our current expectations and projections about future events and financial trends that we believe may affect our financial condition, results of operations and business strategy. They can be affected by inaccurate assumptions, including the risks, uncertainties and assumptions described in "Risk Factors" (some of which are beyond our control). In light of these risks, uncertainties and assumptions, the forward-looking statements in this prospectus may not occur and actual results could differ materially from those anticipated or implied in the forward-looking statements. When you consider these forward-looking statements, you should keep in mind the risk factors and other cautionary statements in this prospectus.

Our forward-looking statements speak only as of the date they are made. Neither we nor any other person assumes responsibility for the future accuracy or completeness of these forward-looking statements. Except as required by

TABLE OF CONTENTS

USE OF PROCEEDS

We estimate that the net proceeds of this offering and the private placement and our expected uses will be as set forth in the following table:

·	Without Over-Allotment Option		Over-Allotment Option Exercised		
Gross proceeds					
Gross proceeds from units offered to the public	\$50	0,000,000	\$57	,500,000	
Gross proceeds from units offered in the private placement	\$ 2	2,800,000	\$ 2	2,800,000	
Offering expenses ⁽¹⁾					
Underwriting discount ⁽²⁾	\$ 2	2,000,000	\$ 2	2,300,000	
Deferred underwriting compensation ⁽³⁾	\$ 1,500,000		\$ 1,725,000		
Legal fees and expenses (including blue sky services and expenses)	\$ 270,000 (4)		\$ 270,000 (4)		
Printing and engraving expenses	\$	40,000	\$	40,000	
Accounting fees and expenses	\$	75,000 ⁽⁵⁾	\$	75,000 ⁽⁵⁾	
SEC registration fee	\$	3,292	\$	3,292	
FINRA registration fee	\$	11,222	\$	11,222	
Miscellaneous expenses	\$	486	\$	486	
Net Proceeds					
Held in trust for our benefit	\$48	3,000,000	\$54	,975,000	
Not held in trust	\$	900,000	\$	900,000	
Total net proceeds	\$48,900,000		\$55,875,000		
Total amount held in trust (including deferred underwriting compensation)	\$49	9,500,000	\$56	5,700,000	
Use of net proceeds not held in trust and up to \$600,000 of after tax interest earned on the trust account that may be released to us					
Legal, accounting and other expenses attendant to the structuring and negotiation of a business combination (6)	\$	150,000		10 %	
Due diligence, identification and research of prospective target businesses and reimbursement of out of pocket expenses of management	\$	300,000		20 %	
Payment to Camden Partners Holdings, LLC for administrative services and support (\$7,500 per month for 2 years)	\$	180,000		12 %	
Legal and accounting fees relating to SEC reporting obligations	\$	150,000		10 %	
Working capital to cover miscellaneous expenses, D&O insurance and reserves	\$	720,000		48 %	
Total	\$	1,500,000	_	100 %	

⁽¹⁾ As of September 30, 2007, \$188,644 of the offering expenses have already been paid from loans to us described below, including legal fees.

⁽²⁾ Represents 4% of the gross proceeds from the sale of the units in this offering.

⁽³⁾ Represents 3% of the gross proceeds from the sale of the units in this offering that will be paid to the underwriters only upon consummation of a business combination, less \$0.24 for each share redeemed for cash in connection with our business combination. If a business combination is not consummated and we are liquidated, such amounts will be distributed among our public stockholders.

⁽⁴⁾ A portion of the legal fees payable to Ellenoff Grossman & Schole LLP, our legal counsel, may be deferred and is contingent on our consummating a business combination.

⁽⁵⁾ Approximately \$25,000 in additional accounting fees may be due in connection with the initial public offering, which such additional fees shall be paid, if at all, from our working capital.

(6) These amounts are expected to be paid to legal, accounting and other outside professional firms to assist in negotiating, structuring and documenting a business combination and the preparation and filing of the related proxy statement. Table 0px Font 10pt

\$49,500,000, or \$56,700,000 if the underwriters' over-allotment option is exercised in full, of the proceeds of the offering and the private placement will be placed in a trust account at J.P. Morgan Chase Bank, N.A. maintained by Continental Stock Transfer & Trust Company, as trustee. Of this amount, up to \$1,500,000 (\$1,725,000 if the underwriters' over-allotment option is exercised in full) will be paid to Morgan Joseph & Co. Inc. as deferred underwriting compensation in connection with this offering, if and only if a business combination is consummated. leaving us with \$48,000,000 (\$54,975,000 if the over-allotment option is exercised in full) with which to consummate a business combination. Other than interest income earned on the trust account, which may be released to us as described in the table above, the proceeds of this offering and the private placement held in the trust account will not be released until the earlier of the completion of a business combination or the liquidation of our trust account to our public stockholders as part of our plan of dissolution and liquidation. The proceeds held in the trust account (exclusive of any funds held for the benefit of the Morgan Joseph & Co. Inc., or used to pay public stockholders who have exercised their redemption rights) may be used as consideration to pay the sellers of a target business with which we ultimately complete a business combination or, if there are insufficient funds not held in trust, to pay other expenses relating to such transaction such as reimbursement to insiders for out-of-pocket expenses, third party due diligence expenses or potential finders fees, in each case only upon the consummation of a business combination. In the event there are funds remaining in the trust account after satisfaction of all of such obligations, such funds may be used to finance operations of the target business or to effect other acquisitions, as determined by our board of directors at that time.

We intend to fund our working capital needs as described below, as well as to pay the costs associated with our plan of dissolution and liquidation including reserves for creditors, if we do not consummate a business combination, from the \$900,000 held outside of the trust account and up to \$600,000 of interest earned on the proceeds being held in the trust account (after taxes payable) which may be released to us.

We have agreed to pay a monthly fee of \$7,500 to Camden Partners Holdings, LLC, our sponsor, for general and administrative services, including but not limited to receptionist, secretarial and general office services. This agreement commences on the date of this prospectus and shall continue until the earliest to occur of: (i) the consummation of a business combination, (ii) [_______, 2009] [24 months from the date of this prospectus] and (iii) the date on which we cease our corporate existence in accordance with our amended and restated certificate of incorporation.

Prior to the closing of a business combination, we have agreed to obtain key man life insurance, of which we will be the sole beneficiary, in the amount of \$2,000,000 on the life of David L. Warnock, our President, Chief Executive Officer and Chairman, for a three year period.

We estimate the costs to identify and research prospective target businesses and the costs related to the business combination, including legal and accounting expenses to structure the transaction, prepare the transaction documents and file the related proxy statement, will be approximately \$400,000.

We expect due diligence of prospective target businesses will be performed by some or all of our officers and directors, and may include engaging market research firms and/or third party consultants. Our officers and directors will not receive any compensation for their due diligence of prospective target businesses, but will be reimbursed for any out-of-pocket expenses (such as travel expenses) incurred in connection with such due diligence activities.

It is also possible we could use a portion of such working capital to make a deposit, down payment or fund a "no-shop" provision with respect to a particular proposed business combination, although we do not have any current intention to do so. In the event we were ultimately required to forfeit such funds (whether as a result of our breach of the agreement relating to such payment or otherwise), we may not have a sufficient amount of working capital available outside of the trust account to conduct due diligence and pay other expenses related to finding another suitable business combination without securing additional financing. Thus, if we were unable to secure additional financing, we would most likely fail to consummate a business combination in the allotted time and would be forced to liquidate.

39

TABLE OF CONTENTS

We may not use all of the proceeds in the trust in connection with a business combination, either because the consideration for the business combination is less than the proceeds in trust or because we finance a portion of the consideration with our capital stock or debt securities. In that event, the proceeds held in the trust account as well as any other net proceeds not expended will be released to us and used to finance the operations of the target businesses, which may include subsequent acquisitions.

As of the date of this prospectus, our sponsor has loaned us a total of \$200,000, which was used to pay a portion of the expenses of this offering, such as SEC registration fees, FINRA registration fees, blue sky fees and certain legal and accounting fees and expenses. This loan is payable, with annual interest equal to 4.9%, on the earlier of April 26,

2008 or the consummation of this offering. The loan will be repaid out of the net proceeds of this offering not being placed in trust.

The net proceeds of this offering and the private placement held in the trust account and not immediately required for the purposes set forth above will be invested only in United States "government securities," defined as any Treasury Bill issued by the United States having a maturity of one hundred and eighty days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the 1940 Act so that we are not deemed to be an investment company under the 1940 Act.

Other than the \$7,500 aggregate per month general and administrative service fees described above, no compensation of any kind (including finder's and consulting fees) will be paid by us or any person or entity to any of our existing stockholders, or any of their affiliates, for services rendered to us prior to or in connection with the consummation of the business combination. However, our existing stockholders will receive reimbursement for any out-of-pocket expenses incurred by them in connection with activities on our behalf, such as identifying potential target businesses and performing due diligence on suitable business combinations. Because the role of present management after a business combination is uncertain, we have no ability to determine what remuneration, if any, will be paid to those persons after a business combination.

A public stockholder will be entitled to receive funds from the trust account (including interest earned on his, her or its portion of the trust account, but net of: (i) taxes paid or payable, if any, on interest income earned on the trust account and (ii) up to an aggregate of \$600,000 of interest income released to us to fund our working capital requirements) in the event of the liquidation of our trust account to our public stockholders or in the event a public stockholder were to seek to redeem such shares for cash in connection with a business combination which the public stockholder voted against and which we actually consummate. In no other circumstances will a public stockholder have any right or interest of any kind to or in the trust account.

40

TABLE OF CONTENTS

DILUTION

The difference between the public offering price per share of common stock, assuming no value is attributed to the warrants included in the units, and the pro forma net tangible book value per share of our common stock after this offering constitutes the dilution to investors in this offering. Net tangible book value per share is determined by dividing our net tangible book value, which is our total tangible assets less total liabilities (including the value of common stock which may be redeemed for cash if voted against the business combination), by the number of outstanding shares of our common stock.

At September 30, 2007, our net tangible book value was \$(198,356) or approximately (\$0.13) per share of common stock. After giving effect to the sale of shares of common stock included in the units sold in the offering and the proceeds of the private placement, the deduction of underwriting discounts and estimated expenses of this offering and the private placement, our pro forma net tangible book value (as decreased by the value of 1,874,999 shares of common stock which may be redeemed for cash) net of underwriting costs waived by Morgan Joseph & Co. Inc. related to the 1,874,999 shares that may be redeemed for cash, as of September 30, 2007 would have been \$34,069,795 or \$5.74 per share, representing an immediate increase in net tangible book value of \$5.87 per share to the existing stockholders and an immediate dilution of \$2.26 per share, or 28.25%, to new investors not exercising their redemption rights.

Our pro forma net tangible book value after this offering has been reduced by approximately \$14,849,992 because if we effect a business combination, the redemption rights to the public stockholders may result in the redemption for cash of up to approximately one share less than 30% of the aggregate number of the shares sold in this offering (1,874,999 shares) at a per-share redemption price equal to \$7.92 (plus a portion of the interest earned on the trust account, but net of (i) taxes payable on interest earned and (ii) up to \$600,000 of interest income released to us to fund our working capital).

The following table illustrates the dilution to the new investors on a per-share basis, assuming no value is attributed to the warrants included in the units (actual dilution to investors may be significantly higher as a result of the exercise of these warrants, particularly if cashless exercise is utilized):

Public offering price	•	\$ 8.00
Net tangible book value before this offering	\$ (0.13)	
Increase attributable to new investors	5.87	
Pro forma net tangible book value after this offering		5.74
Dilution to new investors		\$ 2.26

The following table sets forth information with respect to our existing stockholders prior to and after the private placement and the new investors:

	Shares Pu	Shares Purchased ⁽¹⁾		Total Consideration			Average	
	Number Percentage		Amount		Percentage	Pe	Price er Share	
Existing stockholders	1,562,500	20 %	\$	25,000	0.05%	\$	0.016	
New investors ⁽²⁾	6,250,000	80 %	\$ 50	0,000,000	99.95%	\$	8.00	

Total	7,812,500	100 %	\$50,025,000	100%

- (1) Assumes (i) the sale of 6,250,000 units in this offering but not the exercise of 6,250,000 warrants to purchase shares of our common stock sold as part of such units and (ii) no exercise of the underwriters' over-allotment option.
- (2) Does not include 2,800,000 share of common stock issuable upon exercise of the warrants issued in the private placement.

41

TABLE OF CONTENTS

The pro forma net tangible book value after the offering is calculated as follows:

Numerator:

Net tangible book value before the offering and private placement	\$ (198,356)
Excess of face amount of note payable over carrying amount included	(10,712)
above	
Net proceeds from this offering and the private placement	48,900,000
Offering costs excluded from tangible book value before this offering and the private placement	228,855
Less: Proceeds held in trust subject to redemption for cash at \$7.92 per share ⁽¹⁾	(14,849,992)
	\$ 34,069,795
Denominator:	
Shares of common stock outstanding prior to the offering	1,562,500
Shares of common stock included in the units offered	6,250,000
Less: Shares subject to redemption	(1,874,999)
	5,937,501

- (1) Does not reflect deferred underwriting discount and commissions (\$0.24 per share) that may be distributed to public stockholders.
- (2) Does not reflect the possible forfeiture of up to 195,312 shares by our existing stockholders in the event, and to the extent, holders of more than 23.81% of the shares sold in this offering exercise their redemption rights in connection with a business combination.

42

TABLE OF CONTENTS

CAPITALIZATION

The following table sets forth our capitalization at September 30, 2007 and as adjusted to give effect to the sale of our units in this offering and the application of the estimated net proceeds derived from the sale of our units:

3 11	Septemb	er 30	, 2007
	Actual	As	Adjusted
Notes payable ⁽¹⁾	\$ 189,288	\$	_
Common stock, \$.0001 par value, -0- and 1,874,999 shares which are subject to possible redemption, shares at redemption value ⁽²⁾	\$ _	\$14	,849,992
Stockholders' equity:			
Preferred stock, \$.0001 par value, 1,000,000 shares authorized; none issued or outstanding	\$ _	\$	_
Common stock, \$.0001 par value, 20,000,000 shares authorized; 1,562,500 shares issued and outstanding; 5,937,501 shares issued and outstanding (excluding 1,874,999 shares subject to possible redemption), as adjusted	\$ 156	\$	594
Additional paid-in capital	\$ 42,413	\$34	,091,983

Deficit accumulated during the development stage	\$ (12,070	\$ (22,782)
)	
Total stockholders' equity	\$ 30,499	\$34,069,795
Total capitalization	\$ 219,787	\$48,919,787

- (1) The note payable is due on the earlier of April 26, 2008 or the consummation of this offering. The note, in the face amount of \$200,000, is stated at September 30, 2007 in the amount of \$189,288 to reflect the discounted value of the note at an imputed interest rate of 15% per annum. The "As Adjusted" capitalization reflects the payment of the note at its \$200,000 face amount.
- (2) If we consummate a business combination, the redemption rights afforded to our public stockholders may result in the redemption for cash of approximately \$14,849,992 for up to one share less than 30% of the aggregate number of shares sold in this offering, or approximately 1,874,999 shares at a per-share redemption price equal to \$7.92 per share (plus a portion of the interest earned on the trust account, but net of (i) taxes payable on interest earned and (ii) up to \$600,000 of interest income released to us to fund our working capital), which amount includes \$0.24 per share of deferred underwriting discount and commissions. However, in the event holders of more than 20% of the shares sold in this offering elect to redeem their shares, our existing stockholders have agreed to forfeit a number of the initial 1,562,500 shares of our common stock purchased, up to a maximum of 195,312 shares, so that the existing stockholders will collectively own no more than 23.81% (without regard to any purchase of units in this offering, any open market purchases or private purchases of units by the sponsor directly from us, as set forth elsewhere herein) of our outstanding common stock immediately prior to the consummation of such business combination after giving effect to the redemption.

43

TABLE OF CONTENTS

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Camden Learning Corporation is a newly organized blank check company formed for the purpose of merging with, engaging in a capital stock exchange with, purchasing all or substantially all of the assets of, or engaging in any other similar business combination with one or more operating businesses in the education industry.

We intend to utilize cash derived from the proceeds of this offering and the private placement, our capital stock, debt or a combination of cash, capital stock and debt, in effecting a business combination. The issuance of additional capital stock, including upon conversion of any convertible debt securities we may issue, or the incurrence of debt could have material consequences on our business and financial condition. The issuance of additional shares of our capital stock (including upon conversion of convertible debt securities):

- · may significantly reduce the equity interest of our stockholders;
- will likely cause a change in control if a substantial number of our shares of common stock are issued, which
 may affect, among other things, our ability to use our net operating loss carry forwards, if any, and may also
 result in the resignation or removal of one or more of our present officers and directors;
- · may adversely affect prevailing market prices for our common stock; and
- may subordinate the rights of holders of our common stock if preferred stock is issued with rights senior to those afforded to the common stock.

Similarly, if we issued debt securities, it could result in:

- default and foreclosure on our assets if our operating revenues after a business combination were insufficient to pay our debt obligations;
- acceleration of our obligations to repay the indebtedness even if we have made all principal and interest
 payments when due if the debt security contained covenants that required the maintenance of certain financial
 ratios or reserves and any such covenant were breached without a waiver or renegotiation of that covenant;
- our immediate payment of all principal and accrued interest, if any, if the debt security was payable on demand;
- · covenants that limit our ability to acquire capital assets or make additional acquisitions; and
- our inability to obtain additional financing, if necessary, if the debt security contained covenants restricting our ability to obtain additional financing while such security was outstanding.

We have neither engaged in any operations nor generated any revenues to date. Our entire activity since inception has been to prepare for our proposed fundraising through an offering of our equity securities.

We estimate the net proceeds from the sale of the units in this offering and the warrants in the private placement will be \$49,500,000 (\$56,700,000 if the over-allotment option is exercised in full), after deducting offering expenses of approximately \$3,900,000 including underwriting discount and commissions of approximately \$3,500,000 (or \$4,025,000 if the over-allotment option is exercised in full). Of this amount, \$48,000,000, or \$54,975,000 if the underwriters' over-allotment option is exercised in full, will be available to consummate a business combination.

We will use substantially all of the net proceeds of this offering and the private placement, as well as interest on the funds in the trust account available to us, after taxes and net of up to \$600,000 that may be released to us to fund our working capital, to acquire a target business, including identifying and evaluating prospective acquisition candidates, selecting the target business, and structuring, negotiating and consummating the business combination. The proceeds held in the trust account (exclusive of the deferred underwriting fees payable to Morgan Joseph & Co. Inc., and interest thereon, or used to pay public stockholders who have exercised their redemption rights) may be used as consideration to pay the sellers of a target business with which we ultimately complete a business combination or, if there are insufficient funds not held in trust, to

44

TABLE OF CONTENTS

pay other expenses relating to such transaction such as reimbursement to insiders for out-of-pocket expenses, third party due diligence expenses or potential finders fees, in each case only upon the consummation of a business combination. Any amounts not paid as consideration to the sellers of the target business may be used to finance operations of the target business or to effect other acquisitions, as determined by our board of directors at that time. To the extent our capital stock is used in whole or in part as consideration to effect a business combination, the proceeds held in the trust account as well as any other net proceeds not expended will be released to us and will be used to finance the operations of the target business. We believe that, upon consummation of this offering and the private placement, the \$900,000 available to us outside of the trust account and the \$600,000 of the interest earned on funds in the trust account which may be released to us, will be sufficient to allow us to operate for at least the next 24 months, assuming that a business combination is not consummated during that time. Over this time period, we anticipate approximately \$400,000 of expenses for legal, accounting and other expenses attendant to the due diligence investigations, structuring and negotiating of a business combination, \$180,000 for administrative services and support payable to an affiliated third party (up to \$7,500 per month for 24 months), \$150,000 of expenses in legal and accounting fees relating to our SEC reporting obligations and \$720,000 for general working capital that will be used for miscellaneous expenses and reserves including the costs associated with a plan of dissolution and liquidation if we do not consummate a business combination. Although the rate of interest to be earned on the trust account will fluctuate through the duration of the trust account, and although we are unable to state the exact amount of time it will take to complete a business combination, we anticipate the interest that will accrue on the trust account during the time it will take to identify a target and complete an acquisition, together with up to \$900,000 available to us outside the trust account, will be sufficient to fund our working capital requirements. While we cannot assure you the trust account will yield this rate, we believe such rate is representative of that which we may receive.

We believe there should be sufficient funds available either outside of the trust account or made available to us out of the net interest earned on the trust account and released to us as working capital, to fund the costs and expenses associated with a plan of dissolution and liquidation, although we cannot give any assurances thereof. Our sponsor has agreed to indemnify us for these expenses to the extent there are insufficient funds available from the proceeds not held in the trust account and interest released to us, and its members have agreed to make capital contributions to it in order for it to meet this obligation, as described above under "The Offering — Liquidation If No Business Combination."

We do not believe we will need to raise additional funds following this offering and the private placement in order to meet the expenditures required for operating our business prior to a business combination. However, we may need to raise additional funds through a private offering of debt or equity securities if such funds are required to consummate a business combination. We would only consummate such a fundraising simultaneously with the consummation of a business combination.

In seeking a business combination, we intend to utilize cash derived from the proceeds of this offering and the private placement, as well as our capital stock or debt, or a combination of cash, capital stock and debt, and there is no limit on the issuance of capital stock or incurrence of debt we may undertake in effecting a business combination. In the event a business combination is consummated, all sums remaining in the trust account will be released to us immediately thereafter, and there will be no restriction on our use of such funds.

As of the date of this prospectus, our sponsor has loaned us a total of \$200,000, which was used to pay a portion of the expenses of this offering, such as SEC registration fees, FINRA registration fees, blue sky fees and certain legal and accounting fees and expenses. This loan is payable, with annual interest equal to 4.9%, on the earlier of April 26, 2008 or the consummation of this offering. The loan will be repaid out of the net proceeds of this offering not being placed in trust.

We have agreed to sell to the representative of the underwriters, for \$100, an option to purchase up to a total of 625,000 units. The representative's unit purchase option is exercisable on a cashless basis at \$9.60 per unit commencing one year from the date of the prospectus and it expires five years from the date of the prospectus. The units issuable upon exercise of this option are identical to those being sold in this offering except that the exercise price for the warrants included in these units is \$6.71 per share. If the option is

exercised at any time after the fourth anniversary of the date of this prospectus, the holder will only receive the shares comprising the 625,000 units since the warrants will have expired. The option may only be exercised by the option holder and cannot be redeemed for cash by us or the option holder.

The sale of the option will be accounted for as a cost attributable to the proposed offering. Accordingly, there will be no net impact on our financial position or results of operations, except for the recording of the \$100 proceeds from the sale. We have estimated, based upon a Black-Scholes model, that the fair value of the option on the date of sale would be approximately \$2,430,000, using an expected life of five years, volatility of 59.4%, and a risk-free interest rate of 3.08%. However, because our units do not have a trading history, the volatility assumption is based on information currently available to management. We believe the volatility estimate calculated is a reasonable benchmark to use in estimating the expected volatility of our units. The volatility calculation is based on the most recent trading day average volatility of publicly traded companies providing educational services with market capitalizations less than \$500 million. Although an expected life of five years was used in the calculation, if we do not consummate a business combination within the prescribed time period and we automatically dissolve and subsequently liquidate our trust account, the option will become worthless.

We will seek stockholder approval before we effect any business combination, even if the nature of the acquisition would not ordinarily require stockholder approval under applicable state law. In connection with the vote required for any business combination, all of our existing stockholders, including all of our officers and directors, have agreed to vote the shares of common stock owned by them immediately before this offering in accordance with the majority of the shares of common stock voted by the public stockholders. Any shares acquired in the aftermarket by existing stockholders and their designees will be voted in favor of the business combination. We will proceed with a business combination only if a majority of the shares of common stock cast at the meeting are voted in favor of the business combination and public stockholders owning up to one share less than 30% of the shares sold in this offering exercise their redemption rights described herein. Voting against the business combination alone will not result in redemption of a stockholder's shares into a pro rata share of the trust account. Such stockholder must have also exercised its redemption rights described below. Even if less than 30% of the stockholders exercise their redemption rights, we may be unable to consummate a business combination if such redemption leaves us with funds less than an aggregate fair market value equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction which amount is required for our initial business combination. In such event, we may be forced to either find additional financing to consummate such a business combination, consummate a different business combination or liquidate and dissolve.

46

TABLE OF CONTENTS

PROPOSED BUSINESS

Introduction

We are a blank check company organized under the laws of the State of Delaware on April 10, 2007. We were formed for the purpose of merging with, engaging in a capital stock exchange with, purchasing all or substantially all of the assets of, or engaging in any other similar business combination with one or more operating businesses in the education industry. To date, our efforts have been limited to organizational activities and activities relating to this offering and we have not acquired any business operations. Further, we do not have any specific business combination under consideration or contemplation and we have not, nor has anyone on our behalf, contacted any potential target business or had any discussions, formal or otherwise, with respect to such a transaction. At the first meeting of the board of directors promptly following the closing of this offering, we intend to establish policies and procedures for seeking appropriate business acquisition candidates. As part of our intended processes, we may, following the closing of this offering, create a contact database indicating the materials received by any potential target candidates, when such materials were evaluated, the parties primarily responsible for such evaluation and the reasons such candidate was either rejected or the issues that, upon initial evaluation, require further investigation. As the evaluation process progresses, numerous other factors, which are expected to vary with each potential candidate we evaluate, are expected to be relevant to a final determination of whether to proceed with any particular acquisition candidate.

We believe the experience of our officers and directors in the education industry, investment banking and private equity investments will be beneficial in structuring and consummating a business combination. Our management and board of directors have established an extensive network of relationships from which to identify and generate acquisition opportunities within the education industry. David L. Warnock, our President, Chairman and Chief Executive Officer, has over 24 years of investment experience in the education and business and financial services industries. Mr. Warnock serves on the boards of directors of American Public Education, Inc., New Horizons Worldwide, Inc., Nobel Learning Communities, Inc., Primo Water Corporation and Questar Assessment, Inc., formerly Touchstone Applied Science Associates. Mr. Hughes also serves on the boards of directors of New Horizons Worldwide, Inc. and Questar Assessment, Inc. Dr. Therese Crane, our director, was previously President of Jostens Learning Corporation and its successor, Compass Learning and previously was Vice President of Information and Education Products at America Online. Ronald Tomalis, our director, was previously counselor to the US Secretary of Education and Acting Assistant Secretary of Elementary and Secondary Education, William Jews is a former governor of the Federal Reserve Bank and was the President and Chief Executive Officer of CareFirst Inc./CareFirst Blue Cross Blue Shield from 1993 through 2006, an organization with more than \$5 billion in annual revenues. Mr. Jews has previously been a director of MBNA, MuniMae Inc., Nations Bank, Ecolab, Inc. and Crown Central Petroleum, and currently serves on the boards of directors of The Ryland Group, a national home builder and mortgage provider, Choice Hotels International, a

worldwide lodging franchisor and Fortress International Group, Inc., the parent company of Total Site Solutions, which supplies industry and government with secure data centers and other facilities designed to survive terrorist attacks, natural disasters and blackouts. In addition, we believe the experience of our officers and directors in investment banking and private equity investments will be beneficial in structuring and consummating a business combination.

Investment Approach and Focus

Our executive officers and directors have a broad range of education industry experience to assist them in sourcing, evaluating and executing a proposed business combination. We anticipate our search for potential target businesses will involve making contacts with targets through our executive officers and directors; seeking referrals from our professional network of contacts, including management groups, corporations, banks, private equity funds, consultants, investment bankers and business brokers active in the education industry, in addition to contacting owners of education companies we identify.

Education Industry

The U.S. education industry has continued to show substantial growth in the past decade, due to what we believe to be the importance of developing a skilled workforce. A skilled workforce is increasingly reliant on intellectual capital as the U.S. economy continues its shift to become focused on services rather than manufacturing. While post-secondary graduates are approximately 30% of the U.S. population, more than 85% have

47

TABLE OF CONTENTS

completed their K-12 education according to the National Center for Education Statistics report entitled "Digest of Education Statistics: 2005". International competition, especially in math and science, has driven education legislation, requiring minimum performance levels and allocating funding for supplemental services in underperforming schools. In addition to state and government spending, the U.S. has the second highest level of education funding from private sources in the world at 28%, led only by Korea, according to the Organisation for Economic Co-Operation and Development's report entitled "Education at a Glance 2006". These factors have contributed to the overall increase in education spending with total expenditures for education expected to amount to 7.5% of U.S. gross domestic product in 2003-04, which is approximately 0.6 percentage points higher than in 1993-94 according to the National Center for Education Statistics report entitled "Digest of Education Statistics: 2005". Expenditures for public and private education, from kindergarten through graduate school (excluding postsecondary schools not awarding associate or higher degrees), are estimated at \$827 billion for 2003-04 according to the National Center for Education Statistics report entitled "Digest of Education Statistics: 2005". We expect these factors to continue to drive growth across all sectors of the education industry.

We believe this growth has created significant opportunities for companies engaged in the for-profit education industry serving these students. For-profit, four year, Title IV degree granting institutions increased from 80 in 1993-1994 to 350 in 2003-04, while not-for-profit, four year, Title IV degree granting institutions increased from 2,110 in 1993-1994 to 2,180 in 2003-04 according to the National Center for Education Statistic's report entitled "Digest of Education Statistics: 2005". We believe the growth rate in the for-profit sector will continue to outpace non-profit providers. In addition to enrollment in K-12 and post-secondary education, corporate training and early childcare have shown recent growth, after slowdowns following 2000-2001. The information presented in this prospectus is derived from the most recent publicly available data as published by the respective sources.

Although we may consider a target business in any sector of the education industry, we intend to concentrate our search for a business combination in the following target sectors:

- Early Childcare (pre-school programs and/or day care facilities for pre-school aged children);
- K-12 (kindergarten through twelfth grade);
- Post-secondary (a formal instructional program for students who have completed the requirements for a high school diploma or its equivalent, including programs whose purpose is academic, vocational and continuing professional education); and
- Corporate Training.

While we may need to effect a business combination with more than one target business, which may be in different sectors of the education industry, our initial business acquisition must be with one or more operating businesses the aggregate fair market value of which is, either individually or collectively, at least equal to 80% of our net assets at the time of such transaction (less the deferred underwriting discount and commissions and taxes payable). We do not have any specific business combination under consideration, and we have not had any preliminary contacts or discussions with any target businesses regarding a business combination. At the first meeting of the board of directors following the closing of this offering, we intend to establish policies and procedures for seeking appropriate business acquisition candidates.

Prior to completion of a business combination, we will use our reasonable best efforts to have all vendors, prospective target businesses or other entities that we engage execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account for the benefit of our public stockholders. In the event a vendor, prospective target business or other entity were to refuse to execute such a waiver, we will execute an agreement with that entity only if our management first determines that we would be unable to obtain, on a reasonable basis, substantially similar services or opportunities from another entity willing to execute such a waiver. In addition, we

may elect to forego obtaining waivers only if we receive the approval of our Chief Executive Officer and the approving vote or written consent of at least a majority of our board of directors. Examples of possible instances where we may engage a third party that refused to execute a waiver include the engagement of a third party consultant whose particular expertise or

48

TABLE OF CONTENTS

skills are believed by management to be significantly superior to those of other consultants that would agree to execute a waiver or in cases where management is unable to find a provider of required services willing to provide the waiver.

Competitive Advantages

We believe the experience and contacts of our directors, officers will give us an advantage in sourcing, structuring and consummating a business combination. The future role of our key personnel following a business combination, however, cannot presently be fully ascertained. Specifically, the members of our current management are not obligated to remain with us subsequent to a business combination, and we cannot assure you that the resignation or retention of our current management will be included as a term or condition in any agreement relating to a business combination. In addition, despite the competitive advantages we believe we enjoy, we remain subject to significant competition with respect to identifying and executing a business combination.

Established Deal Sourcing Network

Through our management team and our directors, we believe we have extensive contacts and sources from which to generate acquisition opportunities within the education sector. These contacts and sources include those in government, private and public companies within the education industry, private equity and venture capital funds, investment bankers, attorneys and accountants.

For more information regarding our executive officers and directors, please refer to the more detailed disclosure set forth under the heading "Management" below.

Effecting a Business Combination

General

We are not presently engaged in, and we will not engage in, any substantive commercial business for an indefinite period of time following this offering. We intend to utilize cash derived from the proceeds of this offering and the private placement, our capital stock, debt or a combination of these in effecting a business combination. Although substantially all of the net proceeds of this offering are intended to be generally applied toward effecting a business combination as described in this prospectus, the proceeds are not otherwise being designated for any more specific purposes. Accordingly, prospective investors will invest in us without an opportunity to evaluate the specific merits or risks of any one or more business combinations. A business combination may be with a company which does not need substantial additional capital but which desires to establish a public trading market for its shares, while avoiding what it may deem to be adverse consequences of undertaking a public offering itself. These include time delays, significant expense, loss of voting control and compliance with various federal and state securities laws. In the alternative, we may seek to consummate a business combination with a company that may be financially unstable or in its early stages of development or growth.

We Have Not Identified a Target Business

To date, we have not selected a specific target business on which to concentrate our search for a business combination. None of our officers, directors, promoters or other affiliates have had any preliminary contact or discussions on our behalf with representatives of any prospective target business regarding the possibility of a potential merger, capital stock exchange, asset acquisition or other similar business combination with us. Neither we nor any of our agents or affiliates has yet taken any measure, directly or indirectly, to locate a target business. There has been no due diligence, investigation, discussions, negotiations and/or other similar activities undertaken, directly or indirectly, by us, our affiliates or representatives, or by any third party, with respect to a business combination transaction with us.

Sources of Target Businesses

We anticipate target business candidates will be brought to our attention from various unaffiliated sources, including education industry executives, private equity funds, venture capital funds, investment bankers, attorneys and accountants and other members of the financial community, who may present solicited or unsolicited proposals. We expect such sources to become aware that we are seeking a business combination

49

TABLE OF CONTENTS

candidate by a variety of means, such as publicly available information relating to this offering, public relations and marketing efforts, articles that may be published in industry trade papers discussing our intent on making acquisitions,

and/or direct contact by management to be commenced following the completion of this offering. Our existing stockholders, officers and directors as well as their affiliates may also bring to our attention target business candidates. While our officers and directors make no commitment as to the amount of time they will spend trying to identify or investigate potential target businesses, they believe that the various relationships they have developed over their careers together with their direct inquiry of their contacts will generate a number of potential target businesses that will warrant further investigation. While we do not presently anticipate engaging the services of professional firms that specialize in business acquisitions on any formal basis, we may engage these firms in the future, in which event we may pay a finder's fee or other compensation. The terms of any such arrangements will be negotiated with such persons on arm's length basis and disclosed to our stockholders in the proxy materials we provide in connection with any proposed business combination. In no event, however, will we pay any of our existing officers, directors or stockholders or any entity with which they are affiliated any finder's fee or other compensation for services rendered to us prior to or in connection with the consummation of a business combination. In addition, none of our officers, directors or existing stockholders will receive any finder's fee, consulting fees or any similar fees or other compensation from any other person or entity, including any target company, in connection with any business combination other than any compensation or fees to be received for any services provided following such business combination.

Selection of a Target Business and Structuring of a Business Combination

Subject to the requirement that our initial business combination must be with a target business with an aggregate fair market value that is at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction, our management will have virtually unrestricted flexibility in identifying and selecting a prospective target business in the education industry. In evaluating a prospective target business, our management will consider, among other factors, the following:

- · growth potential;
- financial condition and results of operation;
- · capital requirements;
- the value and extent of intellectual property;
- competitive position;
- stage of development of the products, processes or services;
- degree of current or potential market acceptance of the products, processes or services;
- proprietary features and degree of protection of the products, processes or services; and
- costs associated with effecting the business combination.

These criteria are not intended to be exhaustive. Additionally, at the first meeting of the board of directors promptly following the closing of this offering, we intend to establish policies and procedures for seeking appropriate business acquisition candidates. As part of our intended processes, we may, following the closing of this offering, create a contact database indicating the materials received by any potential target candidates, when such materials were evaluated, the parties primarily responsible for such evaluation and the reasons such candidate was either rejected or the issues that, upon initial evaluation, require further investigation. As the evaluation process progresses, numerous other factors, which are expected to vary with each potential candidate we evaluate, are expected to be relevant to a final determination of whether to proceed with any particular acquisition candidate. Any evaluation relating to the merits of a particular business combination will be based, to the extent relevant, on the above factors as well as other considerations deemed relevant by our management in effecting a business combination consistent with our business objective. In evaluating a prospective target business, we will conduct an extensive due diligence review which will encompass, among other things, meetings with incumbent management, where applicable, and inspection of facilities, as well as review of financial and other information which will be made available to us.

In seeking a business combination, we intend to utilize cash derived from the proceeds of this offering and the private placement, as well as our capital stock or debt, or a combination of cash, capital stock and

50

TABLE OF CONTENTS

debt, and there is no limit on the issuance of capital stock or incurrence of debt we may undertake in effecting a business combination. In the event a business combination is consummated, all sums remaining in the trust account will be released to us immediately thereafter, and there will be no restriction on our use of such funds.

We will endeavor to structure a business combination so as to achieve the most favorable tax treatment to us, the target business and both companies' stockholders. We cannot assure you, however, that the Internal Revenue Service or appropriate state tax authorities will agree with our tax treatment of the business combination.

The time and costs required to select and evaluate a target business and to structure and complete the business combination cannot presently be ascertained with any degree of certainty. Any costs incurred with respect to the identification and evaluation of a prospective target business with which a business combination is not ultimately completed will result in a loss to us and reduce the amount of capital available to otherwise complete a business combination. While we may pay fees or compensation to third parties for their efforts in introducing us to a potential target business, in no event, however, will we pay any of our existing officers, directors or stockholders or any entity with which they are affiliated any finder's fee or other compensation for services rendered to us prior to or in connection

with the consummation of a business combination, other than the \$7,500 payable monthly in the aggregate to Camden Partners Holdings, LLC, one of our affiliates, for certain general and administrative services, including but not limited to receptionist, secretarial and general office services. In addition, none of our officers, directors or existing stockholders will receive any finder's fee, consulting fees or any similar fees from any other person or entity in connection with any business combination involving us other than any compensation or fees that may be received for any services provided following such business combination.

Fair Market Value of Target Business

The initial target business that we acquire must have an aggregate fair market value equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction. There is no limitation on our ability to raise funds privately or through loans that would allow us to acquire a target business or businesses with an aggregate fair market value in an amount greater than 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time such transaction. We have not had any preliminary discussions, or made any agreements or arrangements, with respect to financing arrangements with any third party. The fair market value of such business will be determined by our board of directors based upon standards generally accepted by the financial community, such as actual and potential sales, earnings and cash flow and book value, and the price for which comparable businesses have recently been sold. If our board is not able to independently determine the target business has a sufficient fair market value, we will obtain an opinion from an unaffiliated, independent investment banking firm which is a member of the Financial Industry Regulatory Authority, or FINRA, with respect to the satisfaction of such criteria. Such opinion may or may not be relied upon by the stockholders acquiring units in this offering. While we will consider whether such an opinion may be relied on by our stockholders, it will not be dispositive as to which investment bank we seek such opinion from. Other factors contributing to such a determination are expected to include, among others: reputation of the independent investment bank, specifically their knowledge in our particular industry, timing and cost. We will not be required to obtain an opinion from an investment banking firm as to the fair market value of a proposed business combination if our board of directors independently determines the target business has sufficient fair market value. We will not pursue a business combination with any company that is a portfolio company of, or otherwise affiliated with, or has received financial investment from, any of the private equity firms with which our existing stockholders, executive officers or directors are affiliated.

Probable Lack of Business Diversification

While we may seek to effect business combinations with more than one target business, our initial business combination must be with a target business or target businesses which satisfy the minimum valuation standard at the time of such acquisition, as discussed above. Consequently, it is probable we will have the ability to effect only a single business combination, although this may entail the simultaneous acquisition of several compatible operating businesses or assets. Unlike other entities which may have the resources to

51

TABLE OF CONTENTS

complete several business combinations of entities operating in multiple industries or multiple areas of a single industry, it is probable we will not have the resources to diversify our operations or benefit from the possible spreading of risks or offsetting of losses. By consummating a business combination with only a limited number of entities, our lack of diversification may:

- · leave us solely dependent upon the performance of a single business; and
- result in our dependency upon the development or market acceptance of a single or limited number of products or services.

Additionally, since our business combination may entail the simultaneous acquisitions of several assets or operating businesses at the same time and may be with different sellers, we will need to convince such sellers to agree that the purchase of their assets or closely related businesses is contingent upon the simultaneous closings of the other acquisitions.

Limited Ability to Evaluate the Target Business' Management

Although we intend to closely scrutinize the management of prospective target businesses when evaluating the desirability of effecting a business combination, we cannot assure you that our assessment will prove to be correct. In addition, we cannot assure you that new members that join our management following a business combination will have the necessary skills, qualifications or abilities to help manage a public company. Furthermore, the future role of our officers and directors, in any, in the target businesses cannot presently be stated with any certainty. While it is possible that one or more of our officers and directors will remain associated with us in some capacity following a business combination, it is unlikely that any of them will devote their full efforts to our affairs subsequent to a business combination. Moreover, we cannot assure you that our officers and directors will have significant experience or knowledge relating to the operations of the particular target business acquired.

Opportunity for Stockholder Approval of Business Combination

Prior to the completion of a business combination, we will submit the transaction to our stockholders for approval, even if the nature of the acquisition is such as would not ordinarily require stockholder approval under applicable state law. In connection with seeking stockholder approval of a business combination, we will furnish our stockholders with

proxy solicitation materials prepared in accordance with the Securities Exchange Act of 1934, which, among other matters, will include a description of the operations of the target business and certain required financial information regarding the business.

In connection with the vote required for any business combination, all of our existing stockholders, including all of our officers and directors, have agreed to vote their respective shares of common stock owned by them immediately prior to this offering in accordance with the majority of the shares of common stock voted by the public stockholders. Existing stockholders who purchase shares of common stock in this offering or after this offering have agreed to vote such shares in favor of any proposed business combination. Accordingly, they will not be able to exercise redemption rights with respect to a potential business combination. We will proceed with the business combination only if a majority of the shares of common stock cast at the meeting are voted in favor of the business combination, and public stockholders owning less than 30% of the shares sold in this offering exercise their redemption rights. As a result of our higher redemption threshold, we may have less cash available to complete a business combination. Because we will not know how many stockholders may exercise such redemption rights, we will need to structure a business combination that requires less cash, or we may need to arrange third party financing to help fund the transaction in case a larger percentage of stockholders exercise their redemption rights than we expect. Alternatively, to compensate for the potential shortfall in cash, we may be required to structure the business combination, in whole or in part, using the issuance of our stock as consideration. Accordingly, this increase in the customary redemption threshold may hinder our ability to consummate a business combination in the most efficient manner or to optimize our capital structure. Voting against the business combination alone will not result in redemption of a stockholder's shares into a pro rata share of the trust account. Such stockholder must have also exercised its redemption rights described below.

In connection with any proposed business combination we submit to our stockholders for approval, we will also submit to stockholders a proposal to amend our amended and restated certificate of incorporation to

52

TABLE OF CONTENTS

provide for our perpetual existence, thereby removing the limitation on our corporate life tq24 months from the date of this prospectus]. We will only consummate a business combination if stockholders vote both in favor of such business combination and our amendment to provide for our perpetual existence. The approval of the proposal to amend our amended and restated certificate of incorporation to provide for our perpetual existence would require the affirmative vote of a majority of our outstanding shares of common stock.

Redemption Rights

At the time we seek stockholder approval of any business combination, we will offer each public stockholder the right to have such stockholder's shares of common stock redeemed for cash if the stockholder votes against the business combination, elects to redeem its shares of common stock and the business combination is approved and completed. An eligible stockholder may request redemption at any time after the mailing to our stockholders of the proxy statement and prior to the vote taken with respect to a proposed business combination at a meeting held for that purpose, but the request will not be granted unless the stockholder votes against the business combination, elects to redeem its shares of common stock and the business combination is approved and completed. The actual per-share redemption price will be equal to \$7.92 (plus the interest earned on the trust account, net of any income taxes due on such interest and up to \$600,000 of interest income released to us to fund our working capital), divided by the number of shares of common stock sold in this offering. Because the initial per share redemption price is \$7.92 per share (plus any interest, net of taxes payable and amounts disbursed for working capital purposes), which may be lower than the market price of the common stock on the date of the redemption, there may be a disincentive on the part of public stockholders to exercise their redemption rights.

A stockholder who requests redemption of his or her shares must hold these shares from the record date through the closing date of the business combination. In order to ensure accuracy in determining whether or not the redemption threshold has been met, each redeeming stockholder must continue to hold their shares of common stock until the consummation of the business combination. We will not charge redeeming stockholders any fees in connection with the tender of shares for redemption. If a stockholder votes against the business combination but fails to properly exercise his or her redemption rights, such stockholder will not have his or her shares of common stock redeemed for his or her pro rata distribution of the trust account. Any request for redemption, once made, may be withdrawn at any time up to the date of the meeting. It is anticipated the funds to be distributed to stockholders entitled to redeem their shares who elect redemption will be distributed promptly after completion of a business combination. Public stockholders who redeem their stock into their share of the trust account still have the right to exercise the warrants they received as part of the units. We will not complete any business combination if public stockholders owning 30% or more of the shares sold in this offering exercise their redemption rights. Our existing stockholders are not entitled to redeem any shares of common stock held by them whether acquired by them prior to or after this offering. Even if less than 30% of the stockholders, as described above, exercise their redemption rights, we may be unable to consummate a business combination if such redemption leaves us with funds less than an aggregate fair market value equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction, which amount is required for our initial business combination. In such event, we may be forced to either find additional financing to consummate such a business combination, consummate a different business combination or dissolve, liquidate and wind up.

An eligible stockholder may request redemption at any time after the mailing to our stockholders of the proxy statement and prior to the vote taken with respect to a proposed business combination at a meeting held for that

purpose, but the request will not be granted unless the stockholder votes against the business combination, elects to redeem its shares of common stock and the business combination is approved and completed. Additionally, we may require public stockholders, whether they are a record holder or hold their shares in "street name," to either tender their certificates to our transfer agent at any time through the vote on the business combination or to deliver their shares to the transfer agent electronically using Depository Trust Company's DWAC (Deposit/Withdrawal At Custodian) System, at the holder's option. The proxy solicitation materials we will furnish to stockholders in connection with the vote for any proposed business combination will indicate whether we are requiring stockholders to satisfy such certification and delivery requirements. Accordingly, a stockholder would have from the time we send out our proxy statement through the vote on the business combination to tender his shares if he wishes to seek to exercise his redemption rights, a period

53

TABLE OF CONTENTS

that will not be less than 10, nor more than 60, days. This time period varies depending on the specific facts of each transaction. However, as the delivery process can be accomplished by the stockholder, whether or not he is a record holder or his shares are held in "street name," in a matter of hours (because the transfer is made electronically once final instruction is given to Depository Trust Company) by simply contacting the transfer agent or his broker and requesting delivery of his shares through the DWAC System, we believe this time period is sufficient for an average investor. However, because we do not have any control over this process, it may take significantly longer than we anticipated. Additionally, if the shares of common stock cannot be transferred through the DWAC system, the process may take such number of days required to complete the proper paperwork, obtain the necessary authorizations and consents and to locate and deliver physical stock certificates, if any, Traditionally, in order to perfect redemption rights in connection with a blank check company's business combination, a holder could simply vote against a proposed business combination and check a box on the proxy card indicating such holder was seeking to redeem. After the business combination was approved, the company would contact such stockholder to arrange for him to deliver his certificate to verify ownership. As a result, the stockholder then had an "option window" after the consummation of the business combination during which he could monitor the price of the stock in the market. If the price rose above the redemption price, he could sell his shares in the open market before actually delivering his shares to the company for cancellation. Thus, the redemption right, to which stockholders were aware they needed to commit before the stockholder meeting, would become a right of redemption surviving past the consummation of the business combination and which the Company would be obligated to honor until the redeeming holder delivered his certificate. The requirement for physical or electronic delivery prior to the meeting ensures that a redeeming holder's election to redeem is irrevocable once the business combination is approved. There is a nominal cost associated with the abovereferenced tendering process and the act of certificating the shares or delivering them through the DWAC system. The transfer agent will typically charge the tendering broker \$35 and it would be up to the broker whether or not to pass this cost on to the redeeming holder. This fee would be incurred regardless of whether or not we require holders seeking to exercise redemption rights to tender their shares prior to the meeting. The need to deliver shares is a requirement of redemption regardless of the timing of when such delivery must be effectuated. Accordingly, this would not result in any increased cost to stockholders when compared to the traditional process, however, in the event a stockholder elects redemption of their shares of common stock but the proposed business combination is not approved, a stockholder will have paid \$35 to elect redemption but would not actually have their shares of common stock redeemed. Further, it is possible this tendering process will be cost-prohibitive for stockholders in the event their aggregate holdings of our shares of common stock do not exceed \$35.

If a vote on our initial business combination is held and the business combination is not approved, or is not consummated for any other reason, we may continue to try to consummate a business combination with a different target until twenty four months from the date of this prospectus. Public stockholders voting against our initial business combination who exercised their redemption rights would not be entitled to redeem their shares of common stock into a pro rata share of the aggregate amount then on deposit in the trust account in respect of the unconsummated initial business combination. In such case, if we have required public stockholders to tender their certificates prior to the meeting, we will promptly return such certificates to the tendering public stockholder. Public stockholders would be entitled to receive their pro rata share of the aggregate amount on deposit in the trust account only in the event the initial business combination they voted against was duly approved and subsequently completed, or in connection with our liquidation. If a stockholder redeems his shares of common stock, he will still have the right to exercise the warrants received as part of the units purchased in the offering in accordance with the terms hereof. If the proposed business combination is not consummated then a stockholder's shares will not be redeemed into cash, even if such stockholder elected to redeem.

Liquidation If No Business Combination

Our amended and restated certificate of incorporation also provides that we will continue in existence only until ______, 2009 [twenty four months from the date of this prospectus]. This provision may not be amended except in connection with the consummation of a business combination. If we have not completed a business combination by such date, our corporate existence will cease except for the purposes of winding up our affairs and liquidating, pursuant to Section 278 of the Delaware General Corporation Law. This has the same effect as if our board of directors and stockholders had formally voted to approve our

dissolution pursuant to Section 275 of the Delaware General Corporation Law. Accordingly, limiting our corporate existence to a specified date as permitted by Section 102(b)(5) of the Delaware General Corporation Law removes the necessity to comply with the formal procedures set forth in Section 275 (which would have required our board of directors and stockholders to formally vote to approve our dissolution and liquidation and to have filed a certificate of dissolution with the Delaware Secretary of State).

Our existing stockholders have waived their rights to participate in any liquidation of our trust account or other assets with respect to shares of common stock owned by them prior to this offering. In addition, Morgan Joseph & Co. Inc. has agreed to waive their rights to the \$1,500,000 (\$1,725,000 if the underwriters' over-allotment option is exercised in full) of deferred underwriting discount and commissions deposited in the trust account for their benefit. There will be no distribution from the trust account or otherwise in connection with dissolution with respect to our warrants, which will expire worthless. We estimate our total costs and expenses for implementing and completing our liquidation and dissolution will be between \$25,000 and \$40,000. This amount includes all costs and expenses relating to filing our dissolution in the State of Delaware and the winding up of our company. We believe there should be sufficient funds available, outside of the trust account as well as from the interest earned on the trust account and released to us as working capital, to fund the \$25,000 to \$40,000 in costs and expenses.

If we are unable to consummate a business combination and expend all of the net proceeds of this offering and the private placement, other than the proceeds deposited in the trust account, and without taking into account interest, if any, earned on the trust account, the initial per-share liquidation price to the public stockholders would be equal to \$7.92 per share. The proceeds deposited in the trust account could, however, become subject to the claims of our creditors, which could be prior to the claims of our public stockholders. Although we will use our reasonable best efforts to have all vendors, prospective target businesses or other entities we engage execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account for the benefit of our public stockholders, there is no guarantee that they will execute such agreements or even if they execute such agreements that they would be prevented from bringing claims against the trust account including but not limited to fraudulent inducement, breach of fiduciary responsibility or other similar claims, as well as claims challenging the enforceability of the waiver, in each case in order to gain an advantage with a claim against our assets, including the funds held in the trust account. If any third party refused to execute an agreement waiving such claims to the monies held in the trust account, we would perform an analysis of the alternatives available to us if we chose not to engage such third party and evaluate if such engagement would be in the best interest of our stockholders if such third party refused to waive such claims. We may elect to forego obtaining waivers only if we receive the approval of our Chief Executive Officer and the approving vote or written consent of at least a majority of our board of directors. Examples of possible instances where we may engage a third party that refused to execute a waiver include the engagement of a third party consultant whose particular expertise or skills are believed by management to be significantly superior to those of other consultants that would agree to execute a waiver or in cases where management is unable to find a provider of required services willing to provide the waiver. In order to protect the amounts held in the trust account, our sponsor has agreed to indemnify us for claims of any vendors, service providers, prospective target businesses or creditors that have not executed a valid and binding waiver of any right or claim to the amounts in trust account. As further assurance our sponsor will have the necessary funds required to meet these indemnification obligations, (i) the Camden III Funds have agreed, under our sponsor's limited liability company agreement, to make capital contributions to our sponsor as and when required in order for the sponsor to fulfill its indemnification obligations and (ii) our sponsor has agreed to take all such action reasonably necessary to request its members make such capital contributions. Additionally, in the event either of the Camden III Funds undertakes a liquidating distribution while the

55

TABLE OF CONTENTS

indemnification obligations of the sponsor are outstanding, they have each agreed, in our sponsor's limited liability company agreement, to use reasonable efforts to set aside from such distribution adequate reserves to cover the reasonably anticipated liabilities which may be incurred by our sponsor. We and the representative of the underwriters are named as express third party beneficiaries in and with respect to the provisions of our sponsor's limited liability company agreement which require the Camden III Funds to make such capital contributions and establish such reserves. Although we have a fiduciary obligation to pursue the sponsor to enforce its indemnification obligations, and intend to pursue such actions as and when we deem appropriate, there can be no assurance it or the Camden III Funds will be able to satisfy those obligations, if required to do so.

Under the Delaware General Corporation Law, stockholders may be held liable for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. If the corporation complies with certain procedures set forth in Section 280 of the Delaware General Corporation Law intended to ensure that it makes

reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder's pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution. However, as stated above, it is our intention to make liquidating distributions to our stockholders as soon as reasonably possible after [], 2009 [24 months from the date of the prospectus] and, therefore, we do not intend to comply with those procedures. As such, our stockholders potentially could be liable for any claims to the extent of distributions received by them and liability of our stockholders may extend well beyond the third anniversary of such date. Because we will not be complying with Section 280, Section 281(b) of the Delaware General Corporation Law requires us to adopt a plan of dissolution that will provide for our payment, based on facts known to us at such time, of (i) all existing claims, (ii) all pending claims and (iii) all claims that may be potentially brought against us within the subsequent 10 years. However, because we are a blank check company, rather than an operating company, and our operations will be limited to searching for prospective target businesses to acquire, the only likely claims to arise would be from our vendors (such as accountants, lawyers, investment bankers, etc.) or potential target businesses. As described above, we intend to have all vendors and prospective target businesses execute agreements with us waiving any right, title, interest or claim of any kind in or to any monies held in the trust account. As a result, the claims which could be made against us are significantly limited and the likelihood any claim that would result in any liability extending to the trust is minimal.

Additionally, if we are forced to file a bankruptcy case or an involuntary bankruptcy case is filed against us which is not dismissed, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a "preferential transfer" or a "fraudulent conveyance". As a result, a bankruptcy court could seek to recover all amounts received by our stockholders in our dissolution. Furthermore, because we intend to distribute the proceeds held in the trust account to our public stockholders promptly after [], 2009, this may be viewed or interpreted as giving preference to our public stockholders over any potential creditors with respect to access to or distributions from our assets. Additionally, our board may be viewed as having breached their fiduciary duties to our creditors and/or may have acted in bad faith, and thereby exposing itself and our company to claims of punitive damages, by paying public stockholders from the trust account prior to addressing the claims of creditors. We cannot assure you that claims will not be brought against us for these reasons.

Our public stockholders will be entitled to receive funds from the trust account only in the event of our liquidation or if they seek to redeem their respective shares for cash upon a business combination which the stockholder voted against and which is completed by us. In no other circumstances will a stockholder have any right or interest of any kind to or in the trust account.

Certificate of Incorporation

Our amended and restated certificate of incorporation requires that we obtain the affirmative vote of holders of 95% of the shares purchased in this offering to amend certain provisions of our amended and restated certificate of incorporation. However, the validity of such supermajority voting provisions under

56

TABLE OF CONTENTS

Delaware law has not been settled. A court could conclude that such supermajority voting consent requirement constitutes a practical prohibition on amendment in violation of the stockholders' implicit rights to amend the corporate charter. In that case, certain provisions of the certificate of incorporation would be amendable without such supermajority consent and any such amendment could reduce or eliminate the protection afforded to our stockholders. However, we view the foregoing provisions as obligations to our stockholders, and we will not take any action to waive or amend any of these provisions.

Competition for Target Businesses

In identifying, evaluating and selecting a target business, we may encounter intense competition from other entities having a business objective similar to ours. Currently, there are approximately 65 blank check companies with approximately \$9.3 billion in trust that are seeking to carry out a business plan similar to our business plan and there are likely to be more blank check companies filling registration statements for initial public offerings after the date of this prospectus and prior to our completion of a business combination. Additionally, we may be subject to competition from other companies looking to expand their operations through the acquisition of a target business. Many of these entities are well established and have extensive experience identifying and effecting business combinations directly or through affiliates. Many of these competitors possess greater technical, human and other resources than us and our financial resources will be relatively limited when contrasted with those of many of these competitors. While we believe there are numerous potential target businesses we could acquire with the net proceeds of this offering and the private placement, our ability to compete in acquiring certain sizable target businesses will be limited by our available financial resources. This inherent competitive limitation gives others an advantage in pursuing the acquisition of a target business. Further:

- our obligation to seek stockholder approval of a business combination or obtain the necessary financial
 information to be included in the proxy statement to be sent to stockholders in connection with such business
 combination may delay or prevent the completion of a transaction;
- our obligation to redeem for cash shares of common stock held by our public stockholders in certain instances may reduce the resources available to us for a business combination;

- our outstanding warrants and options, and the future dilution they potentially represent, may not be viewed favorably by certain target businesses; and
- the requirement to acquire assets or one or more operating businesses that has an aggregate fair market value
 equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and
 commissions and taxes payable) at the time of such transaction may require us to acquire several assets or
 closely related operating businesses at the same time, all of which sales would be contingent on the closings of
 the other sales, which could make it more difficult to consummate the business combination.

Any of these factors may place us at a competitive disadvantage in negotiating a business combination. Our management believes, however, our status as a public entity and potential access to the United States public equity markets may give us a competitive advantage over privately-held entities having a similar business objective as us in acquiring a target business with significant growth potential on favorable terms.

If we effect a business combination, there will be, in all likelihood, intense competition from competitors of the target business. We cannot assure you that, subsequent to a business combination, we will have the resources or ability to compete effectively.

Facilities

We maintain executive offices at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 and our telephone number is (410) 878-6800. The cost for this space is included in the \$7,500 per-month fee Camden Partners Holdings, LLC charges us for general and administrative services, including but not limited to receptionist, secretarial and general office services, pursuant to a letter agreement between us and Camden Partners Holdings, LLC. This agreement commences on the date of this prospectus and shall continue until the earliest to occur of: (i) consummation of a business combination, (ii) [] [24 months from the date of this prospectus] and (iii) the date on which we cease our corporate existence in accordance with our amended and restated certificate of incorporation. We believe, based on fees for similar services in the greater Baltimore.

57

TABLE OF CONTENTS

Maryland metropolitan area, that the fee charged by Camden Partners Holdings, LLC is at least as favorable as we could have obtained from an unaffiliated person.

We consider our current office space adequate for our current operations.

Employees

We have two executive officers, one of whom is also a member of our Board of Directors. These individuals are not obligated to contribute any specific number of hours per week and intend to devote only as much time as they deem necessary to our affairs. The amount of time they will devote in any time period will vary based on the availability of suitable target businesses to investigate, although we expect such individuals to devote an average of approximately ten hours per week to our business. We do not intend to have any full time employees prior to the consummation of a business combination.

Periodic Reporting and Financial Information

We will register our units, common stock and warrants under the Securities Exchange Act of 1934, as amended, and have reporting obligations, including the requirement that we file annual and quarterly reports with the SEC. In accordance with the requirements of the Securities Exchange Act of 1934, as amended, our annual reports will contain financial statements audited and reported on by our independent accountants.

We will not acquire an operating business if audited financial statements based on United States generally accepted accounting principles cannot be obtained for such target business. Alternatively, we will not acquire assets if the financial information called for by applicable law cannot be obtained for such assets. Additionally, our management will provide stockholders with the foregoing financial information as part of the proxy solicitation materials sent to stockholders to assist them in assessing each specific target business or assets we seek to acquire. Our management believes that the requirement of having available financial information for the target business or assets may limit the pool of potential target businesses or assets available for acquisition.

Legal Proceedings

To the knowledge of our management, there is no litigation currently pending or contemplated against us or any of our officers or directors in their capacity as such.

Comparison to Offerings of Blank Check Companies

The following table compares and contrasts the terms of our offering and the terms of an offering of blank check companies under Rule 419 promulgated by the SEC assuming that the gross proceeds, underwriting discounts and underwriting expenses for the Rule 419 offering are the same as this offering and that the underwriters will not exercise their over-allotment option. None of the terms of a Rule 419 offering will apply to this offering.

Terms of Our Offering Terms Under a Rule 419 Offering	Terms of Our Offering	Terms Under a Rule 419 Offering
---	-----------------------	---------------------------------

\$49,500,000 of the net offering proceeds and private placement proceeds (including up to \$1,500,000 of deferred underwriting discount and commissions payable to Morgan Joseph & Co. Inc. upon consummation of a business combination) will be deposited into a trust account at J.P. Morgan Chase Bank, N.A. maintained by Continental Stock Transfer & Trust Company.

\$40,000,000 of the offering proceeds would be required to be deposited into either an escrow account with an insured depositary institution or in a separate bank account established by a broker-dealer in which the broker-dealer acts as trustee for persons having the beneficial interests in the account.

58

TABLE OF CONTENTS

Investment of Net Proceeds

Terms of Our Offering
The \$49,500,000 of net offering
proceeds and the private placement
proceeds held in trust will only be
invested in U.S. "government securities,"
within the meaning of Section 2(a)(16) of
the Investment Company Act of 1940
with a maturity of one hundred and eighty
days or less or money market funds
meeting certain criteria.

Limitation on Fair Value or Net Assets of Target Business

Trading of Securities Issued The initial target business or businesses that we acquire must have an aggregate fair market value equal to at least 80% of the amount in our trust account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction.

The units shall commence trading on or promptly after the date of this prospectus. The common stock and warrants comprising the units shall begin to trade separately on the 90th day after the date of this prospectus unless Morgan Joseph & Co. Inc. informs us of its decision to allow earlier separate trading, provided, however, that in no event will Morgan Joseph & Co. Inc. allow separate trading of the common stock and warrants until the business day after (i) we have filed with the SEC a Current Report on Form 8-K, which includes an audited balance sheet reflecting our receipt of the proceeds of this offering and the private placement, including any proceeds we receive from the exercise of the over-allotment option, if such option is exercised on the date of this prospectus, (ii) we file a Current Report on Form 8-K and issue a press release announcing when such separate trading will begin and (iii) the underwriters' over-allotment option has either expired or been exercised in full. Morgan Joseph & Co. Inc. may decide to allow continued trading of the units following such separation.

Terms Under a Rule 419 Offering

Proceeds could be invested only in specified securities such as a money market fund meeting conditions of the Investment Company Act of 1940 or in securities that are direct obligations of, or obligations guaranteed as to principal or interest by, the United States.

We would be restricted from acquiring a target business unless the fair value of such business or net assets to be acquired represent at least 80% of the maximum offering proceeds.

No trading of the units or the underlying common stock and warrants would be permitted until the completion of a business combination. During this period, the securities would be held in the escrow or trust account.

The warrants cannot be exercised until the later of the completion of a business combination or one year from the date of this prospectus and, accordingly, will only be exercised after the trust account has been terminated and distributed. The warrants could be exercised prior to the completion of a business combination, but securities received and cash paid in connection with the exercise would be deposited in the escrow or trust account.

59

TABLE OF CONTENTS

Terms of Our Offering

Election to Remain an Investor

We will give our stockholders the opportunity to vote on the business combination. In connection with seeking stockholder approval, we will send each stockholder a proxy statement containing information required by the SEC. A stockholder following the procedures described in this prospectus is given the right to redeem his or her shares for \$7.92 per share (plus a portion of the interest earned on the trust account, but net of (i) taxes payable on interest earned and (ii) up to \$600,000 of interest income released to us to fund our working capital). However, a stockholder who does not follow these procedures or a stockholder who does not take any action would not be entitled to the return of any funds. Interest will be payable to public stockholders redeeming in connection with a business combination pro rata, net of amounts previously released to us and taxes payable. A business combination must occur within 24 months after the date of this prospectus. If a business combination does not occur in such timeframe, we

Combination Deadline

Business

Release of Funds

The proceeds held in the trust account will not be released until the earlier of the completion of a business combination or upon our failure to complete a business combination within the allotted time except that to the extent the trust account earns interest we are permitted from time to time to receive disbursements of that interest for the purposes of (i) paying taxes on interest earned and (ii) funding working capital up to \$600,000.

will liquidate and return the amounts in

trust to our public stockholders.

Terms Under a Rule 419 Offering

A prospectus containing information required by the SEC would be sent to each investor. Each investor would be given the opportunity to notify the company, in writing, within a period of no less than 20 business days and no more than 45 business days from the effective date of the post-effective amendment, to decide whether he or she elects to remain a stockholder of the company or require the return of his or her investment. If the company has not received the notification by the end of the 45th business day, funds and interest or dividends, if any, held in the trust or escrow account would automatically be returned to the stockholder. Unless a sufficient number of investors elect to remain investors, all of the deposited funds in the escrow account must be returned to all investors and none of the securities will be issued.

If a business combination has not been consummated within 18 months after the effective date of the initial registration statement, funds held in the trust or escrow account would be returned to investors.

The proceeds held in the trust account, including all of the interest earned thereon (after taxes payable) would not be released until the earlier of the completion of a business combination or the failure to effect a business combination within 18 months. See "Risk Factors — Risks associated with our business — You will not be entitled to protections normally afforded to investors of blank check companies." In the event a business combination was not consummated within 18 months. proceeds held in the trust account would be returned within 5 business days of such date.

Interest Earned on Funds in Trust

Up to \$600,000 of the interest earned on the trust account may be released to us to fund our working capital requirements. In addition, interest earned may be disbursed for the purpose of paying taxes on interest earned.

The interest earned on proceeds held in trust (after taxes payable) would be held for the sole benefit of investors, and we would be unable to access such interest for working capital purposes.

60

TABLE OF CONTENTS

MANAGEMENT

Directors and Executive Officers

Our current directors and executive officers are listed below. None of such persons are, or have been, involved with any other blank check companies.

Name	Age	Position
David L. Warnock	49	President, Chief Executive Officer and Chairman
Donald W. Hughes	56	Chief Financial Officer, Secretary
Therese Kreig Crane, Ed.D	57	Director
Ronald Tomalis	44	Director
William Jews	55	Director

David L. Warnock is a partner with Camden Partners and co-founded the firm in 1995. He has over 24 years of investment experience and focuses on investments in the education and business and financial services sectors. He serves on the boards of directors of American Public Education, Inc., a regionally accredited online post-secondary university, New Horizons Worldwide, Inc., one of the largest global IT training companies, Nobel Learning Communities, Inc., a nationwide provider of pre-K through 8th grade private schools and Questar Assessment, Inc., formerly Touchstone Applied Science Associates which provides testing and assessment services for standardized testing, all of which are Camden Partners' portfolio companies. Mr. Warnock served as the Chairman of Nobel from September 2003 through February 2004. Mr. Warnock has previously served on the boards of Concord Career Colleges from 1997 thru 2006 and Children's Comprehensive Services, Inc. from 1993 to 2000. Previously, Mr. Warnock was President of T. Rowe Price Strategic Partners and T. Rowe Price Strategic Partners II. He was also comanager of the T. Rowe Price New Horizons Fund. Mr. Warnock was employed by T. Rowe Price Associates from 1983 to 1995. Upon forming Camden Partners (formerly known as Cahill, Warnock & Company) and until December 31, 1997, Mr. Warnock served as a consultant to the advisory committees of T. Rowe Price Strategic Partners and T. Rowe Price Strategic Partners II.

Mr. Warnock is also involved with numerous non-profit organizations. He is the Chairman of the Center for Fathers, Families, and Workforce Development, as well as Calvert Education Services, the nation's largest non-sectarian homeschooling organization. He also serves on the board of the National Alliance to End Homelessness and the University of Wisconsin Applied Security Analysis Program and is a trustee on the board of the Baltimore Museum of Art. Mr. Warnock earned a B.A. degree from the University of Delaware and a M.S. (in Finance) from the University of Wisconsin. He is a CFA Charterholder.

Donald W. Hughes has been our Chief Financial Officer and Secretary since inception. Since February 1997, Mr. Hughes has served as Executive Vice President and Chief Financial Officer of Camden Partners, Inc. and a member of and Chief Financial Officer of Camden Partners Holdings, LLC, each of which is an affiliate of Camden Learning, LLC, Camden Partners Strategic Fund III-A, L.P. Prior to joining Camden in February 1997, Mr. Hughes served as Vice President, Chief Financial Officer and Secretary of Capstone Pharmacy Services, Inc. from December 1995 and as Executive Vice President and Chief Financial Officer of Broventure Company, Inc., a closely-held investment management company, from July 1984 to November 1995. Mr. Hughes serves on the boards of directors of Questar Assessment, Inc., New Horizons Worldwide, Inc. and the Maryland Food Bank. Mr. Hughes received a B.A. from Lycoming College and an M.S.F. from Loyola College in Maryland, and is a Certified Public Accountant.

Therese Kreig Crane, Ed.D, currently serves in various leadership capacities within the education industry, including as a trustee for the National Education Association Foundation (2003 – present) and the Western Governors University (2001 – present), as Chairman of the Board of Directors of Nobel Learning Communities Inc. (2004 – present) and as a director of Questia Media, Inc. (2001 – present) and Tutor.com.(2005 – present). From 2003 until June 2005, Dr. Crane served on the board of AlphaSmart, a provider of affordable, portable personal learning solutions for the K-12 classroom. In August, 2003, she formed Crane Associates as a sole proprietorship, engaged in the educational technology consulting practice, advising educational technology companies in business strategy, marketing, and sales. Dr. Crane was engaged as a retained consultant by

Infotech Strategies in 2003 and currently serves as the Senior Education Advisor. From 2000 to 2003, Dr. Crane was Vice President, Information and Education Products at America Online. Prior to that, she was President of Jostens Learning Corporation and its successor company, Compass Learning. Dr. Crane also held various positions with Apple Computer, including Senior Vice President, Education of Americas, and was a corporate officer as Apple Computer's Senior Vice President, Worldwide Strategic Market Segments. Dr. Crane started her career as an elementary school classroom teacher. Dr. Crane has a B.S. in elementary education and mathematics from the University of Texas at Austin, an M.Ed. in early childhood education, and an Ed.D. in administrative leadership from the University of North Texas.

Ronald Tomalis is a director, owner and co-founder of The Chartwell Educational Group, an international education consulting firm that serves private, non profit and governmental organizations focusing on pre-K, K-12 and post-secondary education. He has served as a director since July, 2005. Mr. Tomalis advises education companies, non profit organizations, and domestic and international education organizations/agencies on areas of education policy, finance, governance, and management. Mr. Tomalis also served as a director of ELLIS, Inc from 2005 through 2006. From August 2004 to July 2005, Mr. Tomalis was an independent consultant. From June 2001 to August 2004 Mr. Tomalis held various senior positions in the United States Department of Education, including managing the implementation of the No Child Left Behind Law as well as the \$25 billion Title I/II programs. Mr. Tomalis also served as counselor to the United States Secretary of Education and as Acting Assistant Secretary of Elementary and Secondary Education. For six years prior to joining the United States Department of Education, Mr. Tomalis was the Executive Deputy Secretary of Education for the Commonwealth of Pennsylvania. He was appointed to the position by GovernorTomRidge in December of 1995. As Executive Deputy Secretary for Education for the Commonwealth of Pennsylvania, he took on the role of Chief Operating Officer for that department. He was also the principal policy advisor to the Pennsylvania Secretary of Education and spearheaded many of the reform initiatives proposed by Governor Ridge. Mr. Tomalis graduated from DickinsonCollege with a degree in political science.

William Jews retired in December 2006. Prior to such time, Mr. Jews was the President and Chief Executive Officer of CareFirst Inc./CareFirst Blue Cross Blue Shield from 1993 through 2006. With more than \$5 billion in annual revenues, CareFirst and its affiliates and subsidiaries are a combination of not-for-profit and for-profit entities with nearly 3 million customers, including the nation's largest federal health program, served by 6,300 associates in five states and the District of Columbia. From 1990 through 1993, Mr. Jews was the President and Chief Executive Officer of Dimensions Health Corporation, a multi-faceted health care corporation which included two acute care hospitals, a for-profit and not-for-profit nursing home and an emergency ambulatory/surgical center. Mr. Jews currently serves on the boards of directors of The Ryland Group, a national home builder and mortgage provider and Fortress International Group, Inc., the parent company of Total Site Solutions, which supplies industry and government with secure data centers and other facilities designed to survive terrorist attacks, natural disasters and blackouts. He also serves on the board of Choice Hotels International, a worldwide lodging franchisor, including serving on the Nominating/Governance and Diversity committees. He has previously been a director of Ecolab, Inc., MBNA, MuniMae Inc., Nations Bank and Crown Central Petroleum and is a former governor of the Federal Reserve Bank. Mr. Jews received a B.A. in Social and Behavioral Science from The Johns Hopkins University and a Masters in Urban Planning and Policy Analysis, with Health Administration emphasis from MorganStateUniversity, Baltimore, MD.

Our board of directors is divided into two classes with only one class of directors being elected in each year and each class serving a two-year term. The term of office of the first class of directors, consisting of Messrs. Warnock and Tomalis, will expire at our first annual meeting of stockholders. The term of office of the second class of directors, consisting of Ms. Crane and Mr. Jews, will expire at the second annual meeting.

These individuals will play a key role in identifying and evaluating prospective acquisition candidates, selecting the target business and structuring, negotiating and consummating its acquisition. None of these individuals has been a principal of or affiliated with a public company or blank check company that executed a business plan similar to our business plan and none of these individuals is currently affiliated with such an entity. However, we believe the skills and expertise of these individuals, their collective access to acquisition opportunities and ideas, their contacts, and their transaction expertise with public and private companies should enable them to identify and effect an acquisition although we cannot assure you that they will, in fact, be able to do so.

62

TABLE OF CONTENTS

In addition, for a period of no less than two years after the date of the prospectus, we have granted Morgan Joseph & Co. Inc. the right to have an observer present at all meetings of our board of directors until we consummate a business combination. The observer shall be entitled to attend meetings of the board, receive all notices and other correspondence and communications sent by us to members of our board of directors. In addition, such observer shall be entitled to receive, as his/her sole compensation, reimbursement for all costs incurred in attending such meetings.

In order to protect the amounts held in the trust account, our sponsor has agreed to indemnify us for claims of creditors that have not executed a valid and binding waiver of their right to seek payment of amounts due to them out of the trust account. The sole owners and members of our sponsor are Camden Partners Strategic Fund III, L.P. (96.01% ownership of the sponsor) and Camden Partners Strategic Fund III-A, L.P. (3.99% ownership of the sponsor). As further assurance our sponsor will have the necessary funds required to meet these indemnification obligations, (i) the Camden III Funds have agreed, under our sponsor's limited liability company agreement, to make capital contributions to our sponsor as and when required in order for the sponsor to fulfill its indemnification obligations and (ii) our sponsor

has agreed to take all such action reasonably necessary to request its members make such capital contributions. Additionally, in the event either of the Camden III Funds undertakes a liquidating distribution while the indemnification obligations of the sponsor are outstanding, they have agreed, in our sponsor's limited liability company agreement, to use reasonable efforts to set aside from such distribution, adequate reserves to cover the reasonably anticipated liabilities which may be incurred by our sponsor. We and the representative of the underwriters are named as express third party beneficiaries in and with respect to the provisions of our sponsor's limited liability company agreement which require the Camden III Funds to make such capital contributions and establish such reserves. Although we have a fiduciary obligation to pursue the sponsor to enforce its indemnification obligations, and intend to pursue such actions as and when we deem appropriate, there can be no assurance it or the Camden III Funds will be able to satisfy those obligations, if required to do so.

The Camden III Funds were formed in February 2004 and do not terminate until February 2014, unless earlier dissolved. The Camden III Funds are required by their terms to act together with respect to, among other things, dissolution. The Camden III Funds may be dissolved upon any of the following events: (i) at any time limited partners representing 80% in interest of the Camden III Funds taken together vote to dissolve the Camden III Funds upon 90 days prior written notice, (ii) the general partner and a majority in interest of the limited partners of the Camden III Funds upon 90 days prior written notice or (iii) 66-2/3% in interest of the limited partners of the Camden III Funds vote to dissolve the partnership after the general partner or any principal of either partnership (A) is convicted of a felony or securities law violation or has entered into a plea agreement for a securities law violation, in each case in connection with the activities of such partnership or (B) commits an act of bad faith, fraud or gross negligence in connection with the activities of such partnership or breaches a material term of such partnership agreement and such breach is not remedied within 30 days after receipt of notice of such breach. At dissolution, the Camden III Fund's business shall be liquidated in an orderly and timely manner.

Each of the Camden III Funds is owned directly by their respective general and limited partners. The general partner of each Camden III Fund is Camden Partners Strategic III, LLC and the managing member of Camden Partners Strategic III, LLC is Camden Partners Strategic Manager, LLC. David L. Warnock, Donald W. Hughes, Richard M. Johnston and Richard M. Berkeley are the managing members of Camden Partners Strategic Manager, LLC. The limited partners of each Camden III Fund are numerous institutional investors and high net worth individuals. Dispositive and voting power of our securities held by each Camden III Fund is vested solely in its general partner. The Camden III Funds are exempt from registration under the Investment Company Act of 1940 pursuant to either Section 3(c)(1) or 3(c)(7) of such Act and have informed us they expect to continue to qualify for such exemption following this offering.

Additionally, our sponsor has agreed to indemnify and hold us harmless against any and all loss, liability, damage and expense whatsoever (including, but not limited to, any and all legal or other expenses) reasonably incurred in our dissolution and liquidation, and in investigating, preparing or defending against any litigation, whether pending or threatened, or any claim whatsoever) to which we may become subject as a result of our liquidation and dissolution, but only to the extent there are not available funds outside of the trust account sufficient to consummate our dissolution and liquidation.

63

TABLE OF CONTENTS

Board Committees

Our board of directors intends to establish an audit committee and a compensation committee upon consummation of a business combination. At that time our board of directors intends to adopt charters for these committees. Prior to such time we do not intend to establish either one. Accordingly, there will not be a separate committee comprised of some members of our board of directors with specialized accounting and financial knowledge to meet, analyze and discuss solely financial matters concerning potential target businesses. We do not feel a compensation committee is necessary prior to a business combination as there is no salary, fees or other compensation being paid to our officers or directors prior to a business combination other than as disclosed in this prospectus. See "Risk Factor — We do not intend to establish an audit or compensation committee until consummation of a business combination" on page 34.

Code of Conduct

We have adopted a code of conduct and ethics applicable to our directors, officers and employees in accordance with applicable federal securities laws.

Executive Compensation

No executive officer has received any cash compensation for services rendered. No compensation of any kind, including finder's and consulting fees, will be paid to any of our existing stockholders, including our officers and directors, or any of their respective affiliates, for services rendered prior to or in connection with a business combination. However, these individuals will be reimbursed for any out-of-pocket expenses incurred in connection with activities on our behalf such as identifying potential target businesses and performing due diligence on suitable business combinations. There is no limit on the amount of these out-of-pocket expenses and there will be no review of the reasonableness of the expenses by anyone other than our board of directors, which includes persons who may seek reimbursement, or a court of competent jurisdiction if such reimbursement is challenged. If all of our directors are not deemed "independent," we will not have the benefit of independent directors examining the propriety of expenses incurred on our behalf and subject to reimbursement.

We maintain executive offices at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 and our telephone number is (410) 878-6800. The costs for this space is included in the \$7,500 per-month fee Camden Partners Holdings, LLC charges us for general and administrative services, including but not limited to receptionist, secretarial and general office services, pursuant to a letter agreement between us and Camden Partners Holdings, LLC. We believe, based on fees for similar services in the greater Baltimore, Maryland metropolitan area, that the fee charged by Camden Partners Holdings, LLC is at least as favorable as we could have obtained from an unaffiliated person.

Conflicts of Interest

Potential investors should be aware of the following potential conflicts of interest:

- None of our officers or directors is required to commit their full time to our affairs and, accordingly, they may
 have conflicts of interest in allocating management time among various business activities.
- In the course of their other business activities, our officers and directors may become aware of investment and business opportunities which may be appropriate for presentation to us as well as the other entities with which they are affiliated. They may have conflicts of interest in determining to which entity a particular business opportunity should be presented. For a complete description of our management's other affiliations, see the previous section entitled "Directors and Executive Officers."
- Our officers and directors are currently, and may in the future become, affiliated with entities, including other blank check companies, engaged in business activities similar to those intended to be conducted by us.
- Since our directors own shares of our common stock which will be released from escrow only in certain limited situations, and certain of them and their designees are purchasing warrants in the private placement as to which they (as well as our existing stockholders) are waiving their redemption and liquidation rights, our board may have a conflict of interest in determining whether a

64

TABLE OF CONTENTS

particular target business is appropriate to effect a business combination. The personal and financial interests of our directors and officers may influence their motivation in identifying and selecting a target business and completing a business combination timely.

Our sponsor has informed us it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Our sponsor has agreed to vote any such shares of common stock in favor of our initial business combination, representing a possible maximum aggregate of 3.4% of the public shares entitled to vote on the business combination. Our sponsor has entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Such open market purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price per share of not more than the per share amount held in the trust account (less taxes payable) as reported in such 8-K and will be made by a broker-dealer mutually agreed upon by our sponsor and the representative of the underwriters in such amounts and at such times as such broker-dealer may determine, in its sole discretion, so long as the purchase price does not exceed the above-referenced per share purchase price. Our sponsor has agreed to vote any such shares of common stock purchased in the open market in favor of our initial business combination, representing a possible maximum aggregate of 8% of the public shares entitled to vote on the business combination. Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell such shares, provided it will be entitled to participate in any liquidating distributions with respect to the shares purchased in the open market. In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our sponsor has agreed to purchase from us in a private placement a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this co-investment. This co-investment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders.

In general, officers and directors of a corporation incorporated under the laws of the State of Delaware are required to present business opportunities to a corporation if:

- the corporation could financially undertake the opportunity;
- the opportunity is within the corporation's line of business; and
- it would not be fair to the corporation and its stockholders for the opportunity not to be brought to the attention of the corporation.

As a result of multiple business affiliations, our officers and directors may have similar legal obligations relating to presenting business opportunities meeting the above-listed criteria to multiple entities. In addition, conflicts of interest may arise when our board evaluates a particular business opportunity with respect to the above-listed criteria. We cannot assure you any of the above mentioned conflicts will be resolved in our favor.

Each of our officers and directors has pre-existing fiduciary obligations to other businesses of which they are officers

or directors. To the extent they identify business opportunities which may be suitable for the entities to which they owe a pre-existing fiduciary obligation, our officers and directors will honor those fiduciary obligations, subject to the "right of first refusal" described below. Accordingly, they may not present opportunities to us that otherwise may be attractive to us unless the entities to which they owe a pre-existing fiduciary obligation and any successors to such entities have declined to accept such opportunities.

As set forth herein, certain of our directors and officers are directors of companies, both public and private, which may perform business activities in the education industry similar to those which we may perform after consummating a business combination. In order to minimize potential conflicts of interest which may arise from these multiple entity affiliations, each of our officers and directors has agreed, until the earlier

65

TABLE OF CONTENTS

of a business combination, our dissolution and liquidation to our public stockholders of the trust account or such time as he or she ceases to be an officer or director, to present to us for our consideration, prior to presentation to any other entity, any business opportunity which may reasonably be required to be presented to us under Delaware law (as described above), subject to any pre-existing fiduciary or contractual obligations he or she has and the "right of first refusal" described below. We have also adopted a code of ethics that obligates our directors, officers and employees to disclose potential conflicts of interest and prohibits those persons from engaging in such transactions without our consent. At the first meeting of the board of directors promptly following the closing of this offering, we intend to establish policies and procedures for seeking appropriate business acquisition candidates. As part of our intended processes, we may, following the closing of this offering, create a contact database indicating the materials received by any potential target candidates, when such materials were evaluated, the parties primarily responsible for such evaluation and the reasons such candidate was either rejected or the issues that, upon initial evaluation, require further investigation. As the evaluation process progresses, numerous other factors, which are expected to vary with each potential candidate we evaluate, are expected to be relevant to a final determination of whether to move forward with any particular acquisition candidate.

David L. Warnock serves on the boards of directors of American Public Education, Inc. and Nobel Learning Communities, Inc., and both he and Mr. Hughes serve on the boards of directors of New Horizons Worldwide, Inc. and Questar Assessment, Inc., formerly Touchstone Applied Science Associates, all of which are portfolio companies of one or both of the Camden III Funds in the education industry. Both Messrs. Warnock and Hughes have a pre-existing fiduciary duty to each of these companies and may not present opportunities to us that otherwise may be attractive to us unless these entities have declined to accept such opportunities.

Messrs. Warnock and Hughes also have fiduciary obligations to the Camden III Funds. The Camden III Funds are private equity funds focused on investing in micro-cap public and, to a lesser extent, late stage private companies, in the business and financial services, healthcare and education industries. In order to minimize potential conflicts, or the appearance of conflicts, which may arise from the affiliations that Messrs. Warnock and Hughes have with the Camden III Funds, the Camden III Funds have granted us a "right of first refusal" with respect to an acquisition of voting control of any company or business in the education industry whose aggregate fair market value is at least equal to 80% of the balance of the trust account (less the deferred underwriting discounts and commissions and taxes payable), which is the minimum size of a target business for our initial business combination. Pursuant to this right of first refusal, each of the Camden III Funds has agreed that it will present any investment or purchase opportunity in a company meeting these criteria to a committee of our independent directors for our review and that it will not enter into any agreement to purchase or invest in such company until our committee of independent directors has had a reasonable period of time to determine whether or not to pursue such opportunity. This right of first refusal will expire upon the earlier of (i) our consummation of an initial business combination or (ii) 24 months after the consummation of this offering. Furthermore, we have agreed that any target company with respect to which either of the Camden III Funds has initiated any contacts or entered into any discussions, formal or informal, or negotiations regarding such company's acquisition prior to the completion of this offering will not be a potential acquisition target for us, unless such Camden III Fund declines to pursue an investment in such company.

Therese Kreig Crane, Ed.D, currently serves as a trustee for the National Education Association Foundation and the Western Governors University, as Chairman of the Board of Directors of Nobel Learning Communities Inc. and as a director of Questia Inc. and Tutor.com. Ronald Tomalis is a director and owner of The Chartwell Educational Group. While we do not know if any of these entities will be competitive with us because we do not know if any of these entities would consider the business combination we would seek, each of these directors has a pre-existing fiduciary duty to each of these companies and may not present opportunities to us that otherwise may be attractive to us unless these entities have declined to accept such opportunities.

We will not pursue a business combination with any company that is a portfolio company of, or otherwise affiliated with, or has received financial investment from, any of the private equity firms with which our existing stockholders, executive officers or directors are affiliated.

In connection with the vote required for any business combination, all of our existing stockholders, including all of our officers and directors, have agreed to vote their respective shares of common stock in the same manner as a majority of the public stockholders who vote at the special or annual meeting called for the purpose of approving a business combination. In addition, all of our existing stockholders, and the purchasers of our securities in the private placement, have agreed to waive their respective rights to participate in any liquidation of our trust account (except with respect to shares of our common stock acquired by them in connection with this offering or in the aftermarket) in connection with a dissolution occurring upon our failure to consummate a business combination.

67

TABLE OF CONTENTS

PRINCIPAL STOCKHOLDERS

The following table sets forth information as of the date of this prospectus regarding the beneficial ownership of our common stock: (a) before the offering and after the private placement and (b) after the offering and private placement, to reflect the sale of warrants sold in the private placement and the units offered by this prospectus for:

- each person known by us to be the beneficial owner of more than 5% of our outstanding shares of common stock;
- each of our officers and directors; and
- all our officers and directors as a group.

Unless otherwise indicated, we believe that all persons named in the table have sole voting and investment power with respect to all shares of common stock beneficially owned by them.

	Amount and Nature of Beneficial	• •	ntage of Outstanding on Stock
Name and Address of Beneficial Owner ⁽¹⁾	Ownership ⁽²⁾	Before the Offering	After the Offering ⁽³⁾
Camden Learning, LLC ⁽⁴⁾	1,458,334	93.34%	18.67% ⁽⁵⁾
David L. Warnock, President, Chief	1,458,334	93.34%	18.67%
Executive Officer and Chairman ⁽⁴⁾			
Donald W. Hughes, Chief Financial	1,458,334	93.34%	18.67%
Officer and Secretary(4)			
Therese Kreig Crane, Ed.D, Director	34,722	2.22%	.444%
Ronald Tomalis, Director	34,722	2.22%	.444%
William Jews, Director	34,722	2.22%	.444%
All directors and executive officers as a group (5 individuals)	1,562,500	100%	20 %

- (1) Unless otherwise indicated, the business address of each of the individuals is 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 and our telephone number is (410) 878-6800.
- (2) The percentage ownership before and after the offering for all executive officers and directors does not include the shares of common stock underlying the insider warrants sold in the private placement.
- (3) Assumes the sale of 6,250,000 units in this offering but not: (a) the exercise of the 6,250,000 warrants to purchase shares of our common stock included in such units, (b) the exercise of the 2,800,000 warrants sold to our sponsor as described herein, (c) the units underlying the purchase option granted to Morgan Joseph & Co. Inc.
- (4) Camden Learning, LLC is the sponsor, as described herein. The sole owners and members of our sponsor are Camden Partners Strategic Fund III, L.P. (96.01% ownership of the sponsor) and Camden Partners Strategic Fund III-A, L.P. (3.99% ownership of the sponsor). The general partner of each limited partnership is Camden Partners Strategic III, LLC and the managing member of such entity is Camden Partners Strategic Manager, LLC. David L. Warnock, our President, Chief Executive Officer and Chairman, Donald W. Hughes, our Chief Financial Officer and Secretary, Richard M. Johnston and Richard M. Berkeley are the four managing members of Camden Partners Strategic Manager, LLC, which has sole power to direct the vote and disposition of our securities held by the sponsor. Each of Mr. Warnock and Mr. Hughes disclaims beneficial ownership of all shares owned by Camden Learning, LLC.
- (5) If our sponsor purchases an additional 212,500 units in the offering, this percentage will increase to 23.4%.

Our sponsor has informed us that it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Assuming the 212,500 units are purchased in the offering and the \$4,000,000 of common stock are purchased by our sponsor in the open market, and further assuming such open market purchases of stock occur at the initial trust amount per share of \$7.92 per unit, our existing stockholders, collectively, will beneficially own 29.18% of the then issued and outstanding shares of common stock. Assuming neither the 212,500 units nor the \$4,000,000 worth of

our existing stockholders, collectively, will beneficially own approximately 20% of the then issued and outstanding shares of our common stock. Because of this ownership block, these stockholders may be able to effectively influence the outcome of all matters requiring approval by our stockholders, including the election of directors and approval of significant corporate transactions, but other than approval of a business combination. In the event all 212,500 units are purchased in the offering and the additional \$4,000,000 worth of units are purchased on the open market, all of such shares shall be voted in favor of the business combination, representing a maximum aggregate of 9.18% of the shares entitled to vote on any proposed business combination.

If holders of more than 20% of the shares sold in this offering vote against a proposed business combination and seek to exercise their redemption rights and such business combination is consummated, our existing stockholders have agreed to forfeit, on a pro rata basis, and return to us for cancellation, a number of the initial 1,562,500 shares of our common stock purchased, up to a maximum of 195,312 shares, so that the existing stockholders will collectively own no more than 23.81% (without regard to any purchase of units in this offering, any open market purchases or private purchases of units by the sponsor directly from us, as set forth elsewhere herein) of our outstanding common stock immediately prior to the consummation of such business combination after giving effect to the redemption.

In addition, if we take advantage of increasing the size of the offering pursuant to Rule 462(b) under the Securities Act, we may effect a stock dividend in such amount to maintain the existing stockholders' collective ownership at 20% of our issued and outstanding shares of common stock upon consummation of the offering. If we decrease the size of the offering we will effect a reverse split of our common stock in such amount to maintain the existing stockholders allocated ownership at 20% of our issued and outstanding common stock upon the consummation of this offering.

Subject to the possible forfeiture of shares described above, all of the shares of our common stock outstanding prior to the date of this prospectus will be placed in escrow with Continental Stock Transfer & Trust Company, as escrow agent, until the earlier of:

- · one year following consummation of a business combination; or
- the consummation of a liquidation, merger, stock exchange or other similar transaction which results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property subsequent to our consummating a business combination with a target business. Table 0px Font 10pt

During the escrow period, the holders of these shares will not be able to sell or transfer their securities except to their spouses and children or trusts established for their benefit, but will retain all other rights as our stockholders including, without limitation, the right to vote their shares of common stock and the right to receive cash dividends, if declared. If dividends are declared and payable in shares of common stock, such dividends will also be placed in escrow. If we are unable to effect a business combination and liquidate, none of our existing stockholders will receive any portion of the liquidation proceeds with respect to common stock owned by them prior to the date of this prospectus or purchased in the private placement.

The warrants to be purchased by our sponsor in the private placement will contain restrictions prohibiting their transfer until the earlier of the 90th day following consummation of a business combination or our liquidation and will be held in escrow by Morgan Joseph & Co. Inc. until such time.

In addition to the 212,500 units our sponsor has indicated it intends to purchase in the offering, our sponsor has entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Such purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price per share of not more than the per share amount held in the trust account (less taxes payable) as reported in such 8-K and will be made by a broker-dealer mutually agreed upon by our sponsor and the representative of the underwriters in such amounts and at such times as such broker-dealer may determine, in its sole discretion, so long as the

69

TABLE OF CONTENTS

purchase price does not exceed the above-referenced per share purchase price. Our sponsor has agreed to vote all such shares of common stock purchased in the open market in favor of our initial business combination, representing a maximum aggregate of 8% of the shares entitled to vote on any proposed business combination. Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell such shares, provided it will be entitled to participate in any liquidating distributions with respect to the shares purchased in the open market. In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our

sponsor has agreed to purchase from us in a private placement a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this co-investment. This co-investment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders. Our sponsor, whose sole owners are the Camden III Funds, has agreed to such purchases because the managing members of the general partner of the Camden III Funds, including David L. Warnock, our Chairman, President and Chief Executive Officer and Donald W. Hughes, our Chief Financial Officer and Secretary, want the Camden III Funds to have a substantial cash investment in us, including any target business we may acquire.

All of our directors will be deemed to be our "parents" and "promoters" as these terms are defined under the federal securities laws.

70

TABLE OF CONTENTS

CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

In April, 2007, we issued 1,125,000 shares of our common stock as set forth below for an aggregate amount of \$25,000 in cash, at an average purchase price of approximately \$0.02 per share, as follows:

Name	Number of Shares	Relationship to Us
Camden Learning, LLC	1,000,000	Sponsor. Donald W. Hughes and David L. Warnock are among the four managing members of the managing member of Camden Learning, LLC.
Jack L. Brozman	25,000	Director
Therese Kreig Crane, Ed.D	25,000	Director
Ronald Tomalis	25,000	Director
William Jews	25,000	Director
Harry T. Wilkins	25,000	Director

On July 3, 2007, Mr. Wilkins resigned as a director and transferred, for a purchase price of \$.02 per share, an aggregate of 25,000 shares of common stock to Camden Learning, LLC such that our current share ownership is as reflected in the section entitled "Principal Stockholders."

On August 27, 2007, Mr. Brozman resigned as a director and transferred, for a purchase price of \$.02 per share, an aggregate of 25,000 shares of common stock to Camden Learning, LLC such that our current share ownership is as reflected in the section entitled "Principal Stockholders."

Effective November 20, 2007, our board of directors authorized a forward stock split in the form of a stock dividend of 0.3888888 shares of common stock for each outstanding share of common stock, effectively lowering the purchase price to approximately \$0.016 per share.

In addition, if we take advantage of increasing the size of the offering pursuant to Rule 462(b) under the Securities Act, we may effect a stock dividend in such amount to maintain the existing stockholders' collective ownership at 20% of our issued and outstanding shares of common stock upon consummation of the offering. If we reduce the size of the offering we may effect a reverse stock split of our common stock in order to maintain the existing stockholders allocated ownership at 20% of our issued and outstanding common stock upon the consummation of this offering.

The holders of the majority of these shares will be entitled to require us, on up to two occasions, to register these shares pursuant to an agreement to be signed prior to or on the date of this prospectus. The holders of the majority of these shares may elect to exercise these registration rights at any time after the date on which these shares of common stock are released from escrow, which, except in limited circumstances, is not before one year from the consummation of a business combination. In addition, these stockholders have certain "piggy-back" registration rights on registration statements filed subsequent to the date on which these shares of common stock are released from escrow. We will bear the expenses incurred in connection with the filing of any such registration statements.

Our sponsor has agreed to purchase 2,800,000 warrants from us at a purchase price of \$1.00 per warrant in a private placement pursuant to Regulation D of the Securities Act that will occur immediately prior to this offering. We have granted the holders of such warrants demand and "piggy-back" registration rights with respect to the warrants and shares of common stock underlying such warrants at any time commencing on the date we announce we have entered into a letter of intent with respect to a proposed business combination, provided, however, any such registration will not become effective prior to completion of our initial business combination. The demand registration may be exercised by the holders of a majority of such warrants. We will bear the expenses incurred in connection with the filing of any such registration statements. The insider warrants will not be subject to redemption and may be exercised on a "cashless" basis if held by the initial holder thereof or its permitted assigns.

Our sponsor has informed us it intends to purchase 212,500 units in this offering, although it is under no obligation to do so. Our sponsor has agreed to vote any such shares of common stock in favor of our initial

business combination, representing a possible maximum aggregate of 3.4% of the public shares entitled to vote on the business combination. Our sponsor has entered into an agreement with the representative of the underwriters pursuant to which it will place limit orders to purchase up to \$4,000,000 of our common stock in the open market commencing ten business days after we file our current report on Form 8-K announcing our execution of a definitive agreement for a business combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a business combination is to be approved. Such open market purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price per share of not more than the per share amount held in the trust account (less taxes payable) as reported in such 8-K and will be made by a broker-dealer mutually agreed upon by our sponsor and the representative of the underwriters in such amounts and at such times as such broker-dealer may determine, in its sole discretion, so long as the purchase price does not exceed the abovereferenced per share purchase price. Our sponsor has agreed to vote any such shares of common stock purchased in the open market in favor of our initial business combination, representing a possible maximum aggregate of 9% of the public shares entitled to vote on the business combination. Unless a business combination is approved by our stockholders, our sponsor has agreed not to sell such shares, provided it will be entitled to participate in any liquidating distributions with respect to the shares purchased in the open market. In the event our sponsor does not purchase \$4,000,000 of our common stock through those open market purchases, our sponsor has agreed to purchase from us in a private placement a number of units identical to the units offered hereby at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this co-investment. This coinvestment will occur immediately prior to our consummation of a business combination, which will not occur until after the signing of a definitive business combination agreement and the approval of that business combination by a majority of our public stockholders. The co-investment units may not be sold, assigned or transferred unless and until our stockholders approve a business combination.

In order to protect the amounts held in the trust account, our sponsor has agreed to indemnify us for claims of any vendors, service providers, prospective target businesses or creditors that have not executed a valid and binding waiver of any right or claim to the amounts in trust account. As further assurance our sponsor will have the necessary funds required to meet these indemnification obligations, (i) the Camden III Funds have agreed, under our sponsor's limited liability company agreement, to make capital contributions to our sponsor as and when required in order for the sponsor to fulfill its indemnification obligations and (ii) our sponsor has agreed to take all such action reasonably necessary to request its members make such capital contributions. Additionally, in the event either of the Camden III Funds undertakes a liquidating distribution while the indemnification obligations of the sponsor are outstanding, they have agreed, in our sponsor's limited liability company agreement, to use reasonable efforts to set aside from such distribution, adequate reserves to cover the reasonably anticipated liabilities which may be incurred by our sponsor. We and the representative of the underwriters are named as express third party beneficiaries in and with respect to the provisions of our sponsor's limited liability company agreement which require the Camden III Funds to make such capital contributions and establish such reserves. Although we have a fiduciary obligation to pursue the sponsor to enforce its indemnification obligations, and intend to pursue such actions as and when we deem appropriate, there can be no assurance it or the Camden III Funds will be able to satisfy those obligations, if required to do so.

As of the date of this prospectus, our sponsor has loaned us a total of \$200,000, which was used to pay a portion of the expenses of this offering, such as SEC registration fees, FINRA registration fees, blue sky fees and certain legal and accounting fees and expenses. This loan will be payable, with interest on the earlier of April 26, 2008 or the consummation of this offering. The loan will be repaid out of the net proceeds of this offering not being placed in trust.

We maintain executive offices at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 and our telephone number is (410) 878-6800. The costs for this space is included in the \$7,500 per month fee Camden Partners Holdings, LLC charges us for general and administrative services, including but not limited to receptionist, secretarial and general office services, pursuant to a letter agreement between us and Camden Partners Holdings, LLC. This agreement commences on the date of this prospectus and shall continue until the earliest to occur of: (i) consummation of a business combination, (ii) 24 months after the completion of this offering

72

TABLE OF CONTENTS

if no business combination has been consummated and (iii) the date on which we cease our corporate existence in accordance with our amended and restated certificate of incorporation. We believe, based on fees for similar services in the greater Baltimore, Maryland metropolitan area, that the fee charged by Camden Partners Holdings, LLC is at least as favorable as we could have obtained from an unaffiliated person.

We will reimburse our officers and directors for any reasonable out-of-pocket business expenses incurred by them in connection with certain activities on our behalf such as identifying and investigating possible target businesses and business combinations. There is no limit on the amount of accountable out-of-pocket expenses reimbursable by us, which will be reviewed only by our board or a court of competent jurisdiction if such reimbursement is challenged.

Other than the reimbursable out-of-pocket expenses payable to our officers and directors, no compensation or fees

of any kind, including finders and consulting fees, will be paid to any of our existing stockholders, officers or directors who owned our common stock prior to this offering, or to any of their respective affiliates for services rendered to us prior to or with respect to the business combination.

Our existing stockholders will not receive reimbursement for any out-of- pocket expenses incurred by them to the extent that such expenses exceed the amount in the trust account unless the business combination is consummated and there are sufficient funds available for reimbursement after such consummation. The financial interest of such persons could influence their motivation in selecting a target business and thus, there may be a conflict of interest when determining whether a particular business combination is in the stockholders' best interest.

After the consummation of a business combination, if any, to the extent our management remains as officers of the resulting business, we anticipate that our officers and directors may enter into employment or consulting agreements, the terms of which shall be negotiated and which we expect to be comparable to employment or consulting agreements with other similarly-situated companies in the industry in which we consummate a business combination. Further, after the consummation of a business combination, if any, to the extent our directors remain as directors of the resulting business, we anticipate that they will receive compensation comparable to directors at other similarly-situated companies in the industry in which we consummate a business combination.

All ongoing and future transactions between us and any of our officers and directors or their respective affiliates, including loans by our officers and directors, will be on terms believed by us to be no less favorable than are available from unaffiliated third parties and such transactions or loans, including any forgiveness of loans, will require prior approval in each instance by a majority of our uninterested "independent" directors (to the extent we have any) or the members of our board who do not have an interest in the transaction, in either case who had access, at our expense, to our attorneys or independent legal counsel.

73

TABLE OF CONTENTS

DESCRIPTION OF SECURITIES

General

We are authorized to issue 20,000,000 shares of common stock, par value \$.0001, and 1,000,000 shares of preferred stock, par value \$.0001. As of the date of this prospectus, 1,562,500 shares of common stock are outstanding, held by four record holders. No shares of preferred stock are currently outstanding.

Units

Each unit consists of one share of common stock and one warrant. Each warrant entitles the holder to purchase one share of common stock. The common stock and warrants shall begin to trade separately on the 90th day after the date of this prospectus unless Morgan Joseph & Co. Inc. informs us of its decision to allow earlier separate trading, provided that in no event may the common stock and warrants be traded separately until the business day after (i) we file an audited balance sheet reflecting our receipt of the gross proceeds of this offering and the private placement, including any proceeds we receive from the exercise of the over-allotment option, if such option is exercised on the date of this prospectus, (ii) we file a Current Report on Form 8-K and issue a press release announcing when such separate trading will begin and (iii) the expiration of the underwriters over-allotment option or its exercise in full. We will file a Current Report on Form 8-K which includes this audited balance sheet upon the consummation of this offering. The audited balance sheet will reflect proceeds we receive from the exercise of the over-allotment option, if the over-allotment option is exercised on the date of this prospectus. In the event all or any portion of the over-allotment option is exercised after the date of this prospectus, we will file an additional Current Report on Form 8-K to disclose our receipt of the net proceeds from any such exercise.

Common Stock

Our stockholders are entitled to one vote for each share held of record on all matters to be voted on by stockholders. In connection with the vote required for any business combination, all of our existing stockholders, including all of our officers and directors, have agreed to vote their respective shares of common stock owned by them immediately prior to this offering in accordance with the vote of the public stockholders owning a majority of the shares of our common stock cast at the meeting. Our existing stockholders have agreed to vote all the shares of our common stock acquired in this offering or in the aftermarket in favor of any transaction our officers negotiate and present for approval to our stockholders. Our existing stockholders have also agreed to waive their rights to participate in any liquidation occurring upon our failure to consummate a business combination, but only with respect to those shares of common stock acquired by them prior to this offering. Our existing stockholders, officers and directors will vote all of their shares in any manner they determine, in their sole discretion, with respect to any other items that come before a vote of our stockholders.

We will proceed with a business combination only if a majority of the shares of common stock cast at the meeting are voted in favor of the business combination, and public stockholders owning less than 30% of the shares sold in this offering exercise their redemption rights discussed below. Voting against the business combination alone will not result in redemption of a stockholder's shares into a pro rata share of the trust account. Such stockholder must have also exercised its redemption rights described below. Our threshold for redemption has been established at 30% in order for our offering to be competitive with other blank check company offerings. However, to date a 20% threshold has been more typical for offerings of this type. We have selected the higher threshold to reduce the risk of a small group of

stockholders exercising undue influence on the stockholder approval process.

Our board of directors is divided into two classes, each of which will generally serve for a term of two years with only one class of directors being elected in each year. There is no cumulative voting with respect to the election of directors, with the result that the holders of more than 50% of the shares voted for the election of directors can elect all of the directors.

If we are forced to liquidate our trust account, our public stockholders are entitled to share ratably in the trust account, inclusive of any interest, if any, not previously paid to us, after taxes, if any. The term public stockholders means the holders of common stock sold as part of the units in this offering or acquired in the open market, but excludes our officers and directors or their nominees or designees with respect to the shares

74

TABLE OF CONTENTS

owned by them prior to this offering since they have waived their redemption and right to liquidation distributions from our trust account in connection with our dissolution as part of our plan of dissolution and liquidation with respect to these shares.

Our existing stockholders have also agreed to waive their respective rights to participate in any liquidation of the trust account in connection with our dissolution occurring upon our failure to consummate a business combination as well as to vote for any plan of dissolution and liquidation submitted to our stockholders with respect to those shares of common stock acquired by them prior to this offering.

Our stockholders have no redemption, preemptive or other subscription rights and there are no sinking fund or redemption provisions applicable to the common stock, except that public stockholders have the right to have their shares of common stock redeemed for cash equal to their pro rata share of the trust account if they vote against the business combination and the business combination is approved and completed. Public stockholders who redeem their shares of common stock into their share of the trust account still have the right to exercise the warrants that they received as part of the units.

Due to the fact that we currently have 20,000,000 shares of common stock authorized, if we were to enter into a business combination, we may (depending on the terms of such a business combination) be required to increase the number of shares of common stock which we are authorized to issue at the same time as our stockholders vote on the business combination.

Preferred Stock

Our amended and restated certificate of incorporation authorizes the issuance of 1,000,000 shares of blank check preferred stock with such designation, rights and preferences as may be determined from time to time by our board of directors. No shares of preferred stock are being issued or registered in this offering. Accordingly, our board of directors is empowered, without stockholder approval, to issue preferred stock with dividend, liquidation, conversion, voting or other rights which could adversely affect the voting power or other rights of the holders of common stock, although the underwriting agreement prohibits us, prior to a business combination, from issuing preferred stock which participates in any manner in the proceeds of the trust account, or which votes as a class with the common stock on a business combination. We may issue some or all of the preferred stock to effect a business combination. In addition, the preferred stock could be utilized as a method of discouraging, delaying or preventing a change in control of us. Although we do not currently intend to issue any shares of preferred stock, we cannot assure you that we will not do so in the future.

Warrants

No warrants are currently outstanding. Each warrant included in the units sold in this offering and the private placement entitles the registered holder to purchase one share of our common stock at a price of \$5.50 per share, subject to adjustment as discussed below, at any time commencing on the later of:

- · the completion of a business combination; or
- one year from the date of this prospectus.

However, the warrants will be exercisable only if a registration statement relating to the common stock issuable upon exercise of the warrants is effective and current. The warrants will expire on [], 2011 at 5:00 p.m., New York City time.

The warrants may trade separately on the 90th trading day after the date of this prospectus, unless Morgan Joseph & Co. Inc. determines that an earlier date is acceptable; provided, however, that in no event may the common stock and warrants be traded separately until we have filed a Current Report on Form 8-K which includes an audited balance sheet reflecting our receipt of the proceeds of this offering and the private placement, including any proceeds we receive from the exercise of the over-allotment option if such option is exercised on the date of this prospectus. In the event all or any portion of the over-allotment option is exercised after the date of this prospectus, we will file an additional Current Report on Form 8-K to disclose our receipt of the net proceeds from any such exercise.

The warrants comprising part of the units (including any warrants issued to Morgan Joseph & Co. Inc. as part of its unit purchase option) may be redeemed:

- in whole and not in part;
- at a price of \$0.01 per warrant
- at any time while the warrants are exercisable;
- upon not less than 30 days' prior written notice of redemption to each warrant holder; and
- if, and only if, the last closing sales price of our Common Stock equals or exceeds \$11.50 per share for any 20 trading days within a 30 trading day period ending three business days before we send the notice of redemption.

We have established this last criterion to provide warrant holders with a premium to the initial warrant exercise price as well as a degree of liquidity to cushion the market reaction, if any, to our redemption call. If the foregoing conditions are satisfied and we call the warrants for redemption, each warrant holder shall then be entitled to exercise his or her warrant prior to the date scheduled for redemption, however, there can be no assurance that the price of the common stock will exceed the call trigger price or the warrant exercise price after the redemption call is made.

The warrants will be issued in registered form under a warrant agreement between Continental Stock Transfer & Trust Company, as warrant agent, and us. You should review a copy of the warrant agreement, which has been filed as an exhibit to the registration statement of which this prospectus is a part, for a complete description of the terms and conditions applicable to the warrants.

The exercise price and number of shares of common stock issuable on exercise of the warrants may be adjusted in certain circumstances including in the event of a stock dividend, or our recapitalization, reorganization, merger or consolidation. However, the warrants will not be adjusted for issuances of common stock at a price below their respective exercise prices.

The warrants may be exercised upon surrender of the warrant certificate on or prior to the expiration date at the offices of the warrant agent, with the exercise form on the reverse side of the warrant certificate completed and executed as indicated, accompanied by full payment of the exercise price, by certified check payable to us, for the number of warrants being exercised. The warrant holders do not have the rights or privileges of holders of common stock and any voting rights until they exercise their warrants and receive shares of common stock. After the issuance of shares of common stock upon exercise of the warrants, each holder will be entitled to one vote for each share held of record on all matters to be voted on by stockholders.

No warrants will be exercisable and we will not be obligated to issue shares of common stock thereunder unless, at the time a holder seeks to exercise such warrant, a registration statement relating to common stock issuable upon exercise of the warrants is effective and current and the common stock has been registered or qualified or deemed to be exempt under the securities laws of the state of residence of the holder of the warrants. Under the terms of the warrant agreement, we have agreed to meet these conditions and use our best efforts to maintain a current prospectus relating to common stock issuable upon exercise of the warrants until the expiration of the warrants. If we are unable to maintain the effectiveness of such registration statement until the expiration of the warrants or if the common stock is not qualified or exempt from qualification in the jurisdictions in which the holders of the warrants reside, and therefore are unable to deliver registered shares, holders of warrants will not be able to exercise their warrants, the market for the warrants may be limited, and the warrants may expire worthless. In no event will the holder of a warrant be entitled to receive a net-cash settlement, stock or other consideration in lieu of physical settlement in shares of our common stock.

Because the warrants sold in the private placement were originally issued pursuant to an exemption from the registration requirements under the federal securities laws, the holders of the warrants purchased in the private placement will be able to exercise their warrants even if, at the time of exercise, a prospectus relating to the common stock issuable upon exercise of such warrants is not current. As described above, holders of the warrants purchased in this offering will not be able to exercise them unless we have a current registration statement covering the shares issuable upon their exercise.

76

TABLE OF CONTENTS

No fractional shares will be issued upon exercise of the warrants. If, upon exercise of the warrants, a holder would be entitled to receive a fractional interest in a share, we will, upon exercise, round up to the nearest whole number the number of shares of common stock to be issued to the warrant holder.

Unit Purchase Option

We have agreed to issue to Morgan Joseph & Co. Inc., as additional compensation, an option to purchase up to 10% of the number of units sold in the offering, up to a maximum of 625,000 units. The units are identical to those offered by this prospectus except that the exercise price for the warrants included in these units is \$6.71 per share. For a more complete description of the purchase option, see the section entitled "Underwriting — Purchase Option."

We have not paid any dividends on our common stock to date and do not intend to pay dividends prior to the completion of a business combination. The payment of dividends in the future will be contingent upon our revenues and earnings, if any, capital requirements and general financial condition subsequent to completion of a business combination. The payment of any dividends subsequent to a business combination will be within the discretion of our then board of directors. It is the present intention of our board of directors to retain all earnings, if any, for use in our business operations and, accordingly, our board does not anticipate declaring any dividends in the foreseeable future.

Our Transfer Agent and Warrant Agent

The transfer agent for our securities and warrant agent for our warrants is Continental Stock Transfer & Trust Company.

Shares Eligible for Future Sale

Immediately after this offering, we will have 7,812,500 shares of common stock outstanding, or 8,750,000 shares if the underwriters' over-allotment option is exercised in full. Of these shares, the 6,250,000 shares sold in this offering, or 7,187,500 shares if the over-allotment option is exercised in full, will be freely tradable without restriction or further registration under the Securities Act, except for any shares purchased by one of our affiliates within the meaning of Rule 144 under the Securities Act. All of the remaining 1,562,500 shares are restricted securities under Rule 144, in that they were issued in private transactions not involving a public offering, and will not be eligible for sale under Rule 144. Notwithstanding this, all of those shares have been placed in escrow and will not be transferable until one year from the date of consummation of a business combination, and will only be released prior to that date subject to certain limited exceptions such as our liquidation prior to a business combination (in which case the certificate representing such shares will be destroyed), and the consummation of a liquidation, merger, stock exchange or other similar transaction which results in all of our stockholders having the right to exchange their shares of common stock for cash, securities or other property subsequent to our consummating a business combination with a target business.

Rule 144

In general, under Rule 144 as currently in effect, a person who has beneficially owned restricted shares of our common stock for at least one year would be entitled to sell within any three-month period a number of shares that does not exceed the greater of either of the following:

- 1% of the number of shares of common stock then outstanding, which will equal 78,125 shares immediately after this offering (or 87,500 if the underwriters' exercise their over-allotment option in full); and
- the average weekly trading volume of the common stock during the four calendar weeks preceding the filing of a notice on Form 144 with respect to the sale.

Sales under Rule 144 are also limited by manner of sale provisions and notice requirements and to the availability of current public information about us.

Rule 144(k)

Under Rule 144(k), a person who is not deemed to have been one of our affiliates at the time of or at any time during the three months preceding a sale, and who has beneficially owned the restricted shares

77

TABLE OF CONTENTS

proposed to be sold for at least two years, including the holding period of any prior owner other than an affiliate, is entitled to sell their shares without complying with the manner of sale, public information, volume limitation or notice provisions of Rule 144.

SEC Position on Rule 144 Sales

The Securities and Exchange Commission has taken the position that promoters or affiliates of a blank check company and their transferees, both before and after a business combination, would act as an "underwriter" under the Securities Act when reselling the securities of a blank check company. Accordingly, the Securities and Exchange Commission believes that those securities can be resold only through a registered offering and that Rule 144 would not be available for those resale transactions despite technical compliance with the requirements of Rule 144.

Registration Rights

The holders of our 1,562,500 issued and outstanding shares of common stock on the date of this prospectus will be entitled to registration rights pursuant to an agreement to be signed prior to or on the effective date of this offering. The holders of the majority of these shares are entitled to require us, on up to two occasions, to register these shares. The holders of the majority of these shares can elect to exercise these registration rights at any time after the date on which these shares of common stock are released from escrow. In addition, these stockholders have certain "piggy-back" registration rights on registration statements filed subsequent to the date on which these shares of common stock are released from escrow. We will bear the expenses incurred in connection with the filing of any such registration statements.

Our sponsor has agreed to purchase 2,800,000 warrants from us at a purchase price of \$1.00 per warrant in a private placement that will occur immediately prior to this offering. We have granted the holders of such warrants demand and "piggy-back" registration rights with respect to the 2,800,000 warrants and shares of common stock underlying the warrants at any time commencing on the date we announce that we have entered into a letter of intent

with respect to a proposed business combination, provided, however, any such registration shall not become effective until our business combination has been completed. The demand registration may be exercised by the holders of a majority of such warrants. We will bear the expenses incurred in connection with the filing of any such registration statements. The insider warrants will not be subject to redemption and may be exercised on a "cashless" basis if held by the initial holder thereof or its permitted assigns.

Our Amended and Restated Certificate of Incorporation

Our amended and restated certificate of incorporation filed with the State of Delaware contains provisions designed to provide certain rights and protections to our stockholders prior to the consummation of a business combination, including:

- requirement that all proposed business combinations be presented to stockholders for approval regardless of whether or not Delaware law requires such a vote;
- prohibition against completing a business combination if 30% or more of our stockholders exercise their redemption rights in lieu of approving a business combination;
- the right of stockholders voting against a business combination to surrender their shares for a pro rata portion of the trust account in lieu of participating in a proposed business combination;
- a requirement that in the event we do not consummate a business combination by [24 months from the date of this prospectus], the company will dissolve, at which point our purpose and powers will be limited to dissolving, liquidating and winding up; provided, however, that we will reserve our rights under Section 278 of the Delaware General Corporation Law to bring or defend any action, suit or proceeding brought by or against us;
- requirement that our management take all actions necessary to liquidate our trust account to our public stockholders as part of our plan of dissolution and liquidation in the event we do not consummate a business combination by ______ [24 months after the consummation of this offering]

78

TABLE OF CONTENTS

- limitation on stockholders' rights to receive a portion of the trust account so that they may only receive a portion
 of the trust account upon liquidation of our trust account to our public stockholders as part of our plan of
 dissolution and liquidation or upon the exercise of their redemption rights; and
- the bifurcation of our board of directors into two classes and the establishment of related procedures regarding the standing and election of such directors.

Our amended and restated certificate of incorporation and the underwriting agreement we will enter into with Morgan Joseph & Co. Inc. in connection with the consummation of this offering prohibit the amendment or modification of any of the foregoing provisions prior to the consummation of a business combination without the prior approval of 95% of the shares purchased in this offering. Additionally, members of our board of directors and management have undertaken not to propose or vote in favor of any amendment to the foregoing provisions. While these rights and protections have been established for the purchasers of units in this offering, it is nevertheless possible that the prohibition against amending or modifying these rights and protections at any time prior to the consummation of the business combination could be challenged as unenforceable under Delaware law, although pursuant to the underwriting agreement we are prohibited from amending or modifying these rights and protections at any time prior to the consummation of the business combination. We have not sought an unqualified opinion regarding the enforceability of the prohibition on amendment or modification of such provisions because we view these provisions as fundamental terms of this offering. We believe these provisions to be obligations of our company to its stockholders and that investors will make an investment in our company relying, at least in part, on the enforceability of the rights and obligations set forth in these provisions including, without limitation, the prohibition on any amendment or modification of such provisions. As a result, the board of directors will not, and pursuant to the underwriting agreement and other agreements cannot, at any time prior to the consummation of a business combination, propose any amendment to or modification of our amended and restated certificate of incorporation relating to any of the foregoing provisions and will not support, directly or indirectly, or in any way endorse or recommend that stockholders approve an amendment or modification to such provisions.

79

TABLE OF CONTENTS

UNDERWRITING

In accordance with the terms and conditions contained in the underwriting agreement, we have agreed to sell to each of the underwriters named below, and each of the underwriters, for which Morgan Joseph & Co. Inc. is acting as representative, have severally, and not jointly, agreed to purchase on a firm commitment basis, the number of units offered in this offering set forth opposite their respective names below:

Underwriters	Number of Units
Morgan Joseph & Co. Inc.	
Total	6,250,000

A copy of the underwriting agreement has been filed as an exhibit to the registration statement of which this prospectus forms a part.

The underwriters may deliver prospectuses via e-mail both as a PDF document and by a link to the Securities and Exchange Commission's website and websites hosted by the underwriters and other parties, and the prospectus may also be made available on websites maintained by selected dealers and selling group members participating in this offering. The underwriters may agree to allocate a number of units to underwriters and selling group members for sale to their online brokerage account holders. Internet distributions may be allocated by the representative to underwriters and selling group members that may make Internet distributions on the same basis as other allocations.

State Blue Sky Information

We will offer and sell the units to retail customers only in Colorado, Delaware, the District of Columbia, Florida, Georgia, Hawaii, Illinois, Louisiana, Maryland, New York and Rhode Island. We have applied to have the units registered for sale, or we are relying on exemptions from registration in the states mentioned above. In states that require registration, we will not sell the units to retail customers in these states until such registration is effective in each of these states (including in Colorado, pursuant to 11-51-302(6) of the Colorado Revised Statutes).

If you are not an institutional investor, you may purchase our securities in this offering only in the jurisdictions described directly above. Institutional investors in every state except in Idaho may purchase the units in this offering pursuant to exemptions provided to such entities under the Blue Sky laws of various states. The definition of an "institutional investor" varies from state to state but generally includes financial institutions, broker-dealers, banks, insurance companies and other qualified entities.

The National Securities Markets Improvement Act of 1996 ("NSMIA"), which is a federal statute, prevents or preempts the states from regulating transactions in certain securities, which are referred to as "covered securities". This federal statute does allow the states to investigate companies if there is a suspicion of fraud, and if there is a finding of fraudulent activity, then the states can regulate or bar the sale of covered securities in a particular case.

State securities laws either require that a company's securities be registered for sale or that the securities themselves or the transaction under which they are issued, are exempt from registration. When a state law provides an exemption from registration, it is excusing an issuer from the general requirement to register securities before they may be sold in that state. States, may by rule or regulation, place conditions on the use of exemptions, so that certain companies may not be allowed to rely on the exemption for the sale of their securities. If an exemption is not available and the securities the company wishes to sell are not covered securities under the federal statute, then the company must register its securities for sale in the state in question.

We will file periodic and annual reports under the Securities Exchange Act of 1934, as amended. Therefore, under NSMIA, the states and territories of the United States are preempted from regulating the resale by stockholders of the units, from and after the effective date, and the common stock and warrants comprising the units, once they become separately transferable, because our securities will be covered securities. However, NSMIA does allow states and territories of the United States to require notice filings and collect fees with regard to these transactions and a state may suspend the offer and sale of securities within such state if

80

TABLE OF CONTENTS

any such required filing is not made or fee is not paid. As of the date of this prospectus, the following states do not require any notice filings or fee payments and stockholders may resell the units, and the common stock and warrants comprising the units, once they become separately transferable:

Alaska, Arizona, Arkansas, California, Colorado, Connecticut, Delaware, Florida, Georgia, Hawaii, Indiana, Iowa, Kansas, Kentucky, Louisiana, Maine, Massachusetts, Minnesota, Mississippi, Missouri, Nebraska, New Jersey, New Mexico, New York, North Carolina, Ohio, Oklahoma, Pennsylvania, South Carolina, South Dakota, Utah, Virginia, Virgin Islands, Washington, West Virginia, Wisconsin and Wyoming.

Additionally, the stockholders may resell the units, and the common stock and warrants comprising the units, once they become separately transferable, if the proper notice filings have been made and fees paid in the following states: District of Columbia, Illinois, Maryland, Michigan, Montana, New Hampshire, North Dakota, Oregon, Puerto Rico, Rhode Island, Tennessee, Texas and Vermont.

As of the date of this prospectus, we have not determined in which, if any, of these states we will submit the required filings or pay the required fee. Additionally, if any of the states that have not yet adopted a statute, rule or regulation relating to the National Securities Markets Improvement Act adopts such a statute in the future requiring a filing or fee or if any state amends its existing statutes, rules or regulations with respect to its requirements, we would need to comply with those new requirements in order for the securities to continue to be eligible for resale in those jurisdictions.

In addition, we believe that the units, from and after the effective date, and the common stock and warrants comprising the units, once they become separately transferable, may be eligible for sale on a secondary market basis in various states, without any notice filings or fee payments, based upon the availability of an applicable exemption from

the state's registration requirements.

- · commencing 90 days after the date of this prospectus in Nevada; and
- commencing 180 days from the date of this prospectus in Alabama.

Despite the exemption from state registration provided by the National Securities Markets Improvement Act described above, the state of Idaho has advised us that it does not recognize this act as a basis for exempting registration of resales therein of securities issued in blank check offerings.

We do not intend to register the resale of the securities sold in this offering in these states.

Pricing of Securities

We have been advised by the representative that the underwriters propose to offer the units to the public at the initial offering price set forth on the cover page of this prospectus. It may allow some dealers concessions not in excess of \$[] per unit.

Prior to this offering there has been no public market for any of our securities. The public offering price of the units and the terms of the warrants were negotiated between us and the representative. Factors considered in determining the prices and terms of the units, including the common stock and warrants underlying the units, include:

- · the history and prospects of companies whose principal business is the acquisition of other companies;
- · prior offerings of those companies;
- our prospects for acquiring one or more operating businesses at attractive values;
- our capital structure;
- an assessment of our management and their experience in identifying operating companies;
- general conditions of the securities markets at the time of the offering; and
- other factors as were deemed relevant.

81

TABLE OF CONTENTS

However, although these factors were considered, the determination of our offering price is more arbitrary than the pricing of securities for an operating company in a particular industry since the underwriters are unable to compare our financial results and prospects with those of public companies operating in the same industry.

Over-Allotment Option

We have granted to the representative of the underwriters an option, exercisable during the 45-day period commencing on the date of this prospectus, to purchase from us at the offering price, less underwriting discounts, up to an aggregate of 937,500 additional units for the sole purpose of covering over-allotments, if any. The over-allotment option will only be used to cover the net syndicate short position resulting from the initial distribution.

The representative of the underwriters may exercise the over-allotment option if the underwriters sell more units than the total number set forth in the table above.

Commissions and Discounts

The following table shows the public offering price, underwriting discount to be paid by us to the underwriters and the proceeds, before expenses, to us. This information assumes either no exercise or full exercise by the underwriters of their over-allotment option. This information does not reflect the private placement proceeds to be received by us.

	Per	Unit	Without Option	 With Option
Public offering price	\$	8.00	\$50,000,000	\$ 57,500,000
Discount ⁽¹⁾	\$	0.32	\$ 2,000,000	\$ 2,300,000
Deferred discount ⁽²⁾	\$	0.24	\$ 1,500,000	\$ 1,725,000
Proceeds before expenses ⁽³⁾	\$	7.44	\$46,500,000	\$ 53,475,000

- (1) Based upon the underwriters' discount of 4% per unit. Does not include an additional 3% of the gross proceeds from the sale of the units in this offering paid to Morgan Joseph & Co. Inc. only upon the consummation of a business combination (and then only with respect to those units as to which the component shares have not been redeemed for cash) which amounts are reflected in this table as deferred discount. If a business combination is not consummated and we automatically dissolve and subsequently liquidate our trust account, such amounts will not be paid to the underwriters, but rather will be distributed among our public stockholders.
- (2) The underwriters have agreed to forfeit their deferred underwriting discount with respect to those units as to which the underlying shares are redeemed into cash by those stockholders who voted against the business combination and exercised their redemption rights upon consummation of a business combination.
- (3) The offering expenses are estimated at \$400,000.

The underwriters will initially offer the units to be sold in this offering directly to the public at the initial public offering

price set forth on the cover of this prospectus and to selected dealers at the initial public offering price less a selling concession not in excess of \$ per unit. The underwriters may allow, and the selected dealers may reallow, a concession not in excess of \$ per unit on sales to brokers and dealers. After the offering, the underwriters may change the offering price and other selling terms, provided, however, upon execution of the underwriting agreement, there will be no changes to the price and terms of the sale between the underwriters and us. No change in those terms will change the amount of proceeds to be received by us as set forth on the cover of this prospectus.

Purchase Option

We have agreed to sell to the representative, for \$100, an option to purchase up to a total of 625,000 units.

The units issuable upon exercise of this option are identical to those offered by this prospectus except that the exercise price for the warrants included in these units is \$6.71 per share. This option is exercisable on

82

TABLE OF CONTENTS

a cashless basis at \$9.60 per unit commencing on the later of the consummation of a business combination and one year from the date of this prospectus, and expiring five years from the date of this prospectus. The option and the 625,000 units, the 625,000 shares of common stock and the 625,000 warrants underlying such units, and the 625,000 shares of common stock underlying such warrants, have been deemed to be underwriting compensation by the FINRA and are therefore subject to a 180-day lock-up pursuant to Rule 2710(g)(1) of the FINRA Conduct Rules. Morgan Joseph & Co. Inc. will not sell, transfer, assign, pledge, or hypothecate this option or the securities underlying this option, nor will it engage in any hedging, short sale, derivative, put, or call transaction that would result in the effective economic disposition of this option or the underlying securities for a period of 180 days from the effective date of this prospectus.

Additionally, the option may not be sold, transferred, assigned, pledged or hypothecated for a one-year period (including the foregoing 180 day period) following the date of this prospectus except to any underwriter and selected dealer participating in the offering and their bona fide officers or partners. Although the purchase option and its underlying securities have been registered on the registration statement of which this prospectus forms a part, the option grants holders demand and "piggy back" registration rights for periods of five and seven years, respectively, from the date of this prospectus. These rights apply to all of the securities directly and indirectly issuable upon exercise of the option. We will bear all fees and expenses attendant to registering the securities issuable on exercise of the option, other than underwriting commissions incurred and payable by the holders. The exercise price and number of units issuable upon exercise of the option may be adjusted in certain circumstances including in the event of a share dividend, or our recapitalization, reorganization or consolidation.

However, the option exercise price or underlying units will not be adjusted for issuances of shares of common stock at a price below the option exercise price.

We have estimated, based upon a Black-Scholes model, that the fair value of the option on the date of sale would be approximately \$2,400,000, using an expected life of five years, volatility of 59.4%, and a risk-free interest rate of 3.08%. However, because our units do not have a trading history, the volatility assumption is based on information currently available to management. We believe the volatility estimate calculated is a reasonable benchmark to use in estimating the expected volatility of our units. Although an expected life of five years was used in the calculation, if we do not consummate a business combination within the prescribed time period and we automatically dissolve and subsequently liquidate our trust account, the option will become worthless.

Regulatory Restrictions on Purchase of Securities

Rules of the SEC may limit the ability of the underwriters to bid for or purchase our securities before the distribution of the securities is completed. However, the underwriters may engage in the following activities in accordance with the rules:

- Stabilizing Transactions. The underwriters may make bids or purchases for the purpose of pegging, fixing or maintaining the price of our securities.
- Over-Allotments and Syndicate Coverage Transactions. The underwriters may create a short position in our securities by selling more of our securities than are set forth on the cover page of this prospectus. If the underwriters create a short position during the offering, the representative may engage in syndicate covering transactions by purchasing our securities in the open market. The representative may also elect to reduce any short position by exercising all or part of the over-allotment option.
- Penalty Bids. The representative may reclaim a selling concession from a syndicate member when the units originally sold by the syndicate member are purchased in a stabilizing or syndicate covering transaction to cover syndicate short positions.

Stabilization and syndicate covering transactions may cause the price of the securities to be higher than they would be in the absence of these transactions. The imposition of a penalty bid may also have an effect on the prices of the securities if it discourages resales.

Neither we nor the underwriters make any representation or prediction as to the effect the transactions described above may have on the prices of our securities. These transactions may occur on the OTC Bulletin Board, in the overthe-counter market or on any trading market. If any of these transactions are commenced, they may be discontinued without notice at any time.

The distribution of our securities will end upon the underwriters' cessation of selling efforts and stabilization activities, provided, however, in the event the underwriters were to exercise their over-allotment option to purchase securities in excess of their actual syndicate short position, the distribution will not be deemed to have been completed until all of the securities have been sold.

In connection with this offering, the underwriters may distribute prospectuses electronically. No forms of prospectus other than printed prospectuses and electronically distributed prospectuses that are printable in Adobe PDF format will be used in connection with this offering.

Other Terms

For a period of no less than two years after the date of the prospectus, we have granted Morgan Joseph & Co. Inc. the right to have an observer present at all meetings of our board of directors until we consummate a business combination. The observer shall be entitled to attend meetings of the board, receive all notices and other correspondence and communications sent by us to members of our board of directors, but will not have voting rights. In addition, such observer shall be entitled to receive, as his/her sole compensation, reimbursement for all costs incurred in attending such meetings. Morgan Joseph & Co. Inc. has not named its observer as of the date of this prospectus.

Although they are not obligated to do so, any of the underwriters may introduce us to potential target businesses or assist us in raising additional capital, as needs may arise in the future, but there are no preliminary agreements or understandings between any of the underwriters and any potential targets. We are not under any contractual obligation (oral or written) and have no agreement or understanding to engage any of the underwriters to provide any services for us after this offering, but if we do engage any of them in the future we may pay the underwriters a finder's fee or advisory fee for services that would be determined at that time in an arm's length negotiation where the terms would be fair and reasonable to each of the interested parties; provided that no agreement will be entered into and no fee will be paid within 90 days following the date of this prospectus.

Indemnification

We have agreed to indemnify the underwriters against some liabilities, including civil liabilities under the Securities Act of 1933, as amended, or to contribute to payments the underwriters may be required to make in this respect.

LEGAL MATTERS

The validity of the securities offered in this prospectus is being passed upon for us by Ellenoff Grossman & Schole LLP, New York, New York. Such firm has previously represented Morgan Joseph & Co. Inc. on matters unrelated to this offering and expects to do so again in the future. McDermott Will & Emery LLP, New York, New York, is acting as counsel for the underwriters in this offering.

EXPERTS

The financial statements included in this prospectus and in the registration statement have been audited by Eisner LLP, an independent registered public accounting firm, to the extent and for the period set forth in their report (which contains an explanatory paragraph regarding our ability to continue as a going concern) appearing elsewhere in this prospectus and in the registration statement. The financial statements and the report of Eisner LLP are included in reliance upon their report given upon the authority of Eisner LLP as experts in auditing and accounting.

WHERE YOU CAN FIND ADDITIONAL INFORMATION

We have filed with the SEC a registration statement on Form S-1, which includes exhibits, schedules and amendments, under the Securities Act of 1933, as amended, with respect to this offering of our securities.

84

TABLE OF CONTENTS

Although this prospectus, which forms a part of the registration statement, contains all material information included in the registration statement, parts of the registration statement have been omitted as permitted by rules and regulations of the SEC. We refer you to the registration statement and its exhibits for further information about us, our securities and this offering. The registration statement and its exhibits, as well as our other reports filed with the SEC, can be inspected and copied at the SEC's public reference room at 100 F Street, N.E., Room 1580, Washington, D.C. 20549. The public may obtain information about the operation of the public reference room by calling the SEC at 1-800-SEC-0330. In addition, the SEC maintains a web site at http://www.sec.gov which contains the Form S-1 and other reports, proxy and information statements and information regarding issuers that file electronically with the SEC.

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

INDEX TO FINANCIAL STATEMENTS

	Page
Report of Independent Registered Public Accounting Firm	<u>F-2</u>
Financial Statements:	
Balance Sheet as of September 30, 2007	<u>F-3</u>
Statement of Operations for the period from April 10, 2007 (inception) to September 30,	<u>F-4</u>
<u>2007</u>	
Statement of Stockholders' Equity for the period from April 10, 2007 (inception) to	<u>F-5</u>
<u>September 30, 2007</u>	
Statement of Cash Flows for the period from April 10, 2007 (inception) to September 30,	<u>F-6</u>
<u>2007</u>	
Notes to Financial Statements for the period from April 10, 2007 (inception) to September	<u>F-7</u>
<u>30, 2007</u>	

F-1

TABLE OF CONTENTS

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Directors and Stockholders Camden Learning Corporation

We have audited the accompanying balance sheet of Camden Learning Corporation, a corporation in the development stage (the "Company"), as of September 30, 2007, and the related statements of operations, stockholders' equity and cash flows for the period from April 10, 2007 (inception) through September 30, 2007. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Camden Learning Corporation as of September 30, 2007, and the results of its operations and its cash flows for the period from April 10, 2007 (inception) through September 30, 2007 in conformity with accounting principles generally accepted in the United States of America.

The accompanying financial statements have been prepared assuming the Company will continue as a going concern. The Company has generated a net loss, has a deficiency in working capital and has no operations. These factors raise substantial doubt about the Company's ability to continue as a going concern. As discussed in Notes 1 and 4, the Company is in the process of raising capital through both a proposed public offering and a private placement. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Eisner LLP

/s/ Eisner LLP

New York, New York November 21, 2007

F-2

BALANCE SHEET September 30, 2007 ASSETS

Current assets:	
Cash	\$ 35,115
Due from affiliate	1,111
Total current assets	36,226
Deferred offering costs	228,855
Total assets	\$265,081
LIABILITIES AND STOCKHOLDERS' EQUITY	
Current liabilities:	
Accounts payable and accrued expenses	\$ 41,211
Accrued interest expense	4,083
Note payable to affiliate (face amount \$200,000)	189,288
Total current liabilities	\$234,582
Commitments and contingencies	
Stockholders' equity	
Preferred Stock, \$.0001 par value, 1,000,000 shares authorized; none issued or outstanding	_
Common Stock, \$.0001 par value, 20,000,000 shares authorized; 1,562,500 shares issued and outstanding	156
Additional paid-in capital	42,413
Deficit accumulated during the development stage	(12,070
)
Total stockholders' equity	30,499
Total liabilities and stockholders' equity	\$265,081

The accompanying notes are an integral part of the financial statements.

F-3

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

STATEMENT OF OPERATIONS

For the Period from April 10, 2007 (Inception) to September 30, 2007

Formation and operating costs	\$ 1,130
Interest expense	10,940
Net loss	\$ (12,070)
Basic and diluted net loss per share	\$ (0.01)
Weighted average shares outstanding – basic and diluted	 ,562,500

The accompanying notes are an integral part of the financial statements.

F-4

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

STATEMENT OF STOCKHOLDERS' EQUITY
For the Period from April 10, 2007 (Inception) to September 30, 2007

	Common Stock		Additional Paid-In Capital	Deficit Accumulated During the	Total Stockholders' Equity		
	Shares	Α	mount		Development Stage		
Initial capital from founding stockholders	1,562,500	\$	156	\$ 24,844	\$ —	\$	25,000
Discount on note payable to affiliate				17,569			17,569
Net loss during the development stage			_		(12,070)		(12,070)
Balance at September 30, 2007	1,562,500	\$	156	\$ 42,413	\$ (12,070)	\$	30,499

The accompanying notes are an integral part of the financial statements.

F-5

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

STATEMENT OF CASH FLOWS For the Period from April 10, 2007 (inception) to September 30, 2007

Cash flows from operating activities:	
Net loss	\$ (12,070)
Adjustments to reconcile net loss to net cash used in operating activities	
Increase in accrued expenses	1,000
Accretion of interest on note payable	6,857
Advances to affiliate	(1,111)
Accrued interest	 4,083
Net cash used in operating activities	(1,241)
Cash flows from financing activities:	
Proceeds from sale of stock	25,000
Proceeds from note payable to affiliate	200,000
Deferred offering costs	(188,644)
Net cash provided by financing activities	 36,356
Net increase in cash	35,115
Cash at beginning of period	_
Cash at end of period	\$ 35,115
Supplemental Disclosures:	
Non-cash financing activities:	
Additional paid-in capital from discount on note payable to affiliate	\$ 17,569
Increase in deferred offering costs, and in related accounts payable and accrued	\$ 40,211
expenses	

The accompanying notes are an integral part of the financial statements.

F-6

TABLE OF CONTENTS

Note 1 — Organization and Nature of Business Operations

Camden Learning Corporation (the "Company") is a blank check company incorporated in the state of Delaware on April 10, 2007 for the purpose of effecting a merger, capital stock exchange, stock purchase, asset acquisition or other similar business combination with one or more operating businesses in the education industry. The Company is majority owned by Camden Learning, LLC, whose members are Camden Partners Strategic Fund III, LP and Camden Partners Strategic Fund III-A, LP (see Note 4).

At September 30, 2007, the Company had not commenced any operations. All activity through September 30, 2007 relates to the Company's formation and to the proposed public offering described below. The Company has selected December 31 as its fiscal year end.

The financial statements give retroactive effect to a common stock split in the form of a stock dividend of 0.3888888 shares of common stock for each outstanding share of common stock declared and paid as of November 20, 2007.

The Company's ability to commence operations is contingent upon obtaining adequate financial resources through a proposed public offering ("Proposed Offering") which is discussed in Note 3. The Company's management has broad discretion with respect to the specific application of the net proceeds of this Proposed Offering, although substantially all of the net proceeds of the Proposed Offering are intended to be applied toward effecting a merger, capital stock exchange, stock purchase, asset acquisition or other similar business combination with one or more operating businesses in the education industry. As used herein, a "Business Combination" shall mean the merger, capital stock exchange, asset acquisition or other similar business combination with one or more operating businesses in the education industry having, collectively, a fair market value of at least 80.0% of the amount in the Company's trust account, less the deferred underwriting discount and commissions and taxes payable at the time of such transaction. The trust account will be maintained by Continental Stock Transfer & Trust Company pursuant to an investment management trust agreement to be signed upon the date of the prospectus for the Proposed Offering.

Upon closing of the Proposed Offering, approximately 99% of the proceeds (\$49.5 million, or \$56.7 million if the over-allotment option is exercised in full) of this offering will be placed in a trust account invested until the earlier of (i) the consummation of the Company's first Business Combination or (ii) the dissolution of the Company. The proceeds in the trust account include the deferred underwriting discount of \$1,500,000 (\$1,725,000 if the over-allotment option is exercised in full) that will be released to the underwriter if a Business Combination is completed (subject to a \$0.24 per share reduction for public stockholders who exercised their conversion rights). Interest (after taxes) earned on assets held in the trust account will remain in the trust. However, up to \$600,000 of the interest earned on the trust account, and amounts required for payment of taxes on interest earned, may be released to the Company to cover a portion of the Company's operating expenses and expenses incurred in connection with the Company's dissolution and liquidation, if a Business Combination is not consummated.

The Company will seek stockholder approval before it will effect any Business Combination. In connection with the stockholder vote required to approve any Business Combination, the Company's existing stockholders including all of the Company's officers, directors and advisors have agreed to vote the shares of common stock then-owned by them in accordance with the majority of the shares of common stock voted by the Public Stockholders. "Public Stockholders" is defined as the holders of common stock sold as part of the units in the Proposed Offering or in the aftermarket. The Company will proceed with a Business Combination only if a majority of the shares of common stock voted by the Public Stockholders are voted in favor of the Business Combination and Public Stockholders owning up to one share less than 30% of the shares sold in the Public Offering exercise their right to convert their shares into a pro rata share of the aggregate amount then on deposit in the trust account. If a majority of the shares of common stock voted by the Public Stockholders are not voted in favor of a proposed initial Business Combination but 24 months has not yet passed since

F-7

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

NOTES TO FINANCIAL STATEMENTS

Note 1 — Organization and Nature of Business Operations – (continued)

closing of the Proposed Offering, the Company may combine with another Target Business meeting the fair market value criterion described above.

The Company's certificate of incorporation filed with the State of Delaware includes a requirement that all proposed Business Combinations be presented to stockholders for approval; a prohibition against completing a Business Combination if 30% or more of the Company's stockholders exercise their redemption rights in lieu of approving a Business Combination; a provision giving stockholders who vote against a Business Combination the right to redeem their shares for a pro rata portion of the trust account in lieu of participating in a proposed Business Combination; and a requirement that if the Company does not consummate a Business Combination within 24 months from the date of the prospectus for the Proposed Offering, the Company will dissolve and liquidate, including liquidation of the trust account for the benefit of the public stockholders.

Public Stockholders voting against a Business Combination will be entitled to redeem their stock for a pro rata share

of the total amount on deposit in the trust account including the \$0.24 per share deferred underwriter's discount, and including any interest earned net of income taxes on their portion of the trust account, net of up to \$600,000 of the interest less income taxes thereon earned on the trust account which may be released to the Company to cover a portion of the Company's operating expenses if a Business Combination is approved and completed. Public Stockholders who convert their stock into their share of the trust account will continue to have the right to exercise any Warrants they may hold.

The Company will dissolve and promptly distribute only to its Public Stockholders the amount in the trust account, less any income taxes payable on interest income, plus any remaining net assets if the Company does not effect a Business Combination within 24 months after consummation of the Proposed Offering. In the event of dissolution, it is likely that the per share value of the residual assets remaining available for distribution (including trust account assets) will be less than the initial public offering price per share in the Proposed Offering (assuming no value is attributed to the Warrants contained in the units to be offered in the Proposed Offering discussed in Note 3).

The Company's existing stockholders have agreed that, on the date of the prospectus for the Public Offering, they would place the shares they owned before the Public Offering into an escrow account, and with limited exceptions, these shares will not be transferable and will not be released from escrow until one year after consummation of a Business Combination. If the Company is forced to dissolve or liquidate, these shares will be cancelled. Additionally, the insider warrants (see Note 4) will be placed into the escrow account, and subject to limited exceptions, will not be transferable and will not be released from escrow until the 90 th day following the completion of a Business Combination.

If holders of more than 20% of the shares sold in the Public Offering vote against a proposed Business Combination and seek to exercise their redemption rights and the Business Combination is consummated, the Company's existing stockholders have agreed to forfeit, on a pro rata basis, a number of the initial 1,562,500 shares of the Company's common stock purchased, up to a maximum of 195,312 shares, so that the existing stockholders will collectively own no more than 23.81% (without regard to any purchase of units in the Proposed Offering, any open market purchases or private purchases of units directly from the Company) of the Company's outstanding common stock immediately prior to the consummation of the Business Combination.

Note 2 — Summary of Significant Accounting Policies

Loss Per Common Share

Basic and diluted net loss per share is computed by dividing net loss by the weighted average number of common shares outstanding for the period.

F-8

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

NOTES TO FINANCIAL STATEMENTS

Note 2 — Summary of Significant Accounting Policies – (continued)

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of expenses during the reporting period. Actual results could differ from those estimates.

Income Taxes

Deferred income taxes are provided for the differences between the bases of assets and liabilities for financial reporting and income tax purposes. A valuation allowance is established when necessary to reduce deferred tax assets to the amount expected to be realized.

The company recorded a deferred income tax asset for the tax effect of net operating loss carry-forwards and temporary differences, aggregating approximately \$1,800. In recognition of the uncertainty regarding the ultimate amount of income tax benefits to be derived, the Company has recorded a full valuation allowance at September 30, 2007.

The effective tax rate differs from the statutory rate of 34% due to a permanent difference relating to accretion of interest and the increase in the valuation allowance.

Deferred Offering Costs

The costs associated with the Company's proposed initial public offering have been recorded as deferred offering costs and will reduce additional paid in capital if the offering is successful. Should the offering not be consummated, the deferred offering costs will be recognized as an expense of the Company.

Future legal fees of approximately \$175,000 are contingent upon consummation of an initial public offering and/or business combination.

Recently Issued Accounting Pronouncements

In July 2006, the Financial Accounting Standards Board ("FASB") issued Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"), which addresses the accounting for uncertainty in income taxes recognized in the financial statements in accordance with FASB Statement No. 109, "Accounting for Income Taxes." FIN 48 provides guidance on the financial statement recognition and measurement of a tax position taken on the Company's tax return. FIN 48 also provides guidance on classification, interest and penalties, accounting in interim periods, disclosure and transition. FIN 48 is effective for interim periods of fiscal years beginning after December 15, 2006. Adoption of Fin 48 did not have a material impact on the Company's financial position or results of operations.

In September 2006, the FASB issued Statement of Financial Accounting Standards No. 157, "Fair Value Measurements" ("SFAS No. 157"). SFAS No. 157 clarifies the principle that fair value should be based on the assumptions market participants would use when pricing an asset or liability and establishes a fair value hierarchy that prioritizes the information used to develop those assumptions. Under the standard, fair value measurements would be separately disclosed by level within the fair value hierarchy. SFAS No. 157 is effective for financial statements issued for fiscal years beginning after November 15, 2007, and interim periods within those fiscal years, with early adoption permitted. The Company does not believe that SFAS No. 157 would have a material effect on the accompanying financial statements.

In February 2007, FASB issued SFAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities" including an amendment of FASB Statement 115. This statement provides companies with an option to report selected financial assets and liabilities at fair value. This statement is effective for fiscal years

F-9

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

NOTES TO FINANCIAL STATEMENTS

Note 2 — Summary of Significant Accounting Policies – (continued)

beginning after November 15, 2007 with early adoption permitted. The Company is assessing SFAS No. 159 and has not yet determined the impact that the adoption of SFAS No. 159 will have on its results of operations or financial position.

Note 3 — Proposed Public Offering

The Proposed Offering calls for the Company to offer for public sale 6,250,000 units ("Units") at a price of \$8.00 per unit (7,187,500 units if the over-allotment option is exercised in full). Each Unit consists of one share of the Company's common stock, \$.0001 par value, and one warrant. Each warrant will entitle the holder to purchase from the Company one share of common stock at an exercise price of \$5.50 commencing the later of the completion of a Business Combination with a Target Business or one year from the date of the prospectus for the Public Offering and expiring four years from the date of the prospectus, unless earlier redeemed. The warrants will be redeemable at the Company's option, at a price of \$0.01 per warrant upon 30 days' written notice after the warrants become exercisable, only in the event that the last price of the common stock is at least \$11.50 per share for any 20 trading days within a 30 trading day period ending on the third business day prior to the date on which notice of redemption is given.

In accordance with the Warrant Agreement related to the warrants (the "Warrant Agreement"), the Company is only required to use its best efforts to effect the registration of the shares of common stock underlying the Warrants. The Company will not be obligated to deliver securities, and there are no contractual penalties for failure to deliver securities, if a registration statement is not effective at the time of exercise. Additionally, in the event that a registration statement is not effective at the time of exercise, the holder of a warrant shall not be entitled to exercise such warrant and in no event (whether in the case of a registration statement not being effective or otherwise) will the Company be required to net cash settle the warrant exercise. Consequently, the warrants may expire unexercised.

The Company has agreed to sell to the underwriters, for \$100, an option to purchase up to a total of 625,000 units exercisable on a cashless basis at \$9.60 per unit commencing one year from the date of the prospectus and expiring five years from the date of the prospectus. The units issuable upon exercise of this option are identical to those being sold in the offering, except that the public warrants in these units have an exercise price of \$6.71. The sale of the option will be accounted for as a cost attributable to the proposed offering. Accordingly, there will be no net impact on the Company's financial position or results of operations, except for the recording of the \$100 proceeds from the sale. The Company has estimated, based upon a Black-Scholes model, that the fair value of the option on the date of sale would be approximately \$2,430,000, using an expected life of five years, volatility of 59.4%, and a risk-free interest rate of 3.08%. However, because the units do not have a trading history, the volatility assumption is based on information currently available to the Company. The Company believes the volatility estimate calculated is a reasonable benchmark to use in estimating the expected volatility of the units. The volatility calculation is based on the most recent trading day average volatility of publicly traded companies providing educational services with market capitalizations less than \$500 million. Although an expected life of five years was used in the calculation, if the Company does not consummate a Business Combination within the prescribed time period and automatically dissolves and subsequently liquidates the trust account, the option will become worthless.

Note 4 — Note Payable to Affiliate and Related Party Transactions

The Company issued an aggregate \$200,000 unsecured promissory note to Camden Learning, LLC, an affiliate, on April 26, 2007. The note is interest bearing at an annual rate of 4.9% and both principal and interest are payable on the earlier of April 26, 2008 or the consummation of the Public Offering of the Company.

On April 26, 2007 the note was recorded as a liability in the amount of \$182,431, net of a discount in the amount of \$17,569, which has been credited to additional paid-in capital, based on an imputed interest rate

F-10

TABLE OF CONTENTS

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

NOTES TO FINANCIAL STATEMENTS

Note 4 — Note Payable to Affiliate and Related Party Transactions – (continued)

of 15% per annum. The \$17,569 discount will be accreted by charges to interest expense over the term of the note using the interest method. The amount of interest expense recorded through September 30, 2007 totaled \$10,940, including amounts accrued at 4.9% per annum. In its computations of the discount on the note, the Company considered that the loan is unsecured, the Company has no operations and the Company will be able to repay the loan only in the event of a successful public offering, as to which there can be no assurance. In making its computation, the Company also considered the related party nature of the note, the below-market stated interest rate, the equity-like risks associated with the note and the higher interest rates commonly associated with bridge financings.

As of September 30, 2007, the Company is owed \$1,111 from Camden Learning, LLC, the Company's principal stockholder, in conjunction with acquiring the common shares of two former directors of the Company. The entire amount was collected in October 2007.

The Company has agreed to pay up to \$7,500 a month in total for certain general and administrative services, including but not limited to receptionist, secretarial and general office services, to Camden Partners Holdings, LLC. Services will commence on the effective date of the offering and will terminate upon the earlier of (i) the completion of the Company's Business Combination or (ii) the Company's dissolution.

Camden Learning, LLC has agreed to acquire warrants to purchase 2,800,000 shares of Common Stock from the Company at a price of \$1.00 per warrant for a total of \$2,800,000 in a private placement prior to the completion of the offering. The terms of these warrants are identical to the terms of the warrants to be issued in the Proposed Offering, except that these insider warrants will not be subject to redemption and may be exercised on a cashless basis, in each case if held by the initial holder thereof or its permitted assigns, and may not be sold, assigned or transferred prior to the 90th day following consummation of a Business Combination. The holder of these insider warrants will not have any right to any liquidation distributions with respect to shares underlying these warrants if the Company fails to consummate a Business Combination, in which event these warrants will expire worthless. The sale of the warrants to Camden Learning, LLC is not expected to result in the recognition of stock-based compensation expense because they are being sold at or above fair market value.

Camden Learning, LLC has agreed to indemnify the Company for claims of creditors that have not executed a valid and binding waiver of their rights to seek payments of amounts due to them out of the trust account. The Company believes the likelihood of Camden Learning, LLC having to indemnify the trust account is minimal.

The Company's principal stockholder has entered into an agreement with the underwriter pursuant to which it will place limit orders to purchase up to an additional \$4,000,000 of the Company's common stock in the open market commencing the later of (i) ten business days after the Company files its current report on Form 8-K announcing its execution of a definitive agreement for a Business Combination and (ii) 60 calendar days after the end of the restricted period in connection with this offering, as defined under the Securities Exchange Act of 1934, and ending on the business day preceding the record date of the stockholders' meeting at which a Business Combination is to be approved. In the event the Company's principal stockholder does not purchase \$4,000,000 of the Company's common stock in the open market, the stockholder has agreed to purchase from the Company in a private placement a number of units identical to the units to be sold in the Proposed Offering at a purchase price of \$8.00 per unit until it has spent, together with the aforementioned open market purchases, an aggregate of \$4,000,000 for purchase of the Company's common stock.

Note 5 — Common Stock

In April 2007, the Company issued 1,125,000 shares of common stock (1,562,500 shares after giving effect to the stock split described in Note 1) to the Initial Stockholders for an aggregate amount of \$25,000.

CAMDEN LEARNING CORPORATION (a corporation in the development stage)

NOTES TO FINANCIAL STATEMENTS

Note 6 — Preferred Stock

The Company is authorized to issue 1,000,000 shares of preferred stock with such designations, voting and other rights and preferences as may be determined from time to time by the Board of Directors.

F-12

TABLE OF CONTENTS

- -

Until [], 2007, all dealers that effect transactions in these securities, whether or not participating in this offering, may be required to deliver a prospectus. This is in addition to the dealers' obligation to deliver a prospectus when acting as underwriters and with respect to their unsold allotments or subscriptions.

No dealer, salesperson or any other person is authorized to give any information or make any representations in connection with this offering other than those contained in this prospectus and, if given or made, the information or representations must not be relied upon as having been authorized by us. This prospectus does not constitute an offer to sell or a solicitation of an offer to buy any security other than the securities offered by this prospectus, or an offer to sell or a solicitation of an offer to buy any securities by anyone in any jurisdiction in which the offer or solicitation is not authorized or is unlawful.

	Page
Prospectus Summary	1
Summary Financial Data	<u>16</u>
Risk Factors	<u>17</u>
<u>Use of Proceeds</u>	<u>38</u>
<u>Dilution</u>	<u>41</u>
Capitalization	<u>43</u>
Management's Discussion and Analysis of	<u>44</u>
Financial Condition and Results of	
<u>Operations</u>	
<u>Proposed Business</u>	<u>47</u>
<u>Management</u>	<u>61</u>
Principal Stockholders	<u>68</u>
Certain Relationships and Related	<u>71</u>
<u>Transactions</u>	
Description of Securities	<u>74</u>
<u>Underwriting</u>	<u>80</u>
<u>Legal Matters</u>	<u>84</u>
<u>Experts</u>	<u>84</u>
Where You Can Find Additional	<u>84</u>
<u>Information</u>	
Index to Financial Statements	F-
	1

\$50,000,000

CAMDEN LEARNING CORPORATION



6,250,000 Units

PROSPECTUS

MORGAN JOSEPH

Ferris, Baker Watts
Incorporated

Legend Merchant Group

_____, 2007

- -

TABLE OF CONTENTS

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

Item 13. Other Expenses of Issuance and Distribution.

The estimated expenses payable by us in connection with the offering described in this registration statement (other than the underwriting discount and commissions) will be as follows:

Initial Trustees' fee	\$ 1,000.00 ⁽¹⁾
SEC Registration Fee	\$ 3,291.80
FINRA filing fee	\$ 11,222.00
Accounting fees and expenses	50,000.00 ⁽²⁾
Printing and engraving expenses	40,000.00
Legal fees and expenses	250,000.00 ⁽³⁾
Blue sky services and expenses	20,000.00
Miscellaneous	24,486.20 ⁽⁴⁾
Total	\$400,000.00

- (1) In addition to the initial acceptance fee that is charged by Continental Stock Transfer & Trust Company, as trustee following the offering, the registrant will be required to pay to Continental Stock Transfer & Trust Company annual fees of approximately \$3,000 for acting as trustee, approximately \$4,800 for acting as transfer agent of the registrant's common stock, approximately \$2,400 for acting as warrant agent for the registrant's warrants and approximately \$2,400 for acting as escrow agent.
- (2) Approximately \$50,000 in additional accounting fees may be due in connection with the initial public offering, which such additional fees shall be paid, if at all, from our working capital.
- (3) A portion of the legal fees payable to Ellenoff Grossman & Schole LLP, our legal counsel, may be deferred and is contingent on our consummating a business combination.
- (4) This amount represents additional expenses that may be incurred by us in connection with the offering over and above those specifically listed above, including distribution and mailing costs.

Item 14. Indemnification of Directors and Officers.

Our amended and restated certificate of incorporation provides that all of our directors, officers, employees and agents shall be entitled to be indemnified by us to the fullest extent permitted by Section 145 of the Delaware General Corporation Law.

Section 145 of the Delaware General Corporation Law concerning indemnification of officers, directors, employees and agents is set forth below.

"Section 145. Indemnification of officers, directors, employees and agents; insurance.

- (a) A corporation shall have power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe the person's conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that the person's conduct was unlawful.
- (b) A corporation shall have power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to

II-1

TABLE OF CONTENTS

procure a judgment in its favor by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by the person in connection with the defense or settlement of such action or suit if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

- (c) To the extent that a present or former director or officer of a corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in subsections (a) and (b) of this section, or in defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith.
- (d) Any indemnification under subsections (a) and (b) of this section (unless ordered by a court) shall be made by the corporation only as authorized in the specific case upon a determination that indemnification of the present or former director, officer, employee or agent is proper in the circumstances because the person has met the applicable standard of conduct set forth in subsections (a) and (b) of this section. Such determination shall be made, with respect to a person who is a director or officer at the time of such determination, (1) by a majority vote of the directors who are not parties to such action, suit or proceeding, even though less than a quorum, or (2) by a committee of such directors designated by majority vote of such directors, even though less than a quorum, or (3) if there are no such directors, or if such directors so direct, by independent legal counsel in a written opinion, or (4) by the stockholders.
- (e) Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative or investigative action, suit or proceeding may be paid by the corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that such person is not entitled to be indemnified by the corporation as authorized in this section. Such expenses (including attorneys' fees) incurred by former directors and

officers or other employees and agents may be so paid upon such terms and conditions, if any, as the corporation deems appropriate.

- (f) The indemnification and advancement of expenses provided by, or granted pursuant to, the other subsections of this section shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any bylaw, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in such person's official capacity and as to action in another capacity while holding such office.
- (g) A corporation shall have power to purchase and maintain insurance on behalf of any person who is or was director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the corporation would have the power to indemnify such person against such liability under this section.
- (h) For purposes of this section, references to "the corporation" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under this section with respect to the resulting or surviving corporation as such person would have with respect to such constituent corporation if its separate existence had continued.

II-2

TABLE OF CONTENTS

- (i) For purposes of this section, references to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to any employee benefit plan; and references to "serving at the request of the corporation" shall include any service as a director, officer, employee or agent of the corporation which imposes duties on, or involves services by, such director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the corporation" as referred to in this section.
- (j) The indemnification and advancement of expenses provided by, or granted pursuant to, this section shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.
- (k) The Court of Chancery is hereby vested with exclusive jurisdiction to hear and determine all actions for advancement of expenses or indemnification brought under this section or under any bylaw, agreement, vote of stockholders or disinterested directors, or otherwise. The Court of Chancery may summarily determine a corporation's obligation to advance expenses (including attorneys' fees)."

Insofar as indemnification for liabilities arising under the Securities Act of 1933, as amended, may be permitted to our directors, officers, and controlling persons pursuant to the foregoing provisions, or otherwise, we have been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act of 1933, as amended, and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment of expenses incurred or paid by a director, officer or controlling person in a successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, we will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to the court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act of 1933, as amended, and will be governed by the final adjudication of such issue.

Paragraph B of Article Eighth of our amended and restated certificate of incorporation provides:

"The Corporation, to the full extent permitted by Section 145 of the GCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby."

Pursuant to the Underwriting Agreement filed as Exhibit 1.1 to this Registration Statement, we have agreed to indemnify the underwriters, and the underwriters have agreed to indemnify us, against certain civil liabilities that may be incurred in connection with this offering, including certain liabilities under the Securities Act of 1933, as amended.

Item 15. Recent Sales of Unregistered Securities.

(a) During the past three years, we sold the following shares of common stock without registration under the Securities Act of 1933, as amended:

Stockholders Number of Shares

Camden Learning, LLC	1,000,000
Jack L. Brozman	25,000
Therese Kreig Crane, Ed.D	25,000
Ronald Tomalis	25,000
Harry T. Wilkins	25,000
William Jews	25,000

Such shares were issued on April 10, 2007 in connection with our organization pursuant to the exemption from registration contained in Section 4(2) of the Securities Act of 1933, as amended, as they were sold to

II-3

TABLE OF CONTENTS

sophisticated, wealthy non "U.S. Person" individuals. The shares issued to the individuals and entities above were sold for an aggregate offering price of \$25,000 at an average purchase price of approximately \$0.02 per share. No underwriting discounts or commissions were paid with respect to such sales.

Effective November 20, 2007, our board of directors authorized a forward stock split in the form of a stock dividend of 0.3888888 shares of common stock for each outstanting share of common stock, effectively lowering the purchase price to approximately \$0.016 per share.

Our sponsor has agreed to purchase an aggregate of 2.800,000 warrants from us at a purchase price of \$1.00 per warrant in a private placement that will occur immediately prior to this offering in a transaction pursuant to, and in accordance with, Regulation D under the Securities Act of 1933, as amended. The obligation to purchase the warrants undertaken by the sponsor was made pursuant to an amended and restated Subscription Agreement, dated as of November 20, 2007 (the form of which was filed as Exhibit 4.4 to the Registration Statement on Form S-1). Such obligation was made prior to the filing of the Registration Statement, and was undertaken by the sponsor, an institutional investor. Consequently, it is a separate private placement that is not integrated with our public offering. We have granted the holders of such warrants demand and "piggy-back" registration rights with respect to the 2,800,000 shares underlying the warrants at any time commencing on the date we announce that we have entered into a letter of intent with respect to a proposed business combination. The demand registration may be exercised by the holders of a majority of such warrants. We will bear the expenses incurred in connection with the filing of any such registration statements.

In addition, if we take advantage of increasing the size of the offering pursuant to Rule 462(b) under the Securities Act of 1933, as amended, we may effect a stock dividend in such amount to maintain the existing stockholders' collective ownership at 20% of our issued and outstanding shares of common stock upon consummation of the offering. If we decrease the size of the offering we will effect a reverse split of our common stock in such amount to maintain the existing stockholders allocated ownership at 20% of our issued and outstanding common stock upon the consummation of this offering.

Item 16. Exhibits and Financial Statement Schedules.

(a) The following exhibits are filed as part of this Registration Statement:

Exhibit	Description
No.	
1.1	Form of Underwriting Agreement.
3.1	Certificate of Incorporation.*
3.2	Form of Amended and Restated Certificate of Incorporation.
3.3	By-laws.*
4.1	Specimen Unit Certificate.*
4.2	Specimen Common Stock Certificate.*
4.3	Specimen Warrant Certificate.*
4.4	Form of Warrant Agreement between Continental Stock Transfer & Trust Company and the
	Registrant.
4.5	Form of Unit Option Purchase Agreement between the Registrant and Morgan Joseph &
	Co. Inc.
5.1	Opinion of Ellenoff Grossman & Schole LLP.
10.1.1	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and Camden Learning,
	LLC.
10.1.2	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and Donald W.
	Hughes.*
10.1.3	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and David L. Warnock.*
10.1.4	Intentionally Omitted.
10.1.5	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and Therese Kreig
	Crane.*
10.1.6	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and Ronald Tomalis.*
10.1.7	Letter Agreement among the Registrant, Morgan Joseph & Co. Inc. and William Jews.*

- Letter Agreement among the Hegistrant, Morgan Joseph & Co. Inc. and William Jews.

TABLE OF CONTENTS

Exhibit No.	Description
10.3	Form of Securities Escrow Agreement between the Registrant, Continental Stock Transfer
	& Trust Company and the Initial Stockholders.
10.4	Form of Registration Rights Agreement among the Registrant and the Initial Stockholders.
10.5	Lease/Office Services Agreement dated November 20, 2007 by and among the Registrant
	and Camden Partners Holdings, LLC.
10.6	Amended and Restated Subscription Agreement between the Registrant and Sponsor.
10.7	Promissory Note in the amount of \$200,000 dated April 26, 2007 issued in favor of Camden
	Learning, LLC.*
10.8	Right of First Refusal Agreement by and among Camden Learning, LLC, Camden Partners
	Strategic Fund III, L.P. and Camden Partners Strategic Fund III-A, L.P.*
23.1	Consent of Eisner LLP.
23.2	Consent of Ellenoff Grossman & Schole LLP (included in Exhibit 5.1).
99.1	Code of Ethics.*

^{*} Previously filed

Item 17. Undertakings.

- (a) The undersigned registrant hereby undertakes:
- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - i. To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933, as amended;
 - ii. To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement.
 - iii. To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (2) That, for the purpose of determining any liability under the Securities Act of 1933, as amended, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (3) That, for the purpose of determining liability under the Securities Act of 1933, as amended, to any purchaser:
 - i. If the registrant is subject to Rule 430C, each prospectus filed pursuant to Rule 424(b) as part of a registration statement relating to an offering, other than registration statements relying on Rule 430B or other than prospectuses filed in reliance on Rule 430A, shall be deemed to be part of and included in the registration statement as of the date it is first used after effectiveness. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into th registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such first use, supersede or

TABLE OF CONTENTS

- (4) That, for the purpose of determining liability of the registrant under the Securities Act of 1933, as amended, to any purchaser in the initial distribution of the securities: The undersigned registrant undertakes that in a primary offering of securities of the undersigned registrant pursuant to this registration statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:
 - Any preliminary prospectus of the undersigned registrant relating to the offering required to be filed pursuant to Rule 424;
 - ii. Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned registrant or used or referred to by the undersigned registrant;
 - iii. The portion of any other free writing prospectus relating to the offering containing material information about the undersigned registrant or its securities provided by or on behalf of the undersigned registrant; and
 - iv. Any other communication that is an offer in the offering made by the undersigned registrant to the purchaser.
- (5) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned hereby undertakes to provide to the underwriter at the closing specified in the underwriting agreements, certificates in such denominations and registered in such names as required by the underwriter to permit prompt delivery to each purchaser.
- (c) Insofar as indemnification for liabilities arising under the Securities Act of 1933, as amended, may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.
 - (d) The undersigned registrant hereby undertakes that:
 - (1) For purposes of determining any liability under the Securities Act of 1933, as amended, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act of 1933, as amended, shall be deemed to be part of this registration statement as of the time it was declared effective.
 - (2) For the purpose of determining any liability under the Securities Act of 1933, as amended, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

II-6

TABLE OF CONTENTS

SIGNATURE

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this registration statement on Form S-1 to be signed on its behalf by the undersigned, thereunto duly authorized, in Baltimore, Maryland, on the 26th day of November, 2007.

CAMDEN LEARNING CORPORATION

By:/s/ David L. Warnock

Name:

David L. Warnock
Title: Chief Executive Officer
(Principal Executive Officer)

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

Name	Position	Date
/s/ David L. Warnock	President, Chief Executive Officer and Dalidirman (Principal Executive Officer)	November 26, 2007
L. Warnock /s/ Donald W. Hughes	Chief Financial Officer, Secretary	November 26, 2007
W. Hughes	D \Pain cipal Financial and Accounting Officer)	

/s/ Therese Kreig Crane	Director	November 26, 2007
	Therese	
Kreig Crane, Ed.D		
/s/ Ronald Tomalis	Director	November 26, 2007
	Ronald	
Tomalis		
/s/ William Jews	Director	November 26, 2007
-	William	
Jews		
	II-7	

EXHIBIT 1.1

UNDERWRITING AGREEMENT

between

CAMDEN LEARNING CORPORATION

and

MORGAN JOSEPH & CO. INC.

Dated: , 2007

CAMDEN LEARNING CORPORATION

UNDERWRITING AGREEMENT

New York, New York , 2007

Morgan Joseph & Co. Inc. 600 Fifth Avenue, 19th Floor New York, New York 10020

As Representative of the Several Underwriters named in Schedule I hereto

Re: Public Offering of Securities

Ladies and Gentlemen:

The undersigned, Camden Learning Corporation, a Delaware corporation ("<u>Company</u>"), hereby confirms its agreement with Morgan Joseph & Co. Inc. ("<u>Morgan Joseph & Co.</u>" and also referred to herein variously as "you," or the "<u>Representative</u>") and with the other underwriters named on Schedule I hereto for which Morgan Joseph & Co. is acting as Representative (the Representative and the other underwriters being collectively called the "<u>Underwriters</u>" or, individually, an "<u>Underwriter</u>") as follows:

1. Purchase and Sale of Securities.

1.1 Firm Securities.

1.1.1 Purchase of Firm Units. On the basis of the representations and warranties herein contained, but subject to the terms and conditions herein set forth, the Company agrees to issue and sell, severally and not jointly, to the several Underwriters, an aggregate of 6,250,000 units ("Firm Units") of the Company, at a purchase price (net of discounts and commissions) of \$7.44 per Firm Unit. The Underwriters, severally and not jointly, agree to purchase from the Company the number of Firm Units set forth opposite their respective names on Schedule I attached hereto and made a part hereof at a purchase price (net of discounts and commissions) of \$7.44 per Firm Unit. The Firm Units are to be offered initially to the public ("Offering") at the offering price of \$8.00 per Firm Unit. Each Firm Unit consists of one share of the Company's common stock, par value \$.0001 per share ("Common Stock"), and one warrant ("Warrant"). The shares of Common Stock and the Warrants included in the Firm Units will not be separately transferable until 90 days after the effective date ("Effective Date") of the Registration Statement (as defined in Section 2.1.1 hereof) unless the Representative informs the Company, in writing, of its decision to allow earlier separate trading based on its assessment of the relative strengths of the securities markets and small capitalization companies in general, and the trading pattern of, and demand for, the Company's securities in particular, but in no event will the Representative allow separate trading until the business day after (i) the Company has filed with the Securities and Exchange Commission (the "Commission") a Current Report on Form 8-K which includes an audited balance sheet reflecting the Company's receipt of the proceeds of the Offering and the Private Placement (as defined in Section 2.22.4), including any proceeds the Company receives from the exercise of the Over-allotment Option (as defined in Section 1.2.1), if such option is exercised prior to the filing of the Form 8-K, (ii) the Company has filed with the Commission a Current Report on Form 8-K and issued a press release announcing when such separate trading will begin, and (iii) the expiration of the Over-allotment Option or its exercise in full. Each Warrant entitles its holder to exercise it to purchase one share of Common Stock for \$5.50 during the period commencing on the later of the consummation by the Company of its "Business Combination" or one year from the Effective Date of the Registration Statement and terminating on the four-year anniversary of the Effective Date. "Business Combination" shall mean any merger, capital stock exchange, asset acquisition or other similar business combination consummated by the Company with one or more operating businesses in the education industry (as described more fully in the Registration Statement (as defined in Section 2.1.1 below)).

1.1.2 Payment and Delivery. Delivery and payment for the Firm Units shall be made at 10:00 a.m., New York City time, on the fourth Business Day (as defined below) following the effective date of the Registration Statement or at such earlier time as shall be agreed upon by the Representative and the Company at the offices of McDermott Will & Emery LLP ("McDermott") or at such other place as shall be agreed upon by the Representative and the Company. The hour and date of delivery and payment for the Firm Units are referred to herein as the "Closing Date." Payment for the Firm Units shall be made on the Closing Date at the Representative's election by wire transfer in Federal (same day) funds or by certified or bank cashier's check(s) in New York Clearing House funds, payable as follows: \$48,000,000 of the proceeds received by the Company for the Firm Units and the Placement Warrants (as defined in Section 2.22.4) and \$1,500,000 of the Deferred Fees (as defined in Section 1.1.3) shall be deposited (or with respect to the \$2,800,000 of the proceeds from the sale of the Placement Warrants shall have been deposited on or prior to the date hereof) in the trust account established by the Company for the benefit of the public stockholders as described in the Registration Statement ("Trust Account") pursuant to the terms of an Investment Management Trust Agreement ("Trust Agreement") between the Company and Continental Stock Transfer & Trust Company ("CST") and the remaining proceeds shall be paid (subject to Section 3.12 hereof) to the order of the Company upon delivery to you of certificates (in form and substance satisfactory to the Underwriters) representing the Firm Units (or through the facilities of the Depository Trust Company ("DTC") for the account of the Underwriters. The Firm Units shall be registered in such name or names and in such authorized denominations as the Representative may request in writing at least two full Business Days prior to the Closing Date. The Company will permit the Representative to examine and package the Firm Units for delivery, at least one full Business Day prior to the Closing Date. The Company shall not be obligated to sell or deliver the Firm Units except upon tender of payment by the Representative for all the Firm Units. "Business Day" shall mean any day other than a Saturday, a Sunday or a legal holiday or a day on which banking institutions or trust companies are authorized or obligated by law to close in New York City.

1.1.3 <u>Deferral of a Portion of Underwriters' Discount</u>. On the Closing Date and, if applicable, on the Option Closing Date (as defined in Section 1.2.2), Morgan Joseph & Co. agrees to deposit into the Trust Account a portion of the Underwriters' discount equal to \$0.24 per Unit in the Offering and, if applicable, a portion of the discount equal to \$0.24 per Option Unit (as defined in Section 1.2.1) (the "<u>Deferred Fees</u>") until the earlier of the completion of a Business Combination or the liquidation of the Trust Account. Upon the consummation of a Business Combination, Morgan Joseph & Co. shall promptly receive the Deferred Fees, but only with respect to those units as to which the component shares have not been redeemed for cash by those stockholders who voted against the Business Combination and exercised their redemption rights. In the event that the Company is unable to consummate a Business Combination and CST, the trustee of the Trust Account, commences liquidation of the Trust Account, Morgan Joseph & Co. hereby agrees to the following: (i) to forfeit any rights or claims to the Deferred Fees and any interest accrued thereon; and (ii) that the Deferred Fees shall be distributed on a pro-rata basis among the holders of the shares of Common Stock included in the Units sold in the Offering along with any interest accrued thereon, net of taxes.

1.2 Over-Allotment Option.

1.2.1 Option Units. For the purposes of covering any over-allotments in connection with the distribution and sale of the Firm Units, the Underwriters are hereby granted, severally and not jointly, an option to purchase up to an additional 937,500 units from the Company ("Over-allotment Option"). Such additional 937,500 units, the net proceeds of which will be deposited in the Trust Account, are hereinafter referred to as "Option Units." The Firm Units and the Option Units are hereinafter collectively referred to as the "Units," and the Units, the shares of Common Stock and the Warrants included in the Units and the shares of Common Stock issuable upon exercise of the Warrants are hereinafter referred to collectively as the "Public Securities." The purchase price to be paid for the Option Units will be the same price per Option Unit as the price per Firm Unit set forth in Section 1.1.1 hereof.

1.2.2 Exercise of Option. The Over-allotment Option granted pursuant to Section 1.2.1 hereof may be exercised by the Representative as to all (at any time) or any part (from time to time) of the Option Units within 45 days after the Effective Date. The Underwriters will not be under any obligation to purchase any Option Units prior to the exercise of the Over-allotment Option. The Over-allotment Option granted hereby may be exercised by the giving of oral notice to the Company by the Representative, which must be confirmed in writing by overnight mail or facsimile transmission setting forth the number of Option Units to be purchased and the date and time for delivery of and payment for the Option Units (the "Option Closing Date"), which will not be later than five full Business Days after the date of the notice or such other time and in such other manner as shall be agreed upon by the Company and the Representative, at the offices of McDermott or at such other place as shall be agreed upon by the Company and the Representative. Upon exercise of the Over-allotment Option, the Company will become obligated to convey to the Underwriters, and, subject to the terms and conditions set forth herein, the Underwriters will become obligated to purchase, the number of Option Units specified in such notice.

1.2.3 Payment and Delivery. Payment for the Option Units shall be made on the Option Closing Date at the Representative's election by wire transfer in Federal (same day) funds or by certified or bank cashier's check(s) in New York Clearing House funds, payable as follows: \$7.44 per Option Unit, which includes \$0.24 of Deferred Fees per Option Unit, shall be deposited in the Trust Account pursuant to the Trust Agreement and the remaining proceeds shall be paid (subject to Section 3.12 hereof) to the order of the Company upon delivery to you of certificates (in form and substance satisfactory to the Underwriters) representing the Option Units (or through the facilities of DTC) for the account of the Underwriters. The certificates representing the Option Units to be delivered will be in such denominations and registered in such names as the Representative requests not less than two full Business Days prior to the Closing Date or the Option Closing Date, as the case may be, and will be made available to the Representative for inspection, checking and packaging at the aforesaid office of the Company's transfer agent or correspondent not less than one full Business Day prior to such Closing Date.

1.3 Representative's Purchase Option.

1.3.1 Purchase Option. The Company hereby agrees to issue and sell to the Representative (and/or its designees) on the Effective Date an option ("Representative's Purchase Option") for the purchase of an aggregate of 625,000 units ("Representative's Units") for an aggregate purchase price of \$100. Each of the Representative's Units is identical to the Firm Units including the warrants constituting the Units to purchase Common Stock (sometimes referred to as the "Representative's Warrants") except that the exercise price for the Representative's Warrants is \$6.71 per share. The Representative's Purchase Option shall be exercisable, in whole or in part, commencing on the later of the consummation of a Business Combination and one year from the Effective Date and expiring on the five-year anniversary of the Effective Date at an initial exercise price per Representative's Unit of \$9.60 (120% of the initial public offering price of a Unit) and may be exercised on a cashless basis. The Representative's Purchase Option, the Representative's Units, the Common Stock contained within the Representative's Units, the Representative's Warrants and the shares of Common Stock issuable upon exercise of the Representative's Warrants are hereinafter referred to collectively as the "Representative's Securities." The Public Securities and the Representative's Securities are hereinafter referred to collectively as the "Securities." The Representative understands and agrees that there are significant restrictions against transferring the Representative's Securities during the first year after the Effective Date, as set forth in Section 3 of the Representative's Purchase Option.

1.3.2 <u>Payment and Delivery</u>. Delivery and payment for the Representative's Purchase Option shall be made on the Closing Date. The Company shall deliver to the Underwriters, upon payment therefor, certificates for the Representative's Purchase Option in the name or names and in such authorized denominations as the Representative may request.

- 2. Representations and Warranties of the Company. The Company represents and warrants to the Underwriters as follows:
 - 2.1 Filing of Registration Statement.

2.1.1 Pursuant to the Act. The Company has filed with the Commission a registration statement and an amendment or amendments thereto, on Form S-1 (File No. 333-143098), including any related preliminary prospectus ("Preliminary Prospectus"), for the registration of the Securities under the Securities Act of 1933, as amended ("Act"), which registration statement and amendment or amendments have been prepared by the Company in conformity with the requirements of the Act, and the rules and regulations (the "Regulations") of the Commission under the Act. The conditions for use of Form S-1 to register the Offering under the Act, as set forth in the General Instructions to such Form, have been satisfied. Except as the context may otherwise require, such registration statement, as amended, on file with the Commission at the time the registration statement becomes effective (including the prospectus, financial statements, schedules, exhibits and all other documents filed as a part thereof or incorporated therein and all information deemed to be a part thereof as of such time pursuant to Rule 430A of the Regulations), is hereinafter called the "Registration Statement," and the form of the final prospectus dated the Effective Date included in the Registration Statement (or, if applicable, the form of final prospectus filed by the Company with the Commission pursuant to Rule 424 of the Regulations), is hereinafter called the "Prospectus." For purposes of this Agreement, "Time of Sale," as used in the Act, means 4:30 p.m. New York City time, on the date of this Agreement. Prior to the Time of Sale, the Company prepared a preliminary], 2007, for distribution by the Underwriters (the "Sale Preliminary Prospectus"). If the Company has Prospectus, dated [filed, or is required pursuant to the terms hereof to file, a Registration Statement pursuant to Rule 462(b) under the Act registering additional Securities of any type (a "Rule 462(b) Registration Statement"), then, unless otherwise specified, any reference herein to the term "Registration Statement" shall be deemed to include such Rule 462(b) Registration Statement. Other than a Rule 462(b) Registration Statement, which, if filed, becomes effective upon filing, no other document with respect to the Registration Statement has heretofore been filed with the Commission. All of the Public Securities have been registered under the Act pursuant to the Registration Statement or, if any Rule 462(b) Registration Statement is filed, will be duly registered for public sale under the Act with the filing of such Rule 462(b) Registration Statement. The Registration Statement has been declared effective by the Commission on the date hereof. If, subsequent to the date of this Agreement, the Company or the Representative has determined that at the Time of Sale the Sale Preliminary Prospectus includes an untrue statement of a material fact or omitted a statement of material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading and have agreed to provide an opportunity to purchasers of the Firm Units to terminate their old purchase contracts and enter into new purchase contracts, then the Sale Preliminary Prospectus will be deemed to include any additional information available to purchasers at the time of entry into the first such new purchase contract.

2.1.2 <u>Pursuant to the Exchange Act</u>. The Company has filed with the Commission a Form 8-A (File Number 000-52919) providing for the registration under the Securities Exchange Act of 1934, as amended ("<u>Exchange Act</u>"), of the Units, the Common Stock and the Warrants. The registration of the Units, Common Stock and Warrants under the Exchange Act has been declared effective by the Commission on the date hereof.

2.2 No Stop Orders, Etc. Neither the Commission nor, to the best of the Company's knowledge, any state regulatory authority has issued any order or threatened to issue any order preventing or suspending the use of any Sale Preliminary Prospectus or Prospectus or has instituted or, to the best of the Company's knowledge, threatened to institute any proceedings with respect to such an order.

2.3 <u>Disclosures in Registration Statement</u>.

2.3.1 10b-5 Representation. At the time the Registration Statement became effective and at all times subsequent thereto up to the Closing Date and the Option Closing Date, if any, the Registration Statement, the Sale Preliminary Prospectus and the Prospectus do and will contain all material statements that are required to be stated therein in accordance with the Act and the Regulations, and will in all material respects conform to the requirements of the Act and the Regulations; and neither the Registration Statement, the Sale Preliminary Prospectus nor the Prospectus, nor any amendment or supplement thereto, on their respective dates, nor the Sale Preliminary Prospectus as of the Time of Sale (or such subsequent Time of Sale pursuant to Section 2.1.1), does or will contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. When any Preliminary Prospectus or the Sale Preliminary Prospectus was first filed with the Commission (whether filed as part of the Registration Statement for the registration of the Securities or any amendment thereto or pursuant to Rule 424(a) of the Regulations) and when any amendment thereof or supplement thereto was first filed with the Commission, such Preliminary Prospectus or the Sale Preliminary Prospectus and any amendments thereof and supplements thereto complied or will comply in all material respects with the applicable provisions of the Act and the Regulations and did not and will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The representation and warranty made in this Section 2.3.1 does not apply to statements made or statements omitted in reliance upon and in conformity with written information furnished to the Company with respect to the Underwriters by the Representative expressly for use in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus or any amendment thereof or supplement thereto.

2.3.2 Disclosure of Agreements. The agreements and documents described in the Registration Statement. the Sale Preliminary Prospectus and the Prospectus conform to the descriptions thereof contained therein and there are no agreements or other documents required to be described in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus or to be filed with the Commission as exhibits to the Registration Statement, that have not been so described or filed. Each agreement or other instrument (however characterized or described) to which the Company is a party or by which its property or business is or may be bound or affected and (i) that is referred to in the Sale Preliminary Prospectus or the Prospectus, or (ii) is material to the Company's business, has been duly and validly executed by the Company, is in full force and effect and is enforceable against the Company and, to the Company's knowledge, the other parties thereto, in accordance with its terms, except (x) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (y) as enforceability of any indemnification or contribution provision may be limited under the Federal and state securities laws, and (z) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought, and none of such agreements or instruments has been assigned by the Company, and neither the Company nor, to the best of the Company's knowledge, any other party is in breach or default thereunder and, to the best of the Company's knowledge, no event has occurred that, with the lapse of time or the giving of notice, or both, would constitute a breach or default thereunder. To the best of the Company's knowledge, performance by the Company of the material provisions of such agreements or instruments will not result in a violation of any existing applicable law, rule, regulation, judgment, order or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its assets or businesses, including, without limitation, those relating to environmental laws and regulations.

- 2.3.3 <u>Prior Securities Transactions</u>. No securities of the Company have been sold by the Company or by or on behalf of, or for the benefit of, any person or persons controlling, controlled by, or under common control with the Company since the Company's formation, except as disclosed in the Registration Statement.
- 2.3.4 <u>Regulations</u>. The disclosures in the Registration Statement and the Sale Preliminary Prospectus concerning the effects of Federal, State and local regulation on the Company's business as currently contemplated are correct in all material respects and do not omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they were made, not misleading.

2.4 Changes After Dates in Registration Statement.

- 2.4.1 No Material Adverse Change. Since the respective dates as of which information is given in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, except as otherwise specifically stated therein, (i) there has been no material adverse change in the condition, financial or otherwise, or business prospects of the Company, (ii) there have been no material transactions entered into by the Company, other than as contemplated pursuant to this Agreement, and (iii) no member of the Company's management has resigned from any position with the Company.
- 2.4.2 <u>Recent Securities Transactions; Etc.</u> Subsequent to the respective dates as of which information is given in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, and except as may otherwise be indicated or contemplated herein or therein, the Company has not (i) issued any securities or incurred any liability or obligation, direct or contingent, for borrowed money; or (ii) declared or paid any dividend or made any other distribution on or in respect to its equity securities.
- 2.5 <u>Independent Accountants</u>. To the Company's knowledge, Eisner LLP ("Eisner"), whose report is filed with the Commission as part of the Registration Statement, the Sale Preliminary Prospectus and the Prospectus and included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, are independent registered public accountants as required by the Act and the Regulations. Eisner has not, during the periods covered by the financial statements included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, provided to the Company any non-audit services, as such term is used in Section 10A(g) of the Exchange Act.
- 2.6 <u>Financial Statements</u>. The financial statements, including the notes thereto and supporting schedules included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus fairly present the financial position, the results of operations and the cash flows of the Company at the dates and for the periods to which they apply; such financial statements have been prepared in conformity with generally accepted accounting principles, consistently applied throughout the periods involved; and the supporting schedules included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus present fairly the information required to be stated therein. The Registration Statement, the Sale Preliminary Prospectus and the Prospectus disclose all material off-balance sheet transactions, arrangements, obligations (including contingent obligations), and other relationships of the Company with unconsolidated entities or other persons that may have a material current or future effect on the Company's financial condition, changes in financial condition, results of operations, liquidity, capital expenditures, capital resources, or significant components of revenues or expenses. There are no financial statements (historical or pro forma) that are required to be included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus that are not included as required.

2.7 <u>Authorized Capital: Options: Etc.</u> The Company had at the date or dates indicated in the Sale Preliminary Prospectus and the Prospectus duly authorized, issued and outstanding capitalization as set forth in the Registration Statement, the Sale Preliminary Prospectus, and the Prospectus. Based on the assumptions stated in the Registration Statement, the Sale Preliminary Prospectus, and the Prospectus, the Company will have on the Closing Date the adjusted stock capitalization set forth therein. Except as set forth in, or contemplated by the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, on the Effective Date and on the Closing Date, there will be no options, warrants, or other rights to purchase or otherwise acquire any authorized but unissued shares of Common Stock of the Company or any security convertible into shares of Common Stock of the Company, or any contracts or commitments to issue or sell shares of Common Stock or any such options, warrants, rights or convertible securities.

2.8 Valid Issuance of Securities; Etc.

2.8.1 Outstanding Securities. All issued and outstanding securities of the Company (including, without limitation, the Placement Warrants) have been duly authorized and validly issued and are fully paid and non-assessable; the holders thereof have no rights of rescission with respect thereto, and are not subject to personal liability by reason of being such holders; and none of such securities were issued in violation of the preemptive rights of any holders of any security of the Company or similar contractual rights granted by the Company. The Public Securities conform in all material respects to all statements relating thereto contained in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus. The offers and sales of the outstanding securities of the Company were at all relevant times either registered under the Act and the applicable state securities or Blue Sky laws or, based in part on the representations and warranties of the purchasers of such securities, exempt from such registration requirements.

2.8.2 Securities Sold Pursuant to this Agreement. The Securities have been duly authorized and, when issued and paid for, will be validly issued, fully paid and non-assessable; the holders thereof are not and will not be subject to personal liability by reason of being such holders; the Securities are not and will not be subject to the preemptive rights of any holders of any security of the Company or similar contractual rights granted by the Company; and all corporate action required to be taken for the authorization, issuance and sale of the Securities has been duly and validly taken. The form of certificates for the Securities conform to the corporate law of the jurisdiction of the Company's incorporation. The Securities conform in all material respects to all statements with respect thereto contained in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus. When issued, the Representative's Securities will constitute valid and binding obligations of the Company to issue and sell, upon exercise thereof and payment of the respective exercise prices therefor, the number and type of securities of the Company called for thereby in accordance with the terms thereof and such Representative's Securities are enforceable against the Company in accordance with their respective terms, except (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (ii) as enforceability of any indemnification or contribution provision may be limited under the Federal and state securities laws, and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

2.8.3 <u>Placement Warrants</u>. The Placement Warrants (as defined in Section 2.22.4 hereof) constitute valid and binding obligations of the Company to issue and sell, upon exercise thereof and payment of the respective exercise prices therefor, the number and type of securities of the Company called for thereby in accordance with the terms thereof, and such Placement Warrants are enforceable against the Company in accordance with their respective terms, except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification or contribution provision may be limited under federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought. The shares of Common Stock issuable upon exercise of the Placement Warrants have been reserved for issuance upon the exercise of the Placement Warrants and, when issued in accordance with the terms of the Placement Warrants, will be duly and validly authorized, validly issued, fully paid and non-assessable, and the holders thereof are not and will not be subject to personal liability by reason of being such holders.

2.8.4 <u>No Integration</u>. Other than with respect to the Placement Warrants, neither the Company nor any of its affiliates has, prior to the date hereof, made any offer or sale of any securities which are required to be or may be "integrated" pursuant to the Act or the Regulations with the offer and sale of the Securities pursuant to the Registration Statement.

2.9 <u>Registration Rights of Third Parties</u>. Except as set forth in the Sale Preliminary Prospectus and the Prospectus, no holders of any securities of the Company or any rights exercisable for or convertible or exchangeable into securities of the Company have the right to require the Company to register any such securities of the Company under the Act or to include any such securities in a registration statement to be filed by the Company.

2.10 <u>Validity and Binding Effect of Agreements</u>. This Agreement, the Warrant Agreement (as defined in Section 2.21 hereof), the Trust Agreement, the Services Agreement (as defined in Section 3.8.2 hereof), the Securities Escrow Agreement (as defined in Section 2.22.2 hereof) and the Subscription Agreement (as defined in Section 2.22.4 hereof) have been duly and validly authorized by the Company and constitute, and the Representative's Purchase Option, has been duly and validly authorized by the Company and, when executed and delivered, will constitute the valid and binding agreements of the Company, enforceable against the Company in accordance with their respective terms, except (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (ii) as enforceability of any indemnification or contribution provision may be limited under the Federal and state securities laws, and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

2.11 No Conflicts, Etc. The execution, delivery, and performance by the Company of this Agreement, the Warrant Agreement, the Representative's Purchase Option, the Trust Agreement, the Services Agreement, the Securities Escrow Agreement and the Subscription Agreement, the consummation by the Company of the transactions herein and therein contemplated and the compliance by the Company with the terms hereof and thereof do not and will not, with or without the giving of notice or the lapse of time or both (i) result in a breach of, or conflict with any of the terms and provisions of, or constitute a default under, or result in the creation, modification, termination or imposition of any lien, charge or encumbrance upon any property or assets of the Company pursuant to the terms of any agreement or instrument to which the Company is a party except pursuant to the Trust Agreement referred to in Section 2.23 hereof; (ii) result in any violation of the provisions of the Amended and Restated Certificate of Incorporation or the Bylaws of the Company; or (iii) violate any existing applicable law, rule, regulation, judgment, order or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its properties or business.

2.12 No Defaults; Violations. No material default exists in the due performance and observance of any term, covenant or condition of any material license, contract, indenture, mortgage, deed of trust, note, loan or credit agreement, or any other agreement or instrument evidencing an obligation for borrowed money, or any other material agreement or instrument to which the Company is a party or by which the Company may be bound or to which any of the properties or assets of the Company is subject. The Company is not in violation of any term or provision of its Amended and Restated Certificate of Incorporation or Bylaws or in violation of any material franchise, license, permit, applicable law, rule, regulation, judgment or decree of any governmental agency or court, domestic or foreign, having jurisdiction over the Company or any of its properties or businesses.

2.13 Corporate Power; Licenses; Consents.

2.13.1 <u>Conduct of Business</u>. The Company has all requisite corporate power and authority, and has all necessary authorizations, approvals, orders, licenses, certificates and permits of and from all governmental regulatory officials and bodies that it needs as of the date hereof to conduct its business as described in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus. The disclosures in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus concerning the effects of Federal, state and local regulation on this Offering and the Company's business purpose as currently contemplated are correct in all material respects and do not omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

- 2.13.2 <u>Transactions Contemplated Herein</u>. The Company has all corporate power and authority to enter into this Agreement and to carry out the provisions and conditions hereof, and all consents, authorizations, approvals and orders required in connection therewith have been obtained. No consent, authorization or order of, and no filing with, any court, government agency or other body is required for the valid issuance, sale and delivery, of the Securities and the consummation of the transactions and agreements contemplated by this Agreement, the Warrant Agreement, the Representative's Purchase Option, the Trust Agreement, the Securities Escrow Agreement and the Subscription Agreement and as contemplated by the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, except with respect to applicable Federal and state securities laws and the rules and regulations promulgated by the National Association of Securities Dealers, Inc. ("NASD").
- 2.14 <u>D&O Questionnaires</u>. To the best of the Company's knowledge, all information contained in the questionnaires ("<u>Questionnaires</u>") completed by each of the Company's stockholders prior to the Offering ("<u>Existing Stockholders</u>"), directors and officers and provided to the Representative is true and correct and the Company has not become aware of any information which would cause the information disclosed in such questionnaires to become inaccurate or incorrect.
- 2.15 <u>Litigation; Governmental Proceedings</u>. There is no action, suit, proceeding, inquiry, arbitration, investigation, litigation or governmental proceeding pending or, to the best of the Company's knowledge, threatened against, or involving the Company or, to the best of the Company's knowledge, any Existing Stockholder which has not been disclosed, that is required to be disclosed, in the Registration Statement, the Sale Preliminary Prospectus, the Prospectus or the Questionnaires.
- 2.16 <u>Good Standing</u>. The Company has been duly organized and is validly existing as a corporation and is in good standing under the laws of its state of incorporation, and is duly qualified to do business and is in good standing as a foreign corporation in each jurisdiction in which its ownership or lease of property or the conduct of business requires such qualification, except where the failure to qualify would not have a material adverse effect on the assets, business or operations of the Company.
- 2.17 <u>Stop Orders</u>. The Commission has not issued any order preventing or suspending the use of the Registration Statement, any Preliminary Prospectus, the Sale Preliminary Prospectus or the Prospectus or any part thereof and has not threatened to issue any such order.
 - 2.18 Transactions Affecting Disclosure to NASD.

2.18.1 <u>Finder's Fees</u>. There are no claims, payments, arrangements or understandings relating to the payment of a finder's, consulting or origination fee by the Company or any Existing Stockholder with respect to the sale of the Securities hereunder or any other arrangements, agreements or understandings of the Company or, to the Company's knowledge, any Existing Stockholder that may affect the Underwriters' compensation, as determined by the NASD.

2.18.2 <u>Payments Within Twelve Months</u>. The Company has not made any direct or indirect payments (in cash, securities or otherwise) (i) to any person, as a finder's fee, consulting fee or otherwise, in consideration of such person raising capital for the Company or introducing to the Company persons who raised or provided capital to the Company, (ii) to any NASD member or (iii) to any person or entity that has any direct or indirect affiliation or association with any NASD member, within the twelve months prior to the Effective Date, other than payments to the Representative in connection with the Offering.

2.18.3 <u>Use of Proceeds</u>. None of the net proceeds of the Offering and Private Placement will be paid by the Company to any participating NASD member or its affiliates, except as specifically authorized herein and except as may be paid in connection with a Business Combination as contemplated by the Sale Preliminary Prospectus.

2.18.4 <u>Insiders' NASD Affiliation</u>. No officer, director or any beneficial owner of the Company's unregistered securities has any direct or indirect affiliation or association with any NASD member, as determined in accordance with the rules and regulations of the NASD. The Company will advise the Representative and its counsel if it learns that any officer, director or owner of at least 5% of the Company's outstanding Common Stock is or becomes an affiliate or associated person of an NASD member participating in the Offering.

2.19 Foreign Corrupt Practices Act; Patriot Act.

2.19.1 Foreign Corrupt Practices Act. Neither the Company nor any of the Existing Stockholders or any other person acting on behalf of the Company has, directly or indirectly, given or agreed to give any money, gift or similar benefit (other than legal price concessions to customers in the ordinary course of business) to any customer, supplier, employee or agent of a customer or supplier, or official or employee of any governmental agency or instrumentality of any government (domestic or foreign) or any political party or candidate for office (domestic or foreign) or any political party or candidate for office (domestic or foreign) or other person who was, is, or may be in a position to help or hinder the business of the Company (or assist it in connection with any actual or proposed transaction) that (i) might subject the Company to any damage or penalty in any civil, criminal or governmental litigation or proceeding, (ii) if not given in the past, might have had a material adverse effect on the assets, business or operations of the Company as reflected in any of the financial statements contained in the Prospectus or (iii) if not continued in the future, might adversely affect the assets, business, operations or prospects of the Company. The Company's internal accounting controls and procedures are sufficient to cause the Company to comply with the Foreign Corrupt Practices Act of 1977, as amended.

2.19.2 <u>Patriot Act</u>. Neither the Company nor, to the Company's knowledge, any Company affiliates have violated: (i) the Bank Secrecy Act, as amended, (ii) the Money Laundering Control Act of 1986, as amended, or (iii) the Uniting and Strengthening of America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA PATRIOT ACT) Act of 2001, and/or the rules and regulations promulgated under any such law, or any successor law.

2.20 Officers' Certificate. Any certificate signed by any duly authorized officer of the Company and delivered to you or to your counsel shall be deemed a representation and warranty by the Company to the Underwriters as to the matters covered thereby.

2.21 <u>Warrant Agreement</u>. The Company has entered into a warrant agreement with respect to the Warrants and the Representative's Warrants with CST substantially in the form annexed as Exhibit 4.4 to the Registration Statement ("<u>Warrant Agreement</u>").

2.22 Agreements With Existing Stockholders.

2.22.1 <u>Insider Letters</u>. The Company has caused to be duly executed legally binding and enforceable agreements (except (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (ii) as enforceability of any indemnification, contribution or noncompete provision may be limited under the Federal and state securities laws, and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought) annexed as Exhibits 10.1.1-10.1.7 to the Registration Statement ("<u>Insider Letters</u>"), pursuant to which each of the Existing Stockholders of the Company agree to certain matters, including but not limited to, certain matters described as being agreed to by them under the "<u>Proposed Business</u>" section of the Sale Preliminary Prospectus and the Prospectus.

2.22.2 Securities Escrow Agreement. The Company has caused the Existing Stockholders to enter into an escrow agreement ("Securities Escrow Agreement") with CST ("Escrow Agent"), substantially in the form annexed as Exhibit 10.3 to the Registration Statement, whereby (i) the Common Stock owned by the Existing Stockholders (the "Existing Stockholders Shares") will be held in escrow by the Escrow Agent, until one year from the date of consummation of a Business Combination and (ii) the Placement Warrants will be held in escrow by the Escrow Agent until such time that the Company consummates a Business Combination; provided, however, that if the Escrow Agent is notified by the Company that the Company is being liquidated at any time during the applicable Escrow Period (as that term is defined in the Securities Escrow Agreement), then immediately prior to the effectiveness of such liquidation, the Escrow Agent shall promptly destroy the certificates representing the Existing Stockholders Shares and the Placement Warrants. During such escrow period, the Existing Stockholders shall be prohibited from selling or otherwise transferring such shares (except to spouses and children of Existing Stockholders and trusts established for their benefit and as otherwise set forth in the Securities Escrow Agreement) but will retain the right to vote such shares. To the Company's knowledge, the Securities Escrow Agreement is enforceable against each of the Existing Stockholders and will not, with or without the giving of notice or the lapse of time or both, result in a breach of, or conflict with any of the terms and provisions of, or constitute a default under, any agreement or instrument to which any of the Existing Stockholders is a party. The Securities Escrow Agreement shall not be amended, modified or otherwise changed without the prior written consent of the Representative.

2.22.3 No Fiduciary Relationship in Pricing. The Company acknowledges and agrees that (i) the purchase and sale of the Units pursuant to this Agreement is an arm's-length commercial transaction between the Company and the several Underwriters, (ii) in connection therewith and with the process leading to such transaction, each Underwriter is acting solely as a principal and not the agent or fiduciary of the Company, (iii) no Underwriter has assumed an advisory or fiduciary responsibility in favor of the Company with respect to the offering contemplated hereby or the process leading thereto (irrespective of whether such Underwriter has advised or is currently advising the Company on other matters) or any other obligation to the Company except the obligations expressly set forth in this Agreement and (iv) the Company has consulted its own legal and financial advisors to the extent it deemed appropriate. The Company agrees that it will not claim that the Underwriters, or any of them, has rendered advisory services of any nature or respect, or owes a fiduciary or similar duty to the Company, in connection with such transaction or the process leading thereto.

2.22.4 Founding Warrant Purchase Agreement. Camden Learning, LLC, a limited liability company owned by two investment limited partnerships which are indirectly controlled by and partially owned by certain of the Company's officers and directors (the "Sponsor") executed and delivered an agreement, annexed as Exhibit 10.6 of the Registration Statement (the "Subscription Agreement"), pursuant to which such persons, among other things, have purchased an aggregate of 2,800,000 warrants identical to the Warrants (the "Placement Warrants") at a purchase price of \$1.00 per Placement Warrant in a private placement in accordance with Regulation D under the Act occurring immediately prior to the Closing (the "Private Placement"). The Sponsor and the Company have delivered executed copies of the Subscription Agreement and the Sponsor has delivered the purchase price on or before the Effective Date. Pursuant to the Subscription Agreement, (i) \$2,800,000 of the proceeds from the sale of the Placement Warrants will be deposited by the Company in the Trust Account in accordance with the terms of the Trust Agreement prior to the Effective Date, and (ii) the purchasers of the Placement Warrants have waived any and all rights and claims that they may have to any proceeds, and any interest thereon, held in the Trust Account in respect of the Placement Warrants in the event that a Business Combination is not consummated and the Trust Account is liquidated in accordance with the terms of the Trust Agreement. There was no placement agent in the Private Placement and no party shall be entitled to a placement fee or expense allowance from the sale of the Placement Warrants.

2.23 <u>Investment Management Trust Agreement</u>. The Company has entered into the Trust Agreement with respect to certain proceeds of the Offering and the Private Placement substantially in the form annexed as Exhibit 10.2 to the Registration Statement.

2.24 <u>N o Existing Non-Competition Agreements</u>. N o Existing Stockholder, employee, officer or director of the Company is subject to any non-competition agreement or non-solicitation agreement with any employer or prior employer which could materially affect his ability to be an Existing Stockholder, employee, officer and/or director of the Company.

2.25 <u>Investments</u>. The Company is not and, after giving effect to the Offering and sale of the Securities and the application of the proceeds thereof as described in the Sale Preliminary Prospectus and the Prospectus, will not be an "investment company" as defined in the Investment Company Act of 1940, as amended.

- 2.26 <u>Subsidiaries</u>. The Company does not own an interest in any corporation, partnership, limited liability company, joint venture, trust or other business entity.
- 2.27 <u>Related Party Transactions</u>. There are no business relationships or related party transactions involving the Company or any other person required to be described in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus that have not been described as required.
- 2.28 <u>Data</u>. The statistical, industry-related and market-related data included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus are based on or derived from sources which the Company reasonably and in good faith believes are reliable and accurate, and such data agree with the sources from which they are derived.
- 2.29 <u>Business Combinations</u>. The Company does not have any specific Business Combination under consideration or contemplation and the Company has not (nor has anyone on its behalf) contacted any potential target business or had any discussions, formal or otherwise, with respect to such a transaction.
- 2.30 <u>Distribution of Offering Material By the Company</u>. The Company has not distributed and will not distribute, prior to the later of the Closing Date and the completion of the Underwriters' distribution of the Units, any offering material in connection with the offering and sale of the Units other than the Sale Preliminary Prospectus and the Prospectus, in each case as supplemented and amended.
- 2.31 <u>Tax Law Compliance</u>. The Company has filed all necessary federal, state, local and foreign income and franchise tax returns in a timely manner and has paid all taxes required to be paid by the Company and, if due and payable, any related or similar assessment, fine or penalty levied against the Company, except for any taxes, assessments, fines or penalties as may be being contested in good faith and by appropriate proceedings. The Company has made appropriate provisions in the applicable financial statements referred to in Section 2.6 above in respect of all federal, state, local and foreign income and franchise taxes for all current or prior periods as to which the tax liability of the Company has not been finally determined.
- 2.32 <u>Ineligible Issuer</u>. At the time of filing the Registration Statement and at the date hereof, the Company was and is an "ineligible issuer," as defined in Rule 405 under the Securities Act. The Company has not made any offer relating to the Securities that would constitute an "issuer free writing Prospectus," as defined in Rule 433, or that would otherwise constitute a "free writing Prospectus," as defined in Rule 405.
 - 3. Covenants of the Company. The Company covenants and agrees as follows:
- 3.1 <u>Amendments to Registration Statement</u>. The Company will deliver to the Representative, prior to filing, any amendment or supplement to the Registration Statement or Prospectus proposed to be filed after the Effective Date and the Company shall not file any such amendment or supplement to which the Representative shall reasonably object.
 - 3.2 Federal Securities Laws.

- 3.2.1 Compliance. During the time when a Prospectus is required to be delivered under the Act, the Company will use all reasonable efforts to comply with all requirements imposed upon it by the Act, the Regulations and the Exchange Act and by the regulations under the Exchange Act, as from time to time in force, so far as necessary to permit the continuance of sales of or dealings in the Securities in accordance with the provisions hereof and the Prospectus. If at any time when a Sale Preliminary Prospectus or Prospectus relating to the Securities is required to be delivered under the Act, any event shall have occurred as a result of which, in the opinion of counsel for the Company or counsel for the Underwriters, the Sale Preliminary Prospectus or the Prospectus, as then amended or supplemented, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, or if it is necessary at any time to amend the Sale Preliminary Prospectus or the Prospectus to comply with the Act, the Company will notify the Representative promptly and prepare and file with the Commission, subject to Section 3.1 hereof, an appropriate amendment or supplement in accordance with Section 10 of the Act.
- 3.2.2 <u>Filing of Final Prospectus</u>. The Company will file the Prospectus (in form and substance satisfactory to the Representative) with the Commission pursuant to the requirements of Rule 424 of the Regulations.
- 3.2.3 Exchange Act Registration. The Company will use its best efforts to maintain the registration of the Securities under the provisions of the Exchange Act (except in connection with a going-private transaction) for a period of five years from the Effective Date, or until the Company is required to be liquidated if earlier, or, in the case of the Warrants, until the Warrants expire and are no longer exercisable. The Company will not deregister the Securities under the Exchange Act without the prior written consent of the Representative.
- 3.3 <u>Ineligible Issuer</u>. The Company will not make any offer relating to the Securities that would constitute an "issuer free writing Prospectus," as defined in Rule 433, or that would otherwise constitute a "free writing Prospectus," as defined in Rule 405.
- 3.4 <u>Blue Sky Filing</u>. The Company will endeavor in good faith, in cooperation with the Representative, at or prior to the time the Registration Statement becomes effective, to qualify the Securities for offering and sale under the securities laws of such jurisdictions as the Representative may reasonably designate, provided that no such qualification shall be required in any jurisdiction where, as a result thereof, the Company would be subject to service of general process or to taxation as a foreign corporation doing business in such jurisdiction. In each jurisdiction where such qualification shall be effected, the Company will, unless the Representative agrees that such action is not at the time necessary or advisable, use all reasonable efforts to file and make such statements or reports at such times as are or may be required by the laws of such jurisdiction. The Company shall pay all filings fees in connection with the qualification of the securities under the securities laws of such jurisdictions as the Representative may reasonably designate.
- 3.5 <u>Delivery to Underwriters of Preliminary Prospectus</u>, <u>Sale Preliminary Prospectus and Prospectuses</u>. The Company will deliver to each of the several Underwriters, without charge, from time to time during the period when the Prospectus is required to be delivered under the Act or the Exchange Act such number of copies of each Preliminary Prospectus, Sale Preliminary Prospectus and Prospectus as such Underwriters may reasonably request and, as soon as the Registration Statement or any amendment or supplement thereto becomes effective, deliver to you two original executed Registration Statements, including exhibits, and all post-effective amendments thereto and copies of all exhibits filed therewith or incorporated therein by reference and a copy of all original executed consents of certified experts.

3.6 Effectiveness and Events Requiring Notice to the Representative. The Company will use its best efforts to cause the Registration Statement to remain effective and will notify the Representative immediately and confirm the notice in writing (i) of the effectiveness of the Registration Statement and any amendment thereto, (ii) of the issuance by the Commission of any stop order suspending the effectiveness of the Registration Statement or any post-effective amendment thereto or preventing or suspending the use of the Prospectus or of the initiation, or the threatening, of any proceeding for that purpose, (iii) of the issuance by any state securities commission of any proceedings for the suspension of the qualification of the Public Securities for offering or sale in any jurisdiction or of the initiation, or the threatening, of any proceeding for that purpose, (iv) of the mailing and delivery to the Commission for filing of any amendment or supplement to the Registration Statement or Prospectus, (v) of the receipt of any comments or request for any additional information from the Commission, and (vi) of the happening of any event during the period described in Section 3.2.3 hereof that, in the judgment of the Company or its counsel, makes any statement of a material fact made in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus untrue or that requires the making of any changes in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus in order to make the statements therein, in light of the circumstances under which they were made, not misleading. If the Commission or any state securities commission shall enter a stop order or suspend such qualification at any time, the Company will make every reasonable effort to obtain promptly the lifting of such order.

3.7 Review of Quarterly Financial Statements. Until the earlier of five years from the Effective Date, or until such earlier time upon which the Company is required to be liquidated, the Company, at its expense, shall cause its regularly engaged independent registered public accounting firm to review (but not audit) the Company's financial statements for each of the first three fiscal quarters prior to the announcement of quarterly financial information, the filing of the Company's Form 10-Q quarterly report and the mailing of quarterly financial information to stockholders.

3.8 Affiliated Transactions.

3.8.1 <u>Business Combinations</u>. The Company will not consummate a Business Combination with any entity which is affiliated with any Existing Stockholder unless the Company obtains an opinion from an independent investment banking firm that is a member of the NASD that the Business Combination is fair to the Company's stockholders from a financial perspective. No Existing Stockholder or any affiliate of such person shall receive any fees of any type (other than reimbursement of ordinary and customary expenses incurred on behalf of the Company) in connection with any Business Combination.

3.8.2 <u>Administrative Services</u>. The Company has entered into an agreement ("<u>Services Agreement</u>") with the Sponsor ("<u>Affiliate</u>") substantially in the form annexed as Exhibit 10.5 to the Registration Statement pursuant to which the Affiliate will make available to the Company general and administrative services, including office space, utilities, technology and secretarial services for the Company's use for up to \$7.500 per month.

3.8.3 <u>Compensation to Existing Stockholders</u>. Except as set forth above in this Section 3.8, the Company shall not pay any Existing Stockholder or any of their affiliates any fees or compensation from the Company, for services rendered to the Company prior to, or in connection with, the consummation of a Business Combination; provided that the Existing Stockholders shall be entitled to reimbursement from the Company for their reasonable out-of-pocket expenses incurred in connection with seeking and consummating a Business Combination.

3.9 <u>Financial Public Relations Firm</u>. Promptly after the execution of a definitive agreement for a Business Combination, the Company shall retain a financial public relations firm reasonably acceptable to Morgan Joseph & Co. for a term to be agreed on by the Company and Morgan Joseph & Co.

3.10 Reports to the Representative.

3.10.1 Periodic Reports, Etc. For a period of five years from the Effective Date or until such earlier time upon which the Company is required to be liquidated, the Company will furnish to the Representative (Morgan Joseph & Co. Attn: Tina Pappas) and its counsel copies of such financial statements and other periodic and special reports as the Company from time to time furnishes generally to holders of any class of its securities, and promptly furnish to the Representative (i) a copy of each periodic report the Company shall be required to file with the Commission, (ii) a copy of every press release and every news item and article with respect to the Company or its affairs which was released by the Company, (iii) a copy of each Form 8-K or Schedules 13D, 13G, 14D-1 or 13E-4 received or prepared by the Company, (iv) two (2) copies of each registration statement filed by the Company with the Commission under the Securities Act and (v) such additional documents and information with respect to the Company and the affairs of any future subsidiaries of the Company as the Representative may from time to time reasonably request; provided the Representatives shall sign, if requested by the Company, a Regulation FD compliant confidentiality agreement which is reasonably acceptable to the Representatives and their counsel in connection with the Representatives' receipt of such information. Documents filed with the Commission pursuant to its EDGAR system shall be deemed to have been delivered to the Representative pursuant to this Section.

3.10.2 <u>Transfer Sheets</u>. For a period of two years following the Effective Date or until such earlier time upon which the Company is required to be liquidated, the Company shall retain CST or another transfer and warrant agent acceptable to the Representative ("<u>Transfer Agent</u>") and will furnish to the Underwriters at the Company's sole cost and expense, for a period of one year following the Effective Date, such transfer sheets of the Company's securities as the Representative may request, including the daily and monthly consolidated transfer sheets of the Transfer Agent and DTC. The Underwriters acknowledge that CST is an acceptable Transfer Agent.

3.10.3 <u>Secondary Market Trading Maintenance</u>. Unless the Securities are listed or quoted, as the case may be, on the New York Stock Exchange, the American Stock Exchange or quoted on the NASDAQ Global Market or NASDAQ Capital Market, until such earlier time upon which the Company is required to be liquidated or upon consummation of a Business Combination, the Company shall (i) at the beginning of each fiscal quarter, provide the Representative with a written report detailing those states in which the Securities may be traded in non-issuer transactions under the Blue Sky laws of the fifty States and (ii) take such other action as may be reasonably requested by the Representative to obtain a secondary market trading exemption in such other states as may be requested by the Representative.

3.10.4 <u>Trading Reports</u>. During such time as any of the Securities are quoted on the NASD OTC Bulletin Board (or any successor trading market such as the Bulletin Board Exchange) or the Pink Sheets, LLC (or similar publisher of quotations) and no other automated quotation system, the Company shall provide to the Representative, at its expense, such reports published by the NASD or the Pink Sheets, LLC relating to price trading of the Securities, as the Representative shall reasonably request.

3.11 <u>Disqualification of Form S-1</u>. Until the earlier of seven years from the date hereof or until the Warrants have expired and are no longer exercisable, the Company will not take any action or actions which may prevent or disqualify the Company's use of Form S-1 (or other appropriate form) for the registration of the Warrants and the Representative's Warrants under the Act (except in connection with a going-private transaction).

3.12 Payment of Expenses.

3.12.1 General Expenses Related to the Offering. The Company hereby agrees to pay on each of the Closing Date and the Option Closing Date, if any, to the extent not paid at Closing Date, all expenses incident to the performance of the obligations of the Company under this Agreement, including but not limited to (i) the preparation, printing, filing and mailing (including the payment of postage with respect to such mailing) of the Registration Statement, the Preliminary Sale and Final Prospectuses and the printing and mailing of this Agreement and related documents, including the cost of all copies thereof and any amendments thereof or supplements thereto supplied to the Underwriters in quantities as may be required by the Underwriters, (ii) the printing, engraving, issuance and delivery of the Units, the shares of Common Stock and the Warrants included in the Units and the Representative's Purchase Option, including any transfer or other taxes payable thereon, (iii) the listing and qualification of the Public Securities under state or foreign securities or Blue Sky laws, including the costs of printing and mailing the "Preliminary Blue Sky Memorandum," and all amendments and supplements thereto, fees in an amount equal to \$20,000 and disbursements for the Representatives' counsel retained for such purpose; (iv) filing fees incurred in registering the Offering with the NASD; (v) fees, costs and expenses incurred in listing the Company on the Over the Counter Bulletin Board, (vi) fees and disbursements of the transfer and warrant agent, (vii) the Company's expenses associated with "due diligence" and "road show" meetings arranged by the Representative, and (viii) the preparation, binding and delivery of transaction "bibles," in form and style reasonably satisfactory to the Representative and transaction lucite cubes or similar commemorative items in a style and quantity as reasonably requested by the Representative and (ix) all other actual, documented costs and expenses customarily borne by an issuer incident to the performance of its obligations hereunder which are not otherwise specifically provided for in this Section 3.12.1. The Company also agrees that, if requested by the Representative, it will engage and pay up to \$5,000 for an investigative search firm of the Representative's choice to conduct an investigation of the principals of the Company as shall be mutually selected by the Representative and the Company. The Representative may deduct from the net proceeds of the Offering payable to the Company on the Closing Date, or the Option Closing Date, if any, the expenses set forth in this Agreement to be paid by the Company to the Representative and others. If the Company elects not to proceed with the Offering contemplated by this Agreement for any reason whatsoever, then the Company shall reimburse the Underwriters in full for their out of pocket expenses, including, without limitation, its legal fees (up to a maximum of \$25,000) and disbursements and "road show" and due diligence expenses.

- 3.12.2 Expenses Related to Business Combination. The Company further agrees that, in the event the Representative assists the Company in trying to obtain stockholder approval of a proposed Business Combination, the Company agrees to reimburse the Representative for all reasonable out-of-pocket expenses (as agreed upon), including, but not limited to, "road-show" and due diligence expenses.
- 3.13 <u>Application of Net Proceeds</u>. The Company will apply the net proceeds from the Offering and Private Placement received by it in a manner consistent with the application described under the caption "Use of Proceeds" in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus.
- 3.14 <u>Delivery of Earnings Statements to Security Holders</u>. The Company will make generally available to its security holders as soon as practicable, but not later than the first day of the fifteenth full calendar month following the Effective Date, an earnings statement (which need not be certified by independent public or independent certified public accountants unless required by the Act or the Regulations, but which shall satisfy the provisions of Rule 158(a) under Section 11(a) of the Act) covering a period of at least twelve consecutive months beginning after the Effective Date.
- 3.15 Notice to NASD. In the event any person or entity (regardless of any NASD affiliation or association) is engaged to assist the Company in its search for a merger candidate or to provide any other merger and acquisition services, the Company will provide the following to the NASD and to Morgan Joseph & Co. prior to the consummation of the Business Combination: (i) complete details of all services and copies of agreements governing such services; and (ii) justification as to why the person or entity providing the merger and acquisition services should not be considered an "underwriter and related person" with respect to the Company's initial public offering, as such term is defined in Rule 2710 of the NASD's Conduct Rules. The Company also agrees that proper disclosure of such arrangement or potential arrangement will be made in the proxy statement which the Company will file for purposes of soliciting stockholder approval for the Business Combination.

- 3.16 <u>Stabilization</u>. Neither the Company, nor, to its knowledge, any of its employees, directors or stockholders (without the consent of the Representative) has taken or will take, directly or indirectly, any action designed to or that has constituted or that might reasonably be expected to cause or result in, under the Exchange Act, or otherwise, stabilization or manipulation of the price of any security of the Company to facilitate the sale or resale of the Securities.
- 3.17 Existing Lock-Up Agreement. The Company will enforce all existing agreements between the Company and any of its security holders that prohibit the sale, transfer, assignment, pledge or hypothecation of any of the Securities in connection with the Offering. In addition, the Company will direct the Transfer Agent to place stop transfer restrictions upon any such Securities of the Company that are bound by such existing "lock-up" agreements for the duration of the periods contemplated in such agreements.
- 3.18 Internal Controls. The Company will maintain a system of internal accounting controls sufficient to provide reasonable assurances that: (i) transactions are executed in accordance with management's general or specific authorization, (ii) transactions are recorded as necessary in order to permit preparation of financial statements in accordance with generally accepted accounting principles and to maintain accountability for assets, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences.
- 3.19 <u>Accountants</u>. Until the earlier of five years from the Effective Date or until such earlier time upon which the Company is required to be liquidated, the Company shall retain Eisner or another independent registered public accounting firm reasonably acceptable to the Representative.
- 3.20 Form 8-K. The Company shall, on the date hereof, retain its independent registered public accounting firm to audit the financial statements of the Company as of the Closing Date ("Audited Financial Statements") reflecting the receipt by the Company of the proceeds of the Offering and Private Placement. As soon as the Audited Financial Statements become available, the Company shall immediately file a Current Report on Form 8-K with the Commission, which Report shall contain the Company's Audited Financial Statements.
- 3.21 Notice to NASD. For a period of ninety days after the date of the Prospectus, in the event any person or entity (regardless of any NASD affiliation or association but excluding attorneys, accountants, engineers, environmental or labor consultants, investigatory firms, technology consultants and specialists and similar service providers that are not affiliated with or associated with the NASD and are not brokers or finders) is engaged, in writing, to assist the Company in its search for a target business or to provide any other services in connection therewith, the Company will provide the following to the NASD and the Representative prior to the consummation of the Business Combination: (i) complete details of all services and copies of agreements governing such services (which may be appropriately redacted to account for privilege or confidentiality concerns); and (ii) justification as to why the person or entity providing the merger and acquisition services should not be considered an "underwriter and related person" with respect to the IPO, as such term is defined in Rule 2710 of the NASD's Conduct Rules. The Company also agrees that, if required by law, proper disclosure of such arrangement or potential arrangement will be made in the proxy statement which the Company will file for purposes of soliciting stockholder approval for the Business Combination.

- 3.22 <u>NASD</u>. The Company shall advise the NASD if it is aware that any 5% or greater stockholder of the Company becomes an affiliate or associated person of an NASD member participating in the distribution of the Company's Public Securities.
- 3.23 <u>Corporate Proceedings</u>. All corporate proceedings and other legal matters necessary to carry out the provisions of this Agreement and the transactions contemplated hereby shall done to the reasonable satisfaction to counsel for the Underwriters.
- 3.24 <u>Investment Company</u>. The Company shall cause the proceeds of the Offering to be held in the Trust Account to be invested only in "government securities" with specific maturity dates or in money market funds as set forth in the Trust Agreement and disclosed in the Registration Statement, Sale, Preliminary Prospectus or Prospectus. The Company will otherwise conduct its business in a manner so that it will not become subject to the Investment Company Act of 1940, as amended. Furthermore, once the Company consummates a Business Combination, it will be engaged in a business other than that of investing, reinvesting, owning, holding or trading securities.
- 3.25 <u>Business Combination Announcement</u>. Within five Business Days following the consummation by the Company of a Business Combination, the Company shall cause an announcement ("<u>Business Combination Announcement</u>") to be placed, at its cost, in The Wall Street Journal, the New York Times and a third publication to be selected by the Representative announcing the consummation of the Business Combination and indicating that the Representative was the managing underwriter in the Offering in an aggregate amount not to exceed \$10,000. The Company shall supply the Representative with a draft of the Business Combination Announcement and provide the Representative with a reasonable opportunity to comment thereon. The Company will not place the Business Combination Announcement without the final approval of the Representative, which such approval will not be unreasonably withheld.
- 3.26 <u>Colorado Trust Filing</u>. In the event the Securities are registered in the State of Colorado, the Company will cause a Colorado Form ES to be filed with the Commissioner of the State of Colorado no less than 10 days prior to the distribution of the Trust Account in connection with a Business Combination and will do all things necessary to comply with Section 11-51-302 and Rule 51-3.4 of the Colorado Securities Act.
- 3.27 10b5 1 Purchase Agreement. The Sponsor has entered into a letter agreement with the Representative, in accordance with Rule 10b5 1 of the Exchange Act, pursuant to which it will place limit orders for up to \$4,000,000 of the Common Stock commencing ten Business Days after the Company files its Current Report on Form 8-K announcing the execution of a definitive agreement for a Business Combination and ending on the Business Day immediately preceding the meeting date for the meeting of stockholders at which such Business Combination is to be approved. These purchases will be made in accordance with Rule 10b-18 under the Act at a price equal to the per share amount held in the Trust Account (less taxes payable) as reported in such Form 8-K and will be made by a broker dealer mutually agreed upon by the Sponsor and the Representative in such amounts and at such times as such broker dealer may determine, in its sole discretion, so long as the purchase price does not exceed the above-referenced per share purchase price. In the event the Sponsor does not purchase \$4,000,000 of Common Stock through those open market purchases, the Sponsor has agreed to purchase from the Company in a private placement, a number of units identical to the units sold in the Offering at a purchase price of \$8.00 per unit until it has spent an aggregate of \$4,000,000 in the open market purchases described above and this private placement.

- 3.28 <u>Amendments to Certificate of Incorporation</u>. (1) The Company covenants and agrees, that prior to its initial Business Combination it will not seek to amend or modify any of provisions (A) (F) of Article Sixth of its certificate of incorporation.
- (ii) The Company acknowledges that the purchasers of the Public Securities in the Offering shall be deemed to be third party beneficiaries of this Agreement and specifically this Section 3.28.
- (iii) The Representative specifically advises the Company that it will not waive this Section 3.28 under any circumstances.
- 3.29 <u>Private Placement Proceeds</u>. Prior to the Effective Date and execution of this Agreement, the Company shall deposit \$2,800,000 of the proceeds from the Private Placement in the Trust Account and shall provide Morgan Joseph & Co. with evidence of the same.
- 4. <u>Conditions of Underwriters' Obligations</u>. The obligations of the several Underwriters to purchase and pay for the Units, as provided herein, shall be subject to the continuing accuracy of the representations and warranties of the Company as of the date hereof and as of each of the Closing Date and the Option Closing Date, if any, to the accuracy of the statements of officers of the Company made pursuant to the provisions hereof and to the performance by the Company of its obligations hereunder and to the following conditions:

4.1 Regulatory Matters.

- 4.1.1 Effectiveness of Registration Statement. The Registration Statement shall have become effective not later than 5:00 p.m., New York time, on the date of this Agreement or such later date and time as shall be consented to in writing by you, and, at each of the Closing Date and the Option Closing Date, no stop order suspending the effectiveness of the Registration Statement shall have been issued and no proceedings for the purpose shall have been instituted or shall be pending or contemplated by the Commission and any request on the part of the Commission for additional information shall have been complied with to the reasonable satisfaction of McDermott, counsel to the Underwriters.
- 4.1.2 <u>NASD Clearance</u>. By the Effective Date, the Representative shall have received clearance from the NASD as to the amount of compensation allowable or payable to the Underwriters as described in the Registration Statement.

4.1.3 <u>No Blue Sky Stop Orders</u>. No order suspending the sale of the Units in any jurisdiction designated by you pursuant to Section 3.4 hereof shall have been issued on either on the Closing Date or the Option Closing Date, and no proceedings for that purpose shall have been instituted or shall be contemplated.

4.1.4 <u>The OTC Bulletin Board</u>. The Securities shall have been admitted and approved for quotation on the OTC Bulletin Board.

4.2 Company Counsel Matters.

- 4.2.1 <u>Closing Date and Option Closing Date Opinion of Counsel</u>. On the Closing Date and the Option Closing Date, if any, the Representative shall have received the favorable opinion of Ellenoff Grossman & Schole LLP, dated the Closing Date or the Option Closing Date, as the case may be, addressed to the Representative and in form and substance reasonably satisfactory to McDermott, covering the matters set forth on Appendix A hereto.
- 4.2.2 Reliance. In rendering such opinion, such counsel may rely (i) as to matters involving the application of laws other than the laws of the United States and jurisdictions in which they are admitted, to the extent such counsel deems proper and to the extent specified in such opinion, if at all, upon an opinion or opinions (in form and substance reasonably satisfactory to the Representative and McDermott) of other counsel reasonably acceptable to the Representative and McDermott, familiar with the applicable laws, and (ii) as to matters of fact, to the extent they deem proper, on certificates or other written statements of officers of the Company and officers of departments of various jurisdictions having custody of documents respecting the corporate existence or good standing of the Company, provided that copies of any such statements or certificates shall be delivered to the Underwriters' counsel if requested. The opinion of counsel for the Company and any opinion relied upon by such counsel for the Company shall include a statement to the effect that it may be relied upon by counsel for the Underwriters in its opinion delivered to the Underwriters.
- 4.3 <u>Cold Comfort Letter</u>. At the time this Agreement is executed, and at each of the Closing Date and the Option Closing Date, if any, you shall have received a letter, addressed to the Representative and in form and substance satisfactory in all respects (including the non-material nature of the changes or decreases, if any, referred to in clause (iii) below) to you and to McDermott from Eisner dated, respectively, as of the date of this Agreement and as of the Closing Date and the Option Closing Date, if any:
 - (i) Confirming that they are independent accountants with respect to the Company within the meaning of the Act and the applicable Regulations and that they have not, during the periods covered by the financial statements included in the Preliminary Prospectus, Sale Preliminary Prospectus and the Prospectus, provided to the Company any non-audit services, as such term is used in Section 10A(g) of the Exchange Act;
 - (ii) Stating that in their opinion the financial statements of the Company included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus comply as to form in all material respects with the applicable accounting requirements of the Act and the published Regulations thereunder;

(iii) Stating that, on the basis of their review which included a reading of the latest available unaudited interim financial statements of the Company (with an indication of the date of the latest available unaudited interim financial statements), a reading of the latest available minutes of the stockholders and board of directors and the various committees of the board of directors, consultations with officers and other employees of the Company responsible for financial and accounting matters and other specified procedures and inquiries, nothing has come to their attention which would lead them to believe that (a) the unaudited financial statements of the Company included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus do not comply as to form in all material respects with the applicable accounting requirements of the Act and the Regulations or are not fairly presented in conformity with generally accepted accounting principles applied on a basis substantially consistent with that of the audited financial statements of the Company included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, or (b) at a date not later than five days prior to the Effective Date, Closing Date or Option Closing Date, as the case may be, there was any change in the capital stock or long-term debt of the Company, or any decrease in the stockholders' equity of the Company as compared with amounts shown in the , 2007] balance sheet included in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, other than as set forth in or contemplated by the Registration Statement, the Sale Preliminary Prospectus and the Prospectus or, if there was any decrease, setting forth the amount of such decrease, and (c) during the , 2007] to a specified date not later than five days prior to the Effective Date, Closing Date or Option Closing Date, as the case may be, there was any decrease in revenues, net earnings or net earnings per share of Common Stock, in each case as compared with the corresponding period in the preceding year and as compared with the corresponding period in the preceding quarter, other than as set forth in or contemplated by the Registration Statement the Sale Preliminary Prospectus and the Prospectus, or, if there was any such decrease, setting forth the amount of such decrease;

(iv) Setting forth, at a date not later than five days prior to the Effective Date, the amount of liabilities of the Company (including a break-down of commercial papers and notes payable to banks);

(v) Stating that they have compared specific dollar amounts, numbers of shares, percentages of revenues and earnings, statements and other financial information pertaining to the Company set forth in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus in each case to the extent that such amounts, numbers, percentages, statements and information may be derived from the general accounting records, including work sheets, of the Company and excluding any questions requiring an interpretation by legal counsel, with the results obtained from the application of specified readings, inquiries and other appropriate procedures (which procedures do not constitute an examination in accordance with generally accepted auditing standards) set forth in the letter and found them to be in agreement; and

(vi) Statements as to such other matters incident to the transaction contemplated hereby as you may reasonably request.

4.4 Officers' Certificates.

4.4.1 Officers' Certificate. At each of the Closing Date and the Option Closing Date, if any, the Representative shall have received a certificate of the Company signed by the Chairman of the Board or the President and the Secretary or Assistant Secretary of the Company (in their capacities as such), dated the Closing Date or the Option Closing Date, as the case may be, respectively, to the effect that the Company has performed all covenants and complied with all conditions required by this Agreement to be performed or complied with by the Company prior to and as of the Closing Date, or the Option Closing Date, as the case may be, and that the conditions set forth in Section 4.5 hereof have been satisfied as of such date and that, as of Closing Date and the Option Closing Date, as the case may be, the representations and warranties of the Company set forth in Section 2 hereof are true and correct. In addition, the Representative will have received such other and further certificates of officers of the Company (in their capacities as such) as the Representative may reasonably request.

4.4.2 <u>Secretary's Certificate</u>. At each of the Closing Date and the Option Date, if any, the Representative shall have received a certificate of the Company signed by the Secretary or Assistant Secretary of the Company, dated the Closing Date or the Option Date, as the case may be, respectively, certifying (i) that the Bylaws and Certificate of Incorporation of the Company are true and complete, have not been modified and are in full force and effect, (ii) that the resolutions relating to the public offering contemplated by this Agreement are in full force and effect and have not been modified, (iii) all correspondence between the Company or its counsel and the Commission, and (iv) as to the incumbency of the officers of the Company. The documents referred to in such certificate shall be attached to such certificate.

4.5 No Material Changes. Prior to and on each of the Closing Date and the Option Closing Date, if any, (i) there shall have been no material adverse change or development involving a prospective material adverse change in the condition or prospects or the business activities, financial or otherwise, of the Company from the latest dates as of which such condition is set forth in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, (ii) no action suit or proceeding, at law or in equity, shall have been pending or threatened against the Company or any Existing Stockholder before or by any court or Federal or state commission, board or other administrative agency wherein an unfavorable decision, ruling or finding may materially adversely affect the business, operations, prospects or financial condition or income of the Company, except as set forth in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus, (iii) no stop order shall have been issued under the Act and no proceedings therefor shall have been initiated or threatened by the Commission, and (iv) the Registration Statement, the Sale Preliminary Prospectus and the Prospectus and any amendments or supplements thereto shall contain all material statements which are required to be stated therein in accordance with the Act and the Regulations and shall conform in all material respects to the requirements of the Act and the Regulations, and neither the Registration Statement, the Sale Preliminary Prospectus nor the Prospectus nor any amendment or supplement thereto shall contain any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

4.6 Delivery of Agreements.

- 4.6.1 <u>Effective Date Deliveries</u>. On the Effective Date, the Company shall have delivered to the Representative executed copies of the Securities Escrow Agreement, the Trust Agreement, the Warrant Agreement, the Services Agreement, and all of the Insider Letters and Subscription Agreement.
- 4.6.2 <u>Closing Date Deliveries</u>. On the Closing Date, the Company shall have delivered to the Representative executed copies of the Representative's Purchase Option.
- 4.7 Opinion of Counsel for the Underwriters. All proceedings taken in connection with the authorization, issuance or sale of the Securities as herein contemplated shall be reasonably satisfactory in form and substance to you and to McDermott and you shall have received from such counsel a favorable opinion, dated the Closing Date and the Option Closing Date, if any, with respect to such of these proceedings as you may reasonably require. On or prior to the Effective Date, the Closing Date and the Option Closing Date, as the case may be, counsel for the Underwriters shall have been furnished such documents, certificates and opinions as they may reasonably require for the purpose of enabling them to review or pass upon the matters referred to in this Section 4.7, or in order to evidence the accuracy, completeness or satisfaction of any of the representations, warranties or conditions herein contained.
- 4.8 <u>Secondary Market Trading</u>. Unless the Public Securities are listed or quoted, as the case may be, on the New York Stock Exchange, the American Stock Exchange, the NASDAQ Global Market, or the NASDAQ Capital Market, the Company shall take such other action as may be reasonably requested by the Representative to obtain a secondary market trading exemption in such other states as may be requested by the Representative.
 - 5. Indemnification.
 - 5.1 Indemnification of Underwriters.

5.1.1 General. Subject to the conditions set forth below, the Company agrees to indemnify and hold harmless each of the Underwriters, and each dealer selected by you that participates in the offer and sale of the Units (each a "Selected Dealer") and each of their respective directors, officers and employees and each person, if any, who controls any such Underwriter ("controlling person") within the meaning of Section 15 of the Act or Section 20(a) of the Exchange Act, against any and all loss. liability, claim, damage and expense whatsoever as incurred to which they or any of them may become subject under the Act, the Exchange Act or any other statute or at common law or otherwise or under the laws of foreign countries, arising out of or based upon any untrue statement or alleged untrue statement of a material fact contained in (i) any Preliminary Prospectus, the Registration Statement, Sale Preliminary Prospectus or the Prospectus (as from time to time each may be amended and supplemented, including, but not limited to any information deemed to be a part thereof pursuant to Rule 430A, Rule 430B or Rule 430C); (ii) any materials or information provided to investors by, or with the approval of, the Company in connection with the marketing of the offering of the Securities, including any "road show" or investor presentations made to investors by the Company (whether in person or electronically); (iii) any application or other document or written communication (in this Section 5, collectively called "application") executed by the Company or based upon written information furnished by the Company in any jurisdiction in order to qualify the Securities under the securities laws thereof or filed with the Commission, any state securities commission or agency or the OTC Bulletin Board; or (iv) any post-effective amendments to the Registration Statement or Prospectus or new Registration Statement or Prospectus filed by the Company with the Commission, any state securities commission or agency or the OTC Bulletin Board, or the omission or alleged omission from any Preliminary Prospectus, the Registration Statement, the Sale Preliminary Prospectus or the Prospectus or subsequent filing by the Company under clause (iv) of a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, and to reimburse each Underwriter, each Selected Dealer and each of their respective directors, officers and employees and each controlling person, if any, for any and all expenses (including the fees and disbursements or counsel chosen by Morgan Joseph & Co.) as such expenses are incurred by such Underwriter, such Selected Dealer or each of their respective directors, officers and employees or such controlling person in connection with investigating, defending, settling, compromising or paying any such loss, claim damage, liability, expense or action; provided however, that the foregoing indemnity agreement shall not apply to any loss, claim, damage, liability or expenses to the extent, but only to the extent, arising out of or based upon (x) any untrue statement or alleged untrue statement or omission or alleged omission made in reliance upon and in conformity with written information furnished to the Company with respect to an Underwriter by or on behalf of such Underwriter expressly for use in any Preliminary Prospectus, the Registration Statement, Sale Preliminary Prospectus or the Prospectus, or any amendment or supplement thereof, or in any application, as the case may be, or (y) the use of the Sale Preliminary Prospectus or Prospectus in violation of any stop order or other notice received by any Underwriter indicating the then-current Prospectus is not to be used in connection with the sale of any Securities or (z) an Underwriter otherwise failing in its prospectus delivery obligations. With respect to any untrue statement or omission or alleged untrue statement or omission made in the Preliminary Prospectus, the indemnity agreement contained in this paragraph shall not inure to the benefit of any Underwriter to the extent that any loss, liability, claim, damage or expense of such Underwriter results from the fact that a copy of the Prospectus was not given or sent to the person asserting any such loss, liability, claim or damage at or prior to the written confirmation of sale of the Securities to such person as required by the Act and the Regulations, and if the untrue statement or omission has been corrected in the Prospectus, unless such failure to deliver the Prospectus was a result of non-compliance by the Company with its obligations under Section 3.4 hereof. The Company agrees promptly to notify the Representative of the commencement of any litigation or proceedings against the Company or any of its officers, directors or controlling persons in connection with the issue and sale of the Securities or in connection with the Registration Statement, the Sale Preliminary Prospectus or the Prospectus. The indemnity agreement set forth in this Section 5.1 shall be in addition to any liabilities that the Company may otherwise have.

5.2 Indemnification of the Company. Each Underwriter, severally and not jointly, agrees to indemnify and hold harmless the Company, each of its directors, each of its officers who signed the Registration Statement and each person, if any, who controls the Company within the meaning of Section 15 of the Act or Section 20 of the Exchange Act against any and all loss, liability, claim, damage and expense described in the foregoing indemnity from the Company to the several Underwriters, as incurred, but only with respect to untrue statements or omissions, or alleged untrue statements or omissions made in any Preliminary Prospectus, the Registration Statement, the Sale Preliminary Prospectus, the Prospectus or any amendment or supplement thereto or in any application, in reliance upon, and in strict conformity with, written information furnished to the Company with respect to such Underwriter by or on behalf of the Underwriter expressly for use in such Preliminary Prospectus, the Registration Statement, the Sale Preliminary Prospectus, the Prospectus or any amendment or supplement thereto or in any such application; and to reimburse the Company or any such director, officer or controlling person, if any, for any and all expenses as such expenses are reasonably incurred, in connection with investigating, defending, settling, compromising or paying any such loss, claim damage, liability, expense or action; provided, however, that the obligation of each Underwriter to indemnify the Company (including any director, officer or controlling person thereof), shall be limited to the commissions received by such Underwriter in connection with the Securities underwritten by it. The Company hereby acknowledges that the only information that the Underwriters have furnished to the Company expressly for use in the Preliminary Prospectus, the Registration Statement, the Sale Preliminary Prospectus, the Prospectus or any amendment or supplement thereto or in any such application, are the statements set forth in the paragraphs entitled "Pricing of Securities" and "Commissions and Discounts" under the caption "Underwriting" in the Prospectus. The indemnity agreement set forth in this Section 5.2 shall be in addition to any liabilities that each Underwriter may otherwise have.

5.3 Notifications and Other Indemnification Procedures. Promptly after receipt by an indemnified party under this Section 5 of notice of the commencement of any action, such indemnified party will, if a claim in respect thereof is to be made against an indemnifying party under this Section 5, notify the indemnifying party in writing of the commencement thereof, but the failure to so notify the indemnifying party (i) will not relieve it from liability under paragraph 5.1 or 5.2 above unless and to the extent it did not otherwise learn of such action and such failure results in the forfeiture by the indemnifying party of substantial rights and defenses and (ii) will not, in any event, relieve the indemnifying party from any obligations to any indemnified party other than the indemnification obligation provided in paragraph 5.1 or 5.2 above. In case any such action is brought against any indemnified party and such indemnified party seeks or intends to seek indemnity from an indemnifying party, the indemnifying party will be entitled to participate in, and, to the extent that it shall elect, jointly with all other indemnifying parties similarly notified, by written notice delivered to the indemnified party promptly after receiving the aforesaid notice from such indemnified party, to assume the defense thereof with counsel satisfactory to such indemnified party; provided, however, if the defendants in any such action include both the indemnified party and the indemnifying party and the indemnified party shall have reasonably concluded that a conflict may arise between the positions of the indemnifying party and the indemnified party in conducting the defense of any such action or that there may be legal defenses available to it and/or other indemnified parties that are different from or additional to those available to the indemnifying party, the indemnified party or parties shall have the right to select separate counsel to assume such legal defenses and to otherwise participate in the defense of such action on behalf of such indemnified party or parties. Upon receipt of notice from the indemnifying party to such indemnified party of such indemnifying party's election so to assume the defense of such action and approval by the indemnified party of counsel, the indemnifying party will not be liable to such indemnified party under this Section 5 for any legal or other expenses subsequently incurred by such indemnified party in connection with the defense thereof unless (i) the indemnified party shall have employed separate counsel in accordance with the provision of the preceding sentence reasonably approved by the indemnifying party (or by Morgan Joseph & Co. in the case of Section 5.2), representing the indemnified parties who are parties to such action or (ii) the indemnifying party shall not have employed counsel satisfactory to the indemnified party to represent the indemnified party within a reasonable time after notice of commencement of the action, in each of which cases the fees and expenses of counsel shall be at the expense of the indemnifying party.

5.4 <u>Settlements</u>. The indemnifying party under this Section 5 shall not be liable for any settlement of any proceeding effected without its written consent, which shall not be withheld, delayed or conditioned unreasonably, but if settled with such consent or if there is a final judgment for the plaintiff, the indemnifying party agrees to indemnify the indemnified party against any loss, claim, damage, liability or expense by reason of such settlement or judgment. Notwithstanding the foregoing sentence, if at any time an indemnified party shall have requested an indemnifying party to reimburse the indemnified party for fees and expenses of counsel as contemplated by Section 5.3 hereof, the indemnifying party agrees that it shall be liable for any settlement of any proceeding effected without its written consent if (i) such settlement is entered into more than 30 days after receipt by such indemnifying party of the aforesaid request and (ii) such indemnifying party shall not have reimbursed the indemnified party in accordance with such request prior to the date of such settlement. No indemnifying party shall, without the prior written consent of the indemnified party, effect any settlement, compromise or consent to the entry of judgment in any pending or threatened action, suit or proceeding in respect of which any indemnified party is or could have been a party and indemnity was or could have been sought hereunder by such indemnified party, unless such settlement, compromise or consent (x) includes an unconditional release of such indemnified party from all liability on claims that are the subject matter of such action, suit or proceeding and (y) does not include a statement as to or an admission of fault, culpability or a failure to act, by or on behalf of any indemnified party.

5.5 Contribution.

5.5.1 Contribution Rights. In order to provide for just and equitable contribution under the Act in any case in which (i) any person entitled to indemnification under this Section 5 makes claim for indemnification pursuant hereto but it is judicially determined (by the entry of a final judgment or decree by a court of competent jurisdiction and the expiration of time to appeal or the denial of the last right of appeal) that such indemnification may not be enforced in such case notwithstanding the fact that this Section 5 provides for indemnification in such case, or (ii) contribution under the Act, the Exchange Act or otherwise may be required on the part of any such person in circumstances for which indemnification is provided under this Section 5, then, and in each such case, the Company and the Underwriters shall contribute to the aggregate losses, liabilities, claims, damages and expenses of the nature contemplated by said indemnity agreement incurred by the Company and the Underwriters, as incurred, in such proportions that the Underwriters are responsible for that portion represented by the percentage that the underwriting discount appearing on the cover page of the Prospectus bears to the initial offering price appearing thereon and the Company is responsible for the balance; provided, that, no person guilty of a fraudulent misrepresentation (within the meaning of Section 11(f) of the Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation. If the allocation provided by the immediately preceding sentence is unavailable for any reason, the Company and the Underwriters shall contribute in such proportion as is appropriate to reflect the relative fault of the Company and the Underwriters in connection with the actions or omissions which resulted in such loss, claim, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of the Company and the Underwriters shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company or the Underwriters and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. Notwithstanding the provisions of this Section 5.5.1, no Underwriter shall be required to contribute any amount in excess of the underwriting commissions received by such Underwriter in connection with the Securities underwritten by it and distributed to the public. For purposes of this Section, each director, officer and employee of an Underwriter or the Company, as applicable, and each person, if any, who controls an Underwriter or the Company, as applicable, within the meaning of Section 15 of the Act shall have the same rights to contribution as the Underwriters or the Company, as applicable.

5.5.2 Contribution Procedure. Within fifteen days after receipt by any party to this Agreement (or its representative) of notice of the commencement of any action, suit or proceeding, such party will, if a claim for contribution in respect thereof is to be made against another party ("contributing party"), notify the contributing party of the commencement thereof, but the omission to so notify the contributing party will not relieve it from any liability which it may have to any other party other than for contribution hereunder. In case any such action, suit or proceeding is brought against any party, and such party notifies a contributing party or its representative of the commencement thereof within the aforesaid fifteen days, the contributing party will be entitled to participate therein with the notifying party and any other contributing party similarly notified. Any such contributing party shall not be liable to any party seeking contribution on account of any settlement of any claim, action or proceeding effected by such party seeking contribution without the written consent of such contributing party. The contribution provisions contained in this Section are intended to supersede, to the extent permitted by law, any right to contribution under the Act, the Exchange Act or otherwise available. The Underwriters' obligations to contribute pursuant to this Section 5.3 are several and not joint.

6. Default by an Underwriter.

- 6.1 <u>Default Not Exceeding 10% of Firm Units or Option Units</u>. If any Underwriter or Underwriters shall default in its or their obligations to purchase the Firm Units or the Option Units, if the Over-Allotment Option is exercised, hereunder, and if the number of the Firm Units or Option Units with respect to which such default relates does not exceed in the aggregate 10% of the number of Firm Units or Option Units that all Underwriters have agreed to purchase hereunder, then such Firm Units or Option Units to which the default relates shall be purchased by the non-defaulting Underwriters in proportion to their respective commitments hereunder.
- 6.2 <u>Default Exceeding 10% of Firm Units or Option Units</u>. In the event that the default addressed in Section 6.1 above relates to more than 10% of the Firm Units or Option Units, you may in your discretion arrange for yourself or for another party or parties to purchase such Firm Units or Option Units to which such default relates on the terms contained herein. If within one Business Day after such default relating to more than 10% of the Firm Units or Option Units you do not arrange for the purchase of such Firm Units or Option Units, then the Company shall be entitled to a further period of one Business Day within which to procure another party or parties satisfactory to you to purchase said Firm Units or Option Units on such terms. In the event that neither you nor the Company arrange for the purchase of the Firm Units or Option Units to which a default relates as provided in this Section 6, this Agreement will be terminated by you or the Company without liability on the part of the Company (except as provided in Sections 3.12 and 5 hereof) or the several Underwriters (except as provided in Section 5 hereof); provided, however, that if such default occurs with respect to the Option Units, this Agreement will not terminate as to the Firm Units; and provided further that nothing herein shall relieve a defaulting Underwriter of its liability, if any, to the other several Underwriters and to the Company for damages occasioned by its default hereunder.
- 6.3 <u>Postponement of Closing Date</u>. In the event that the Firm Units or Option Units to which the default relates are to be purchased by the non-defaulting Underwriters, or are to be purchased by another party or parties as aforesaid, you or the Company shall have the right to postpone the Closing Date or Option Closing Date for a reasonable period, but not in any event exceeding five Business Days, in order to effect whatever changes may thereby be made necessary in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus or in any other documents and arrangements, and the Company agrees to file promptly any amendment to the Registration Statement or the Prospectus that in the opinion of counsel for the Underwriters may thereby be made necessary. The term "<u>Underwriter</u>" as used in this Agreement shall include any party substituted under this Section 6 with like effect as if it had originally been a party to this Agreement with respect to such Securities.
- 7. Right to Appoint Observer. Until the consummation of a Business Combination, upon notice from Morgan Joseph & Co. to the Company, Morgan Joseph & Co. shall have the right to send a representative (who need not be the same individual from meeting to meeting) to observe each meeting of the Board of Directors of the Company; provided that such representative shall sign a Regulation FD compliant confidentiality agreement which is reasonably acceptable to Morgan Joseph & Co. and its counsel in connection with such representative's attendance at meetings of the Board of Directors; and provided further that upon written notice to Morgan Joseph & Co., the Company may exclude the representative from meetings where, in the written opinion of counsel for the Company, the representative's presence would destroy the attorney-client privilege. The Company agrees to give Morgan Joseph & Co. written notice of each such meeting and to provide Morgan Joseph & Co. with an agenda and minutes of the meeting no later than it gives such notice and provides such items to the other directors and to reimburse the representative of Morgan Joseph & Co. for his reasonable out-of-pocket expenses incurred in connection with his attendance at the meeting, including but not limited to, food, lodging and transportation.

8. Additional Covenants.

- 8.1 Additional Shares or Options. The Company hereby agrees that until the consummation of a Business Combination, it shall not issue any shares of Common Stock (except with respect to any exercise of Warrants) or any options or other securities convertible into Common Stock, or any shares of preferred stock or other Securities of the Company which participate in any manner in the Trust Account or which vote as a class with the Common Stock on a Business Combination.
- 8.2 Trust Account Waiver Acknowledgment. The Company hereby agrees that it will use its reasonable best efforts prior to engaging in discussions with any operating business which the Company seeks to acquire ("Target Business") or obtaining the services of any vendor to acknowledge in writing whether through a letter of intent, memorandum of understanding or other similar document (and subsequently acknowledges the same in any definitive document replacing any of the foregoing, that (a) it has read the Prospectus and understands that the Company has established the Trust Account, initially in an amount of \$49,500,000 (without giving effect to any exercise of the Over-allotment Option) for the benefit of the public stockholders and that, except for a portion of the interest earned on the amounts held in the Trust Account, the Company may disburse monies from the Trust Account only (i) to the public stockholders in the event they vote against a Business Combination and elect to redeem their IPO Shares (as defined below in Section 8.5), (ii) to the holders of the IPO Shares upon the liquidation of the Company if the Company fails to consummate a Business Combination, (iii) after or concurrently with the consummation of a Business Combination, or (iv) to the Company only with respect to the interest income, net of taxes, to fund working capital and (b) for and in consideration of the Company (i) agreeing to evaluate such Target Business for purposes of consummating a Business Combination with it or (ii) agreeing to engage the services of the vendor, as the case may be, such Target Business or vendor agrees that it does not have any right, title, interest or claim of any kind in or to any monies in the Trust Account ("Claim") and waives any Claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with the Company and will not seek recourse against the Trust Account for any reason whatsoever. The foregoing letters shall substantially be in the form attached hereto as Exhibits A and B respectively. The Company may forego obtaining such waivers only if the Company shall have received the approval of its Chief Executive Officer and the approving vote or written consent of at least a majority of its Board of Directors.

8.3 <u>Insider Letters</u>. The Company shall not take any action or omit to take any action which would cause a breach of any of the Insider Letters or the warrant purchase agreements executed between each Existing Stockholder and Morgan Joseph & Co. and will not allow any amendments to, or waivers of, such Insider Letters without the prior written consent of the Representative.

8.4 <u>Certificate of Incorporation and Bylaws</u>. The Company shall not take any action or omit to take any action that would cause the Company to be in breach or violation of its Certificate of Incorporation or Bylaws. Prior to the consummation of a Business Combination, the Company will not amend certain provisions of its Certificate of Incorporation (as set forth in the Certificate of Incorporation) without: (a) the prior written consent of holders holding not less than 95% of the Public Securities and (b) providing prior written notice to the Representative of its intention to amend such provision(s).

8.5 Acquisition/Liquidation Procedure. The Company agrees: (i) prior to the consummation of any Business Combination, it will submit such transaction to the Company's stockholders for their approval ("Business Combination Vote") even if the nature of the acquisition is such as would not ordinarily require stockholder approval under the laws of the state of Delaware; and (ii) in the event that the Company does not effect a business combination within twenty-four months from the consummation of the offering (the "Termination Date"), this shall trigger an automatic winding-up of the Company and the trust account will be liquidated to holders of IPO Shares in the manner described in the Sale Preliminary Prospectus and the Prospectus as soon as reasonably practicable, and subject to the requirements of the laws of the state of Delaware. For purposes of this Section 8.5, the term "IPO Shares" means the shares of Common Stock contained in the Public Securities.

8.5.1 Upon liquidation of the Trust Account, subject to the requirements of the laws of the State of Delaware, the Company will distribute only to the holders of IPO Shares an aggregate sum equal to the Company's Liquidation Value, which sum shall be distributed pro rata among the holders of the IPO Shares. The Company's "Liquidation Value" means: (i) all principal and accrued interest contained within the Trust Account, less any amounts previously distributed to the Company out of the interest earned on the Trust Account pursuant to the terms of the Trust Agreement (after payment of, or provision for, applicable taxes and claims of creditors) PLUS (ii) all cash and other liquid assets (which shall be reduced to cash as part of the Company's winding up) then held by the Company outside of the Trust Account, all as distributed in amounts to the holders as determined by CST, as trustee of the Trust Account. Only holders of IPO Shares as of the record date for the distribution shall be entitled to receive liquidating distributions with respect to the IPO Shares they beneficially own and the Company shall pay no liquidating distributions with respect to any other shares of capital stock of the Company, including the shares of Common Stock held by the Existing Stockholders prior to the Offering (but shall include Common Stock purchased by Existing Stockholders in or after the Offering).

8.5.2 With respect to the Business Combination Vote, the Company shall use its commercially reasonable efforts to cause the Existing Stockholders to vote all Common Stock (i) owned by them (either directly or indirectly) prior to the Offering in accordance with the majority of the votes cast by the holders of the IPO Shares and (ii) purchased by them in or following the IPO "for" a Business Combination.

8.5.3 At the time the Company seeks approval of any potential Business Combination (prior to the confirmation of its initial Business Combination), the Company will offer each of the holders of the IPO Shares the right to redeem their IPO Shares into a pro rata share of the Trust Account (the "Conversion Price"). If holders of less than 30% in interest of the Company's Common Stock vote against such approval of a Business Combination, the Company may, but will not be required to, proceed with such Business Combination. If the Company elects to so proceed, it will redeem shares of Common Stock, based upon the Conversion Price, from those holders of Common Stock who affirmatively requested such redemption and who voted against the Business Combination as provided under the laws of the State of Delaware. If holders of 30% or more in interest of the Common Stock vote against approval of any potential Business Combination and seek to redeem their shares of Common Stock, the Company will not proceed with such Business Combination and will liquidate or seek another Business Combination, if permitted.

8.6 Rule 419. The Company agrees that it will use its best efforts to prevent the Company from becoming subject to Rule 419 under the Act prior to the consummation of any Business Combination, including but not limited to using its best efforts to prevent any of the Company's outstanding securities from being deemed to be a "penny stock" as defined in Rule 3a-51-1 under the Exchange Act during such period.

8.7 <u>Affiliated Transactions</u>. The Company shall cause each of the Existing Stockholders to agree that, in order to minimize potential conflicts of interest which may arise from multiple affiliations, the Existing Stockholders will present to the Company for its consideration, prior to presentation to any other person or company, any suitable opportunity to acquire an operating business, until the earlier of the consummation by the Company of a Business Combination, the liquidation of the Company or until such time as the Existing Stockholders cease to be an officer or director of the Company, subject to any pre-existing fiduciary or contractual obligations the Existing Stockholders might have.

8.8 <u>Target Net Assets</u>. The Company agrees that its initial Business Combination must be with one or more Target Businesses that have an aggregate fair market value equal to at least 80% of the amount in the Company's Trust Account (less Deferred Fees, including interest thereon, held in the Trust Account) at the time of such Business Combination. The fair market value of each Target Business must be determined by the Board of Directors of the Company based upon standards generally accepted by the financial community, such as actual and potential sales, earnings and cash flow and book value. If the Board of Directors of the Company is not able to independently determine that the Target Business has a sufficient fair market value at the time of such transaction, or if the Target Business is affiliated with any of the existing stockholders, the Company will obtain an opinion from an unaffiliated, independent investment banking firm which is a member of the NASD with respect to the satisfaction of such criteria. The Company is not required to obtain an opinion from an investment banking firm as to the fair market value if the Company's Board of Directors independently determines that the Target Business does have sufficient fair market value.

8.9 <u>Proxy and Other Information</u>. The Company shall provide counsel to the Representative with ten copies of all proxy information and all related material filed with the Commission in connection with a Business Combination concurrently with such filing with the Commission. In addition, the Company shall furnish any other state in which its initial public offering was registered, such information as may be requested by such state.

9. Representations and Agreements to Survive Delivery. Except as the context otherwise requires, all representations, warranties and agreements contained in this Agreement shall be deemed to be representations, warranties and agreements as of the Closing Dates and such representations, warranties and agreements of the Underwriters and the Company, including the indemnity agreements contained in Section 5 hereof, shall remain operative and in full force and effect regardless of any investigation made by or on behalf of any Underwriter, the Company or any controlling person, and shall survive termination of this Agreement or the issuance and delivery of the Securities to the several Underwriters until the earlier of the expiration of any applicable statute of limitations and the seventh anniversary of the later of the Closing Date or the Option Closing Date, if any, at which time the representations, warranties and agreements shall terminate and be of no further force and effect.

10. Effective Date of This Agreement and Termination Thereof.

10.1 <u>Effective Date</u>. This Agreement shall become effective on the Effective Date at the time the Registration Statement is declared effective by the Commission.

10.2 Termination. You shall have the right to terminate this Agreement at any time prior to the Closing Date, (i) if any domestic or international event or act or occurrence has materially disrupted, or in your opinion will in the immediate future materially disrupt, general securities markets in the United States; or (ii) if trading on the New York Stock Exchange, the American Stock Exchange, the NASDAQ Global Market, the NASDAQ Capital Market or on the NASD OTC Bulletin Board (or successor trading market) shall have been suspended, or minimum or maximum prices for trading shall have been fixed, or maximum ranges for prices for securities shall have been fixed, or maximum ranges for prices for securities shall have been required on the American Stock Exchange, the NASDAQ Global Market, the NASDAQ Capital Market or on the NASD OTC Bulletin Board (or successor trading market) or by order of the Commission or any other government authority having jurisdiction, or (iii) if the United States shall have become involved in a new war or an increase in major hostilities, or (iv) if a banking moratorium has been declared by a New York State or Federal authority, or (v) if a moratorium on foreign exchange trading has been declared which materially adversely impacts the United States securities market, or (vi) if the Company shall have sustained a material loss by fire, flood, accident, hurricane, earthquake, theft, sabotage or other calamity or malicious act which, whether or not such loss shall have been insured, will, in your opinion, make it inadvisable to proceed with the delivery of the Units, or (vii) if any of the Company's representations, warranties or covenants hereunder are materially breached, or (viii) if the Representative shall have become aware after the date hereof of such a material adverse change in the conditions or prospects of the Company, or such adverse material change in general market conditions, including without limitation as a result of terrorist activities after the date hereof, as in the Representative's judgment would make it impracticable to proceed with the offering, sale and/or delivery of the Units or to enforce contracts made by the Underwriters for the sale of the Securities.

10.3 <u>Indemnification</u>. Notwithstanding any contrary provision contained in this Agreement, any election hereunder or any termination of this Agreement, and whether or not this Agreement is otherwise carried out, the provisions of Section 5 shall not be in any way effected by such election or termination or failure to carry out the terms of this Agreement or any part hereof.

11. Miscellaneous.

11.1 <u>Notices</u>. All communications hereunder, except as herein otherwise specifically provided, shall be in writing and shall be mailed, delivered or telecopied and confirmed and shall be deemed given when so delivered or telecopied and confirmed or if mailed, two days after such mailing

If to the Representative:

Morgan Joseph & Co. Inc. 600 Fifth Avenue 19th Floor New York, New York 10020 Attn: Tina Pappas Facsimile: (212-218-3725

Copy to:

McDermott Will & Emery LLP 340 Madison Avenue New York, New York 10173 Attn: Joel Rubinstein, Esq. Facsimile: (212) 547-5444

If to the Company:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202 Attn: President

Facsimile: (410) 878-6850

Copy to:

Ellenoff Grossman & Schole LLP 370 Lexington Avenue New York, NY 10017 Attn: Douglas S. Ellenoff, Esq. Facsimile: (212) 370-7889

- 11.2 <u>Headings</u>. The headings contained herein are for the sole purpose of convenience of reference, and shall not in any way limit or affect the meaning or interpretation of any of the terms or provisions of this Agreement.
- 11.3 <u>Amendment</u>. This Agreement may only be amended by a written instrument executed by each of the parties hereto.
- 11.4 <u>Entire Agreement</u>. This Agreement (together with the other agreements and documents being delivered pursuant to or in connection with this Agreement) constitute the entire agreement of the parties hereto with respect to the subject matter hereof and thereof, and supersede all prior agreements and understandings of the parties, oral and written, with respect to the subject matter hereof.
- 11.5 <u>Binding Effect</u>. This Agreement shall inure solely to the benefit of and shall be binding upon the Representative, the Underwriters, the Company and the controlling persons, directors and officers referred to in Section 5 hereof, and their respective successors, legal representatives and assigns, and no other person shall have or be construed to have any legal or equitable right, remedy or claim under or in respect of or by virtue of this Agreement or any provisions herein contained. This agreement and all conditions and provisions hereof are intended to be for the sole and exclusive benefit of the parties hereto and said controlling persons and their respective successors, officers, directors, heirs and legal representatives, and it is not for the benefit of any other person. The term "successors and assigns" shall not include a purchaser, in its capacity as such, of securities from any of the Underwriters. The Company acknowledges and agrees that: (i) the sale and issuance of the securities pursuant to this Agreement is an arm's-length commercial transaction between the Company and the Underwriters; (ii) in connection therewith and with the process leading to the offering, the Underwriters are acting solely as a principal and not the agent or fiduciary of the Company; (iii) no Underwriter has assumed an advisory or fiduciary responsibility in favor of the Company with respect to the offering contemplated hereby or the process leading thereto, including any negotiation related to the pricing of the securities; and (iv) the Company has consulted its own legal and financial advisors to the extent it has deemed appropriate in connection with this Agreement and the offering.
- 11.6 Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The Company hereby agrees that any action, proceeding or claim against it arising out of, or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum. Any such process or summons to be served upon the Company may be served by transmitting a copy thereof by registered or certified mail, return receipt requested, postage prepaid, addressed to it at the address set forth in Section 11.1 hereof. Such mailing shall be deemed personal service and shall be legal and binding upon the Company in any action, proceeding or claim. The Company agrees that the prevailing party(ies) in any such action or proceeding and/or incurred in connection with the preparation therefor.

11.7 Execution in Counterparts. This Agreement may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which shall be deemed to be an original, but all of which taken together shall constitute one and the same agreement, and shall become effective when one or more counterparts has been signed by each of the parties hereto and delivered to each of the other parties hereto. Delivery of a signed counterpart of this Agreement by fax or email/pdf transmission shall constitute valid and sufficient delivery thereof.

11.8 <u>Waiver, Etc.</u> The failure of any of the parties hereto to at any time enforce any of the provisions of this Agreement shall not be deemed or construed to be a waiver of any such provision, nor to in any way affect the validity of this Agreement or any provision hereof or the right of any of the parties hereto to thereafter enforce each and every provision of this Agreement. No waiver of any breach, non-compliance or non-fulfillment of any of the provisions of this Agreement shall be effective unless set forth in a written instrument executed by the party or parties against whom or which enforcement of such waiver is sought; and no waiver of any such breach, non-compliance or non-fulfillment shall be construed or deemed to be a waiver of any other or subsequent breach, non-compliance or non-fulfillment.

[Remainder of page intentionally left blank]

If the foregoing correctly sets forth the understanding between the Underwriters and the Company, please so indicat in the space provided below for that purpose, whereupon this letter shall constitute a binding agreement between us.				
	Very truly yours, CAMDEN LEARNING CORPORATION			
	By: Name: David L. Warnock Title: President and Chief Executive Officer			
Accepted on the date first above written.				
MORGAN JOSEPH & CO. INC.				

By:____ Name: [_

Title: Managing Director

CAMDEN LEARNING CORPORATION

6,250,000 Units

Number of Firm Units to be Purchased			

APPENDIX A

- 1. Based solely on a certificate of good standing dated within 10 days of the date hereof, the Company has been duly organized and is validly existing as a corporation and is in good standing under the laws of its state of incorporation.
- 2. All issued and outstanding securities of the Company (including, without limitation, the Placement Warrants) have been duly authorized and validly issued and are fully paid and non-assessable; the holders thereof are not subject to personal liability by reason of being such holders; and none of such securities were issued in violation of the preemptive rights of any stockholder of the Company arising by operation of law or under the Certificate of Incorporation or Bylaws of the Company. The offers and sales of the outstanding Common Stock were at all relevant times either registered under the Act or exempt from such registration requirements. The authorized, and to the extent such counsel's knowledge, outstanding capital stock of the Company is as set forth in the Prospectus.
- 3. The Securities have been duly authorized and, when issued and paid for, will be validly issued, fully paid and non-assessable; the holders thereof are not and will not be subject to personal liability solely by reason of being such holders. The Securities are not and will not be subject to the preemptive rights of any holders of any security of the Company arising by operation of law or under the Certificate of Incorporation or Bylaws of the Company. When issued, the Representative's Purchase Option, the Representative's Warrants and the Warrants will constitute valid and binding obligations of the Company to issue and sell, upon exercise thereof and payment therefor, the number and type of securities of the Company called for thereby and such Warrants, the Representative's Purchase Option, and the Representative's Warrants, when issued, in each case, are enforceable against the Company in accordance with their respective terms, except (a) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (b) as enforceability of any indemnification or contribution provision may be limited under the Federal and state securities laws, and (c) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought. The certificates representing the Securities are in due and proper form.
- 4. The Placement Warrants constitute valid and binding obligations of the Company to issue and sell, upon exercise thereof and payment therefor, the number and type of securities of the Company called for thereby, and such Placement Warrants are enforceable against the Company in accordance with their respective terms, except: (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally; (ii) as enforceability of any indemnification or contribution provision may be limited under federal and state securities laws; and (iii) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought. A sufficient number of shares of Common Stock have been reserved for issuance upon exercise of the Placement Warrants. The shares of Common Stock underlying the Placement Warrants will, upon exercise of the Warrants and payment of the exercise price thereof, be duly and validly issued, fully paid and non-assessable.

- 5. This Agreement, the Warrant Agreement, the Services Agreement, the Trust Agreement, the Securities Escrow Agreement and the Subscription Agreement have each been duly and validly authorized and, when executed and delivered by the Company, constitute, and the Representative's Purchase Option has been duly and validly authorized by the Company and, when executed and delivered, will constitute, the valid and binding obligations of the Company, enforceable against the Company in accordance with their respective terms, except (a) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally, (b) as enforceability of any indemnification or contribution provisions may be limited under the Federal and state securities laws, and (c) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.
- 6. The execution, delivery and performance of this Agreement, the Warrant Agreement, the Representative's Purchase Option, the Securities Escrow Agreement, the Trust Agreement, the Services Agreement and the Subscription Agreement and compliance by the Company with the terms and provisions thereof and the consummation of the transactions contemplated thereby, and the issuance and sale of the Securities, do not and will not, with or without the giving of notice or the lapse of time, or both, (a) to such counsel's knowledge, conflict with, or result in a breach of, any of the terms or provisions of, or constitute a default under, or result in the creation or modification of any lien, security interest, charge or encumbrance upon any of the properties or assets of the Company pursuant to the terms of, any mortgage, deed of trust, note, indenture, loan, contract, commitment or other agreement or instrument filed as an exhibit to the Registration Statement, (b) result in any violation of the provisions of the Certificate of Incorporation or the Bylaws of the Company or (c) to such counsel's knowledge, violate any United States statute or any judgment, order or decree, rule or regulation applicable to the Company of any court, United States Federal, state or other regulatory authority or other governmental body having jurisdiction over the Company, its properties or assets.
- 7. The Registration Statement, the Sale Preliminary Prospectus and the Prospectus and any post-effective amendments or supplements thereto (other than the financial statements included therein, as to which no opinion need be rendered) each as of their respective dates appeared on their face to comply as to form in all material respects with the requirements of the Act and Regulations. The Securities and all other securities issued or issuable by the Company conform in all material respects to the description thereof contained in the Registration Statement and the Prospectus. The descriptions in the Registration Statement, the Sale Preliminary Prospectus and in the Prospectus, insofar as such statements constitute a summary of statutes, legal matters, contracts, documents or proceedings referred to therein, fairly present in all material respects the information required to be shown with respect to such statutes, legal matters, contracts, documents and proceedings, and such counsel does not know of any statutes or legal or governmental proceedings required to be described in the Sale Preliminary Prospectus and the Prospectus that are not described in the Registration Statement, the Sale Preliminary Prospectus or included as exhibits to the Registration Statement that are not described or included as required.

- 8. Based solely on a notice of effectiveness received from the Securities and Exchange Commission, the Registration Statement is effective under the Act. To such counsel's knowledge, no stop order suspending the effectiveness of the Registration Statement has been issued and no proceedings for that purpose have been instituted or are pending or threatened under the Act or applicable state securities laws.
- 9. The Company is not and, after giving effect to the offering and sale of the Securities and the application of the proceeds thereof as described in the Registration Statement and the Prospectus, will not be, an "investment company" as defined in the Investment Company Act of 1940, as amended.

The opinion of counsel shall further include a statement to the effect that such counsel has participated in conferences with officers and other representatives of the Company, the Underwriters and the independent registered public accounting firm of the Company, at which conferences the contents of the Registration Statement, the Sale Preliminary Prospectus and the Prospectus contained therein and related matters were discussed and, although such counsel is not passing upon and does not assume any responsibility for the accuracy, completeness or fairness of the statements contained in the Registration Statement, the Sale Preliminary Prospectus and the Prospectus contained therein (except as otherwise set forth in the foregoing opinion), solely on the basis of the foregoing without independent check and verification, no facts have come to the attention of such counsel which lead them to believe that the Registration Statement or any amendment thereto, at the time the Registration Statement or amendment became effective, contained an untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein not misleading or the Prospectus or any amendment or supplement thereto, at the time they were filed pursuant to Rule 424(b) or at the date of such counsel's opinion, contained an untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statement therein, in light of the circumstances under which they were made, not misleading (except that such counsel need express no opinion with respect to the financial information and statistical data and information included in the Registration Statement, the Sale Preliminary Prospectus or the Prospectus).

EXHIBIT A

FORM OF TARGET BUSINESS LETTER

Camden	Learning	Corne	ration

Gentlemen:

Reference is made to the Final Prospectus of Camden Learning Corporation (the "<u>Company</u>"), dated _, 2007 (the "<u>Prospectus</u>"). Capitalized terms used and not otherwise defined herein shall have the meanings assigned to them in Prospectus.

We have read the Prospectus and understand that the Company has established the Trust Account, initially in an amount of at least \$_ for the benefit of the Public Stockholders and the Underwriters of the Company's initial public offering (the "<u>Underwriters</u>") and that, except for a portion of the interest earned on the amounts held in the Trust Account, the Company may disburse monies from the Trust Account only: (i) to the Public Stockholders in the event they elect to redeem their IPO Shares, (ii) to the Public Stockholders upon the liquidation of the Company if the Company fails to consummate a Business Combination or (iii) to the Company and the Underwriters after or concurrently with the consummation of a Business Combination.

For and in consideration of the Company agreeing to evaluate the undersigned for purposes of consummating a Business Combination with it, the undersigned hereby agrees that it does not have any right, title, interest or claim of any kind in or to any monies in the Trust Account (each, a "Claim") and hereby waives any Claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with the Company and will not seek recourse against the Trust Account for any reason whatsoever.

Print Name of Target Business

Authorized Signature of Target Business

EXHIBIT B

FORM OF VENDOR LETTER

Camden Learning Corporation

Gentlemen:

Reference is made to the Final Prospectus of Camden Learning Corporation (the "<u>Company</u>"), dated _, 2007 (the "<u>Prospectus</u>"). Capitalized terms used and not otherwise defined herein shall have the meanings assigned to them in Prospectus.

We have read the Prospectus and understand that the Company has established the Trust Account, initially in an amount of at least \$_ for the benefit of the Public Stockholders and the Underwriters of the Company's initial public offering (the "<u>Underwriters</u>") and that, except for a portion of the interest earned on the amounts held in the Trust Account, the Company may disburse monies from the Trust Account only: (i) to the Public Stockholders in the event they elect to redeem their IPO Shares, (ii) to the Public Stockholders upon the liquidation of the Company if the Company fails to consummate a Business Combination or (iii) to the Company and the Underwriters after or concurrently with the consummation of a Business Combination.

For and in consideration of the Company agreeing to engage the services of the undersigned, the undersigned hereby agrees

waives any Claim it may have in the frecourse against the Trust Account for	uture as a result of, or ar	ising out of, any service	•	, <u> </u>
Print Name of Vendor				
Authorized Signature of Vendor	_			
		-46-		

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF

CAMDEN LEARNING CORPORATION

Camden Learning Corporation, a Delaware corporation (the "Corporation"), does hereby certify as follows:

- 1. The name of the Corporation is Camden Learning Corporation. The date of filing of its original Certificate of Incorporation with the Secretary of State was April 10, 2007 under the name of Camden Learning Corporation.
- 2. This Amended and Restated Certificate of Incorporation of Camden Learning Corporation, in the form attached hereto as Exhibit A, has been duly adopted in accordance with the provisions of Sections 228, 242 and 245 of the Delaware General Corporation Law by the directors and stockholders of the Corporation.
- 3. This Amended and Restated Certificate of Incorporation restates, integrates and amends the original Certificate of Incorporation of the Corporation.
- 4. This Amended and Restated Certificate of Incorporation shall be effective on the date of filing with the Secretary of State of the State of Delaware.
- 5. The text of the original Certificate of Incorporation of the Corporation is hereby amended and restated to read in its entirety as set forth on Exhibit A attached hereto and incorporated herein by reference.

IN WITNESS WHEREOF, the Corporation has	caused this	Amended and Restated	Certificate of	Incorporation	to be duly
executed on its behalf by an authorized officer on this	day of _	, 2007.			

CAMDEN LEARNING CORPORATION

By: /s/ David L. Warnock

Name: David L. Warnock

Title: President and Chief Executive Officer

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF CAMDEN LEARNING CORPORATION

FIRST: The name of the corporation is Camden Learning Corporation (the "Corporation").

SECOND: The address of the Corporation's registered office in the State of Delaware is National Registered Agents, Inc., 160 Greentree Drive, Suite 101, Dover, Delaware 19904, County of Kent. The name of the Corporation's registered agent at such address is National Registered Agents, Inc.

THIRD: The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the Delaware General Corporation Law, as amended from time to time (the "DGCL"). In addition to the powers and privileges conferred upon the Corporation by law and those incidental thereto, the Corporation shall possess and may exercise all the powers and privileges which are necessary or convenient to the conduct, promotion or attainment of the business or purposes of the Corporation; provided, however, that in the event a Business Combination (as defined below) is not consummated prior to the Termination Date (as defined below), then the purposes of the Corporation shall automatically, with no action required by the Board of Directors or the stockholders, on the Termination Date be limited to effecting and implementing the dissolution and liquidation of the Corporation and the taking of any other actions expressly required to be taken herein on or after the Termination Date and the Corporation's powers shall thereupon be limited to those set forth in Section 278 of the DGCL and as otherwise may be necessary to implement the limited purposes of the Corporation as provided herein. This Article Third may not be amended without the affirmative vote of at least 95% of the IPO Shares (as defined below) unless such amendment is in connection with, and becomes effective upon, the consummation of a Business Combination.

FOURTH: The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is 21,000,000, of which 20,000,000 shares shall be Common Stock of the par value of \$.0001 per share and 1,000,000 shares shall be Preferred Stock of the par value of \$.0001 per share.

A. Preferred Stock. The Board of Directors is expressly granted authority to issue shares of the Preferred Stock, in one or more series, and to fix for each such series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issue of such series (a "Preferred Stock Designation") and as may be permitted by the DGCL. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, without a separate vote of the holders of the Preferred Stock, or any series thereof, unless a vote of any such holders is required pursuant to any Preferred Stock Designation.

B. Common Stock. Except as otherwise required by law or as otherwise provided in any Preferred Stock Designation, the holders of the Common Stock shall exclusively possess all voting power and each share of Common Stock shall have one vote.

FIFTH: The Corporation's existence shall terminate on [], 2009 (the "Termination Date"). This provision may only be amended in connection with, and such amendment shall become effective upon, the consummation of a Business Combination. A proposal to so amend this section shall be submitted to stockholders in connection with any proposed Business Combination pursuant to Article Sixth below. This Article Fifth may not be amended without the affirmative vote of at least 95% of the IPO Shares unless such amendment is in connection with, and becomes effective upon, the consummation of a Business Combination.

SIXTH: The following provisions (A) through (F) shall apply during the period commencing upon the filing of this Amended and Restated Certificate of Incorporation and terminating upon the earlier to occur of: (i) the consummation of Business Combination or (ii) the Termination Date and may not be amended prior thereto without the affirmative vote of at least 95% of the IPO Shares unless such amendment is in connection with, and becomes effective upon, the consummation of a Business Combination. A "Business Combination" shall mean the merger, capital stock exchange, asset acquisition or other similar business combination between the Corporation and one or more operating businesses in the education industry having, collectively, a fair market value (as calculated in accordance with the requirements set forth below) of at least 80% of the amount in the Trust Account (less the deferred underwriting discount and commissions and taxes payable) at the time of such transaction.

For purposes of this Article Sixth, the fair market value of an acquisition proposed for a Business Combination shall be determined by the Board of Directors based upon financial standards generally accepted by the financial community, such as actual and potential sales, earnings and cash flow and book value. If the Board of Directors of the Corporation is not able to independently determine the fair market value of the target business, the Corporation shall obtain an opinion with regard to such fair market value from an unaffiliated, independent investment banking firm that is a member of the National Association of Securities Dealers, Inc. The Corporation will not consummate a Business Combination with any entity that is a portfolio company of, or otherwise affiliated with, or has received financial investment from, any of the private equity firms with which the Corporation's existing stockholders, executive officers or directors are affiliated.

A. Immediately after the Corporation's initial public offering (the "IPO"), the amount of the net offering proceeds received by the Corporation in the IPO (including the proceeds of any exercise of the underwriter's over-allotment option) specified in the Corporation's registration statement on Form S-1 filed with the Securities and Exchange Commission (the "Registration Statement") shall be deposited and thereafter held in a trust account established by the Corporation (the "Trust Account"). Neither the Corporation nor any officer, director or employee of the Corporation shall disburse any of the proceeds held in the Trust Account until the earlier of (i) a Business Combination or (ii) the Termination Date, in each case in accordance with the terms of the investment management trust agreement governing the Trust Account; provided, however, that (x) a portion of the interest earned on the Trust Account as described in the Registration Statement may be released to the Corporation to cover operating expenses, and (y) the Corporation shall be entitled to withdraw such amounts from the Trust Account as would be required to pay taxes on the interest earned on the Trust Account.

- B. Prior to the consummation of any Business Combination, the Corporation shall submit such Business Combination to its stockholders for approval regardless of whether the Business Combination is of a type which normally would require such stockholder approval under the DGCL. In the event a majority of the shares cast at the meeting to approve the Business Combination are voted for the approval of such Business Combination, the Corporation shall be authorized to consummate the Business Combination; provided, however, that the Corporation shall not consummate any Business Combination if holders of an aggregate of 30% or more in interest of the IPO Shares exercise their redemption rights described in paragraph C below.
- C. In the event that a Business Combination is approved in accordance with the above paragraph B and is consummated by the Corporation, any stockholder of the Corporation holding shares of Common Stock issued in the IPO (the "IPO Shares") who voted against the Business Combination may, contemporaneous with such vote, demand the Corporation redeem his IPO Shares for cash. If so demanded, the Corporation shall, promptly after consummation of the Business Combination, redeem such shares at a per share redemption price equal to the amount held in the Trust Account as of two business days prior to the consummation of the Business Combination (net of taxes payable), divided by the total number of IPO Shares, which shall in no event be less than \$7.92 per share.
- D. The holders of IPO Shares shall be entitled to receive distributions from the Trust Account only (i) in the event that the Corporation has not consummated a Business Combination by the Termination Date or (ii) in the event they demand redemption of their IPO Shares in accordance with subparagraph C and a Business Combination is approved in accordance with subparagraph B. The Corporation shall pay no liquidating distributions with respect to any shares of capital stock of the Corporation other than IPO Shares. In no other circumstances shall a holder of IPO Shares have any right or interest of any kind in or to the Trust Account. A holder of securities issued in the private placement concurrently with or prior to the consummation of the IPO shall not have any right or interest of any kind in or to the Trust Account.

E. Unless and until the Corporation has consummated a Business Combination as permitted under this Article Sixth, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, stock purchase, asset acquisition or otherwise.

F. The Board of Directors shall be divided into two classes: Class A and Class B. The number of directors in each class shall be as nearly equal as possible. Prior to the IPO, there shall be elected two Class A directors for a term expiring at the Corporation's first Annual Meeting of Stockholders and two Class B directors for a term expiring at the Corporation's second Annual Meeting of Stockholders. Commencing at the first Annual Meeting of Stockholders, and at each annual meeting thereafter, directors elected to succeed those directors whose terms expire shall be elected for a term of office to expire at the second succeeding annual meeting of stockholders after their election. Except as the DGCL may otherwise require, in the interim between annual meetings of stockholders or special meetings of stockholders called for the election of directors and/or the removal of one or more directors and the filling of any vacancy in that connection, newly created directorships and any vacancies in the Board of Directors, including unfilled vacancies resulting from the removal of directors for cause, may be filled by the vote of a majority of the remaining directors then in office, although less than a quorum (as defined in the Corporation's Bylaws), or by the sole remaining director. All directors shall hold office until the expiration of their respective terms of office and until their successors shall have been elected and qualified. A director elected to fill a vacancy resulting from the death, resignation or removal of a director shall serve for the remainder of the full term of the director whose death, resignation or removal shall have created such vacancy and until his successor shall have been elected and qualified.

SEVENTH: The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

- A. Election of directors need not be by ballot unless the by-laws of the Corporation so provide.
- B. The Board of Directors shall have the power, without the assent or vote of the stockholders, to make, alter, amend, change, add to or repeal the by-laws of the Corporation.
- C. The directors in their discretion may submit any contract or act for approval or ratification at any annual meeting of the stockholders or at any meeting of the stockholders called for the purpose of considering any such act or contract, and any contract or act that shall be approved or be ratified by the vote of the holders of a majority of the stock of the Corporation which is represented in person or by proxy at such meeting and entitled to vote thereat (provided that a lawful quorum of stockholders be there represented in person or by proxy) shall be as valid and binding upon the Corporation and upon all the stockholders as though it had been approved or ratified by every stockholder of the Corporation, whether or not the contract or act would otherwise be open to legal attack because of directors' interests, or for any other reason.
- D. In addition to the powers and authorities hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation; subject, nevertheless, to the provisions of the statutes of Delaware, of this Certificate of Incorporation, and to any by-laws from time to time made by the stockholders; provided, however, that no by-law so made shall invalidate any prior act of the directors which would have been valid if such by-law had not been made.

EIGHTH: A. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty by such director as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the DGCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the DGCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. Any repeal or modification of this paragraph A by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.

B. The Corporation, to the full extent permitted by Section 145 of the DGCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys' fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.

NINTH: Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation.

TENTH: The Corporation hereby elects not to be governed by Section 203 of the DGCL.

EXHIBIT 4.4

WARRANT AGREEMENT

This Warrant Agreement (this "Agreement") is made as of November [], 2007, by and between Camden Learning Corporation, a Delaware corporation having its principal place of business at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 ("Company"), and Continental Stock Transfer & Trust Company, a New York corporation with offices at 17 Battery Place, New York, New York 10004 (the "Warrant Agent").

WHEREAS, the Company is engaged in a public offering (the "Public Offering") of Units ("Units") and, in connection therewith, has determined to issue and deliver up to (i) 7,187,500 Warrants (the "Public Warrants") to the public investors, each of such Public Warrants evidencing the right of the holder thereof to purchase one share of common stock, par value \$.0001 per share, of the Company's Common Stock ("Common Stock") for \$5.50, subject to adjustment as described herein and (ii) 625,000 Warrants to Morgan Joseph & Co. Inc. ("Morgan Joseph") as representative of the underwriters (the "Underwriters") or its designees (the "Underwriter's Warrants"), with each of such Underwriter's Warrants evidencing the right of the holder thereof to purchase one share of Common Stock for \$6.71, subject to adjustment as described herein;

WHEREAS, immediately prior to the completion of the Public Offering, the Company shall sell and issue 2,800,000 Warrants in a private placement (the "Private Warrants") pursuant to that certain Amended and Restated Subscription Agreement dated November 20, 2007 (the "Subscription Agreement"), each of such Private Warrants evidencing the right of the holder thereof to purchase one share of Common Stock (the Public Warrants, the Underwriter's Warrants and the Private Warrants are collectively referred to herein as the "Warrants");

WHEREAS, the Company has filed with the Securities and Exchange Commission (the "SEC") a Registration Statement, No. 333-143098 on Form S-1, as amended ("Registration Statement") for the registration under the Securities Act of 1933, as amended ("Act") of, among other securities, the Public Warrants, the Underwriter's Warrants and the Common Stock issuable upon exercise of each of the Public Warrants and the Underwriter's Warrants; and

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing to so act, in connection with the issuance, registration, transfer, exchange, redemption and exercise of the Warrants; and

WHEREAS, the Company desires to provide for the form and provisions of the Warrants, the terms upon which they shall be issued and exercised, and the respective rights, limitation of rights, and immunities of the Company, the Warrant Agent, and the holders of the Warrants; and

WHEREAS, all acts and things have been done and performed which are necessary to make the Warrants, when executed on behalf of the Company and countersigned by or on behalf of the Warrant Agent, as provided herein, the valid, binding and legal obligations of the Company, and to authorize the execution and delivery of this Agreement.

NOW, THEREFORE, in consideration of the mutual agreements herein contained, the parties hereto agree as follows:

1. <u>Appointment of Warrant Agent</u>. The Company hereby appoints the Warrant Agent to act as agent for the Company for the Warrants, and the Warrant Agent hereby accepts such appointment and agrees to perform the same in accordance with the terms and conditions set forth in this Agreement.

2. Warrants.

- 2.1 Form of Warrant. Each Warrant shall be issued in registered form only. The Public Warrants and the Underwriter's Warrants shall be in substantially the form of Exhibit A hereto (with the exception that the Underwriter's Warrants shall have an exercise price of \$6.71 per share of Common Stock) and the Private Warrants shall be in substantially the form of Exhibit B hereto, the provisions of each of which are incorporated herein, and shall be signed by, or bear the facsimile signature of, the Chief Executive Officer or President and Chief Financial Officer, Treasurer, Secretary or Assistant Secretary of the Company and shall bear a facsimile of the Company's seal. In the event the person whose facsimile signature has been placed upon any Warrant shall have ceased to serve in the capacity in which such person signed the Warrant before such Warrant is issued, it may be issued with the same effect as if he or she had not ceased to be such at the date of issuance.
- 2.2 <u>Effect of Countersignature</u>. Unless and until countersigned by the Warrant Agent pursuant to this Agreement, a Warrant shall be invalid and of no effect and may not be exercised by the holder thereof.

2.3 Registration.

- 2.3.1 <u>Warrant Register</u>. The Warrant Agent shall maintain books ("Warrant Register") for the registration of original issuance and the registration of transfer of the Warrants. Upon the initial issuance of the Warrants, the Warrant Agent shall issue and register the Warrants in the names of the respective holders thereof in such denominations and otherwise in accordance with instructions delivered to the Warrant Agent by the Company.
- 2.3.2 Registered Holder. Prior to due presentment for registration of transfer of any Warrant, the Company and the Warrant Agent may deem and treat the person in whose name such Warrant shall be registered upon the Warrant Register ("registered holder"), as the absolute owner of such Warrant and of each Warrant represented thereby (notwithstanding any notation of ownership or other writing on the Warrant Certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.
- 2.4 <u>Detachability of Public Warrants</u>. The securities comprising the Units will begin to trade separately on the 90th trading day after the effective date of the Registration Statement unless Morgan Joseph informs the Company of its decision to allow earlier trading (the "Detachment Date"), provided that in no event will Morgan Joseph allow the separate trading of the securities comprising the Units until (i) the Company files with the SEC a Current Report on Form 8-K, which includes an audited balance sheet reflecting the receipt by the Company of the gross proceeds of the sale of the Private Warrants and the Public Offering, including the proceeds received by the Company from the exercise of the Underwriters' over-allotment option, if the over-allotment option is exercised on the date of the effective date of the Registration Statement, (ii) the Company issues a press release and files with the SEC a Current Report on Form 8-K announcing when such separate trading will begin, and (iii) the date on which separate trading begins is a business day following the earlier to occur of the exercise of the Underwriters' over-allotment option or its exercise in full (as described more fully in the Registration Statement).

3. Terms and Exercise of Warrants.

3.1 Warrant Price. Each Public Warrant, Private Warrant and Underwriter's Warrant shall, when countersigned by the Warrant Agent, entitle the registered holder thereof, subject to the provisions of such Warrant and this Warrant Agreement, to purchase from the Company the number of shares of Common Stock stated therein, at the price of \$5.50, \$5.50 and \$6.71, respectively, per whole share, subject to the adjustments provided in this Section 3.1 and Section 4 hereof. The term "Warrant Price" as used in this Warrant Agreement refers to the price per share at which Common Stock may be purchased at the time a Warrant is exercised. The Company in its sole discretion may lower the Warrant Price at any time prior to the Expiration Date for a period of not less than ten business days, provided that any such reduction shall be identical among all of the Warrants. The Private Warrants may be exercised on a "cashless" basis provided that at the time of exercise they are held by the original purchaser thereof, or their permitted assigns. In the event the Private Warrants are exercised on a "cashless" basis the holder thereof shall surrender his or her Private Warrant for that number of shares of Common Stock equal to the quotient obtained by dividing (x) the product of the number of shares of Common Stock underlying the Private Warrant, multiplied by the difference between the Warrant Price and the Fair Market Value (as defined below) by (y) the Fair Market Value. The "Fair Market Value" shall mean the average reported last sale price of the Common Stock for the 10 trading days ending on the third business day prior to the date on which notice of exercise is received by the Company.

3.2 Duration of Warrants.

3.2.1 Public Warrants and Underwriter's Warrants. A Public Warrant or Underwriter's Warrant may be exercised only during the period commencing on the later of: (i) the consummation by the Company of a merger, capital stock exchange, asset acquisition or other similar business combination (as described more fully in the Registration Statement, "Business Combination"), or (ii) [], 2008, and terminating at 5:00 p.m., New York City time on the earlier to occur of (x) [], 2011 or (y) the date fixed for redemption of the Warrants as provided in Section 6 of this Agreement. Notwithstanding the foregoing, no Public Warrant or Underwriter's Warrant shall be exercisable unless, at the time of exercise, a registration statement relating to the Common Stock issuable upon the exercise of such Public Warrant or Underwriter's Warrant is effective and current and a prospectus is available for use by the holders thereof and the Common Stock has been qualified or deemed to be exempt under the securities laws of the state of residence of the holder of such Public Warrants or Underwriter's Warrants.

3.2.2 Private Warrants. A Private Warrant may be exercised only during the period following consummation of a Business Combination by the Company and terminating at 5:00 p.m., New York City time on the earlier to occur of (x) [], 2011 or (y) the date fixed for redemption of the Warrants as provided in Section 6 of this Agreement. The Private Warrants are not subject to redemption so long as they are held by their initial purchasers or their permitted designees.

3.2.3 General. The period during which a Warrant may be exercised shall be deemed the "Exercise Period" and the termination of such Exercise Period shall be deemed the "Expiration Date". Except with respect to the right to receive the Redemption Price (as set forth in Section 6 hereunder), each Warrant not exercised on or before the Expiration Date shall become void, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at the close of business on the Expiration Date. The Company in its sole discretion may extend the duration of the Warrants by delaying the Expiration Date; provided, however, the Company will provide notice to registered holders of the Warrants of such extension of not less than 20 days and, further provided that any such extension shall be identical in duration among all of the Warrants.

3.3 Exercise of Warrants.

3.3.1 Payment. Subject to the provisions of the Warrants and this Warrant Agreement, a Warrant, when countersigned by the Warrant Agent, may be exercised by the registered holder thereof by surrendering it at the office of the Warrant Agent, or at the office of its successor as Warrant Agent, in the Borough of Manhattan, City and State of New York, with the subscription form, as set forth in the Warrant, duly executed by paying in full, in lawful money of the United States, in cash, good certified check or good bank draft payable to the order of the Company, the Warrant Price for each full share of Common Stock as to which the Warrant is exercised and any and all applicable taxes due in connection with the exercise of the Warrant, the exchange of the Warrant for the Common Stock and the issuance of the Common Stock.

3.3.2 <u>Issuance of Certificates</u>. As soon as practicable after the exercise of any Warrant and the clearance of the funds in payment of the Warrant Price, the Company shall issue to the registered holder of such Warrant a certificate or certificates for the number of full shares of Common Stock to which he, she or it is entitled, registered in such name or names as may be directed by him, her or it, and if such Warrant shall not have been exercised in full, a new countersigned Warrant for the number of shares as to which such Warrant shall not have been exercised. Notwithstanding the foregoing, the Company shall not be obligated to deliver any securities pursuant to the exercise of a Warrant unless (i) a registration statement under the Act with respect to the Common Stock issuable upon such exercise is effective, or (ii) in the opinion of counsel to the Company, the exercise of the Warrants is exempt from the registration requirements of the Act and such securities are qualified for sale or exempt from qualification under applicable securities laws of the states or other jurisdictions in which the registered holders reside. Warrants may not be exercised by, or securities issued to, any registered holder in any state in which such exercise or issuance would be unlawful. In no event will the Company be required to provide the registered holder of a warrant with a net-cash settlement or other consideration in lieu of physical settlement in shares of Common Stock, regardless of whether the Common Stock underlying the Warrants is registered pursuant to an effective registration statement.

- 3.3.3 <u>Valid Issuance</u>. All shares of Common Stock issued upon the proper exercise of a Warrant in conformity with this Agreement shall be validly issued, fully paid and nonassessable.
- 3.3.4 <u>Date of Issuance</u>. Each person in whose name any such certificate for shares of Common Stock is issued shall for all purposes be deemed to have become the holder of record of such shares on the date on which the Warrant was surrendered and payment of the Warrant Price was made, irrespective of the date of delivery of such certificate, except that, if the date of such surrender and payment is a date when the stock transfer books of the Company are closed, such person shall be deemed to have become the holder of such shares at the close of business on the next succeeding date on which the stock transfer books are open.

4. Adjustments.

- 4.1 Stock Dividends Split Ups. If after the date hereof, and subject to the provisions of Section 4.6 below, the number of outstanding shares of Common Stock is increased by a stock dividend payable in shares of Common Stock, or by a split up of shares of Common Stock, or other similar event, then, on the effective date of such stock dividend, split up or similar event, the number of shares of Common Stock issuable on exercise of each Warrant shall be increased in proportion to such increase in outstanding shares of Common Stock.
- 4.2 <u>Aggregation of Shares</u>. If after the date hereof, and subject to the provisions of Section 4.6, the number of outstanding shares of Common Stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Common Stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of shares of Common Stock issuable on exercise of each Warrant shall be decreased in proportion to such decrease in outstanding shares of Common Stock.
- 4.3 Adjustments in Exercise Price. Whenever the number of shares of Common Stock purchasable upon the exercise of the Warrants is adjusted, as provided in Section 4.1 and 4.2 above, the Warrant Price shall be adjusted (to the nearest cent) by multiplying such Warrant Price immediately prior to such adjustment by a fraction (x) the numerator of which shall be the number of shares of Common Stock purchasable upon the exercise of the Warrants immediately prior to such adjustment, and (y) the denominator of which shall be the number of shares of Common Stock so purchasable immediately thereafter.
- 4.4 Replacement of Securities upon Reorganization, etc. In case of any reclassification or reorganization of the outstanding shares of Common Stock (other than a change covered by Section 4.1 or 4.2 hereof or that solely affects the par value of such shares of Common Stock), or in the case of any merger or consolidation of the Company with or into another corporation (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding shares of Common Stock), or in the case of any sale or conveyance to another corporation or entity of the assets or other property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the Warrant holders shall thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the Warrants and in lieu of the shares of Common Stock of the Company immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the Warrant holder would have received if such Warrant holder had exercised his, her or its Warrant(s) immediately prior to such event; and if any reclassification also results in a change in shares of Common Stock covered by Section 4.1 or 4.2, then such adjustment shall be made pursuant to Sections 4.1, 4.2, 4.3 and this Section 4.4. The provisions of this Section 4.4 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers.

- 4.5 Notices of Changes in Warrant. Upon every adjustment of the Warrant Price or the number of shares issuable on exercise of a Warrant, the Company shall give written notice thereof to the Warrant Agent, which notice shall state the Warrant Price resulting from such adjustment and the increase or decrease, if any, in the number of shares purchasable at such price upon the exercise of a Warrant, setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based. Upon the occurrence of any event specified in Sections 4.1, 4.2, 4.3 or 4.4, then, in any such event, the Company shall give written notice to the Warrant holder, at the last address set forth for such holder in the Warrant Register, of the record date or the effective date of the event. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.
- 4.6 <u>No Fractional Shares</u>. Notwithstanding any provision contained in this Warrant Agreement to the contrary, the Company shall not issue fractional shares upon exercise of Warrants. If, by reason of any adjustment made pursuant to this Section 4, the holder of any Warrant would be entitled, upon the exercise of such Warrant, to receive a fractional interest in a share, the Company shall, upon such exercise, round up to the nearest whole number the number of the shares of Common Stock to be issued to the Warrant holder.
- 4.7 Form of Warrant. The form of Warrant need not be changed because of any adjustment pursuant to this Section 4, and Warrants issued after such adjustment may state the same Warrant Price and the same number of shares as is stated in the Warrants initially issued pursuant to this Agreement. However, the Company may at any time in its sole discretion make any change in the form of Warrant that the Company may deem appropriate and that does not affect the substance thereof, and any Warrant thereafter issued or countersigned, whether in exchange or substitution for an outstanding Warrant or otherwise, may be in the form as so changed.

5. Transfer and Exchange of Warrants.

5.1 <u>Transfer of Warrants</u>. Prior to the Detachment Date, the Public Warrants may be transferred or exchanged only together with the Unit in which such Warrant is included, and only for the purpose of effecting, or in conjunction with, a transfer or exchange of such Unit. Furthermore, each transfer of a Unit on the register relating to such Units shall operate also to transfer the Warrants included in such Unit. From and after the Detachment Date this Section 5.1 will have no further force and effect.

- 5.2 <u>Registration of Transfer</u>. The Warrant Agent shall register the transfer, from time to time, of any outstanding Warrant upon the Warrant Register, upon surrender of such Warrant for transfer, properly endorsed with signatures properly guaranteed and accompanied by appropriate instructions for transfer. Upon any such transfer, a new Warrant representing an equal aggregate number of Warrants shall be issued and the old Warrant shall be cancelled by the Warrant Agent. The Warrants so cancelled shall be delivered by the Warrant Agent to the Company from time to time upon request.
- 5.3 <u>Procedure for Surrender of Warrants</u>. Warrants may be surrendered to the Warrant Agent, together with a written request for exchange or transfer, and thereupon the Warrant Agent shall issue in exchange therefor one or more new Warrants as requested by the registered holder of the Warrants so surrendered, representing an equal aggregate number of Warrants; provided, however, in the event a Warrant surrendered for transfer bears a restrictive legend, the Warrant Agent shall not cancel such Warrant and issue new Warrants in exchange therefor until the Warrant Agent has received an opinion of counsel for the Company stating such transfer may be made and indicating whether the new Warrants must also bear a restrictive legend.
- 5.4 <u>Fractional Warrants</u>. The Warrant Agent shall not be required to effect any registration of transfer or exchange which will result in the issuance of a warrant certificate for a fraction of a warrant.
 - 5.5 Service Charges. No service charge shall be made for any exchange or registration of transfer of Warrants.
- 5.6 <u>Warrant Execution and Countersignature</u>. The Warrant Agent is hereby authorized to countersign and deliver, in accordance with the terms of this Agreement, the Warrants required to be issued pursuant to the provisions of this Section 5, and the Company, whenever required by the Warrant Agent, will supply the Warrant Agent with Warrants duly executed on behalf of the Company for such purpose.
- 5.7 <u>Private Warrants</u>. Notwithstanding anything herein to the contrary, the Warrant Agent shall not register for transfer any Private Warrants until the 90th day after the consummation of the Company's initial business combination, except for (a) transfers of Private Warrants resulting from the death of any of the holders thereof, (b) transfers by operation of law, (c) any transfer for estate planning purposes to persons immediately related to the transferor by blood, marriage or adoption, or (d) transfers to any trust solely for the benefit of such transferor and/or the persons described in the preceding clause, on condition that prior to such registration for transfer, the Warrant Agent shall be presented with written documentation pursuant to which each permitted transferee or the trustee or legal guardian for each permitted transferee agrees to be bound by the terms of the Subscription Agreement.

6. Redemption.

6.1 Redemption. Not less than all of the outstanding Public Warrants and the Underwriter's Warrants may be redeemed, at the option of the Company, at any time after they become exercisable and prior to their expiration, at the office of the Warrant Agent, upon the notice referred to in Section 6.3, at the price of \$.01 per Warrant ("Redemption Price"), provided that the last sales price of the Common Stock has been equal to or greater than \$11.50 per share, on each of twenty (20) trading days within any thirty (30) trading day period ending on the third business day prior to the date on which notice of redemption is given. Notwithstanding the foregoing, the Registration Statement must be current in order for the Company to exercise its redemption rights pursuant to this Section 6. The provisions of this Section 6.1 may not be modified, amended or deleted without the prior written consent of Morgan Joseph. The Private Warrants are not subject to this Section 6 provided they are held by the initial purchasers thereof, or their permitted designees.

- 6.2 <u>Date Fixed for, and Notice of, Redemption</u>. In the event the Company shall elect to redeem all of the Warrants, the Company shall fix a date for the redemption. Notice of redemption shall be mailed by first class mail, postage prepaid, by the Company not less than 30 days prior to the date fixed for redemption to the registered holders of the Warrants to be redeemed at their last addresses as they shall appear on the Warrant Register. Any notice mailed in the manner herein provided shall be conclusively presumed to have been duly given whether or not the registered holder received such notice.
- 6.3 Exercise After Notice of Redemption. The Warrants may be exercised in accordance with Section 3 of this Warrant Agreement at any time after notice of redemption shall have been given by the Company pursuant to Section 6.2 hereof and prior to the time and date fixed for redemption. On and after the redemption date, the record holder of the Warrants shall have no further rights except to receive, upon surrender of the Warrants, the Redemption Price.
- 6.4 <u>Outstanding Warrants Only</u>. The Company understands that the redemption rights provided for by this Section 6 apply only to outstanding Warrants. To the extent a person holds rights to purchase Warrants, such purchase rights shall not be extinguished by redemption. However, once such purchase rights are exercised, the Company may redeem the Warrants issued upon such exercise provided that the criteria for redemption are met, including the opportunity of the Warrant holder to exercise prior to redemption pursuant to Section 6.3. The provisions of this Section 6.4 may not be modified, amended or deleted without the prior written consent of Morgan Joseph.

7. Other Provisions Relating to Rights of Holders of Warrants.

- 7.1 No Rights as Stockholder. A Warrant does not entitle the registered holder thereof to any of the rights of a stockholder of the Company, including, without limitation, the right to receive dividends, or other distributions, exercise any preemptive rights to vote or to consent or to receive notice as stockholders in respect of the meetings of stockholders or the election of directors of the Company or any other matter.
- 7.2 Lost, Stolen, Mutilated, or Destroyed Warrants. If any Warrant is lost, stolen, mutilated, or destroyed, the Company and the Warrant Agent may on such terms as to indemnity or otherwise as they may in their discretion impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination, tenor, and date as the Warrant so lost, stolen, mutilated, or destroyed. Any such new Warrant shall constitute a substitute contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated, or destroyed Warrant shall be at any time enforceable by anyone.

- 7.3 <u>Reservation of Common Stock</u>. The Company shall at all times reserve and keep available a number of its authorized but unissued shares of Common Stock that will be sufficient to permit the exercise in full of all outstanding Warrants issued pursuant to this Warrant Agreement.
- 7.4 Registration of Common Stock. The Company agrees that prior to the commencement of the Exercise Period, it shall file with the SEC a post-effective amendment to the Registration Statement, or a new registration statement, for the registration, under the Act, of the Common Stock issuable upon exercise of the Warrants, and it shall take such action as is necessary to qualify for sale, in those states in which the Warrants were initially offered by the Company, the Common Stock issuable upon exercise of the Warrants. In either case, the Company will use its best efforts to cause the same to become effective on or prior to the commencement of the Exercise Period and to use its best efforts to maintain the effectiveness of such registration statement until the expiration of the Warrants in accordance with the provisions of this Warrant Agreement; provided, however, the Company shall not be obligated to deliver Common Stock and shall not have penalties for failure to deliver Common Stock if a registration statement is not effective at the time of exercise by the holder. In addition, the Company agrees to use its reasonable efforts to register such securities under the blue sky laws of the states of residence of the exercising warrant holders to the extent an exemption is not available. The provisions of this Section 7.4 may not be modified, amended or deleted without the prior written consent of Morgan Joseph. Notwithstanding the foregoing, a Warrant can expire unexercised regardless of whether a registration statement is current under the Act with respect to the Common Stock issuable upon exercise of the Warrants. In no event will the registered holder of a warrant be entitled to receive a net-cash settlement or shares of Common Stock or other consideration as of result of the Company's non-compliance with this Section 7.4.

8. Concerning the Warrant Agent and Other Matters.

8.1 <u>Payment of Taxes</u>. The Company will from time to time promptly pay all taxes and charges that may be imposed upon the Company or the Warrant Agent in respect of the issuance or delivery of shares of Common Stock upon the exercise of Warrants, but the Company shall not be obligated to pay any transfer taxes in respect of the Warrants or such shares.

8.2 Resignation, Consolidation, or Merger of Warrant Agent.

8.2.1 Appointment of Successor Warrant Agent. The Warrant Agent, or any successor to it hereafter appointed, may resign its duties and be discharged from all further duties and liabilities hereunder after giving sixty (60) days' notice in writing to the Company. If the office of the Warrant Agent becomes vacant by resignation or incapacity to act or otherwise, the Company shall appoint in writing a successor Warrant Agent in place of the Warrant Agent. If the Company shall fail to make such appointment within a period of 30 days after it has been notified in writing of such resignation or incapacity by the Warrant Agent or by the holder of the Warrant (who shall, with such notice, submit his Warrant for inspection by the Company), then the holder of any Warrant may apply to the Supreme Court of the State of New York for the County of New York for the appointment of a successor Warrant Agent. Any successor Warrant Agent, whether appointed by the Company or by such court, shall be a corporation organized and existing under the laws of the State of New York, in good standing and having its principal office in the Borough of Manhattan, City and State of New York, and authorized under such laws to exercise corporate trust powers and subject to supervision or examination by federal or state authority. After appointment, any successor Warrant Agent shall be vested with all the authority, powers, rights, immunities, duties, and obligations of its predecessor Warrant Agent with like effect as if originally named as Warrant Agent hereunder, without any further act or deed; but if for any reason it becomes necessary or appropriate, the predecessor Warrant Agent shall execute and deliver, at the expense of the Company, an instrument transferring to such successor Warrant Agent all the authority, powers, and rights of such predecessor Warrant Agent hereunder; and upon request of any successor Warrant Agent the Company shall make, execute, acknowledge, and deliver any and all instruments in writing for more fully and effectually vesting in and confirming to such successor Warrant Agent all such authority, powers, rights, immunities, duties, and obligations.

- 8.2.2 <u>Notice of Successor Warrant Agent</u>. In the event a successor Warrant Agent shall be appointed, the Company shall give notice thereof to the predecessor Warrant Agent and the transfer agent for the Common Stock not later than the effective date of any such appointment.
- 8.2.3 Merger or Consolidation of Warrant Agent. Any corporation into which the Warrant Agent may be merged or with which it may be consolidated or any corporation resulting from any merger or consolidation to which the Warrant Agent shall be a party shall be the successor Warrant Agent under this Warrant Agreement without any further act.

8.3 Fees and Expenses of Warrant Agent.

- 8.3.1 <u>Remuneration</u>. The Company agrees to pay the Warrant Agent reasonable remuneration for its services as such Warrant Agent hereunder as set forth on Exhibit C hereto, and will reimburse the Warrant Agent upon demand for all expenditures that the Warrant Agent may reasonably incur in the execution of its duties hereunder.
- 8.3.2 <u>Further Assurances</u>. The Company agrees to perform, execute, acknowledge, and deliver or cause to be performed, executed, acknowledged, and delivered all such further and other acts, instruments, and assurances as may reasonably be required by the Warrant Agent for the carrying out or performing of the provisions of this Warrant Agreement.

8.4 Liability of Warrant Agent.

8.4.1 <u>Reliance on Company Statement</u>. Whenever in the performance of its duties under this Warrant Agreement, the Warrant Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a statement signed by the Chief Executive Officer or Chief Operating Officer of the Company and delivered to the Warrant Agent. The Warrant Agent may rely upon such statement for any action taken or suffered in good faith by it pursuant to the provisions of this Warrant Agreement.

- 8.4.2 <u>Indemnity</u>. The Warrant Agent shall be liable hereunder only for its own negligence, willful misconduct or bad faith. The Company agrees to indemnify the Warrant Agent and save it harmless against any and all liabilities, including judgments, costs and reasonable counsel fees, for anything done or omitted by the Warrant Agent in the execution of this Warrant Agreement except as a result of the Warrant Agent's negligence, willful misconduct, or bad faith.
- 8.4.3 Exclusions. The Warrant Agent shall have no responsibility with respect to the validity of this Warrant Agreement or with respect to the validity or execution of any Warrant (except its countersignature thereof); nor shall it be responsible for any breach by the Company of any covenant or condition contained in this Warrant Agreement or in any Warrant; nor shall it be responsible to make any adjustments required under the provisions of Section 4 hereof or responsible for the manner, method, or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment; nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any shares of Common Stock to be issued pursuant to this Warrant Agreement or any Warrant or as to whether any shares of Common Stock will when issued be valid and fully paid and nonassessable.
- 8.5 <u>Acceptance of Agency</u>. The Warrant Agent hereby accepts the agency established by this Warrant Agreement and agrees to perform the same upon the terms and conditions herein set forth and among other things, shall account promptly to the Company with respect to Warrants exercised and concurrently account for, and pay to the Company, all moneys received by the Warrant Agent for the purchase of shares of the Company's Common Stock through the exercise of Warrants.
- 8.6 <u>Waiver</u>. The Warrant Agent hereby waives any and all right, title, interest or claim of any kind ("Claim") in or to any distribution of the Trust Account (as defined in that certain Investment Management Trust Agreement, dated as of the date hereof, by and between the Company and the Warrant Agent as trustee thereunder), and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim against the Trust Account for any reason whatsoever.

9. Miscellaneous Provisions.

- 9.1 <u>Successors</u>. All the covenants and provisions of this Warrant Agreement by or for the benefit of the Company or the Warrant Agent shall bind and inure to the benefit of their respective successors and assigns.
- 9.2 <u>Notices</u>. Any notice or other communication required or which may be given hereunder shall be in writing and either be delivered personally or by private national courier service, or be mailed, certified or registered mail, return receipt requested, postage prepaid, and shall be deemed given when so delivered personally or, if sent by private national courier service, on the next business day after delivery to the courier, or, if mailed, two business days after the date of mailing, as follows:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202 Attn: David L. Warnock, President

Any notice, statement or demand authorized by this Warrant Agreement to be given or made by the holder of any Warrant or by the Company to or on the Warrant Agent shall be sufficiently given when so delivered if by hand or overnight delivery or if sent by certified mail or private courier service five days after deposit of such notice, postage prepaid, addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004 Attn: []

with a copy in each case to:

McDermott Will & Emery LLP 340 Madison Avenue New York, New York 10173 Attn: Joel L. Rubinstein, Esq.

and

Ellenoff Grossman & Schole LLP 370 Lexington Avenue New York, New York 10017 Attn: Douglas Ellenoff, Esq.

and

Morgan Joseph & Co. Inc. 600 Fifth Avenue, 19th floor New York, New York 10020

Attn: Gordon Pollock, Managing Director

9.3 Applicable law. The validity, interpretation, and performance of this Warrant Agreement and of the Warrants shall be governed in all respects by the laws of the State of New York, without giving effect to conflict of laws. The Company hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Warrant Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum. Any such process or summons to be served upon the Company may be served by transmitting a copy thereof by registered or certified mail, return receipt requested, postage prepaid, addressed to it at the address set forth in Section 9.2 hereof. Such mailing shall be deemed personal service and shall be legal and binding upon the Company in any action, proceeding or claim.

- 9.4 <u>Persons Having Rights under this Warrant Agreement</u>. Nothing in this Warrant Agreement expressed and nothing that may be implied from any of the provisions hereof is intended, or shall be construed, to confer upon, or give to, any person or corporation other than the parties hereto and the registered holders of the Warrants and, for the purposes of Sections 6.1, 6.4, 7.4, 9.2 and 9.8 hereof, Morgan Joseph, any right, remedy, or claim under or by reason of this Warrant Agreement or of any covenant, condition, stipulation, promise, or agreement hereof. Morgan Joseph shall be deemed to be a third-party beneficiary of this Warrant Agreement with respect to Sections 6.1, 6.4, 7.4, 9.2 and 9.8 hereof. All covenants, conditions, stipulations, promises, and agreements contained in this Warrant Agreement shall be for the sole and exclusive benefit of the parties hereto (and Morgan Joseph with respect to the Sections 6.1, 6.4, 7.4, 9.2 and 9.8 hereof) and their successors and assigns and of the registered holders of the Warrants.
- 9.5 Examination of the Warrant Agreement. A copy of this Warrant Agreement shall be available at all reasonable times at the office of the Warrant Agent in the Borough of Manhattan, City and State of New York, for inspection by the registered holder of any Warrant. The Warrant Agent may require any such holder to submit his Warrant for inspection by it.
- 9.6 <u>Counterparts</u>. This Warrant Agreement may be executed in any number of counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.
- 9.7 Effect of Headings. The Section headings herein are for convenience only and are not part of this Warrant Agreement and shall not affect the interpretation thereof.
- 9.8 Amendments. This Warrant Agreement may be amended by the parties hereto without the consent of any registered holder for the purpose of curing any ambiguity, or of curing, correcting or supplementing any defective provision contained herein or adding or changing any other provisions with respect to matters or questions arising under this Warrant Agreement as the parties may deem necessary or desirable and that the parties deem shall not adversely affect the interest of the registered holders. All other modifications or amendments, including any amendment to increase the Warrant Price or shorten the Exercise Period, shall require the written consent of each of Morgan Joseph and the registered holders of a majority of the then outstanding Warrants. Notwithstanding the foregoing, the Company may lower the Warrant Price or extend the duration of the Exercise Period in accordance with Sections 3.1 and 3.2, respectively, without such consent.
- 9.9 <u>Severability</u>. This Warrant Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Warrant Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Warrant Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.

IN WITNESS WHEREOF, this Warrant Agreement has been duly executed by the parties hereto as of the day and year first above written.

Attest:	CAMDEN LEARNING CORPORATION
	By: David L. Warnock President
Attest:	CONTINENTAL STOCK TRANSFER & TRUST COMPANY
	Name: Title:
	15

EX-4.5 9 v095534_ex4-5.htm **EXHIBIT 4.5**

THE REGISTERED HOLDER OF THIS PURCHASE OPTION BY ITS ACCEPTANCE HEREOF, AGREES THAT IT WILL NOT SELL, TRANSFER OR ASSIGN THIS PURCHASE OPTION EXCEPT AS HEREIN PROVIDED AND THE REGISTERED HOLDER OF THIS PURCHASE OPTION AGREES THAT IT WILL NOT SELL, TRANSFER, ASSIGN, PLEDGE OR HYPOTHECATE THIS PURCHASE OPTION FOR A PERIOD OF ONE YEAR FOLLOWING THE EFFECTIVE DATE (DEFINED BELOW) TO ANYONE OTHER THAN (I) MORGAN JOSEPH & CO. INC ("MORGAN JOSEPH"), OR AN UNDERWRITER OR A SELECTED DEALER IN CONNECTION WITH THE OFFERING, OR (II) A BONA FIDE OFFICER OR PARTNER OF MORGAN JOSEPH OR OF ANY SUCH UNDERWRITER OR SELECTED DEALER.

THIS PURCHASE OPTION IS NOT EXERCISABLE PRIOR TO THE LATER OF THE CONSUMMATION BY CAMDEN LEARNING CORPORATION ("COMPANY") OF A SHARE CAPITAL EXCHANGE, ASSET ACQUISITION OR OTHER SIMILAR BUSINESS COMBINATION WITH ONE OR MORE OPERATING BUSINESSES ("BUSINESS COMBINATION") (AS DESCRIBED MORE FULLY IN THE COMPANY'S REGISTRATION STATEMENT (DEFINED HEREIN)) OR , 2008. VOID AFTER 5:00 P.M. NEW YORK CITY LOCAL TIME, , 2012.

UNIT PURCHASE OPTION
FOR THE PURCHASE OF
625,000 UNITS
OF
CAMDEN LEARNING CORPORATION

1. Purchase Option.

THIS CERTIFIES THAT, in consideration of \$100.00 duly paid by or on behalf of Morgan Joseph or its registered assigns (the "Holder"), as registered owner of this Purchase Option, to Camden Learning Corporation (the "Company"), the Holder is entitled. at any time or from time to time upon the later of the consummation of a Business Combination or , 2008 ("Commencement Date"), and at or before 5:00 p.m., New York City local time, 2012 ("Expiration Date"), but not thereafter, to subscribe for, purchase and receive, in whole or in part, up to 625,000 units ("Units") of the Company, each Unit consisting of one share of common stock of the Company, par value \$0.0001 per share ("Common Share(s)"), and one warrant ("Warrant(s)") expiring four years from the effective date ("Effective Date") of the registration statement ("Registration Statement") pursuant to which Units are offered for sale to the public ("Offering"). Except as otherwise specifically set forth herein, each Warrant is the same as the warrants included in the Units being registered for sale to the public by way of the Registration Statement ("Public Warrants"). If the Expiration Date is a day on which banking institutions are authorized by law to close, then this Purchase Option may be exercised on the next succeeding day which is not such a day in accordance with the terms herein. During the period ending on the Expiration Date, the Company agrees not to take any action that would terminate the Purchase Option. This Purchase Option is initially exercisable at \$9.60 per Unit so purchased; provided, however, that upon the occurrence of any of the events specified in Section 6 hereof, the rights granted by this Purchase Option, including the exercise price per Unit and the number of Units (and Common Shares and Warrants) to be received upon such exercise, shall be adjusted as therein specified. The term "Exercise Price" shall mean the initial exercise price or the adjusted exercise price of the Warrant, depending on the context.

Exercise.

- 2.1 <u>Exercise Form.</u> In order to exercise this Purchase Option, the exercise form attached hereto must be duly executed and completed and delivered to the Company, together with this Purchase Option and payment of the Exercise Price for the Units being purchased payable in cash or by certified check or official bank check or wire transfer of immediately available funds to an account designated by the Company. If the subscription rights represented hereby shall not be exercised at or before 5:00 p.m., New York City local time, on the Expiration Date this Purchase Option shall become and be void without further force or effect, and all rights represented hereby shall cease and expire.
- 2.2 <u>Legend</u>. Each certificate for the securities purchased under this Purchase Option shall bear a legend as follows unless such securities have been registered under the Securities Act of 1933, as amended ("Act"):

"The securities represented by this certificate have not been registered under the Securities Act of 1933, as amended ("Act") or applicable state law. Neither the securities nor any interest therein may be offered for sale, sold or otherwise transferred except pursuant to an effective registration statement under the Act, or pursuant to an exemption from registration under the Act and applicable state law which, in the opinion of counsel to the Company, is available."

2.3 <u>Cashless Exercise</u>.

Determination of Amount. In lieu of the payment of the Exercise Price multiplied by the number of Units for which this Purchase Option is exercisable (and in lieu of being entitled to receive Common Shares and Warrants) in the manner required by Section 2.1, the Holder shall have the right (but not the obligation) to convert any exercisable but unexercised portion of this Purchase Option into Units (the "Conversion Right") as follows: upon exercise of the Conversion Right, the Company shall deliver to the Holder (without payment by the Holder of any of the Exercise Price in cash) that number of Common Shares and Warrants comprising that number of Units equal to the quotient obtained by dividing (x) the "Value" (as defined below) of the portion of the Purchase Option being converted by (y) the Current Market Value (as defined below) of the portion of the Purchase Option being converted. The "Value" of the portion of the Purchase Option being converted shall equal the remainder derived from subtracting (a) (i) the Exercise Price multiplied by (ii) the number of Units underlying the portion of this Purchase Option being converted from (b) the Current Market Value of a Unit multiplied by the number of Units underlying the portion of the Purchase Option being converted. As used herein, the term "Current Market Value" per Unit at any date means: (A) in the event that neither the Units nor Warrants are still trading, the remainder derived from subtracting (x) the exercise price of the Warrants multiplied by the number of Common Shares issuable upon exercise of the Warrants underlying one Unit from (y) (i) the Current Market Price of the Common Shares multiplied by (ii) the number of Common Shares underlying one Unit, which shall include the Common Shares underlying the Warrants included in such Unit; (B) in the event that the Units, Common Shares and Warrants are still trading, (i) if the Units are listed on a national securities exchange or quoted on the Nasdaq Global Market, Nasdaq Capital Market or OTC Bulletin Board (or successor such as the Bulletin Board Exchange), the average of the last sale price of the Units in the principal trading market for the Units as reported by the exchange, Nasdag or FINRA, as the case may be, for the ten (10) Trading Days ending on the third business day prior to exercise; or (ii) if the Units are not listed on a national securities exchange or quoted on the Nasdag Global Market, Nasdag Capital Market or the OTC Bulletin Board (or successor exchange), but is traded in the residual over-the-counter market, the average of the closing bid price for Units for the ten (10) Trading Days ending on the third business day prior to exercise for which such quotations are reported by the Pink Sheets, LLC or similar publisher of such quotations; and (C) in the event that the Units are not still trading but the Common Shares and Warrants underlying the Units are still trading, the Current Market Price of the Common Shares plus the product of (x) the Current Market Price of the Warrants and (y) the number of Common Shares underlying the Warrants included in one Unit. The "Current Market Price" shall mean (i) if the Common Shares (or Warrants, as the case may be) are listed on a national securities exchange or quoted on the Nasdaq Global Market, Nasdaq Capital Market or OTC Bulletin Board (or successor such as the Bulletin Board Exchange), the last sale price of the Common Shares (or Warrants) in the principal trading market for the Common Shares as reported by the exchange, Nasdaq or FINRA, as the case may be, on the last trading day preceding the date in question; (ii) if the Common Shares (or Warrants, as the case may be) are not listed on a national securities exchange or quoted on the Nasdag Global Market, Nasdag Capital Market or the OTC Bulletin Board (or successor exchange), but are traded in the residual over-the-counter market, the closing bid price for the Common Shares (or Warrants) on the last trading day preceding the date in question for which such quotations are reported by the Pink Sheets, LLC or similar publisher of such quotations; and (iii) if the fair market value of the Common Shares cannot be determined pursuant to clause (i) or (ii) above, such price as the Board of Directors of the Company shall determine, in good faith.

2.3.2 <u>Mechanics of Cashless Exercise</u> . The Conversion Right may be exercised by the Holder on any business day on or after the Commencement Date and not later than the Expiration Date by delivering the Purchase Option with the duly executed exercise form attached hereto with the cashless exercise section completed to the Company, exercising the Conversion Right and specifying the total number of Units the Holder will purchase pursuant to such Cashless Exercise Right.		
2.3.3 <u>Warrant Exercise</u> . Any warrants underlying the Units shall be issued pursuant to and subject to the terms and conditions set forth in the Warrant Agreement, dated as of, 2007, between the Company and Continental Stock Transfer & Trust Company, acting as Warrant Agent (the "Warrant Agreement"); provided, that the exercise price of the Warrants shall be as set forth herein.		
2.4 Effective Registration Statement. The Warrants underlying this Purchase Option are exercisable only during those periods of time in which the Company maintains the effectiveness of the Registration Statement. If the Company fails to maintain the effectiveness of the Registration Statement, the Warrants underlying this Purchase Option may expire worthless.		
- 3 -		

Transfer.

- 3.1 General Restrictions. The Holder of this Purchase Option, by its acceptance hereof, agrees that it will not sell, transfer, assign, pledge or hypothecate this Purchase Option or its underlying shares for a period of one year following the Effective Date to anyone other than (i) Morgan Joseph or an underwriter or a selected dealer in connection with the Offering, or (ii) a bona fide officer or partner of Morgan Joseph or of any such underwriter or selected dealer in accordance with the Financial Industry Regulatory Authority ("FINRA") Conduct Rule 2710(g)(1) or cause this Purchase Option or the Securities issuable hereunder to be the subject of any hedging, short sale, derivative, put or call transaction that would result in the effective economic disposition of this Purchase Option or the Securities hereunder, except as provided for in FINRA Rule 2710(g)(2). On and after the first anniversary of the Effective Date, transfers to others may be made subject to compliance with or exemptions from applicable securities laws. In order to make any permitted assignment, the Holder must deliver to the Company the assignment form attached hereto duly executed and completed, together with the Purchase Option and payment of all transfer taxes, if any, payable in connection therewith. The Company shall within five business days transfer this Purchase Option on the books of the Company and shall execute and deliver a new Purchase Option or Purchase Options of like tenor to the appropriate assignee(s) expressly evidencing the right to purchase the aggregate number of Units purchasable hereunder or such portion of such number as shall be contemplated by any such assignment.
- 3.2 Restrictions Imposed by the Act. The securities evidenced by this Purchase Option shall not be transferred unless and until (i) the Company has received the opinion of counsel for the Holder that the securities may be transferred pursuant to an exemption from registration under the Act and applicable state securities laws, the availability of which is established to the reasonable satisfaction of the Company (the Company hereby agreeing that the opinion of McDermott Will & Emery LLP shall be deemed satisfactory evidence of the availability of an exemption), or (ii) a registration statement or a post-effective amendment to the Registration Statement relating to such securities has been filed by the Company and declared effective by the Securities and Exchange Commission (the "Commission") and compliance with applicable state securities law has been established.

4. New Purchase Options to be Issued.

- 4.1 <u>Partial Exercise or Transfer</u>. Subject to the restrictions in Section 3 hereof, this Purchase Option may be exercised or assigned in whole or in part. In the event of the exercise or assignment hereof in part only, upon surrender of this Purchase Option for cancellation, together with the duly executed exercise or assignment form and except in the case of an exercise of this Purchase Option contemplated by Section 2.3 hereof, funds sufficient to pay any Exercise Price and/or transfer tax, the Company shall cause to be delivered to the Holder without charge a new Purchase Option of like tenor to this Purchase Option in the name of the Holder evidencing the right of the Holder to purchase the number of Units purchasable hereunder as to which this Purchase Option has not been exercised or assigned.
- 4.2 <u>Lost Certificate</u>. Upon receipt by the Company of evidence satisfactory to it of the loss, theft, destruction or mutilation of this Purchase Option and of reasonably satisfactory indemnification or the posting of a bond, the Company shall execute and deliver a new Purchase Option of like tenor and date. Any such new Purchase Option executed and delivered as a result of such loss, theft, mutilation or destruction shall constitute a substitute contractual obligation on the part of the Company.

5. Registration Rights.

5.1 <u>Demand Registration</u>.

- Grant of Right. The Company, upon written demand ("Initial Demand Notice") of the Holder(s) of a majority in interest of the Purchase Options and/or the underlying Units and/or the underlying securities ("Majority Holders"), agrees to register (the "Demand Registration") under the Act, all or any portion of the Purchase Options requested by the Majority Holders in the Initial Demand Notice and all of the securities underlying such Purchase Options, including the Units, Common Shares, the Warrants and the Common Shares underlying the Warrants (collectively, the "Registrable Securities"). On such occasion, the Company will file a registration statement or a post-effective amendment to the Registration Statement covering the Registrable Securities within sixty days after receipt of the Initial Demand Notice and use its best efforts to have such registration statement or post-effective amendment declared effective as soon as possible thereafter. The demand for registration may be made at any time during a period of five years beginning on the Effective Date. The Initial Demand Notice shall specify the number of shares of Registrable Securities proposed to be sold and the intended method(s) of distribution thereof. The Company will notify all Holders of the Purchase Options and/or Registrable Securities of the demand within ten (10) days from the date of the receipt of any such Initial Demand Notice. Each holder of Registrable Securities who wishes to include all or a portion of such holder's Registrable Securities in the Demand Registration (each such holder including shares of Registrable Securities in such registration, a "Demanding Holder") shall so notify the Company within fifteen (15) days after the receipt by the holder of the notice from the Company. Upon any such request, the Demanding Holders shall be entitled to have their Registrable Securities included in the Demand Registration, subject to Section 5.1.4. The Company shall not be obligated to effect more than one Demand Registration under this Section 5.1 in respect of the Registrable Securities.
- 5.1.2 <u>Effective Registration</u>. A registration will not count as a Demand Registration until the registration statement filed with the Commission with respect to such Demand Registration has been declared effective and the Company has complied with all of its obligations under this Agreement with respect thereto; provided, however, that if, after such registration statement has been declared effective, the offering of Registrable Securities pursuant to a Demand Registration is interfered with by any stop order or injunction of the Commission or any other governmental agency or court, the registration statement with respect to such Demand Registration will be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders thereafter elect to continue the offering.
- 5.1.3 <u>Underwritten Offering</u>. If the Majority Holders so elect and such holders so advise the Company as part of the Initial Demand Notice, the offering of such Registrable Securities pursuant to such Demand Registration shall be in the form of an underwritten offering. In such event, the right of any holder to include its Registrable Securities in such registration shall be conditioned upon such holder's participation in such underwriting and the inclusion of such holder's Registrable Securities in the underwriting to the extent provided herein. All Demanding Holders proposing to distribute their securities through such underwriting shall enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by the Majority Holders.

- Reduction of Offering. If the managing underwriter or underwriters for a Demand Registration that is to be an underwritten offering advises the Company and the Demanding Holders in writing that the dollar amount or number of shares of Registrable Securities which the Demanding Holders desire to sell, taken together with all other Common Shares or other securities which the Company desires to sell and the Common Shares, if any, as to which registration has been requested pursuant to written contractual piggy-back registration rights held by other shareholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of shares that can be sold in such offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of shares, as applicable, the "Maximum Number of Shares"), then the Company shall include in such registration: (i) first, the Registrable Securities as to which Demand Registration has been requested by the Demanding Holders (pro rata in accordance with the number of shares that each such Person has requested be included in such registration, regardless of the number of shares held by each such Person (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Shares; (ii) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (i), the Common Shares or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (iii) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (i) and (ii), the Common Shares or other securities registrable pursuant to the terms of the Registration Rights Agreement between the Company and the initial investors in the Company, dated as of, 2007 (the "Registration Rights Agreement" and such registrable securities, the "Investor Securities") as to which "piggy-back" registration has been requested by the holders thereof, Pro Rata, that can be sold without exceeding the Maximum Number of Shares; and (iv) fourth, to the extent that the Maximum Number of Shares have not been reached under the foregoing clauses (i), (ii), and (iii), the Common Shares or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Shares.
- 5.1.5 <u>Withdrawal</u>. If a majority in interest of the Demanding Holders disapprove of the terms of any underwriting or are not entitled to include all of their Registrable Securities in any offering, such majority-in-interest of the Demanding Holders may elect to withdraw from such offering by giving written notice to the Company and the underwriter or underwriters of their request to withdraw prior to the effectiveness of the registration statement filed with the Commission with respect to such Demand Registration. In such event, the Company need not seek effectiveness of such Registration Statement for the benefit of the initial investors in the Company. If the majority-in-interest of the Demanding Holders withdraws from a proposed offering relating to a Demand Registration, then such registration shall not count as a Demand Registration provided for in Section 5.1.

- 5.1.6 <u>State Securities Laws Compliance</u>. The Company agrees to use its reasonable best efforts to qualify or register the Registrable Securities in such states as are reasonably requested by the Majority Holders; provided, however, that in no event shall the Company be required to register the Registrable Securities in a state in which such registration would cause (i) the Company to be obligated to qualify to do business in such state, or would subject the Company to taxation as a foreign corporation doing business in such jurisdiction or (ii) the principal shareholders of the Company to be obligated to escrow their shares of share capital of the Company.
- 5.1.7 <u>Period of Effectiveness.</u> The Company shall cause any registration statement filed pursuant to the Demand Registration to remain effective until the first to occur of (i) sale or transfer of all the Registrable Securities included in such registration statement and (ii) nine months from the effective date of such registration statement, which period shall be extended by the number of days in such period that the Company has advised the Demanding Holders they cannot sell their Registrable Securities under the registration statement.

5.2 Piggy-Back Registration.

- 5.2.1 Piggy-Back Rights. If at any time during the seven year period commencing on the Effective Date the Company proposes to file a registration statement under the Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, by the Company for its own account or for shareholders of the Company for their account (or by the Company and by shareholders of the Company including, without limitation, pursuant to Section 5.1), other than a registration statement (i) filed in connection with any employee share option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing shareholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall (x) give written notice of such proposed filing to the holders of Registrable Securities as soon as practicable but in no event less than ten (10) days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing underwriter or underwriters, if any, of the offering, and (y) offer to the holders of Registrable Securities in such notice the opportunity to register the sale of such number of shares of Registrable Securities as such holders may request in writing within fifteen (15) days following receipt of such notice (a "Piggy-Back Registration"). The Company shall cause such Registrable Securities to be included in such registration and shall use its best efforts to cause the managing underwriter or underwriters of a proposed underwritten offering to permit the Registrable Securities requested to be included in a Piggy-Back Registration on the same terms and conditions as any similar securities of the Company and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All holders of Registrable Securities proposing to distribute their securities through a Piggy-Back Registration that involves an underwriter or underwriters shall enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such Piggy-Back Registration.
- 5.2.2 Reduction of Offering. If the managing underwriter or underwriters for a Piggy-Back Registration that is to be an underwritten offering advises the Company and the holders of Registrable Securities in writing that the dollar amount or number of Common Shares which the Company desires to sell, taken together with Common Shares, if any, as to which registration has been demanded pursuant to written contractual arrangements with persons other than the holders of Registrable Securities hereunder, the Registrable Securities as to which registration has been requested under this Section 5.2, and the Common Shares, if any, as to which registration has been requested pursuant to the written contractual piggy-back registration rights of other shareholders of the Company, exceeds the Maximum Number of Shares, then the Company shall include in any such registration:

- (a) If the registration is undertaken for the Company's account: (A) first, the Common Shares or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the Common Shares or other securities, if any, comprised of Registrable Securities and Investor Securities, as to which registration has been requested pursuant to the applicable written contractual piggy-back registration rights of such security holders, Pro Rata, that can be sold without exceeding the Maximum Number of Shares; and (C) third, to the extent that the Maximum Number of shares has not been reached under the foregoing clauses (A) and (B), the Common Shares or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual piggy-back registration rights with such persons and that can be sold without exceeding the Maximum Number of Shares;
- (b) If the registration is a "demand" registration undertaken at the demand of holders of Investor Securities, (A) first, the Common Shares or other securities for the account of the demanding persons, Pro Rata, that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the Common Shares or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (C) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A) and (B), the shares of Registrable Securities, Pro Rata, as to which registration has been requested pursuant to the terms hereof, that can be sold without exceeding the Maximum Number of Shares; and (D) fourth, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A), (B) and (C), the Common Shares or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual arrangements with such persons, that can be sold without exceeding the Maximum Number of Shares; and
- (c) If the registration is a "demand" registration undertaken at the demand of persons other than either the holders of Registrable Securities or of Investor Securities, (A) first, the Common Shares or other securities for the account of the demanding persons that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the Common Shares or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (C) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A) and (B), collectively the Common Shares or other securities comprised of Registrable Securities and Investor Securities, Pro Rata, as to which registration has been requested pursuant to the terms hereof and of the Registration Rights Agreement, as applicable, that can be sold without exceeding the Maximum Number of Shares; and (D) fourth, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A), (B) and (C), the Common Shares or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual arrangements with such persons, that can be sold without exceeding the Maximum Number of Shares.

- 5.2.3 <u>Withdrawal</u>. Any holder of Registrable Securities may elect to withdraw such holder's request for inclusion of Registrable Securities in any Piggy-Back Registration by giving written notice to the Company of such request to withdraw prior to the effectiveness of the Registration Statement. The Company (whether on its own determination or as the result of a withdrawal by persons making a demand pursuant to written contractual obligations) may withdraw a registration statement at any time prior to the effectiveness of the registration statement. Notwithstanding any such withdrawal, the Company shall pay all expenses incurred by the holders of Registrable Securities in connection with such Piggy-Back Registration as provided in Section 5.4.1.
- 5.3 No Net-Cash Settlement or Damages Upon Failure of Registration. In no event shall the registered holder of this Purchase Option and the Warrants underlying the Purchase Option be entitled to (i) net-cash settlement of this Purchase Option or the Warrants underlying the Purchase Option, regardless of whether any or all of the Registrable Securities have been registered by the Company pursuant to an effective registration statement, or (ii) receive any damages if any or all of the Registrable Securities have not been registered by the Company pursuant to an effective registration statement, subject to the requirement that the Company use its best efforts to have a registration statement or post-effective amendment filed pursuant to this Section declared effective as soon as possible after receiving the Initial Demand Notice. In the event there is no effective registration statement related to the issuance or exercise of the Warrants contained within the Units, that portion of the Units may not be exercised by the holder and therefore may expire and be worthless.

5.4 General Terms.

5.4.1 Registration Expenses. The Company shall bear all fees and expenses incurred in connection with any Demand Registration or any Piggy-Back Registration, and all expenses incurred in performing or complying with its other obligations under this Agreement, whether or not the Registration Statement becomes effective or whether any or all Demanding Holders of Registrable Securities withdraw from any Registration Statement, including, without limitation: (i) all registration and filing fees; (ii) fees and expenses of compliance with securities or "blue sky" laws (including fees and disbursements of counsel in connection with blue sky qualifications of the Registrable Securities); (iii) printing expenses; (iv) the Company's internal expenses (including, without limitation, all salaries and expenses of its officers and employees); (v) the fees and expenses incurred in connection with the listing of the Registrable Securities; (vi) FINRA fees; (vii) fees and disbursements of counsel for the Company and fees and expenses for independent certified public accountants retained by the Company (including the expenses or costs associated with the delivery of any opinions or comfort letters); (viii) the fees and expenses of any special experts retained by the Company in connection with such registration and (ix) the fees and expenses of one legal counsel selected by the holders of a majority-in-interest of the Registrable Securities included in such registration. The Company shall have no obligation to pay any underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions shall be borne by such holders. Additionally, in an underwritten offering, all selling shareholders and the Company shall bear the expenses of the underwriter pro rata in proportion to the respective amount of shares each is selling in such offering.

- 5.4.2 Indemnification. The Company shall indemnify the Holder(s) of the Registrable Securities to be sold pursuant to any registration statement hereunder and each person, if any, who controls such Holder(s) within the meaning of Section 15 of the Act or Section 20(a) of the Securities Exchange Act of 1934, as amended ("Exchange Act"), against all loss, claim, damage, expense or liability (including all reasonable attorneys' fees and other expenses reasonably incurred in investigating, preparing or defending against litigation, commenced or threatened, or any claim whatsoever whether arising out of any action between the underwriter and the Company or between the underwriter and any third party or otherwise) to which any of them may become subject under the Act, the Exchange Act or otherwise, arising from such registration statement but only to the same extent and with the same effect as the provisions pursuant to which the Company has agreed to indemnify the underwriters contained in Section 5 of the Underwriting Agreement between the Company, Morgan Joseph and the other underwriters named therein dated the Effective Date. The Holder(s) of the Registrable Securities to be sold pursuant to such registration statement, and their successors and assigns, shall severally, and not jointly, indemnify the Company, its officers and directors and each person, if any, who controls the Company within the meaning of Section 15 of the Act or Section 20(a) of the Exchange Act, against all loss, claim, damage, expense or liability (including all reasonable attorneys' fees and other expenses reasonably incurred in investigating, preparing or defending against any claim whatsoever) to which they may become subject under the Act, the Exchange Act or otherwise, arising from information furnished by or on behalf of such Holders, or their successors or assigns, in writing, for specific inclusion in such registration statement to the same extent and with the same effect as the provisions contained in Section 5 of the Underwriting Agreement pursuant to which the underwriters have agreed to indemnify the Company.
- 5.4.3 <u>Exercise of Purchase Options</u>. Nothing contained in this Purchase Option shall be construed as requiring the Holder(s) to exercise their Purchase Options or Warrants underlying such Purchase Options prior to or after the initial filing of any registration statement or the effectiveness thereof.
- 5.4.4 <u>Documents Delivered to Holders</u>. The Company shall furnish Morgan Joseph, as representative of the Holders participating in any of the foregoing offerings, a signed counterpart, addressed to the participating Holders, of (i) an opinion of counsel to the Company, dated the effective date of such registration statement (and, if such registration includes an underwritten public offering, an opinion dated the date of the closing under any underwriting agreement related thereto), and (ii) a "cold comfort" letter dated the effective date of such registration statement (and, if such registration includes an underwritten public offering, a letter dated the date of the closing under the underwriting agreement) signed by the independent public accountants who have issued a report on the Company's financial statements included in such registration statement, in each case covering substantially the same matters with respect to such registration statement (and the prospectus included therein) and, in the case of such accountants' letter, with respect to events subsequent to the date of such financial statements, as are customarily covered in opinions of issuer's counsel and in accountants' letters delivered to underwriters in underwritten public offerings of securities. The Company shall also deliver promptly to Morgan Joseph, as representative of the Holders participating in the offering, the correspondence and memoranda described below and copies of all correspondence between the Commission and the Company, its counsel or auditors and all memoranda relating to discussions with the Commission or its staff with respect to the registration statement and permit Morgan Joseph, as representative of the Holders, to do such investigation, upon reasonable advance notice, with respect to information contained in or omitted from the registration statement as it deems reasonably necessary to comply with applicable securities laws or rules of FINRA. Such investigation shall include access to books, records and properties and opportunities to discuss the business of the Company with its officers and independent auditors, all to such reasonable extent and at such reasonable times and as often as Morgan Joseph, as representative of the Holders, shall reasonably request. The Company shall not be required to disclose any confidential information or other records to Morgan Joseph, as representative of the Holders, or to any other person, until and unless such persons shall have entered into reasonable confidentiality agreements (in form and substance reasonably satisfactory to the Company), with the Company with respect thereto.

- 5.4.5 <u>Documents to be Delivered by Holder(s)</u>. Each Holder participating in any of the foregoing offerings shall furnish to the Company a completed and executed questionnaire provided by the Company requesting information customarily sought of selling securityholders.
- Underwriting Agreement. The Company shall enter into an underwriting agreement with the managing 5.4.6 underwriter(s), if any, selected by any Holders whose Registrable Securities are being registered pursuant to this Section 5, which managing underwriter shall be reasonably acceptable to the Company. Such agreement shall be reasonably satisfactory in form and substance to the Company and its legal counsel, each Holder and such managing underwriters, and shall contain such representations, warranties and covenants by the Company and such other terms as are customarily contained in agreements of that type used by the managing underwriter. The Holders shall be parties to any underwriting agreement relating to an underwritten sale of their Registrable Securities and may, at their option, require that any or all the representations, warranties and covenants of the Company to or for the benefit of such underwriters shall also be made to and for the benefit of such Holders. Such Holders shall not be required to make any representations or warranties to or agreements with the Company or the underwriters except as they may relate to such Holders and their intended methods of distribution. Such Holders, however, shall agree to such covenants and indemnification and contribution obligations for selling shareholders as are customarily contained in agreements of that type used by the managing underwriter. Further, such Holders shall execute appropriate custody agreements and otherwise cooperate fully in the preparation of the registration statement and other documents relating to any offering in which they include securities pursuant to this Section 5. Each Holder shall also furnish to the Company such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as shall be reasonably required to effect the registration of the Registrable Securities.

- 5.4.7 <u>Rule 144 Sale.</u> Notwithstanding anything contained in this Section 5 to the contrary, the Company shall have no obligation pursuant to Sections 5.1 or 5.2 for the registration of Registrable Securities held by any Holder (i) where such Holder would then be entitled to sell under Rule 144 within any three-month period (or such other period prescribed under Rule 144 as may be provided by amendment thereof) all of the Registrable Securities then held by such Holder, and (ii) where the number of Registrable Securities held by such Holder is within the volume limitations under paragraph (e) of Rule 144 (calculated as if such Holder were an affiliate within the meaning of Rule 144).
- 5.4.8 <u>Supplemental Prospectus</u>. Each Holder agrees, that upon receipt of any notice from the Company of the happening of any event as a result of which the prospectus included in the registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances then existing, such Holder will immediately discontinue disposition of Registrable Securities pursuant to the registration statement covering such Registrable Securities until such Holder's receipt of the copies of a supplemental or amended prospectus, and, if so desired by the Company, such Holder shall deliver to the Company (at the expense of the Company) or destroy (and deliver to the Company a certificate of such destruction) all copies, other than permanent file copies then in such Holder's possession, of the prospectus covering such Registrable Securities current at the time of receipt of such notice.

6. Adjustments.

- 6.1 <u>Adjustments to Exercise Price and Number of Securities</u>. The Exercise Price and the number of Units underlying the Purchase Option shall be subject to adjustment from time to time as hereinafter set forth:
- 6.1.1 Share Dividends Split-Ups. If after the date hereof, and subject to the provisions of Section 6.3 below, the number of outstanding Common Shares is increased by a share dividend payable in Common Shares or by a split-up of Common Shares or other similar event, then, on the effective date thereof, the number of Common Shares underlying each of the Units purchasable hereunder shall be increased in proportion to such increase in outstanding shares. In such case, the number of Common Shares, and the exercise price applicable thereto, underlying the Warrants underlying each of the Units purchasable hereunder shall be adjusted in accordance with the terms of the Warrants. For example, if the Company declares a two-for-one share dividend and at the time of such dividend this Purchase Option is for the purchase of one Unit at \$9.60 per whole Unit (each Warrant underlying the Units is exercisable for \$6.71 per share), upon effectiveness of the dividend, this Purchase Option will be adjusted to allow for the purchase of one Unit at \$9.60 per Unit, each Unit entitling the holder to receive two Common Shares and two Warrants (each Warrant exercisable for \$3.36 per share).
- 6.1.2 <u>Extraordinary Dividend</u>. If the Company, at any time while this Purchase Option is outstanding and unexpired, shall pay a dividend or make a distribution in cash, securities or other assets to the holders of Common Stock (or other shares of the Company's capital stock receivable upon exercise of the Purchase Option), other than (i) as described in Sections 6.1.1, 6.1.3 or 6.1.4, (ii) regular quarterly or other periodic dividends, (iii) in connection with the redemption rights of the holders of Common Stock upon consummation of the Company's initial Business Combination or (iv) in connection with the Company's liquidation and the distribution of its assets upon its failure to consummate a Business Combination (any such non-excluded event being referred to herein as an "Extraordinary Dividend"), then the Exercise Price shall be decreased, effective immediately after the effective date of such Extraordinary Dividend, by the amount of cash and/or the fair market value (as determined by the Company's Board of Directors, in good faith) of any securities or other assets paid on each share of Common Stock in respect of such Extraordinary Dividend.

- 6.1.3 Aggregation of Shares. If after the date hereof, and subject to the provisions of Section 6.3, the number of outstanding Common Shares is decreased by a consolidation, combination or reclassification of Common Shares or other similar event, then, on the effective date thereof, the number of Common Shares underlying each of the Units purchasable hereunder shall be decreased in proportion to such decrease in outstanding shares. In such case, the number of Common Shares, and the exercise price applicable thereto, underlying the Warrants underlying each of the Units purchasable hereunder shall be adjusted in accordance with the terms of the Warrants.
- 6.1.4 Replacement of Securities Upon Reorganization, Etc. In case of any reclassification or reorganization of the outstanding Common Shares other than a change covered by Section 6.1.1 or 6.1.3 hereof or that solely affects the par value of such Common Shares, or in the case of any merger or consolidation of the Company with or into another corporation (other than a consolidation or merger in which the Company is the continuing corporation and that does not result in any reclassification or reorganization of the outstanding Common Shares), or in the case of any sale or conveyance to another corporation or entity of the property of the Company as an entirety or substantially as an entirety in connection with which the Company is dissolved, the Holder of this Purchase Option shall have the right thereafter (until the expiration of the right of exercise of this Purchase Option) to receive upon the exercise hereof, for the same aggregate Exercise Price payable hereunder immediately prior to such event, the kind and amount of shares or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, by a Holder of the number of Common Shares of the Company obtainable upon exercise of this Purchase Option and the underlying Warrants immediately prior to such event; and if any reclassification also results in a change in Common Shares covered by Section 6.1.1 or 6.1.3, then such adjustment shall be made pursuant to Sections 6.1.1, 6.1.3 and this Section 6.1.4. The provisions of this Section 6.1.4 shall similarly apply to successive reclassifications, reorganizations, mergers or consolidations, sales or other transfers.
- 6.1.5 <u>Changes in Form of Purchase Option</u>. This form of Purchase Option need not be changed because of any change pursuant to this Section, and Purchase Options issued after such change may state the same Exercise Price and the same number of Units as are stated in the Purchase Options initially issued pursuant to this Agreement. The acceptance by any Holder of the issuance of new Purchase Options reflecting a required or permissive change shall not be deemed to waive any rights to an adjustment occurring after the Commencement Date or the computation thereof.

- 6.1.6 Adjustments of Warrants. To the extent the price of the Warrants are lowered pursuant to Section 3.1 of the Warrant Agreement, the price of the Warrants underlying the Purchase Option shall be reduced on identical percentage terms. To the extent the duration of the Warrants is extended pursuant to Section 3.2 of the Warrant Agreement, the duration of the Warrants underlying the Purchase Option shall be extended on identical terms.
- 6.2 <u>Substitute Purchase Option</u>. In case of any consolidation of the Company with, or merger of the Company into, another corporation (other than a consolidation or merger which does not result in any reclassification or change of the outstanding Common Shares), the corporation formed by such consolidation or merger shall execute and deliver to the Holder a supplemental Purchase Option providing that the holder of each Purchase Option then outstanding or to be outstanding shall have the right thereafter (until the stated expiration of such Purchase Option) to receive, upon exercise of such Purchase Option, the kind and amount of shares and other securities and property receivable upon such consolidation or merger, by a holder of the number of Common Shares of the Company for which such Purchase Option might have been exercised immediately prior to such consolidation, merger, sale or transfer. Such supplemental Purchase Option shall provide for adjustments which shall be identical to the adjustments provided in Section 6. The above provision of this Section shall similarly apply to successive consolidations or mergers.
- 6.3 <u>Elimination of Fractional Interests</u>. The Company shall not be required to issue certificates representing fractions of Common Shares or Warrants upon the exercise of the Purchase Option, nor shall it be required to issue scrip or pay cash in lieu of any fractional interests, it being the intent of the parties that all fractional interests shall be eliminated by rounding any fraction up or down to the nearest whole number of Warrants, Common Shares or other securities, properties or rights.
- 7. Reservation and Listing. The Company shall at all times reserve and keep available out of its authorized Common Shares, solely for the purpose of issuance upon exercise of the Purchase Options or the Warrants underlying the Purchase Option, such number of Common Shares or other securities, properties or rights as shall be issuable upon the exercise thereof. The Company covenants and agrees that, upon exercise of the Purchase Options and payment of the Exercise Price therefor, all Common Shares and other securities issuable upon such exercise shall be duly and validly issued, fully paid and non-assessable and not subject to preemptive rights of any shareholder. The Company further covenants and agrees that upon exercise of the Warrants underlying the Purchase Options and payment of the respective Warrant exercise price therefor, all Common Shares and other securities issuable upon such exercise shall be duly and validly issued, fully paid and non-assessable and not subject to preemptive rights of any shareholder. As long as the Purchase Options shall be outstanding, the Company shall use its best efforts to cause all (i) Units and Common Shares issuable upon exercise of the Purchase Options, (iii) Warrants issuable upon exercise of the Purchase Options and (iv) Common Shares issuable upon exercise of the Warrants included in the Units issuable upon exercise of the Purchase Option to be listed (subject to official notice of issuance) on all securities exchanges (or, if applicable on the Nasdaq Global Market, Capital Market, OTC Bulletin Board or any successor trading market) on which the Units, the Common Shares or the Public Warrants issued to the public in connection herewith may then be listed and/or quoted.

8. Notice Requirements.

- 8.1 <u>Holder's Right to Receive Notice</u>. Nothing herein shall be construed as conferring upon the Holders the right to vote or consent as a shareholder for the election of directors or any other matter, or as having any rights whatsoever as a shareholder of the Company. If, however, at any time prior to the expiration of the Purchase Options and their exercise, any of the events described in Section 8.2 shall occur, then, in one or more of said events, the Company shall give written notice of such event at least fifteen days prior to the date fixed as a record date or the date of closing the transfer books for the determination of the shareholders entitled to such dividend, distribution, conversion or exchange of securities or subscription rights, or entitled to vote on such proposed dissolution, liquidation, winding up or sale. Such notice shall specify such record date or the date of the closing of the transfer books, as the case may be. Notwithstanding the foregoing, the Company shall deliver to each Holder a copy of each notice given to the other shareholders of the Company at the same time and in the same manner that such notice is given to the shareholders.
- 8.2 Events Requiring Notice. The Company shall be required to give the notice described in this Section 8 upon one or more of the following events: (i) if the Company shall take a record of the holders of its Common Shares for the purpose of entitling them to receive a dividend or distribution payable otherwise than in cash, or a cash dividend or distribution payable otherwise than out of retained earnings, as indicated by the accounting treatment of such dividend or distribution on the books of the Company, or (ii) the Company shall offer to all the holders of its Common Shares any additional shares of share capital of the Company or securities convertible into or exchangeable for shares of share capital of the Company, or any option, right or warrant to subscribe therefor, or (iii) a dissolution, liquidation or winding up of the Company (other than in connection with a consolidation or merger) or a sale of all or substantially all of its property, assets and business shall be proposed.
- 8.3 <u>Notice of Change in Exercise Price</u>. The Company shall, promptly after an event requiring a change in the Exercise Price pursuant to Section 6 hereof, send notice to the Holders of such event and change ("Price Notice"). The Price Notice shall describe the event causing the change and the method of calculating same and shall be certified as being true and accurate by the Company's President and Chief Financial Officer.
- 8.4 <u>Transmittal of Notices</u>. All notices, requests, consents and other communications under this Purchase Option shall be in writing and shall be deemed to have been duly made when hand delivered, or mailed by express mail or private courier service: (i) if to the registered Holder of the Purchase Option, to the address of such Holder as shown on the books of the Company, or (ii) if to the Company, to the following address or to such other address as the Company may designate by notice to the Holders:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202 Attn: David L. Warnock, President

9. <u>Miscellaneous</u>.

- 9.1 <u>Amendments.</u> The Company and Morgan Joseph may from time to time supplement or amend this Purchase Option without the approval of any of the Holders in order to cure any ambiguity, to correct or supplement any provision contained herein that may be defective or inconsistent with any other provisions herein, or to make any other provisions in regard to matters or questions arising hereunder that the Company and Morgan Joseph may deem necessary or desirable and that the Company and Morgan Joseph deem shall not adversely affect the interest of the Holders. All other modifications or amendments shall require the written consent of and be signed by the party against whom enforcement of the modification or amendment is sought.
- 9.2 <u>Headings</u>. The headings contained herein are for the sole purpose of convenience of reference, and shall not in any way limit or affect the meaning or interpretation of any of the terms or provisions of this Purchase Option.
- 9.3 <u>Entire Agreement</u>. This Purchase Option (together with the other agreements and documents being delivered pursuant to or in connection with this Purchase Option) constitutes the entire agreement of the parties hereto with respect to the subject matter hereof, and supersedes all prior agreements and understandings of the parties, oral and written, with respect to the subject matter hereof.
- 9.4 <u>Binding Effect</u>. This Purchase Option shall inure solely to the benefit of and shall be binding upon, the Holder and the Company and their permitted assignees, respective successors, legal representative and assigns, and no other person shall have or be construed to have any legal or equitable right, remedy or claim under or in respect of or by virtue of this Purchase Option or any provisions herein contained.
- 9.5 Governing Law; Submission to Jurisdiction. This Purchase Option shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflict of laws. The Company hereby agrees that any action, proceeding or claim against it arising out of, or relating in any way to this Purchase Option shall be brought and enforced in the courts of the State of New York, located in New York County or of the United States of America for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. The Company hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum. Any process or summons to be served upon the Company may be served by transmitting a copy thereof by registered or certified mail, return receipt requested, postage prepaid, addressed to it at the address set forth in Section 8 hereof. Such mailing shall be deemed personal service and shall be legal and binding upon the Company in any action, proceeding or claim. The Company and the Holder agree that the prevailing party(ies) in any such action shall be entitled to recover from the other party(ies) all of its reasonable attorneys' fees and expenses relating to such action or proceeding and/or incurred in connection with the preparation therefor.
- 9.6 <u>Waiver, Etc.</u> The failure of the Company or the Holder to at any time enforce any of the provisions of this Purchase Option shall not be deemed or construed to be a waiver of any such provision, nor to in any way affect the validity of this Purchase Option or any provision hereof or the right of the Company or any Holder to thereafter enforce each and every provision of this Purchase Option. No waiver of any breach, non-compliance or non-fulfillment of any of the provisions of this Purchase Option shall be effective unless set forth in a written instrument executed by the party or parties against whom or which enforcement of such waiver is sought; and no waiver of any such breach, non-compliance or non-fulfillment shall be construed or deemed to be a waiver of any other or subsequent breach or non-compliance.

- 9.7 <u>Execution in Counterparts.</u> This Purchase Option may be executed in one or more counterparts, and by the different parties hereto in separate counterparts, each of which shall be deemed to be an original, but all of which taken together shall constitute one and the same agreement, and shall become effective when one or more counterparts has been signed by each of the parties hereto and delivered to each of the other parties hereto.
- 9.8 <u>Exchange Agreement</u>. As a condition of the Holder's receipt and acceptance of this Purchase Option, Holder agrees that, at any time prior to the complete exercise of this Purchase Option by Holder, if the Company and Morgan Joseph enter into an agreement ("Exchange Agreement") pursuant to which they agree that all outstanding Purchase Options will be exchanged for securities or cash or a combination of both, then Holder shall agree to such exchange and become a party to the Exchange Agreement.

IN WITNESS WHEREOF, the Company has caused this day of 2007.	s Purchase Option to be signed by its duly authorized officer as of the
	CAMDEN LEARNING CORPORATION

Ву:	
	Name: Title
- 18	} -

Form to be used to exercise Purchase Option:

Camden Learning Corporation

500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202
Attn: David L. Warnock, President
Date: , 20
The undersigned hereby elects irrevocably to exercise all or a portion of the within Purchase Option and to purchase Units of Camden Learning Corporation and hereby makes payment of \$ (at the rate of \$ per Unit) in payment of the Exercise Price pursuant thereto. Please issue the Common Shares and Warrants as to which this Purchase Option is exercised in accordance with the instructions given below.
or
The undersigned hereby elects irrevocably to convert its right to purchase Units purchasable under the within Purchase Option by surrender of the unexercised portion of the attached Purchase Option (with a "Value" of \$ based on a "Market Price" of \$ Please issue the securities comprising the Units as to which this Purchase Option is exercised in accordance with the instructions given below.
NOTICE: The signature to this assignment must correspond with the name as written upon the face of the purchase option every particular, without alteration or enlargement or any change whatever.
Signature(s) Guaranteed:
THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15).
INSTRUCTIONS FOR REGISTRATION OF SECURITIES
Name
(Print in Block Letters)
Address
- 19 -

Form to be used to assign Purchase Option:

ASSIGNMENT

(To be executed by the registered Holder to effect a transfer of the within Purchase Option):

FOR VALUE RECEIVED, does hereby sell, assign and transfer unto the right to purchase Units of Camden Learning Corporation ("COMPANY") evidenced by the within Purchase Option and does hereby authorize the Company to transfer such right on the books of the Company.

Dated: , 20
Signature
NOTICE: The signature to this assignment must correspond with the name as written upon the face of the purchase option in every particular, without alteration or enlargement or any change whatever.

Signature(s) Guaranteed:

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15).

November 26, 2007

Camden Learning Corporation 500 East Pratt Street Suite 1200 Baltimore, MD 21202

Ladies and Gentlemen:

Reference is made to the Registration Statement on Form S-1 (File No. 333-143098), as amended (the "Registration Statement") filed by Camden Learning Corporation (the "Company"), a Delaware corporation, under the Securities Act of 1933, as amended (the "Act"), covering (i) 6,250,000 units, with each unit consisting of one share of the Company's common stock, par value \$.0001 per share (the "Common Stock"), and one warrant, each to purchase one share of the Company's Common Stock (the "Warrants," and the shares of Common Stock underlying the Warrants, the "Warrant Shares") to the underwriters for whom Morgan Joseph & Co. Inc. ("Morgan Joseph") is acting as representative (collectively, the "Underwriters"), (ii) up to 937,500 units which the Underwriters will have a right to purchase from the Company to cover over-allotments, if any (the "Over-Allotment Units," and collectively with the 6,250,000 units to be sold pursuant to the terms of the Registration Statement, the "Units"), (iii) up to 7,187,500 shares of Common Stock underlying the Units, (iv) up to 7,187,000 Warrants underlying the Units, (v) up to 7,187,500 Warrant Shares, (vi) one (1) Unit Purchase Option (the "Purchase Option") pursuant to which Morgan Joseph shall have the right to purchase for its own account or that of its designees up to an aggregate of 625,000 Units, (vii) up to 625,000 Units (the "PO Units") underlying the Purchase Option, (vii) up to 625,000 shares of Common Stock and up to 625,000 Warrants issued as part of the PO Units (the "PO Shares" and the "PO Warrants," respectively) and (viii) up to 625,000 shares of Common Stock issuable upon exercise of the PO Warrants (the "PO Warrant Shares").

We have examined such documents and considered such legal matters as we have deemed necessary and relevant as the basis for the opinion set forth below. With respect to such examination, we have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as reproduced or certified copies, and the authenticity of the originals of those latter documents. As to questions of fact material to this opinion, we have, to the extent deemed appropriate, relied upon certain representations of certain officers and employees of the Company.

Based upon the foregoing, we are of the opinion that:

- 1. Units. When the Registration Statement becomes effective under the Act and when the offering is completed as contemplated by the Registration Statement, such Units will be validly issued, fully paid and non-assessable.
- **2. Common Stock**. When the Registration Statement becomes effective under the Act and when the offering is completed as contemplated by the Registration Statement, the shares of Common Stock will be validly issued, fully paid and non-assessable.

- 3. Warrants and Warrant Shares. When the Registration Statement becomes effective under the Act, when the terms of the warrant agreement under which the Warrants are to be issued (the "Warrant Agreement") are duly established and the Warrant Agreement is duly executed and delivered, when the terms of the Warrants underlying the Units and of their issuance and sale are duly established in conformity with the Warrant Agreement and when such Warrants are duly executed and authenticated in accordance with the Warrant Agreement and issued, delivered, sold and paid for as part of the Units, as contemplated by the Registration Statement, such Warrants will be legally binding obligations of the Company in accordance with their terms, except (a) as such enforceability may be limited by bankruptcy, insolvency, reorganization or similar laws affecting creditors' rights generally and by general equitable principles (regardless of whether enforceability is considered in a proceeding in equity or at law); (b) as enforceability of any indemnification or contribution provision may be limited under the Federal and state securities laws, and (c) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to the equitable defenses and to the discretion of the court before which any proceeding therefor may be brought (collectively, the "Exceptions") and such Warrants will be duly issued, fully paid and non-assessable, and the Warrant Shares underlying such Warrants, when duly issued, delivered, sold and paid for upon exercise of such Warrants, as contemplated by the Warrant Agreement and the Registration Statement, such Warrants will be validly issued, fully paid and non-assessable.
- **4. Purchase Option**. When the Registration Statement becomes effective under the Act, when the terms of the Purchase Option and of its issuance and sale are duly established and the Purchase Option is duly executed and issued, delivered, sold and paid for, as contemplated by the Registration Statement, such Purchase Option will be a legally binding obligation of the Company in accordance with its terms, except for the Exceptions and such Purchase Option will be validly issued, fully paid and non-assessable.
- **5. PO Units**. When the Registration Statement has become effective under the Act, when the terms of the PO Units and of their issuance and sale are duly established, and when such PO Units are duly executed and issued, delivered, sold and paid for upon exercise of the Purchase Option, as contemplated by the Purchase Option and the Registration Statement, such PO Units will be validly issued, fully paid and non-assessable.
- **6. PO Shares**. When the Registration Statement becomes effective under the Act, the terms of the PO Shares underlying the PO Units and the sale thereof are duly established, and such PO Shares are duly issued, delivered, sold and paid for as part of such PO Units, as contemplated by the Purchase Option and the Registration Statement, the PO Shares will be validly issued, fully paid and non-assessable.
- 7. PO Warrants and PO Warrant Shares. When the Registration Statement becomes effective under the Act, when the terms of the Warrant Agreement are duly established and the Warrant Agreement is duly executed and delivered, when the terms of the PO Warrants underlying the PO Units and of their issuance and sale are duly established in conformity with the Warrant Agreement and when such PO Warrants are duly executed and authenticated in accordance with the Warrant Agreement and issued, delivered, sold and paid for as part of the PO Units, as contemplated by the Purchase Option and the Registration Statement, such PO Warrants will be validly issued, fully paid and non-assessable, and the PO Warrant Shares underlying such PO Warrants, when duly issued, delivered, sold and paid for upon exercise of such PO Warrants, as contemplated by the Warrant Agreement and the Registration Statement, such PO Warrants will be validly issued, fully paid and non-assessable.

We are opining solely on all applicable statutory provisions of Delaware corporate law, including the rules and regulations underlying those provisions, all applicable provisions of the Delaware Constitution, all applicable judicial and regulatory determinations in connection therewith and, as to the Warrants and the Purchase Option constituting legally binding obligations of the Company, solely with respect to the laws of the State of New York. Our opinion is based on these laws as in effect on the date hereof. We express no opinion as to whether the laws of any jurisdiction are applicable to the subject matter hereof. We are not rendering any opinion as to compliance with any federal or state law, rule or regulation relating to securities, or to the sale or issuance thereof.

We hereby consent to the use of this opinion as an exhibit to the Registration Statement, to the use of our name as your counsel and to all references made to us in the Registration Statement and in the Prospectus forming a part thereof. In giving this consent, we do not hereby admit that we are in the category of persons whose consent is required under Section 7 of the Act, or the rules and regulations promulgated thereunder. This opinion is given as of the effective date of the Registration Statement, and we are under no duty to update the opinions contained herein.

Very truly yours,

/s/ Ellenoff Grossman & Schole LLP

Ellenoff Grossman & Schole LLP

November 26, 2007

Camden Learning Corporation 500 East Pratt Street Suite 1200 Baltimore, MD 21202

Morgan Joseph & Co. Inc. 600 Fifth Avenue, 19 th floor New York, NY 10020

Re: Camden Learning Corporation (the "Company")

Gentlemen:

The undersigned, in consideration of Morgan Joseph & Co. Inc. (" **Morgan Joseph**") entering into a letter of intent (" **Letter of Intent**") to underwrite an initial public offering of the securities of the Company (" **IPO**") and embarking on the IPO process, hereby agrees as follows (certain capitalized terms used herein are defined in paragraph XV hereof):

- I. (1) In the event the Company fails to consummate a Business Combination within 24 months from the effective date (the " **Effective Date**") of the registration statement relating to the IPO (the "**Termination Date**"), the undersigned shall, in accordance with all applicable requirements of the Delaware General Corporation Law (the "**DGCL**"), (i) cause the Trustee to liquidate the Trust Account to the holders of the IPO Shares and (ii) take all reasonable actions within its power to cause the Company to liquidate as soon as reasonably practicable.
- (2) Except with respect to any of the IPO Shares acquired by the undersigned in connection with or following the IPO, the undersigned hereby (a) waives any and all right, title, interest or claim of any kind ("Claim") in or to all funds in the Trust Account and any remaining net assets of the Company upon liquidation of the Trust Account and dissolution of the Company, (b) waives any Claim the undersigned may have in the future as a result of, or arising out of, any contracts or agreements with the Company and (c) agrees the undersigned will not seek recourse against the Trust Account for any reason whatsoever.
- (3) The undersigned agrees to indemnify and hold harmless the Company against any and all loss, liability, claims, damage and expense whatsoever (including, but not limited to, any and all legal or other expenses reasonably incurred in investigating, preparing or defending against any litigation, whether pending or threatened, or any claim whatsoever) to which the Company may become subject as a result of any claim by any vendor, service provider, prospective target business, creditor or other entity that is owed money by the Company for services rendered or products sold to the Company or the claims of any target businesses, subject to the following limitations: (i) such indemnification will only be made insofar as the Company did not obtain a waiver from such party of such party's rights or claims to the Trust Account as required under the Underwriting Agreement entered into between the Company and Morgan Joseph and (ii) such indemnification will be made only to the extent necessary to ensure that such loss, liability, claim, damage or expense does not reduce the amount in the Trust Account (as defined in the Letter of Intent) below the amount necessary in order for each holder of IPO Shares to receive a liquidation amount of at least \$7.92 per IPO Share owned by such holder.

- (4) In the event that the Company does not consummate a Business Combination and must liquidate and its remaining net assets are insufficient to complete such liquidation, the undersigned agrees to advance such funds necessary to complete such liquidation and agrees not to seek repayment for such expenses.
- (5) The undersigned agrees to take all such action reasonably necessary to request its members make capital contributions to it in order for it to be able to satisfy its obligations under Sections (3) and (4) above.
- II. In order to minimize potential conflicts of interest which may arise from multiple affiliations, the undersigned agrees, in accordance with the Right of First Refusal Agreement, to present to the Company for its consideration, prior to presentation to any other person or entity, any suitable opportunity to acquire an operating business, until the earlier of (i) the consummation by the Company of a Business Combination or (ii) the dissolution of the Company, subject to any pre-existing fiduciary and contractual obligations the undersigned might have.
- **III.** The undersigned acknowledges and agrees the Company will not consummate any Business Combination which involves a company which is affiliated with any of the Insiders unless the Company obtains an opinion from an independent investment banking firm which is a member of the National Association of Securities Dealers, Inc. and is reasonably acceptable to Morgan Joseph that the Business Combination is fair to the Company's stockholders from a financial perspective.
- **IV.** (1) Neither the undersigned, any member of the family of the undersigned, nor any affiliate of the undersigned ("**Affiliate**") will be entitled to receive, and no such person will accept, any compensation for services rendered to the Company by the Company prior to the consummation of a Business Combination; provided that commencing on the Effective Date, the undersigned shall be allowed to charge the Company \$7,500 per month, to compensate it for certain general and administrative services, including but not limited to receptionist, secretarial and general office services, that it will provide to the Company.
- (2) The undersigned shall be entitled to reimbursement from the Company for its out-of-pocket expenses incurred in connection with seeking and consummating a Business Combination, only (a) from funds not held in the Trust Account or (b) upon the consummation of a Business Combination.
- **V.** (1) Neither the undersigned, any member of the family of the undersigned, nor any Affiliate will be entitled to receive or accept a finder's fee or any other compensation in the event the undersigned, any member of the family of the undersigned or any Affiliate originates a Business Combination.
- (2) The undersigned's Questionnaire previously furnished to the Company and Morgan Joseph is true and accurate in all respects as of the date first written above.
 - (3) The undersigned represents and warrants:
 - (a) it is not subject to or a respondent in any legal action for, any injunction relating to, or any cease-and-desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction;
 - (b) it has never been convicted of or pleaded guilty to any crime (i) involving any fraud or (ii) relating to any financial transaction or handling of funds of another person, or (iii) pertaining to any dealings in any securities, and is not currently a defendant in any such criminal proceeding; and
 - (c) it has never been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked.

- VI. The undersigned has full right and power, without violating any agreement by which it is bound, to enter into this letter agreement.
- VII. The undersigned authorizes any employer, financial institution or consumer credit reporting agency to release to Morgan Joseph and its legal representatives or agents (including any investigative search firm retained by Morgan Joseph) any information they may have about the undersigned's background and finances ("Information"). Neither Morgan Joseph nor its agents shall be violating the undersigned's right of privacy in any manner in requesting and obtaining the Information and the undersigned hereby releases them from liability for any damage whatsoever in that connection; provided, that Morgan Joseph shall maintain the confidentiality of any information received pursuant thereto, and further shall not transfer, or cause or permit the transfer of, such information to any other person or party, or use such information other than in connection with the IPO, in each case without the express written consent of the undersigned.
- **VIII.** In connection with the vote required to consummate a Business Combination, the undersigned agrees it will vote (A) all shares of common stock owned by it (either directly or indirectly) prior to the IPO (the "**Insider Shares**"), in accordance with the majority of the votes cast by the holders of the IPO Shares and (B) all shares of common stock acquired in or following the IPO "for" a Business Combination.
- **IX.** The undersigned hereby agrees not to propose, or vote in favor of, any amendment to the Company's Certificate of Incorporation that requires the affirmative vote of at least 95% of the IPO Shares. This paragraph may not be modified or amended under any circumstances.
- **X.** The undersigned will escrow all of the Insider Shares beneficially owned by it, if any, for the period commencing on the Effective Date and ending on the first anniversary following a Business Combination, subject to the terms of a Securities Escrow Agreement which the Company will enter into with the undersigned and an escrow agent acceptable to the Company.
- **XI.** The undersigned will escrow its warrants purchased in a private placement immediately prior to the IPO until the 90th day after the date of the consummation of the initial Business Combination, subject to the terms of a Securities Escrow Agreement which the Company will enter into with the undersigned and Continental Stock Transfer & Trust Company.
- **XII.** The undersigned hereby waives its right to exercise redemption rights with respect to any Insider Shares owned or to be owned by the undersigned, directly or indirectly, and agrees that it will not seek redemption with respect to such shares in connection with any vote to approve a Business Combination.
- XIII. The undersigned hereby agrees to place limit orders to purchase up to \$4,000,000 of Company common stock in the open market commencing ten business days after the Company files a Current Report on form 8-K announcing the execution of a definitive agreement for a Business Combination and ending on the business day immediately preceding the date of the meeting of stockholders at which a Business Combination is to be approved. The undersigned acknowledges such purchases will be made in accordance with Rule 10b-18 under the Securities Exchange Act of 1934, as amended, at a price of not more than the per share amount held in the Trust Account (less taxes payable) as reported in such 8-K and will be made by a broker-dealer mutually agreed upon by the undersigned and Morgan Joseph in such amounts and at such times as such broker-dealer may determine, in its sole discretion, so long as the purchase price does not exceed the above-referenced per share purchase price. To the extent that the aggregate amount of such purchases is less than \$4,000,000, the undersigned agrees to purchase from the Company, and the Company agrees to sell to the undersigned, a number of units identical to the units offered in the IPO at a price of \$8.00 per unit in a private placement to be completed immediately prior to the consummation of a business combination until it has purchased, together with the above-referenced open market purchases, an aggregate of \$4,000,000 of Company common stock.
- XIV. This letter agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflicts of law principles that would result in the application of the substantive laws of another jurisdiction. The undersigned hereby (i) agrees that any action, proceeding or claim against him arising out of or relating in any way to this letter agreement (a "Proceeding") shall be brought and enforced in the federal courts of the United States of America for the Southern District of New York, and irrevocably submits to the jurisdiction of such courts, which jurisdiction shall be exclusive, (ii) waives any objection to the exclusive jurisdiction of such courts and any objection that such courts represent an inconvenient forum and (iii) irrevocably agrees to appoint Ellenoff Grossman & Schole LLP as agent for the service of process in the State of New York to receive, for the undersigned and on its behalf, service of process in any Proceeding. If for any reason such agent is unable to act as such, the undersigned will promptly notify the Company and Morgan Joseph and appoint a substitute agent acceptable to each of the Company and Morgan Joseph within 30 days and nothing in this letter will affect the right of either party to serve process in any other manner permitted by law.

XV. As used herein, (i) a "Business Combination" shall mean a merger, capital stock exchange, asset acquisition or other similar
business combination between the Company and one or more operating businesses in the education industry; (ii) "Insiders" shall
mean all officers, directors and stockholders of the Company immediately prior to the IPO; (iii) "IPO Shares" shall mean the shares of
Common Stock issued in the Company's IPO; (iv) "Trust Account" shall mean the trust account in which most of the proceeds to the
Company of the IPO will be deposited and held for the benefit of the holders of the IPO shares, as described in greater detail in the
prospectus relating to the IPO; (v) "Trustee" shall mean Continental Stock Transfer & Trust Company and (vi) "Right of First Refusal
Agreement" refers to such agreement executed by all the officers and directors in connection with the transactions contemplated
hereby.

XVI. This letter agreement shall supersede any other letter agreement signed by the undersigned with respect to the subject matter hereof.

[Signature Page to Follow]

O	1	
Camaen	Learning,	LLC

By: Camden Partners Strategic III, LLC

Its: Manager

By: Camden Partners Strategic Manager, LLC Its: Managing Member

Ву:

Donald W. Hughes Managing Member EX-10.2 12 v095534_ex10-2.htm **EXHIBIT 10.2**

INVESTMENT MANAGEMENT TRUST AGREEMENT

This Agreement is made as of [], 2007 by and between Camden Learning Corporation (the "Company") whose principal office is located at 500 East Pratt Street, Suite 1200, Baltimore, Maryland 21202 and Continental Stock Transfer & Trust Company ("Trustee") located at 17 Battery Place, New York, New York 10004.

WHEREAS, the Company's Registration Statement on Form S-1, File No. 333-143098 ("Registration Statement"), for its initial public offering of securities ("IPO") has been declared effective on November [], 2007 by the Securities and Exchange Commission ("Effective Date"); and

WHEREAS, the Company has completed a private placement of 2,800,000 Warrants (the "Private Warrants") prior to the completion of the IPO for a purchase price of \$2,800,000; and

WHEREAS, Morgan Joseph & Co. Inc. ("Morgan Joseph") is acting as the representative of the underwriters in the IPO (the "Underwriters"); and

WHEREAS, as described in the Company's Registration Statement, in accordance with the Company's Amended and Restated Certificate of Incorporation, \$[49,500,000] of the net proceeds of the IPO and the sale of the Private Warrants (\$[] if the Underwriters' over-allotment option is exercised in full), will be delivered to the Trustee as of [], 2007 to be deposited and held in a trust account for the benefit of the Company, the holders of the common stock, par value \$.0001 per share, of the Company ("Common Stock") included in the units of the Company's securities issued in the IPO (the "Units") and Morgan Joseph and in the event the securities offered in the IPO are registered in Colorado, pursuant to Section 11-51-302 (6) of the Colorado Revised Statutes ("CRS"), a copy of which is attached hereto, and made a part hereof. The amount to be delivered to the Trustee will be referred to herein as the "Property," the stockholders for whose benefit the Trustee shall hold the Property will be referred to as the "Public Stockholders," and the Public Stockholders, the Company and Morgan Joseph and the Underwriters will be referred to together as the "Beneficiaries";

WHEREAS, a portion of the Property consists of \$1,500,000 (or \$1,725,000 if the Underwriters' over-allotment is exercised in full) attributable to the Underwriters' discount (the "Deferred Discount") which the Underwriters have agreed to deposit in the Trust Account (as defined below); and

WHEREAS, the Company and the Trustee desire to enter into this Agreement to set forth the terms and conditions pursuant to which the Trustee shall hold the Property.

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and agreements herein contained, the parties hereto agree as follows:

- 1. Agreements and Covenants of Trustee. The Trustee hereby agrees and covenants to:
- (a) Hold the Property in trust for the Beneficiaries in accordance with the terms of this Agreement, including, without limitation, with respect to the Public Stockholders, the terms of Section 11-51-302(6) of the CRS in segregated trust accounts (the "Trust Account") established by the Trustee with JP Morgan Chase Bank, N.A. and at a brokerage institution selected by the Trustee;
 - (b) Manage, supervise and administer the Trust Account subject to the terms and conditions set forth herein;
- (c) In a timely manner, upon the written instruction of the Company, to invest and reinvest the Property in any "Government Security" or in money market funds selected by the Company meeting the conditions specified in Rule 2a-7 promulgated under the Investment Company Act of 1940, as amended, as determined by the Company. As used herein, "Government Security" means any Treasury Bill issued by the United States, having a maturity of one hundred and eighty days or less;
- (d) Collect and receive, when due, all principal and income arising from the Property, which shall become part of the "Property," as such term is used herein:

- (e) Promptly notify the Company and Morgan Joseph of all communications received by it with respect to any Property requiring action by the Company;
- (f) Supply any necessary information or documents as may be requested by the Company in connection with the Company's preparation of the tax returns for the Trust Account or the Company;
- (g) Participate in any plan or proceeding for protecting or enforcing any right or interest arising from the Property if, as and when instructed by the Company and/or Morgan Joseph to do so;
- (h) Render to the Company and to such other person as the Company may instruct, monthly written statements of the activities of and amounts in the Trust Account reflecting all receipts and disbursements of the Trust Account;
- (i) If there is any income or other tax obligation relating to the income from the Property in the Trust Account, then, from time to time, at the written instruction of the Company, the Trustee shall promptly to the extent there is not sufficient cash in the Trust Account to pay such tax obligation, liquidate such assets held in the Trust Account as shall be designated by the Company in writing; and
- (j) Commence liquidation of the Trust Account only after and promptly after receipt of, and only in accordance with, the terms of a letter ("Termination Letter"), in a form substantially similar to that attached hereto as either Exhibit A or Exhibit B hereto, signed on behalf of the Company by its President or Chairman of the Board and Secretary or Assistant Secretary or other authorized officer of the Company, and complete the liquidation of the Trust Account and distribute the Property in the Trust Account only as directed in the Termination Letter and the other documents referred to therein; provided, however, that in the event that a Termination Letter has not been received by the Trustee by the 24-month anniversary of the effective date of the Registration Statement ("Last Date"), the Trust Account shall be liquidated in accordance with the procedures set forth in the Termination Letter attached as Exhibit B hereto and distributed to the stockholders of record on the Last Date. In all cases, the Trustee shall provide Morgan Joseph with a copy of any Termination Letters and/or any other correspondence that it receives with respect to any proposed withdrawal from the Trust Account promptly after it receives same. The provisions of this Section 1(j) may not be modified, amended or deleted under any circumstances.

2. <u>Limited Distributions of Income on Property</u>.

- (a) If there is any income tax obligation relating to the income from the Property in the Trust Account, then, at the written instruction of the Company, the Trustee shall disburse to the Company by wire transfer or check (as directed by the Company in its instruction letter), out of the Property in the Trust Account, the amount indicated by the Company as required to pay income taxes and disburse to the Company by wire transfer out of the Property in the Trust Account, the amount indicated by the Company as owing in respect of such income tax.
- (b) Upon written request from the Company containing certification that such distribution pursuant to this Section 2(b) shall only be used to fund the working capital requirements of the Company and the costs related to identifying, researching and acquiring a prospective target businesses, included, without limitation, the expenses incurred in connection with the Company's dissolution, in each case as described in the prospectus that forms a part of the Registration Statement, the Trustee shall distribute to the Company an amount equal to up to \$600,000 of the income earned on the Property in the Trust Account, net of taxes payable, through the last day of the month immediately preceding the date of receipt of the Company's written request.
 - (c) Upon receipt of the Termination Letter, the Trustee shall liquidate the Trust Account in accordance with Section 1(j).
 - (d) Except as provided in this Section 2, no other distributions from the Trust Account shall be permitted.
 - 3. Agreements and Covenants of the Company. The Company hereby agrees and covenants:
- (a) To provide all instructions to the Trustee hereunder in writing, signed by the Company's Chief Executive Officer and Chief Financial Officer. In addition, except with respect to its duties under paragraphs 1(i) and 1(j), the Trustee shall be entitled to rely on, and shall be protected in relying on, any verbal or telephonic advice or instruction which it in good faith believes to be given by any one of the persons authorized above to give written instructions, provided that the Company and/or Morgan Joseph shall promptly confirm such instructions in writing;

- (b) To hold the Trustee harmless and indemnify the Trustee from and against any and all expenses, including reasonable counsel fees and disbursements, or loss suffered by the Trustee in connection with any action, suit or other proceeding brought against the Trustee involving any claim, or in connection with any claim or demand which in any way arises out of or relates to this Agreement, the services of the Trustee hereunder, or the Property or any income earned from investment of the Property, except for expenses and losses resulting from the Trustee's gross negligence or willful misconduct. Promptly after the receipt by the Trustee of notice of demand or claim or the commencement of any action, suit or proceeding, pursuant to which the Trustee intends to seek indemnification under this paragraph, it shall notify the Company in writing of such claim (hereinafter referred to as the "Indemnified Claim"). The Trustee shall have the right to conduct and manage the defense against such Indemnified Claim, provided that the Trustee shall obtain the consent of the Company with respect to the selection of counsel, which consent shall not be unreasonably withheld. The Trustee may not agree to settle any Indemnified Claim without the prior written consent of the Company which consent shall not be unreasonably withheld. The Company may participate in such action with its own counsel at its own expense;
- (c) Pay the Trustee an initial acceptance fee, an annual fee and a transaction processing fee for each disbursement made pursuant to Sections 1(i) and 2(b) as set forth on Schedule A hereto, which fees shall be subject to modification by the parties from time to time. It is expressly understood that the Property shall not be used to pay such fees and further agreed that said transaction processing fees shall be deducted by the Trustee from the disbursements made to the Company pursuant to Section 2(b). The Company shall pay the Trustee the initial acceptance fee and first year's fee at the consummation of the IPO and thereafter on the anniversary of the Effective Date. The Trustee shall refund to the Company the annual fee (on a pro rata basis) with respect to any period after the liquidation of the Trust Account. The Company shall not be responsible for any other fees or charges of the Trustee except as set forth in this Section 3(c) and as may be provided in Section 3(b) hereof (it being expressly understood that the Property shall not be used to make any payments to the Trustee under such Sections);
- (d) That, in the event that the Company consummates a Business Combination and the Trust Account is liquidated in accordance with Section 1(j) hereof, an independent party designated by Morgan Joseph shall act as the inspector of election to certify the results of the stockholder vote. As used in this Agreement, the term "Business Combination" means the acquisition by the Company, through merger, capital stock exchange, asset or stock acquisition of, or similar business combination with, one or more entities with an operating business in the education industry, as more fully described in the prospectus forming a part of the Registration Statement;
- (e) That the Officers Certificate referenced in Sections 1(j) hereof shall require the Company's Chief Executive Officer and Chief Financial Officer to each certify the following (wherever applicable): (1) prior to the Last Date, the Company has entered into a Business Combination with a target business, the terms of which are consistent with the requirements set forth in the Registration Statement; and (2) the Board of Directors (the "Board") pursuant to the unanimous written consent of the Board has approved the Business Combination. A copy of such consent and the definitive agreement relating to the Business Combination so approved shall be attached as an exhibit to the Officers Certificate:
- (f) In connection with any vote of the Company's stockholders regarding a Business Combination, provide to the Trustee an affidavit or certificate of a firm regularly engaged in the business of soliciting proxies and tabulating stockholder votes verifying the vote of the Company's stockholders regarding such Business Combination;
- (g) Within five business days after the Underwriters' over-allotment option (or any unexercised portion thereof) expires or is exercised in full, to provide the Trustee notice in writing (with a copy to the Underwriters) of the total amount of the Deferred Discount, which shall in no event be less than \$1,500,000; and
 - 4. <u>Limitations of Liability</u>. The Trustee shall have no responsibility or liability to:
- (a) Take any action with respect to the Property, other than as directed in Sections 1 and 2 hereof and the Trustee shall have no liability to any party except for liability arising out of its own gross negligence or willful misconduct;
- (b) Institute any proceeding for the collection of any principal and income arising from, or institute, appear in or defend any proceeding of any kind with respect to, any of the Property unless and until it shall have received written instructions from the Company given as provided herein to do so and the Company shall have advanced or guaranteed to it funds sufficient to pay any expenses incident thereto;
 - (c) Change the investment of any Property, other than in compliance with Section 1(c);

- (d) Refund any depreciation in principal of any Property;
- (e) Assume that the authority of any person designated by the Company to give instructions hereunder shall not be continuing unless provided otherwise in such designation, or unless the Company shall have delivered a written revocation of such authority to the Trustee;
- (f) The other parties hereto or to anyone else for any action taken or omitted by it, or any action suffered by it to be taken or omitted, in good faith and in the exercise of its own best judgment, except for its gross negligence or willful misconduct. The Trustee may rely conclusively and shall be protected in acting upon any order, notice, demand, certificate, opinion or advice of counsel (including counsel chosen by the Trustee), statement, instrument, report or other paper or document (not only as to its due execution and the validity and effectiveness of its provisions, but also as to the truth and acceptability of any information therein contained) which is believed by the Trustee, in good faith, to be genuine and to be signed or presented by the proper person or persons. The Trustee shall not be bound by any notice or demand, or any waiver, modification, termination or rescission of this Agreement or any of the terms hereof, unless evidenced by a written instrument delivered to the Trustee signed by the proper party or parties and, if the duties or rights of the Trustee are affected, unless it shall give its prior written consent thereto;
- (g) Verify the correctness of the information set forth in the Registration Statement or to confirm or assure that any acquisition made by the Company or any other action taken by it is as contemplated by the Registration Statement;
- (h) Pay Prepare, execute and file tax reports, income or other tax returns and pay any taxes with respect to income and activities relating to the Trust Account, regardless of whether such tax is payable by the Trust Account or the Company (including but not limited to income tax obligations), it being expressly understood that as set forth in Section 1(i), if there is any income or other tax obligation relating to the Trust Account or the Property in the Trust Account, as determined from time to time by the Company and regardless of whether such tax is payable by the Company or the Trust, at the written instruction of the Company, the Trustee shall make funds available in cash from the Property in the Trust Account an amount specified by the Company as owing to the applicable taxing authority, which amount shall be paid directly to the Company by electronic funds transfer, account debit or other method of payment, and the Company shall forward such payment to the taxing authority; and
- (i) Verify calculations, qualify or otherwise approve Company requests for distributions pursuant to Section 1(i), 2(a) or 2(b) above.
- 5. No Right of Set-Off. The Trustee waives any right of set-off or any right, title, interest or claim of any kind that the Trustee may have against the Property held in the Trust Account. In the event the Trustee has a claim against the Company under this Agreement, including, without limitation, under Section 3(b), the Trustee will pursue such claim solely against the Company and not against the Property held in the Trust Account.
 - 6. <u>Termination</u>. This Agreement shall terminate as follows:
- (a) If the Trustee gives written notice to the Company that it desires to resign under this Agreement, the Company shall use its reasonable efforts to locate a successor trustee during which time the Trustee shall continue to act in accordance with the terms of this Agreement. At such time the Company notifies the Trustee that a successor trustee has been appointed by the Company and has agreed to become subject to the terms of this Agreement, the Trustee shall transfer the management of the Trust Account to the successor trustee, including, but not limited to, the transfer of copies of the reports and statements relating to the Trust Account, whereupon this Agreement shall terminate; provided, however, that, in the event the Company does not locate a successor trustee within ninety days of receipt of the resignation notice from the Trustee, the Trustee may, but shall not be obligated to, submit an application to have the Property deposited with the United States District Court for the Southern District of New York and upon such deposit, the Trustee shall be immune from any liability whatsoever that arises due to any actions or omissions to act by any party after such deposit;
- (b) At such time that the Trustee has completed the liquidation of the Trust Account in accordance with the provisions of Section 1(j) hereof, and distributed the Property in accordance with the provisions of the Termination Letter, this Agreement shall terminate except with respect to Section 3(b); or
- (c) At such time that the Trustee has completed the liquidation of the Trust Account and distributed the Property in accordance with Sections 1(j) hereof, this Agreement shall terminate except with respect to Section 3(b).

7. Miscellaneous.

- (a) The Company and the Trustee each acknowledge that the Trustee will follow the security procedures set forth below with respect to funds transferred from the Trust Account. The Company and the Trustee will each restrict access to confidential information relating to such security procedures to authorized persons. Each party must notify the other party immediately if it has reason to believe unauthorized persons may have obtained access to such information, or of any change in its authorized personnel. In executing funds transfers, the Trustee will rely upon account numbers or other identifying numbers of a beneficiary, beneficiary's bank or intermediary bank, rather than names. The Trustee shall not be liable for any loss, liability or expense resulting from any error in an account number or other identifying number, provided it has accurately transmitted the numbers provided.
- (b) This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without giving effect to conflict of laws principles that would result in the application of the substantive laws of another jurisdiction. It may be executed in several counterparts, each one of which shall constitute an original, and together shall constitute but one instrument. Facsimile signatures shall constitute original signatures for all purposes of this Agreement.
- (c) This Agreement contains the entire agreement and understanding of the parties hereto with respect to the subject matter hereof. This Agreement or any provision hereof may only be changed, amended or modified by a writing signed by each of the parties hereto; provided, however, that no such change, amendment or modification may be made without the prior written consent of Morgan Joseph, who, along with each other Underwriter, the parties specifically agree, is and shall be a third party beneficiary for purposes of this Agreement; and provided further, any amendment to Section 1(j) shall require the vote or consent of holders of 95% of the shares of Common Stock sold in the IPO. As to any claim, cross-claim or counterclaim in any way relating to this Agreement, each party waives the right to trial by jury.
- (d) The parties hereto consent to the jurisdiction and venue of any state or federal court located in the State and County of New York for purposes of resolving any disputes hereunder. The parties hereto irrevocably submit to such jurisdiction, which jurisdiction shall be exclusive, and hereby waive any objection to such exclusive jurisdiction and accept such venue, and waive any objection that such courts represent an inconvenient forum.
- (e) Any notice, consent or request to be given in connection with any of the terms or provisions of this Agreement shall be in writing and shall be sent by express mail or similar private courier service, by certified mail (return receipt requested), by hand delivery or by facsimile transmission:

if to the Trustee, to:

Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004 Attn: Steven G. Nelson and Frank A. Di Paolo Fax No.: (212) 509 5150

if to the Company, to:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202 Attn: David L. Warnock, President

Fax No.: (410) 878-6850

in either case with a copy to:

Morgan Joseph & Co. Inc. 600 Fifth Avenue, 19th floor New York, New York 10020 Attn: Gordon Pollock, Managing Director

Fax No.: (212) 218-3719

and

Ellenoff, Grossman & Schole LLP 370 Lexington Avenue New York, New York 10017 Attn: Douglas S. Ellenoff, Esq. Fax No.: (212) 370-7889

and

Title: Chief Executive Officer

McDermott Will & Emery LLP 340 Madison Avenue New York, New York 10173 Attn: Joel L. Rubinstein, Esq. Fax No.: (212) 547-5444

- (f) This Agreement may not be assigned by the Trustee without the prior written consent of the Company and Morgan Joseph.
- (g) Each of the Trustee and the Company hereby represents that it has the full right and power and has been duly authorized to enter into this Agreement and to perform its respective obligations as contemplated hereunder. The Trustee acknowledges and agrees that it shall not make any claims or proceed against the Trust Account, including by way of set-off, and shall not be entitled to any funds in the Trust Account under any circumstance.

IN WITNESS WHEREOF, the parties have duly executed this Investment Management Trust Agreement as of the date first written above.

CONTINENTAL STOCK TRANSFER & TRUST COMPANY, as Trustee

By: _____
Name:
Title:

CAMDEN LEARNING CORPORATION

By: _____
Name: David L. Warnock

3

[Letterhead of Company]

[Insert date]

Date.

Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004

Attn: Stever	n G. Nelson and Frank A. Di Paolo
Re:	Trust Account No. [] Termination Letter
Gentlemen:	
"Company" his is to adv Business") t Company sh	suant to Section 1(j) of the Investment Management Trust Agreement between Camden Learning Corporation and Continental Stock Transfer & Trust Company ("Trustee"), dated as of [
Business Co he IPO Sha Shares have of Article 6 o Agreement,	coordance with paragraph 2 of Article 6 of the Amended and Restated Certificate of Incorporation of the Company, the ombination has been approved by the stockholders of the Company and by the Public Stockholders holding a majority of res cast at the meeting relating to the Business Combination, and Public Stockholders holding less than 30% of the IPO e voted against the Business Combination and given notice of exercise of their redemption rights described in paragraph of the Amended and Restated Certificate of Incorporation of the Company. Pursuant to Section 3(f) of the Trust we are providing you with [an affidavit] [a certificate] of, which verifies the vote of the Company's a in connection with the Business Combination. In accordance with the terms of the Trust Agreement, we hereby authorize

On the Consummation Date (i) counsel for the Company shall deliver to you written notification that (a) the Business Combination has been consummated or will, concurrently with your transfer of funds to the accounts as directed by the Company, be consummated, and (b) the provisions of Section 11-51-302(6) and Rule 51-3.4 of the CRS have been met, to the extent applicable, and (ii) the Company shall deliver to you written instructions with respect to the transfer of the funds held in the Trust Account ("Instruction Letter"). You are hereby directed and authorized to transfer the funds held in the Trust Account immediately upon your receipt of the counsel's letter and the Instruction Letter in accordance with the terms of the Instruction Letter. In the event that certain deposits held in the Trust Account may not be liquidated by the Consummation Date without penalty, you will notify the Company of the same and the Company shall direct you as to whether such funds should remain in the Trust Account and be distributed after the Consummation Date to the Company or be distributed immediately and the penalty incurred. Upon the distribution of all the funds in the Trust Account pursuant to the terms hereof, the Trust Agreement shall be terminated.

you to commence liquidation of the Trust Account to the effect that, on the Consummation Date, all of funds held in the Trust Account will be immediately available for transfer to the account or accounts that the Company shall direct in writing on the Consummation

In the event that the Business Combination is not consummated on the Consummation Date described in the notice thereof and we have not notified you on or before the original Consummation Date of a new Consummation Date, then the funds held in the Trust Account shall be reinvested as provided in the Trust Agreement on the business day immediately following the Consummation Date as set forth in the notice.

	Very truly yours,
	CAMDEN LEARNING CORPORATION
	Ву:
Cc: Morgan Joseph & Co. Inc.	David L. Warnock Chief Executive Officer

[Letterhead of Company]

П	n	s	ei	t	d	at	te]
	•	·	v	•	•		

Continental Stock Transfer
& Trust Company
17 Battery Place
New York, New York 10004

New York, N	ew York 10004		
Attn: Stever	G. Nelson and Frank	A. Di Paolo	
Re:	Trust Account No. [] Termination Letter	
Gentlemen:			
("Company") advise you tl company wit	and Continental Stock hat the Company has b hin the time frame spe	of the Investment Management Trust Agreement between Camden Learning Co Transfer & Trust Company ("Trustee"), dated as of [], 2007 ("Trust Agreen unable to effect a Business Combination (as defined in the Trust Agreementified in the Amended and Restated Certificate of Incorporation of the Company to its initial public offering.	greement"), this is to the start of the star
The Compar Company ar immediate tr Designated I Paying Ager to oversee th	ny has appointed [nd the "Designated Pay ransfer (the "Transfer D Paying Agent that the f nt may commence distr ne Designated Paying /	It o serve as its Designated Paying Agent; accordingly ing Agent" in writing as to when all of the funds in the Trust Account will be availate"). The Designated Paying Agent shall thereafter notify you as to the account unds in the Trust Account should be transferred to on the Transfer Date so that ibution of such funds in accordance with the Company's instructions. You shall agent's distribution of the funds. Upon the payment to the Designated Paying Agement shall terminate in accordance with the terms thereof.	r, you will notify the ilable for It or accounts of the the Designated have no obligation
		Very truly yours,	
		CAMDEN LEARNING CORPORATION	
		Ву:	
Cc: Morgan	Joseph & Co. Inc.	David L. Warnock Chief Executive Officer	_

AUTHORIZED INDIVIDUAL(S) FOR TELEPHONE CALL BACK

AUTHORIZED TELEPHONE NUMBER(S)

Company:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202

Attn: David L. Warnock, President (410) 878-6820

Morgan Joseph

600 Fifth Avenue, 19th floor New York, New York 10020

Attn: Gordon Pollock, Managing Director (212) 218-3700

Trustee:

Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004

Attn: Frank A. Di Paolo (212) 845-3270

SCHEDULE A

Schedule of fees pursuant to Section 3(c) of Investment Management Trust Agreement between Camden Learning Corporation and Continental Stock Transfer & Trust Company

Fee Item	Time and method of payment	Amount
Initial acceptance fee	Initial closing of IPO by wire transfer	\$1,000.00
Annual fee	First year, initial closing of IPO by wire transfer; thereafter on the anniversary of the effective date of the IPO by wire transfer or check	\$3,000.00
Transaction processing fee for disbursements to Company under Sections 2(a) and 2(b)	Deduction by Trustee from disbursement made to Company under Section 2(b)	\$250.00

		Agreed:
		Camden Learning Corporation
Dated: [], 2007	By: /s/
		David L. Warnock Chief Executive Officer
		Continental Stock Transfer & Trust Co.
		Ву:
		Authorized Officer

SECURITIES ESCROW AGREEMENT

SECURITIES ESCROW AGREEMENT, dated as of November [], 2007 (the "Agreement") by and among Camden Learning Corporation, a Delaware corporation (the "Company"), the undersigned parties listed as Initial Stockholders on the signature page hereto (collectively, the "Initial Stockholders") and Continental Stock Transfer & Trust Company, a New York corporation (the "Escrow Agent").

WHEREAS, the Company has entered into an Underwriting Agreement, dated November [], 2007 ("Underwriting Agreement") with Morgan Joseph & Co. Inc. ("Morgan Joseph") acting as representative of the several underwriters (collectively, the "Underwriters"), pursuant to which, among other matters, the Underwriters have agreed to purchase 6,250,000 units (not including the underwriters' over-allotment option) ("Units") of the Company. Each Unit consists of one share of the Company's common stock, par value \$.0001 per share (the "Common Stock"), and one warrant ("Warrant"), each Warrant to purchase one share of Common Stock, all as more fully described in the Company's definitive Prospectus, dated November [], 2007 ("Prospectus") comprising part of the Company's Registration Statement on Form S-1 (File No. 333-143098) under the Securities Act of 1933, as amended (the "Registration Statement"), declared effective on November [], 2007 (the "Effective Date").

WHEREAS, the Initial Stockholders have agreed, as a condition of the Underwriters' obligation to purchase the Units pursuant to the Underwriting Agreement and to offer them to the public, to deposit all of their shares of Common Stock, as set forth opposite their respective names in Exhibit A attached hereto (collectively the "Escrow Shares"), in escrow as hereinafter provided;

WHEREAS, the Company has entered into an Amended and Restated Subscription Agreement with one of the Initial Stockholders (the "Initial Warrantholder"), dated November 20, 2007 (the "Subscription Agreement"), pursuant to which the Initial Warrantholder has agreed to purchase 2,800,000 warrants (the "Private Warrants") in a private placement transaction;

WHEREAS, the Initial Warrantholder has agreed as a condition of the sale of the Private Warrants to deposit the Private Warrants (together with the Escrow Shares, the "Escrow Securities"), with the Escrow Agent as hereinafter provided; and

WHEREAS, the Company and the Initial Stockholders desire that the Escrow Agent accept the Escrow Securities, in escrow, to be held and disbursed as hereinafter provided.

IT IS AGREED:

- 1. <u>Appointment of Escrow Agent</u>. The Company and the Initial Stockholders hereby appoint the Escrow Agent to act in accordance with and subject to the terms of this Agreement and the Escrow Agent hereby accepts such appointment and agrees to act in accordance with and subject to such terms.
- 2. <u>Deposit of Escrow Securities</u>. On or before the Effective Date, each of the Initial Stockholders shall deliver to the Escrow Agent certificates representing his, her or its respective Escrow Shares, to be held and disbursed subject to the terms and conditions of this Agreement. Each Initial Stockholder acknowledges and agrees that the certificates representing his or her Escrow Securities will bear a legend to reflect the deposit of such Escrow Securities under this Agreement.

3. Disbursement of the Escrow Securities.

- 3.1 The Escrow Agent shall hold the Escrow Shares and the Escrow Warrants until the termination of their respective Escrow Period (as defined below). In the case of the Escrow Shares, the "Escrow Period" shall be the period beginning on the date the certificates representing the Shares are deposited with the Escrow Agent and ending on the date that is twelve (12) months following the consummation of the initial Business Combination (as such term is defined in the Registration Statement). In the case of the Escrow Warrants, the "Escrow Period" shall be the period beginning on the date the certificates representing the Warrants are deposited with the Escrow Agent and ending on the 90th day after the date of the consummation of the initial Business Combination. On the termination date of the applicable Escrow Period, the Escrow Agent shall, upon written instructions from the Company or Company's counsel, disburse each of the Initial Stockholder's Escrow Securities to such Initial Stockholder; provided, however, that if the Escrow Agent is notified by the Company pursuant to Section 6.7 hereof that the Company is being liquidated at any time during the Escrow Period, then the Escrow Agent shall promptly destroy the certificates representing the Escrow Securities; provided further, that if, after the Company consummates a Business Combination (as such term is defined in the Registration Statement), it (or the surviving entity) subsequently consummates a liquidation, merger, stock exchange or other similar transaction which results in all of its stockholders of such entity having the right to exchange their shares of Common Stock for cash, securities or other property, then the Escrow Agent will, upon consummation of such transaction, release the Escrow Securities to the Initial Stockholders so that they can similarly participate. The Escrow Agent shall have no further duties hereunder after the disbursement or destruction of the Escrow Securities in accordance with this Section 3.
- 3.2 Upon written instructions from the Company advising that a Business Combination has been consummated and that public stockholders holding 20% or more of the shares of Common Stock issued pursuant to the Registration Statement exercise the right to redeem their shares for cash as described in the Registration Statement, the Escrow Agent will release and deliver to the Company for cancellation on a pro rata basis certificates representing that number of Escrow Shares (not to exceed 195,312 in the aggregate) which results in the Initial Stockholders collectively owning no more than 23.81% of the Company's outstanding Common Stock immediately prior to the consummation of the Business Combination after giving effect to the redemption. Such instructions shall set forth both the number of shares the Company is redeeming and the number of Escrow Shares to be delivered to the Company for cancellation.
 - 4. Rights of Initial Stockholders in Escrow Securities.
- 4.1 <u>Voting Rights as a Stockholder</u>. Subject to the terms of the Insider Letters described in Section 4.4 hereof and except as herein provided, the Initial Stockholders shall retain all of their rights as stockholders of the Company during the Escrow Period, including, without limitation, the right to vote such shares.
- 4.2 <u>Dividends and Other Distributions in Respect of the Escrow Securities</u>. During the Escrow Period, all dividends payable in cash with respect to the Escrow Securities shall be paid to the Initial Stockholders, but all dividends payable in stock or other non-cash property ("Non-Cash Dividends") shall be delivered to the Escrow Agent to hold in accordance with the terms hereof. As used herein, the term "Escrow Securities" shall be deemed to include the Non-Cash Dividends distributed thereon, if any.
- 4.3 Restrictions on Transfer. During the Escrow Period, no sale, transfer or other disposition may be made of any or all of the Escrow Securities except (i) by gift to a member of Initial Stockholder's immediate family or to a trust or other entity, the beneficiary of which is an Initial Stockholder or a member of an Initial Stockholder's immediate family, (ii) by virtue of the laws of descent and distribution upon death of any Initial Stockholder, (iii) pursuant to a qualified domestic relations order, (iv) to an entity that is an Initial Stockholder, (v) to any person or entity controlled by, or under common control with, an Initial Stockholder or (vi) with respect to an Initial Stockholder who is an individual, to an entity controlled by such Initial Stockholder; provided, however, that such permitted transfers may be implemented only upon the respective transferee's written agreement to be bound by the terms and conditions of this Agreement and of the Insider Letter signed by the Initial Stockholder transferring the Escrow Securities. During the Escrow Period, the Initial Stockholders shall not pledge or grant a security interest in the Escrow Securities or grant a security interest in their rights under this Agreement.

4.4 <u>Insider Letters</u>. Each of the Initial Stockholders has executed a letter agreement with Morgan Joseph and the Company, dated as of the Effective Date, and which is filed as an exhibit to the Registration Statement ("Insider Letter"), respecting the rights and obligations of such Initial Stockholder in certain events, including, but not limited to, the liquidation of the Company.

5. Concerning the Escrow Agent.

- 5.1 Good Faith Reliance. The Escrow Agent shall not be liable for any action taken or omitted by it in good faith and in the exercise of its own best judgment, and may rely conclusively and shall be protected in acting upon any order, notice, demand, certificate, opinion or advice of counsel (including counsel chosen by the Escrow Agent), statement, instrument, report or other paper or document (not only as to its due execution and the validity and effectiveness of its provisions, but also as to the truth and acceptability of any information therein contained) which is believed by the Escrow Agent to be genuine and to be signed or presented by the proper person or persons. The Escrow Agent shall not be bound by any notice or demand, or any waiver, modification, termination or rescission of this Agreement unless evidenced by a writing delivered to the Escrow Agent signed by the proper party or parties and, if the duties or rights of the Escrow Agent are affected, unless it shall have given its prior written consent thereto.
- 5.2 <u>Indemnification</u>. The Escrow Agent shall be indemnified and held harmless by the Company from and against any expenses, including counsel fees and disbursements, or loss suffered by the Escrow Agent in connection with any action, suit or other proceeding involving any claim which in any way, directly or indirectly, arises out of or relates to this Agreement, the services of the Escrow Agent hereunder, or the Escrow Securities held by it hereunder, other than expenses or losses arising from the gross negligence or willful misconduct of the Escrow Agent. Promptly after the receipt by the Escrow Agent of notice of any demand or claim or the commencement of any action, suit or proceeding, the Escrow Agent shall notify the other parties hereto in writing. In the event of the receipt of such notice, the Escrow Agent, in its sole discretion, may commence an action in the nature of interpleader in an appropriate court to determine ownership or disposition of the Escrow Securities or it may deposit the Escrow Securities with the clerk of any appropriate court or it may retain the Escrow Securities pending receipt of a final, non appealable order of a court having jurisdiction over all of the parties hereto directing to whom and under what circumstances the Escrow Securities are to be disbursed and delivered. The provisions of this Section 5.2 shall survive in the event the Escrow Agent resigns or is discharged pursuant to Sections 5.5 or 5.6 below.
- 5.3 <u>Compensation</u>. The Escrow Agent shall be entitled to reasonable compensation from the Company for all services rendered by it hereunder, as set forth on Exhibit B hereto. The Escrow Agent shall also be entitled to reimbursement from the Company for all expenses paid or incurred by it in the administration of its duties hereunder including, but not limited to, all counsel, advisors' and agents' fees and disbursements and all taxes or other governmental charges.

- 5.4 <u>Further Assurances</u>. From time to time on and after the date hereof, the Company and the Initial Stockholders shall deliver or cause to be delivered to the Escrow Agent such further documents and instruments and shall do or cause to be done such further acts as the Escrow Agent shall reasonably request to carry out more effectively the provisions and purposes of this Agreement, to evidence compliance herewith or to assure itself that it is protected in acting hereunder.
- 5.5 <u>Resignation</u>. The Escrow Agent may resign at any time and be discharged from its duties as escrow agent hereunder by its giving the other parties hereto written notice and such resignation shall become effective as hereinafter provided. Such resignation shall become effective at such time that the Escrow Agent shall turn over to a successor escrow agent appointed by the Company and approved by Morgan Joseph, the Escrow Securities held hereunder. If no new escrow agent is so appointed within the 60 day period following the giving of such notice of resignation, the Escrow Agent may deposit the Escrow Securities with any court it deems appropriate.
- 5.6 <u>Discharge of Escrow Agent</u>. The Escrow Agent shall resign and be discharged from its duties as escrow agent hereunder if so requested in writing at any time by the other parties hereto, jointly, provided, however, that such resignation shall become effective only upon acceptance of appointment by a successor escrow agent as provided in Section 5.5.
- 5.7 <u>Liability</u>. Notwithstanding anything herein to the contrary, the Escrow Agent shall not be relieved from liability hereunder for its own gross negligence or its own willful misconduct.

6. Miscellaneous.

- 6.1 Governing Law. This Agreement shall for all purposes be deemed to be made under and shall be construed in accordance with the laws of the State of New York. Each of the parties hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. Each of the parties hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum.
- 6.2 <u>Third Party Beneficiaries</u>. Each of the Initial Stockholders hereby acknowledges that the Underwriters, including, without limitation, Morgan Joseph, are third party beneficiaries of this Agreement and this Agreement may not be modified or changed without the prior written consent of Morgan Joseph.
- 6.3 <u>Entire Agreement</u>. This Agreement contains the entire agreement of the parties hereto with respect to the subject matter hereof and, except as expressly provided herein, may not be changed or modified except by an instrument in writing signed by the party to the charged.
- 6.4 <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation thereof.
- 6.5 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the respective parties hereto and their legal representatives, successors and assigns.

6.6 Notices. Any notice or other communication required or which may be given hereunder shall be in writing and either be delivered personally or by private national courier service, or be mailed, certified or registered mail, return receipt requested, postage prepaid, and shall be deemed given when so delivered personally or, if sent by private national courier service, on the next business day after delivery to the courier, or, if mailed, two business days after the date of mailing, as follows:

If to the Company, to:

Camden Learning Corporation 500 East Pratt Street, Suite 1200 Baltimore, Maryland 21202 Attn: David L. Warnock, President

Fax No.: (410) 878-6850

If to a Stockholder, to his address set forth in Exhibit A.

and if to the Escrow Agent, to:

Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004 Attn: Steven Nelson and Frank DiPialo Fax No.: (212) 509-5150

A copy of any notice sent hereunder shall be sent to:

McDermott Will Emery LLP 340 Madison Avenue New York, New York 10173 Attn: Joel L. Rubinstein, Esq.

and:

Morgan Joseph & Co. Inc. 600 Fifth Avenue, 19th floor New York, New York 10020

Attn: Gordon Pollock, Managing Director

and:

Ellenoff, Grossman & Schole LLP 370 Lexington Avenue New York, New York 10017 Attn: Douglas S. Ellenoff, Esq.

The parties may change the persons and addresses to which the notices or other communications are to be sent by giving written notice to any such change in the manner provided herein for giving notice.

- 6.7 <u>Liquidation of Company</u>. The Company shall give the Escrow Agent written notification of the liquidation and dissolution of the Company in the event that the Company fails to consummate a Business Combination within the time period(s) specified in the Prospectus.
- 6.8 <u>Waiver</u>. Notwithstanding anything herein to the contrary, the Escrow Agent hereby waives any and all right, title, interest or claim of any kind ("Claim") in or to any distribution of the Trust Account, and hereby agrees not to seek recourse, reimbursement, payment or satisfaction for any Claim against the Trust Account for any reason whatsoever.
- 6.9 <u>Counterparts</u>. This Agreement may be executed in several counterparts each one of which shall constitute an original and may be delivered by facsimile transmission and together shall constitute one instrument.

[remainder of page intentionally left blank]

WITNESS the execution of this Agreement as of the date first above written.

CAMDEN LEARNING CORPORATION

David L. Warnock, Chief Executive	e Officer and President	
CONTINENTAL STOCK TRANSFER	& TRUST COMPANY	
Ву:		
Name:		
Title:		
	INITIAL STOCKHOLDERS:	
	CAMDEN LEARNING, LLC	
	By: Camden Partners Strategic III, LLC Its: Manager	
	By: Camden Partners Strategic Manager, LLC Its: Managing Member	
	Ву:	
	Donald W. Hughes Managing Member	
	Address:500 East Pratt Street Baltimore, MD 21202	
	Therese Kreig Crane, Ed.D	
	Ronald Tomalis	
	William Jews	

EXHIBIT A

Name and Address of Initial Stockholder	Number of Shares	Number of Warrants	Stock Certificate Number
Camden Learning, LLC	1,458,334	2,800,000	
Therese Kreig Crane, Ed.D	34,722	0	
Ronald Tomalis	34,722	0	
William Jews	34,722	0	
	8		

EXHIBIT B

Escrow Agent Fees

\$1,800 annually for acting agent escrow fee.

Initial acceptance fee and first year agent fee to be paid at closing.

EXHIBIT 10.4

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "Agreement") is entered into as of the [] day of [], 2007, by and among Camden Learning Corporation, a Delaware corporation (the "Company"), and the undersigned parties listed under Investors on the signature page hereto (each, an "Investor" and collectively, the "Investors").

WHEREAS, the Investors currently hold all of the issued and outstanding securities of the Company;

WHEREAS, the Investors currently hold an aggregate of 2,800,000 warrants ("Warrants"), exercisable into an aggregate of 2,800,000 shares of the Company's Common Stock (as defined below) ("Warrant Shares"), each of the Warrants and the Warrant Shares shall be referred to herein as the "Warrant Securities";

WHEREAS, the Investors and the Company desire to enter into this Agreement to provide the Investors with certain rights relating to the registration of shares of Common Stock and Warrant Securities held by them;

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. <u>DEFINITIONS</u>. The following capitalized terms used herein have the following meanings:

"Agreement" means this Agreement, as amended, restated, supplemented, or otherwise modified from time to time.

"Business Combination" means the acquisition of direct or indirect ownership through a merger, capital stock exchange, asset or stock acquisition or other similar type of transaction, of an operating business or businesses having collectively, a fair market value of at least 80% of the Company's net assets at the time of such transaction; provided, however, that, any acquisition of multiple operating businesses shall occur contemporaneously with one another.

"Commission" means the Securities and Exchange Commission, or any other federal agency then administering the Securities Act or the Exchange Act.

"Common Stock" means the common stock, par value \$0.0001 per share, of the Company.

"Company" is defined in the preamble to this Agreement.

"Demand Registration" is defined in Section 2.1.1.

"Demanding Holder" is defined in Section 2.1.1.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"Form S-3" is defined in Section 2.2.4.

"Indemnified Party" is defined in Section 4.3.

"Indemnifying Party" is defined in Section 4.3.

"Investor" is defined in the preamble to this Agreement.

"Investor Indemnified Party" is defined in Section 4.1.

"Maximum Number of Shares" is defined in Section 2.1.4.

"Notices" is defined in Section 6.3.

"Piggy-Back Registration" is defined in Section 2.2.1.

"<u>Purchase Option</u>" means the option to purchase 625,000 units, each unit consisting of one share of Common Stock and one Common Stock purchase warrant, issued to Morgan Joseph or its registered assignees in connection with the Company's initial public offering or securities (as may be transferred from time to time in accordance with its terms).

"<u>Purchase Option Securities</u>" means the securities underlying the Purchase Option that have been granted registration rights by the Company pursuant to the Purchase Option.

"Register," "registered" and "registration" mean a registration with respect to the Registrable Securities effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registrable Securities" mean the 1,562,500 shares of Common Stock, 2,800,000 Warrants and 2,800,000 Warrant Shares owned or held by Investors prior to the effective date of the Company's initial public offering of securities, in each case that are eligible for registration under the Securities Act and the terms of the Securities Escrow Agreement. Registrable Securities include any warrants, shares of capital stock or other securities of the Company issued as a dividend or other distribution with respect to or in exchange for or in replacement of such shares of Common Stock. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when: (a) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (b) such securities shall have been otherwise transferred, new certificates for them not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of them shall not require registration under the Securities Act; (c) such securities shall have ceased to be outstanding, or (d) the Securities and Exchange Commission makes a definitive determination to the Company that the Registrable Securities are saleable under Rule 144(k).

"Registration Statement" means a registration statement filed by the Company with the Commission in compliance with the Securities Act and the rules and regulations promulgated thereunder for a public offering and sale of Common Stock (other than a registration statement on Form S-4 or Form S-8, or their successors, or any registration statement covering only securities proposed to be issued in exchange for securities or assets of another entity).

"Release Date I" means the date on which shares of Common Stock are disbursed from escrow pursuant to Section 3 of the Securities Escrow Agreement.

"Release Date II" means the date on which the Company publicly announces that it has entered into a letter of intent with respect to a proposed Business Combination.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"Securities Escrow Agreement" means that certain Securities Escrow Agreement dated as of [], 2007 by and among the parties hereto and Continental Stock Transfer & Trust Company.

"<u>Underwriter</u>" means a securities dealer who purchases any Registrable Securities as principal in an underwritten offering and not as part of such dealer's market-making activities.

- 2. REGISTRATION RIGHTS.
- 2.1 Demand Registration.

2.1.1. Request for Registration. At any time and from time to time on or after each of Release Date I as it relates to the 1,562,500 shares of Common Stock and Release Date II as it relates to the Warrant Securities, as applicable, the holders of a majority-in-interest of the 1,562,500 shares of Common Stock or Warrant Securities, as the case may be, held by the Investors or the transferees of the Investors, may make a written demand for registration under the Securities Act of all or part of their Registrable Securities (a "Demand Registration"). Any demand for a Demand Registration shall specify the number and type of Registrable Securities proposed to be sold and the intended method(s) of distribution thereof. The Company will notify all holders of Registrable Securities of the demand within ten (10) days from the receipt of the Demand Registration, and each holder of Registrable Securities who wishes to include all or a portion of such holder's Registrable Securities in the Demand Registration (each such holder including shares of Registrable Securities in such registration, a "Demanding Holder") shall so notify the Company within fifteen (15) days after the receipt by the holder of the notice from the Company. Upon any such request, the Demanding Holders shall be entitled to have their Registrable Securities included in the Demand Registration, subject to Section 2.1.4 and the provisos set forth in Section 3.1.1. The Company shall not be obligated to effect more than an aggregate of two (2) Demand Registrations with respect to the 1,562,500 shares of Common Stock and two (2) Demand Registrations with respect to the Warrant Securities under this Section 2.1.1 in respect of Registrable Securities. In no event shall a registration statement that has been filed with respect to the Warrant Securities be declared effective until the Company has completed its initial Business Combination.

- 2.1.2. Effective Registration. A registration will not count as a Demand Registration until the Registration Statement filed with the Commission with respect to such Demand Registration has been declared effective and the Company has complied with all of its obligations under this Agreement or otherwise with respect thereto; provided, however, if, after such Registration Statement has been declared effective, the offering of Registrable Securities pursuant to a Demand Registration is interfered with by any stop order or injunction of the Commission or any other governmental agency or court, the Registration Statement with respect to such Demand Registration will be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders thereafter elect to continue the offering; provided, further, the Company shall not be obligated to file a second Registration Statement until a Registration Statement that has been filed is counted as a Demand Registration or is terminated.
- 2.1.3. <u>Underwritten Offering</u>. If a majority-in-interest of the Demanding Holders so elect and such holders so advise the Company as part of their written demand for a Demand Registration, the offering of such Registrable Securities pursuant to such Demand Registration shall be in the form of an underwritten offering. In such event, the right of any holder to include its Registrable Securities in such registration shall be conditioned upon such holder's participation in such underwriting and the inclusion of such holder's Registrable Securities in the underwriting to the extent provided herein. All Demanding Holders proposing to distribute their securities through such underwriting shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such underwriting by a majority-in-interest of the holders initiating the Demand Registration.
- 2.1.4. Reduction of Offering. If the managing Underwriter or Underwriters for a Demand Registration that is to be an underwritten offering advises the Company and the Demanding Holders in writing that the dollar amount or number of shares of Registrable Securities which the Demanding Holders desire to sell, taken together with all other shares of Common Stock or other securities which the Company desires to sell and the shares of Common Stock, if any, as to which registration has been requested pursuant to written contractual piggy-back registration rights held by other shareholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of shares that can be sold in such offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of shares, as applicable, the "Maximum Number of Shares"), then the Company shall include in such registration: (i) first, the Registrable Securities as to which Demand Registration has been requested by the Demanding Holders (pro rata in accordance with the number of shares of Registrable Securities which such Demanding Holders have requested be included in such registration, regardless of the number of shares of Registrable Securities held by each Demanding Holder) that can be sold without exceeding the Maximum Number of Shares; (ii) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (i), the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (iii) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (i) and (ii), the shares of Common Stock for the account of other persons that the Company is obligated to register pursuant to written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Shares; and (iv) fourth, to the extent that the Maximum Number of Shares have not been reached under the foregoing clauses (i), (ii), and (iii), the shares of Common Stock that other shareholders desire to sell that can be sold without exceeding the Maximum Number of Shares.

2.1.5. <u>Withdrawal</u>. If a majority-in-interest of the Demanding Holders disapprove of the terms of any underwriting or are not entitled to include all of their Registrable Securities in any offering, such majority-in-interest of the Demanding Holders may elect to withdraw from such offering by giving written notice to the Company and the Underwriter or Underwriters of their request to withdraw prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Demand Registration. In such event, the Company need not seek effectiveness of such Registration Statement for the benefit of other Investors. If the majority-in-interest of the Demanding Holders withdraws from a proposed offering relating to a Demand Registration, then such registration shall not count as a Demand Registration provided for in Section 2.1.1.

2.2 Piggy-Back Registration.

2.2.1. Piggy-Back Rights. If at any time on or after Release Date I as it relates to the 1,562,500 shares of Common Stock and Release Date II as it relates to the Warrant Securities, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, by the Company for its own account or for shareholders of the Company for their account (or by the Company and by shareholders of the Company including, without limitation, pursuant to Section 2.1), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing shareholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall (x) give written notice of such proposed filing to the holders of Registrable Securities as soon as practicable but in no event less than ten (10) days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, of the offering, and (y) offer to the holders of Registrable Securities in such notice the opportunity to register the sale of such number of shares of Registrable Securities as such holders may request in writing within five (5) days following receipt of such notice (a "Piggy-Back Registration"). The Company shall cause such Registrable Securities to be included in such registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed underwritten offering to permit the Registrable Securities requested to be included in a Piggy-Back Registration to be included on the same terms and conditions as any similar securities of the Company and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All holders of Registrable Securities proposing to distribute their securities through a Piggy-Back Registration that involves an Underwriter or Underwriters shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such Piggy-Back Registration.

2.2.2. Reduction of Offering. If the managing Underwriter or Underwriters for a Piggy-Back Registration that is to be an underwritten offering advises the Company and the holders of Registrable Securities in writing that the dollar amount or number of shares of Common Stock which the Company desires to sell, taken together with shares of Common Stock, if any, as to which registration has been demanded pursuant to written contractual arrangements with persons other than the holders of Registrable Securities hereunder, the Registrable Securities as to which registration has been requested under this Section 2.2, and the shares of Common Stock, if any, as to which registration has been requested pursuant to the written contractual piggy-back registration rights of other shareholders of the Company, exceeds the Maximum Number of Shares, then the Company shall include in any such registration:

(i) If the registration is undertaken for the Company's account: (A) first, the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the shares of Common Stock or other securities, if any, including the Registrable Securities, and Purchase Option Securities as to which registration has been requested pursuant to the applicable written contractual piggy-back registration rights of such security holders (pro rata in accordance with the number of shares of Common Stock which each such person has actually requested to be included in such registration, regardless of the number of shares of Common Stock with respect to which such persons have the right to request such inclusion) that can be sold without exceeding the Maximum Number of Shares; and (C) third, to the extent that the Maximum Number of shares has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual piggy-back registration rights with such persons (pro rata in accordance with the number of shares of Common Stock which each such person has actually requested to be included in such registration, regardless of the number of shares of Common Stock with respect to which such persons have the right to request such inclusion) that can be sold without exceeding the Maximum Number of Shares; and

(ii) If the registration is a "demand" registration undertaken at the demand of persons other than the holders of Registrable Securities or pursuant to contractual arrangements with such persons, (A) first, the shares of Common Stock for the account of the demanding persons that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (C) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A) and (B), the Registrable Securities and Purchase Option Securities as to which registration has been requested under this Section 2.2 (pro rata in accordance with the number of shares of Registrable Securities or Purchase Options Securities held by each such holder); and (D) fourth, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A), (B) and (C), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to register, if any, as to which registration has been requested pursuant to written contractual arrangements with such persons that can be sold without exceeding the Maximum Number of Shares.

- 2.2.3. Withdrawal. Any holder of Registrable Securities may elect to withdraw such holder's request for inclusion of Registrable Securities in any Piggy-Back Registration by giving written notice to the Company of such request to withdraw prior to the effectiveness of the Registration Statement. The Company (whether on its own determination or as the result of a withdrawal by persons making a demand pursuant to written contractual obligations) may also elect to withdraw a registration statement at any time prior to the effectiveness of the Registration Statement. Notwithstanding any such withdrawal, the Company shall pay all expenses incurred by the holders of Registrable Securities in connection with such Piggy-Back Registration as provided in Section 3.3.
- 2.2.4. Registrations on Form S-3. The holders of Registrable Securities may at any time and from time to time, request in writing that the Company register the resale of any or all of such Registrable Securities on Form S-3 or any similar short-form registration which may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an underwritten offering. Upon receipt of such written request, the Company will promptly give written notice of the proposed registration to all other holders of Registrable Securities, and, as soon as practicable thereafter, effect the registration of all or such portion of such holder's or holders' Registrable Securities as are specified in such request, together with all or such portion of the Registrable Securities of any other holder or holders joining in such request as are specified in a written request given within fifteen (15) days after receipt of such written notice from the Company; provided, however, that the Company shall not be obligated to effect any such registration pursuant to this Section 2.2: (i) if Form S-3 is not available for such offering; or (ii) if the holders of the Registrable Securities, together with the holders of any other securities of the Company entitled to inclusion in such registration, propose to sell Registrable Securities and such other securities (if any) at any aggregate price to the public of less than \$500,000. Registrations effected pursuant to this Section 2.2 shall not be counted as Demand Registrations effected pursuant to Section 2.1.
- 2.3 No Net Cash Settlement Value. In connection with the exercise of the Warrants, the Company will not be obligated to deliver securities, and there are no contractual penalties for failure to deliver securities, if a registration statement is not effective at the time of exercise; however, the Company may satisfy its obligation by delivering unregistered shares of Common Stock. In no event will the Company be required to net cash settle an exercise of a Warrant.

3. REGISTRATION PROCEDURES.

- 3.1 <u>Filings: Information</u>. Whenever the Company is required to effect the registration of any Registrable Securities pursuant to Section 2, the Company shall use its best efforts to effect the registration and sale of such Registrable Securities in accordance with the intended method(s) of distribution thereof as expeditiously as practicable, and in connection with any such request:
- 3.1.1. Filing Registration Statement. The Company shall, as expeditiously as possible and in any event within sixty (60) days after receipt of a request for a Demand Registration pursuant to Section 2.1, prepare and file with the Commission a Registration Statement on any form for which the Company then qualifies or which counsel for the Company shall deem appropriate and which form shall be available for the sale of all Registrable Securities to be registered thereunder in accordance with the intended method(s) of distribution thereof, and shall use its best efforts to cause such Registration Statement to become and remain effective for the period required by Section 3.1.3; provided, however, that the Company shall have the right to defer any Demand Registration for up to thirty (30) days, and any Piggy-Back Registration for such period as may be applicable to deferment of any demand registration to which such Piggy-Back Registration relates, in each case if the Company shall furnish to the holders a certificate signed by the Chief Executive Officer of the Company stating that, in the good faith judgment of the Board of Directors of the Company, it would be materially detrimental to the Company and its shareholders for such Registration Statement to be effected at such time; provided further, however, that the Company shall not have the right to exercise the right set forth in the immediately preceding proviso more than once in any 365-day period in respect of a Demand Registration hereunder.

- 3.1.2. <u>Copies</u>. The Company shall, prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish without charge to the holders of Registrable Securities included in such registration, and such holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the prospectus included in such Registration Statement (including each preliminary prospectus), and such other documents as the holders of Registrable Securities included in such registration or legal counsel for any such holders may request in order to facilitate the disposition of the Registrable Securities owned by such holders.
- 3.1.3. Amendments and Supplements. The Company shall prepare and file with the Commission such amendments, including post-effective amendments, and supplements to such Registration Statement and the prospectus used in connection therewith as may be necessary to keep such Registration Statement effective and in compliance with the provisions of the Securities Act until all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement (which period shall not exceed the sum of one hundred eighty (180) days plus any period during which any such disposition is interfered with by any stop order or injunction of the Commission or any governmental agency or court) or such securities have been withdrawn.
- 3.1.4. Notification. After the filing of a Registration Statement, the Company shall promptly, and in no event more than two (2) business days after such filing, notify the holders of Registrable Securities included in such Registration Statement of such filing, and shall further notify such holders promptly and confirm such advice in writing in all events within two (2) business days of the occurrence of any of the following: (i) when such Registration Statement becomes effective; (ii) when any post-effective amendment to such Registration Statement becomes effective; (iii) the issuance or threatened issuance by the Commission of any stop order (and the Company shall take all actions required to prevent the entry of such stop order or to remove it if entered); and (iv) any request by the Commission for any amendment or supplement to such Registration Statement or any prospectus relating thereto or for additional information or of the occurrence of an event requiring the preparation of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of the securities covered by such Registration Statement, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and promptly make available to the holders of Registrable Securities included in such Registration Statement any such supplement or amendment; except that before filing with the Commission a Registration Statement or prospectus or any amendment or supplement thereto, including documents incorporated by reference, the Company shall furnish to the holders of Registrable Securities included in such Registration Statement and to the legal counsel for any such holders, copies of all such documents proposed to be filed sufficiently in advance of filing to provide such holders and legal counsel with a reasonable opportunity to review such documents and comment thereon, and the Company shall not file any Registration Statement or prospectus or amendment or supplement thereto, including documents incorporated by reference, to which such holders or their legal counsel shall object.

- 3.1.5. <u>State Securities Laws Compliance</u>. The Company shall use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other Governmental Authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this Section 3.1.5 or subject itself to taxation in any such jurisdiction.
- 3.1.6. Agreements for Disposition. The Company shall enter into customary agreements (including, if applicable, an underwriting agreement in customary form) and take such other actions as are reasonably required in order to expedite or facilitate the disposition of such Registrable Securities. The representations, warranties and covenants of the Company in any underwriting agreement which are made to or for the benefit of any Underwriters, to the extent applicable, shall also be made to and for the benefit of the holders of Registrable Securities included in such registration statement. No holder of Registrable Securities included in such registration statement shall be required to make any representations or warranties in the underwriting agreement except, if applicable, with respect to such holder's organization, good standing, authority, title to Registrable Securities, lack of conflict of such sale with such holder's material agreements and organizational documents, and with respect to written information relating to such holder that such holder has furnished in writing expressly for inclusion in such Registration Statement. Holders of Registrable Securities shall agree to such covenants and indemnification and contribution obligations for selling stockholders as are customarily contained in agreements of that type. Further, such holders shall cooperate fully in the preparation of the registration statement and other documents relating to any offering in which they include securities pursuant to Section 2 hereof. Each holder shall also furnish to the Company such information regarding itself, the Registrable Securities held by such holder and the intended method of disposition of such securities as shall be reasonably required to effect the registration of the Registrable Securities.

- 3.1.7. <u>Cooperation</u>. The principal executive officer of the Company, the principal financial officer of the Company, the principal accounting officer of the Company and all other officers and members of the management of the Company shall cooperate fully in any offering of Registrable Securities hereunder, which cooperation shall include, without limitation, the preparation of the Registration Statement with respect to such offering and all other offering materials and related documents, and participation in meetings with Underwriters, attorneys, accountants and potential investors.
- 3.1.8. Records. The Company shall make available for inspection by the holders of Registrable Securities included in such Registration Statement, any Underwriter participating in any disposition pursuant to such registration statement and any attorney, accountant or other professional retained by any holder of Registrable Securities included in such Registration Statement or any Underwriter, all financial and other records, pertinent corporate documents and properties of the Company, as shall be necessary to enable them to exercise their due diligence responsibility, and cause the Company's officers, directors and employees to supply all information requested by any of them in connection with such Registration Statement.
- 3.1.9. Opinions and Comfort Letters. The Company shall furnish to each holder of Registrable Securities included in any Registration Statement a signed counterpart, addressed to such holder, of (i) any opinion of counsel to the Company delivered to any Underwriter and (ii) any comfort letter from the Company's independent public accountants delivered to any Underwriter. In the event no legal opinion is delivered to any Underwriter, the Company shall furnish to each holder of Registrable Securities included in such Registration Statement, at any time that such holder elects to use a prospectus, an opinion of counsel to the Company to the effect that the Registration Statement containing such prospectus has been declared effective and that no stop order is in effect.
- 3.1.10. <u>Earnings Statement</u>. The Company shall comply with all applicable rules and regulations of the Commission and the Securities Act, and make available to its shareholders, as soon as practicable, an earnings statement covering a period of twelve (12) months, beginning within three (3) months after the effective date of the registration statement, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder.
- 3.1.11. <u>Listing</u>. The Company shall use its best efforts to cause all Registrable Securities included in any registration to be listed on such exchanges or otherwise designated for trading in the same manner as similar securities issued by the Company are then listed or designated or, if no such similar securities are then listed or designated, in a manner satisfactory to the holders of a majority of the Registrable Securities included in such registration.
- 3.2 Obligation to Suspend Distribution. Upon receipt of any notice from the Company of the happening of any event of the kind described in Section 3.1.4(iv), or, in the case of a resale registration on Form S-3 pursuant to Section 2.3 hereof, upon any suspension by the Company, pursuant to a written insider trading compliance program adopted by the Company's Board of Directors, of the ability of all "insiders" covered by such program to transact in the Company's securities because of the existence of material non-public information, each holder of Registrable Securities included in any registration shall immediately discontinue disposition of such Registrable Securities pursuant to the Registration Statement covering such Registrable Securities until such holder receives the supplemented or amended prospectus contemplated by Section 3.1.4(iv) or the restriction on the ability of "insiders" to transact in the Company's securities is removed, as applicable, and, if so directed by the Company, each such holder will deliver to the Company all copies, other than permanent file copies then in such holder's possession, of the most recent prospectus covering such Registrable Securities at the time of receipt of such notice.

- 3.3 Registration Expenses. The Company shall bear all costs and expenses incurred in connection with any Demand Registration pursuant to Section 2.1, any Piggy-Back Registration pursuant to Section 2.2, and any registration on Form S-3 effected pursuant to Section 2.3, and all expenses incurred in performing or complying with its other obligations under this Agreement, whether or not the Registration Statement becomes effective or whether any or all Holders of Registrable Securities withdraw from any Registration Statement, including, without limitation: (i) all registration and filing fees; (ii) fees and expenses of compliance with securities or "blue sky" laws (including fees and disbursements of counsel in connection with blue sky qualifications of the Registrable Securities); (iii) printing expenses; (iv) the Company's internal expenses (including, without limitation, all salaries and expenses of its officers and employees); (v) the fees and expenses incurred in connection with the listing of the Registrable Securities as required by Section 3.1.11; (vi) National Association of Securities Dealers, Inc. fees; (vii) fees and disbursements of counsel for the Company and fees and expenses for independent certified public accountants retained by the Company (including the expenses or costs associated with the delivery of any opinions or comfort letters requested pursuant to Section 3.1.9); (viii) the fees and expenses of any special experts retained by the Company in connection with such registration and (ix) the fees and expenses of one legal counsel selected by the holders of a majority-in-interest of the Registrable Securities included in such registration. The Company shall have no obligation to pay any underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions shall be borne by such holders. Additionally, in an underwritten offering, all selling shareholders and the Company shall bear the expenses of the underwriter pro rata in proportion to the respective amount of shares each is selling in such offering.
- 3.4 <u>Information</u>. The holders of Registrable Securities shall provide such information as may reasonably be requested by the Company, or the managing Underwriter, if any, in connection with the preparation of any Registration Statement, including amendments and supplements thereto, in order to effect the registration of any Registrable Securities under the Securities Act pursuant to Section 2 and in connection with the Company's obligation to comply with federal and applicable state securities laws.
- 3.5 <u>Holder Obligations</u>. No holder of Registrable Securities may participate in any underwritten offering pursuant to this Section 3 unless such holder (i) agrees to sell only such holder's Registrable Securities on the basis reasonably provided in any underwriting agreement, and (ii) completes, executes and delivers any and all questionnaires, powers of attorney, custody agreements, indemnities, underwriting agreements and other documents reasonably required by or under the terms of any underwriting agreement or as reasonably requested by the Company.

4. INDEMNIFICATION AND CONTRIBUTION.

4.1 Indemnification by the Company. The Company agrees to indemnify and hold harmless each Investor and each other holder of Registrable Securities, and each of their respective officers, employees, affiliates, directors, partners, members, attorneys and agents, and each person, if any, who controls an Investor and each other holder of Registrable Securities (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) (each, an "Investor Indemnified Party"), from and against any expenses, losses, judgments, claims, damages or liabilities, whether joint or several, arising out of or based upon any untrue statement (or allegedly untrue statement) of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to such Registration Statement, or arising out of or based upon any omission (or alleged omission) to state a material fact required to be stated therein or necessary to make the statements therein not misleading, or any violation by the Company of the Securities Act or any rule or regulation promulgated thereunder applicable to the Company and relating to action or inaction required of the Company in connection with any such registration; and the Company shall promptly reimburse the Investor Indemnified Party for any legal and any other expenses reasonably incurred by such Investor Indemnified Party in connection with investigating and defending any such expense, loss, judgment, claim, damage, liability or action; provided, however, that (a) the Company will not be liable in any such case to the extent that any such expense, loss, claim, damage or liability arises out of or is based upon (i) any untrue statement or allegedly untrue statement or omission or alleged omission made in such Registration Statement, preliminary prospectus, final prospectus, or summary prospectus, or any such amendment or supplement, in reliance upon and in conformity with information furnished to the Company, in writing, by such selling holder expressly for use therein or (ii) for the use by any selling holder of a prospectus in violation of any stop order or other suspension of the Registration Statement of which the Company made the selling holder aware; and (b) the foregoing indemnity shall not inure to the benefit of any Investor Indemnified Party if a copy of the Prospectus (as then amended or supplemented if the Company shall have furnished any amendments or supplements thereto) was not sent or given by or on behalf of the applicable selling holder to the person asserting such expense, loss, claim, damage or liability who purchased the Registrable Securities from such selling holder, if required by law so to have been delivered at or prior to the written confirmation of the sale of the Registrable Securities to such person, and if the Prospectus (as so amended or supplemented) would have cured the defect giving rise to such expense, loss, claim, damage or liability, unless such failure is the result of noncompliance by the Company with Section 3.1.3 hereof. The Company also shall indemnify any Underwriter of the Registrable Securities or Purchase Option Securities, their officers, employees, affiliates, directors, partners, members and agents and each person who controls such Underwriter on substantially the same basis as that of the indemnification provided above in this Section 4.1.

- 4.2 Indemnification by Holders of Registrable Securities. Each selling holder of Registrable Securities will, in the event that any registration is being effected under the Securities Act pursuant to this Agreement of any Registrable Securities held by such selling holder, indemnify and hold harmless the Company, each of its directors and officers and each Underwriter (if any), and each other person, if any, who controls another selling holder or such Underwriter or the Company within the meaning of the Securities Act or Section 20 of the Exchange Act, against any losses, claims, judgments, damages or liabilities, whether joint or several, insofar as such losses, claims, judgments, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or allegedly untrue statement of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to the Registration Statement, or arise out of or are based upon any omission or the alleged omission to state a material fact required to be stated therein or necessary to make the statement therein not misleading, if the statement or omission was made in reliance upon and in conformity with information furnished in writing to the Company by such selling holder expressly for use therein or for the use by any Investor Indemnified Party of a prospectus in violation of any stop order or other suspension of the Registration Statement, and shall reimburse the Company, its directors and officers, and each other selling holder or such controlling person for any legal or other expenses reasonably incurred by any of them in connection with investigation or defending any such loss, claim, damage, liability or action. Each selling holder's indemnification obligations hereunder shall be several and not joint and shall be limited to the amount of any net proceeds actually received by such selling holder in connection with the sale of the Registrable Securities by such selling holder pursuant to the Registration Statement containing such untrue statement.
- 4.3 Conduct of Indemnification Proceedings. Promptly after receipt by any person of any notice of any loss, claim, damage or liability or any action in respect of which indemnity may be sought pursuant to Section 4.1 or 4.2, such person (the "Indemnified Party") shall, if a claim in respect thereof is to be made against any other person for indemnification hereunder, notify such other person (the "Indemnifying Party") in writing of the loss, claim, judgment, damage, liability or action; provided, however, that the failure by the Indemnified Party to notify the Indemnifying Party shall not relieve the Indemnifying Party from any liability which the Indemnifying Party may have to such Indemnified Party hereunder, except and solely to the extent the Indemnifying Party is actually prejudiced by such failure. If the Indemnified Party is seeking indemnification with respect to any claim or action brought against the Indemnified Party, then the Indemnifying Party shall be entitled to participate in such claim or action, and, to the extent that it wishes, jointly with all other Indemnifying Parties, to assume control of the defense thereof with counsel satisfactory to the Indemnified Party. After notice from the Indemnifying Party to the Indemnified Party of its election to assume control of the defense of such claim or action, the Indemnifying Party shall not be liable to the Indemnified Party for any legal or other expenses subsequently incurred by the Indemnified Party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that in any action in which both the Indemnified Party and the Indemnifying Party are named as defendants, the Indemnified Party shall have the right to employ separate counsel (but no more than one such separate counsel) to represent the Indemnified Party and its controlling persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought by the Indemnified Party against the Indemnifying Party, with the fees and expenses of such counsel to be paid by such Indemnifying Party if, based upon the written opinion of counsel of such Indemnified Party, representation of both parties by the same counsel would be inappropriate due to actual or potential differing interests between them. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, consent to entry of judgment or effect any settlement of any claim or pending or threatened proceeding in respect of which the Indemnified Party is or could have been a party and indemnity could have been sought hereunder by such Indemnified Party, unless such judgment or settlement includes an unconditional release of such Indemnified Party from all liability arising out of such claim or proceeding.

4.4 Contribution.

- 4.4.1. If the indemnification provided for in the foregoing Sections 4.1, 4.2 and 4.3 is unavailable to any Indemnified Party in respect of any loss, claim, damage, liability or action referred to herein, then each such Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such loss, claim, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the Indemnified Parties and the Indemnifying Parties in connection with the actions or omissions which resulted in such loss, claim, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of any Indemnified Party and any Indemnifying Party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by such Indemnified Party or such Indemnifying Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.
- 4.4.2. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 4.4 were determined by <u>pro rata</u> allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the immediately preceding Section 4.4.1. The amount paid or payable by an Indemnified Party as a result of any loss, claim, damage, liability or action referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses incurred by such Indemnified Party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this Section 4.4, no holder of Registrable Securities shall be required to contribute any amount in excess of the dollar amount of the net proceeds (after payment of any underwriting fees, discounts, commissions or taxes) actually received by such holder from the sale of Registrable Securities which gave rise to such contribution obligation. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

5. UNDERWRITING AND DISTRIBUTION.

5.1 Rule 144. The Company covenants that it shall file any reports required to be filed by it under the Securities Act and the Exchange Act and shall take such further action as the holders of Registrable Securities may reasonably request, all to the extent required from time to time to enable such holders to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such Rules may be amended from time to time, or any similar Rule or regulation hereafter adopted by the Commission.

6. MISCELLANEOUS.

- 6.1 Other Registration Rights. The Company represents and warrants that except for the securities issued or issuable upon exercise of the Purchase Option to be issued to Morgan Joseph or its assigns, no person, other than a holder of the Registrable Securities has any right to require the Company to register any shares of the Company's capital stock for sale or to include shares of the Company's capital stock in any registration filed by the Company for the sale of shares of capital stock for its own account or for the account of any other person.
- 6.2 <u>Assignment; No Third Party Beneficiaries</u>. This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part. This Agreement and the rights, duties and obligations of the holders of Registrable Securities hereunder may be freely assigned or delegated by such holder of Registrable Securities in conjunction with and to the extent of any transfer of Registrable Securities by any such holder in accordance with applicable law. This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties, to Morgan Joseph and their respective successors and the permitted assigns of the Investor or holder of Registrable Securities or of any assignee of the Investor or holder of Registrable Securities. This Agreement is not intended to confer any rights or benefits on any persons that are not party hereto other than as expressly set forth in Article 4 and this Section 6.2.
- 6.3 Notices. All notices, demands, requests, consents, approvals or other communications (collectively, "Notices") required or permitted to be given hereunder or which are given with respect to this Agreement shall be in writing and shall be personally served, delivered by reputable air courier service with charges prepaid, or transmitted by hand delivery, telegram, telex or facsimile, addressed as set forth below, or to such other address as such party shall have specified most recently by written notice. Notice shall be deemed given on the date of service or transmission if personally served or transmitted by telegram, telex or facsimile; provided, that if such service or transmission is not on a business day or is after normal business hours, then such notice shall be deemed given on the next business day. Notice otherwise sent as provided herein shall be deemed given on the next business day following timely delivery of such notice to a reputable air courier service with an order for next-day delivery.

To the Company: Camden Learning Corporation

500 East Pratt Street

Suite 1200

Baltimore, MD 21202 Attn: David L. Warnock

; or

with a copy to: Ellenoff Grossman & Schole LLP

370 Lexington Avenue New York, New York 10017

Attn: Douglas S. Ellenoff, Esq.

; or

To an Investor, to: [Name of Investor]

c/o Camden Learning Corporation

500 East Pratt Street

Suite 1200

Baltimore, MD 21202 Attn: David L. Warnock

- 6.4 <u>Severability</u>. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.
- 6.5 <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original, and all of which taken together shall constitute one and the same instrument. Facsimile signatures shall be deemed to be original signatures for all purposes of this Agreement.
- 6.6 Entire Agreement. This Agreement (including all agreements entered into pursuant hereto and all certificates and instruments delivered pursuant hereto and thereto) constitutes the entire agreement of the parties with respect to the subject matter hereof and supersede all prior and contemporaneous agreements, representations, understandings, negotiations and discussions between the parties, whether oral or written.
- 6.7 <u>Modifications and Amendments</u>. No amendment, modification or termination of this Agreement shall be binding upon any party unless executed in writing by such party.
- 6.8 <u>Titles and Headings</u>. Titles and headings of sections of this Agreement are for convenience only and shall not affect the construction of any provision of this Agreement.
- 6.9 <u>Waivers and Extensions</u>. Any party to this Agreement may waive any right, breach or default which such party has the right to waive, provided that such waiver will not be effective against the waiving party unless it is in writing, is signed by such party, and specifically refers to this Agreement. Waivers may be made in advance or after the right waived has arisen or the breach or default waived has occurred. Any waiver may be conditional. No waiver of any breach of any agreement or provision herein contained shall be deemed a waiver of any preceding or succeeding breach thereof nor of any other agreement or provision herein contained. No waiver or extension of time for performance of any obligations or acts shall be deemed a waiver or extension of the time for performance of any other obligations or acts.

- 6.10 Remedies Cumulative. In the event that the Company fails to observe or perform any covenant or agreement to be observed or performed under this Agreement, the Investor or any other holder of Registrable Securities may proceed to protect and enforce its rights by suit in equity or action at law, whether for specific performance of any term contained in this Agreement or for an injunction against the breach of any such term or in aid of the exercise of any power granted in this Agreement or to enforce any other legal or equitable right, or to take any one or more of such actions, without being required to post a bond. None of the rights, powers or remedies conferred under this Agreement shall be mutually exclusive, and each such right, power or remedy shall be cumulative and in addition to any other right, power or remedy, whether conferred by this Agreement or now or hereafter available at law, in equity, by statute or otherwise.
- 6.11 <u>Governing Law</u>. This Agreement shall be governed by, interpreted under, and construed in accordance with the internal laws of the State of New York applicable to agreements made and to be performed within the State of New York, without giving effect to any choice-of-law provisions thereof that would compel the application of the substantive laws of any other jurisdiction. Each of the parties hereby agrees that any action, proceeding or claim against it arising out of or relating in any way to this Agreement shall be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York (each, a "New York Court"), and irrevocably submits to such jurisdiction, which jurisdiction shall be exclusive. Each of the parties hereby waives any objection to such exclusive jurisdiction and that such courts represent an inconvenient forum.
- 6.12 <u>Waiver of Trial by Jury</u>. Each party hereby irrevocably and unconditionally waives the right to a trial by jury in any action, suit, counterclaim or other proceeding (whether based on contract, tort or otherwise) arising out of, connected with or relating to this Agreement, the transactions contemplated hereby, or the actions of the Investor in the negotiation, administration, performance or enforcement hereof.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties have caused this Registration Rights Agreement to be executed and delivered by their duly authorized representatives as of the date first written above.
CAMDEN LEARNING CORPORATION

By:

Name: David L. Warnock

Title: Chief Executive Officer and President

IN WITNESS WHEREOF, the parties have caused this Registration Rights Agreement to be executed and delivered by their duly authorized representatives as of the date first written above.

INVESTORS: CAMDEN PARTNERS HOLDINGS, LLC
Ву:
Donald W. Hughes Manager Address:500 East Pratt Street Baltimore, MD 21202
David L. Warnock
Therese Kreig Crane, Ed.D
Ronald Tomalis
William Jews
19

SCHEDULE OF BUYERS

Investors Address

and Facsimile Number Investor Camden Learning, LLC 500 East Pratt Street Suite 1200 Baltimore, MD 21202 Number of Shares: 1,458,334 Number of Warrants: 2,800,000 Attn: Donald W. Hughes Facsimile Number: (410) 878-6850 Therese Kreig Crane, Ed.D c/o Camden Learning Corporation 500 East Pratt Street Suite 1200 Number of Shares: 34,722 Baltimore, MD 21202 Number of Warrants: 0 Attn: Donald W. Hughes Facsimile Number: (410) 878-6850 **Ronald Tomalis** c/o Camden Learning Corporation 500 East Pratt Street Number of Shares: 34,722 Suite 1200 Baltimore, MD 21202 Number of Warrants: 0 Attn: Donald W. Hughes Facsimile Number: (410) 878-6850 William Jews c/o Camden Learning Corporation 500 East Pratt Street Suite 1200 Number of Shares: 34,722 Baltimore, MD 21202 Number of Warrants: 0 Attn: Donald W. Hughes Facsimile Number: (410) 878-6850

November 20, 2007

Camden Partners Holdings, LLC 500 East Pratt Street Suite 1200 Baltimore, MD 21202

Re: CAMDEN LEARNING CORPORATION

Gentlemen:

This letter will confirm our agreement, that commencing on the effective date ("Effective Date") of the registration statement ("Registration Statement") of the initial public offering ("IPO") of the securities of Camden Learning Corporation (the "Company") and continuing until the earlier to occur of: (i) the consummation of a Business Combination (as described in the Registration Statement), (ii) failure to effect a Business Combination within 24 months from the consummation of the IPO and (iii) the date on which we determine to dissolve and liquidate our trust account as part of our plan of dissolution and liquidation, Camden Partners Holdings, LLC ("Holdings LLC") shall make available to the Company certain general and administrative services, including but not limited to receptionist, secretarial and general office services. In exchange therefore, the Company shall pay Holdings LLC the sum of \$7,500 per month on the Effective Date and continuing monthly thereafter.

Very truly yours,

CAMDEN LEARNING CORPORATION

By: /s/ David L. Warnock

Name: David L. Warnock Title: Chief Executive Officer

Agreed to and Accepted by:

CAMDEN PARTNERS HOLDINGS, LLC

By: /s/ Donald W. Hughes

Donald W. Hughes Managing Member

AMENDED AND RESTATED SUBSCRIPTION AGREEMENT

AMENDED AND RESTATED SUBSCRIPTION AGREEMENT (this "Agreement") made as of this 20th day of November, 2007 for the benefit of Camden Learning Corporation, a Delaware corporation (the "Company"), having its principal place of business at 500 East Pratt Street, Suite 1200, Baltimore, MD 21202 by Camden Learning, LLC ("Subscriber").

WHEREAS, the Company and the Subscriber entered into a subscription agreement dated as of May 16, 2007 (the "Original Agreement"), and wish to amend and restate the Original Agreement as set forth herein;

WHEREAS, the Company desires to sell on a private placement basis (the "Offering") an aggregate of 2,800,000 warrants (the "Warrants") of the Company for a purchase price of \$1.00 per Warrant. Each Warrant is exercisable to purchase one share of Common Stock at an exercise price of \$5.50 per share during the period commencing on the later of: (i) one year from the date of the prospectus relating to the Company's IPO (as defined below) and (ii) the completion of a Business Combination (as defined in Section 5 below) and expiring on the fourth anniversary of the date of the prospectus relating to the Company's IPO (as defined below);

WHEREAS, Subscriber wishes to purchase the Warrants and the Company wishes to accept such subscription.

NOW, THEREFORE, in consideration of the premises and the mutual covenants hereinafter set forth and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Company and Subscriber hereby agree as follows

1. Agreement to Subscribe

- 1.1. <u>Purchase and Issuance of the Warrants</u>. Upon the terms and subject to the conditions of this Agreement, Subscriber hereby agrees to purchase from the Company, and the Company hereby agrees to sell to the Subscriber, on the Closing Date, the Warrants for an aggregate purchase price of \$2,800,000 (the "Purchase Price").
- 1.2. <u>Delivery of the Purchase Price</u>. Upon execution of this Agreement, the undersigned is hereby bound to fulfill its obligations hereunder and hereby irrevocably commits to deliver into a trust account at a financial institution to be chosen by the Company, maintained by Continental Stock Transfer & Trust Company, acting as Trustee, on the date of Closing (as hereinafter defined), the Purchase Price in immediately available funds by certified bank check, wire transfer or such other form of payment as shall be acceptable to the Trustee, in its sole and absolute discretion, at the Closing.
- 1.3. <u>Closing</u>. The closing (the "Closing") of the Offering, shall take place at the offices of the Company, prior to the effective date of the registration statement pursuant to which the Company proposes to register its initial public offering (the "IPO") of 6,250,000 units of Common Stock and Warrants (the "Closing Date").

2. Representations and Warranties of the Subscriber

Subscriber represents and warrants to the Company that:

- 2.1. <u>No Government Recommendation or Approval</u>. Subscriber understands that no United States federal or state agency has passed upon or made any recommendation or endorsement of the Company or the Offering of the Warrants or the Common Stock underlying the Warrants (the "Warrant Shares" and, collectively with the Warrants, the "Securities").
- 2.2. Regulation D Offering. Subscriber represents that it is an "accredited investor" as such term is defined in Rule 501(a) of Regulation D under the Securities Act of 1933, as amended (the "Securities Act") and acknowledges the sale contemplated hereby is being made in reliance on a private placement exemption to "Accredited Investors" within the meaning of Section 501(a) of Regulation D under the Securities Act or similar exemptions under state law; and, accordingly, such securities will be "restricted securities" within the meaning of Rule 144(a)(3) under the Securities Act, and therefore may not be offered, pledged or sold by it, directly or indirectly, in the United States without registration under United States federal and state securities laws and Subscriber understands the certificates representing the such securities will contain a legend in respect of such restrictions.
- 2.3. <u>Intent</u>. Subscriber is purchasing the Warrants solely for investment purposes, for the Subscriber's own account and not for the account or benefit of any U.S. Person, and not with a view towards the distribution thereof and Subscriber has no present arrangement to sell the Securities to or through any person or entity. Subscriber shall not engage in hedging transactions with regard to the Warrants and the underlying securities unless in compliance with the Securities Act.
- 2.4. Restrictions on Transfer. Subscriber understands the Warrants are being offered in a transaction not involving a public offering in the United States within the meaning of the Securities Act. The Securities have not been registered under the Securities Act, and, if in the future the Subscriber decides to offer, resell, pledge or otherwise transfer the Securities, such Securities may be offered, resold, pledged or otherwise transferred only (A) pursuant to an effective registration statement filed under the Securities Act, (B) pursuant to an exemption from registration under Rule 144 promulgated under the Securities Act, if available, or (C) pursuant to any other exemption from the registration requirements of the Securities Act, and in each case in accordance with any applicable securities laws of any state or any other jurisdiction. Subscriber agrees that if any transfer of its Securities or any interest therein is proposed to be made, as a condition precedent to any such transfer, Subscriber may be required to deliver to the Company an opinion of counsel satisfactory to the Company. Absent registration or another exemption from registration, the Subscriber agrees that it will not resell the Securities. Subscriber further understands and acknowledges the Securities and Exchange Commission (the "SEC") has taken the position the Subscriber would be considered a promoter under the Securities Act and that promoters or affiliates of a blank check company and their transferees, both before and after a business combination, would act as "underwriters" under the Securities Act when reselling the securities of that blank check company. Accordingly, Rule 144 promulgated under the Securities Act will not be available to the Subscriber for the resale of the Securities despite technical compliance with the requirements of Rule 144, in which event the resale transactions would need to be made through a registered offering.

2.5. Sophisticated Investor.

- (i) Subscriber is sophisticated in financial matters and is able to evaluate the risks and benefits of the investment in the Securities.
- (ii) Subscriber is aware that an investment in the Warrants is highly speculative and subject to substantial risks because, among other things, none of the Securities have been registered under the Securities Act and therefore cannot be sold unless subsequently registered under the Securities Act or an exemption from such registration is available. Subscriber is able to bear the economic risk of its investment in the Securities for an indefinite period of time. Notwithstanding the foregoing, Subscriber further understands and acknowledges the SEC has taken the position that the Subscriber is considered a promoter under the Securities Act and that promoters or affiliates of a blank check company and their transferees, both before and after a Business Combination, would act as an "underwriter" under the Securities Act when reselling the securities of that blank check company. Accordingly, Rule 144 promulgated under the Securities Act would not be available for the resale of the Securities despite technical compliance with the requirements of Rule 144, in which event the resale transactions would need to be made through a registered offering.
- 2.6. <u>Independent Investigation</u>. Subscriber, in making the decision to purchase the Warrants, has relied upon an independent investigation of the Company and has not relied upon any information or representations made by any third parties or upon any oral or written representations or assurances from the Company, its officers, directors or employees or any other representatives or agents of the Company, other than as set forth in this Agreement. Subscriber is familiar with the business, operations and financial condition of the Company and has had an opportunity to ask questions of, and receive answers from, the Company's officers and directors concerning the Company and the terms and conditions of the offering of the Warrants and has had full access to such other information concerning the Company as the Subscriber has requested. Subscriber confirms that all documents that it has requested have been made available and that the Subscriber has been supplied with all of the additional information concerning this investment which Subscriber has requested.
- 2.7. <u>Authority</u>. This Agreement has been validly authorized, executed and delivered by Subscriber and is a valid and binding agreement enforceable in accordance with its terms, subject to the general principles of equity and to bankruptcy or other laws affecting the enforcement of creditors' rights generally. The execution, delivery and performance of this Agreement by Subscriber does not and will not conflict with, violate or cause a breach of any agreement, contract or instrument to which Subscriber is a party.
- 2.8. No Legal Advice from Company. Subscriber acknowledges it has had the opportunity to review this Agreement and the transactions contemplated by this Agreement and the other agreements entered into between the parties hereto with the Subscriber's own legal counsel and investment and tax advisors. Except for any statements or representations of the Company made in this Agreement and the other agreements entered into between the parties hereto, Subscriber is relying solely on such counsel and advisors and not on any statements or representations of the Company or any of its representatives or agents for legal, tax or investment advice with respect to this investment, the transactions contemplated by this Agreement or the securities laws of any jurisdiction.

- 2.9. Reliance on Representations and Warranties. Subscriber understands the Warrants are being offered and sold to Subscriber in reliance on exemptions from the registration requirements under the Securities Act, and analogous provisions in the laws and regulations of various states, and that the Company is relying upon the truth and accuracy of the representations, warranties, agreements, acknowledgments and understandings of the Subscriber set forth in this Agreement in order to determine the applicability of such provisions.
- 2.10. <u>No Advertisements</u>. The undersigned is not subscribing for the Warrants as a result of or subsequent to any advertisement, article, notice or other communication published in any newspaper, magazine, or similar media or broadcast over television or radio, or presented at any seminar or meeting.
- 2.11. <u>Legend</u>. Subscriber acknowledges and agrees the certificates evidencing the Warrants and the Warrant Shares shall bear a restrictive legend (the "Legend"), in form and substance as set forth in Section 4 hereof, prohibiting the offer, sale, pledge or transfer of the securities, except (i) pursuant to an effective registration statement covering these securities under the Securities Act or (ii) pursuant to any other exemptions from the registration requirements under the Securities Act and such laws which, in the opinion of counsel for this Company, is available.
 - 3. Representations and Warranties of the Company

The Company represents and warrants to Subscriber that:

- 3.1. <u>Valid Issuance of Capital Stock</u>. The total number of shares of all classes of capital stock which the Company has authority to issue is 20,000,000 shares of Common Stock and 1,000,000 shares of Preferred Stock. As of the date hereof, the Company has 1.562.500 shares of Common Stock and no shares of Preferred Stock issued and outstanding. All of the issued shares of capital stock of the Company have been duly authorized, validly issued, and are fully paid and non-assessable.
- 3.2. <u>Organization and Qualification</u>. The Company is a corporation duly incorporated and existing in good standing under the laws of the state of Delaware and has the requisite corporate power to own its properties and assets and to carry on its business as now being conducted.
- 3.3. Authorization; Enforcement. (i) The Company has the requisite corporate power and authority to enter into and perform its obligations under this Agreement and to issue the Warrants and the underlying securities in accordance with the terms hereof, (ii) the execution, delivery and performance of this Agreement by the Company and the consummation by it of the transactions contemplated hereby have been duly authorized by all necessary corporate action, and no further consent or authorization of the Company or its Board of Directors or stockholders is required, and (iii) this Agreement constitutes valid and binding obligations of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, moratorium, reorganization, or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by equitable principles of general application and except as enforcement of rights to indemnity and contribution may be limited by federal and state securities laws or principles of public policy.

3.4. No Conflicts. The execution, delivery and performance of this Agreement and the consummation by the Company of the transactions contemplated hereby do not (i) result in a violation of the Company's Certificate of Incorporation or Bylaws or (ii) conflict with, or constitute a default under any agreement, indenture or instrument to which the Company is a party. Other than any SEC or state securities filings which may be required to be made by the Company subsequent to the Closing, and any registration statement which may be filed pursuant thereto, the Company is not required under federal, state or local law, rule or regulation to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency or self-regulatory entity in order for it to perform any of its obligations under this Agreement or issue the Common Stock in accordance with the terms hereof.

4. Legends

4.1. <u>Legend</u>. The Company will issue the Warrants, and when issued, the Warrant Shares, purchased by the Subscriber in the name of the Subscriber. The Securities will bear the following Legend and appropriate "stop transfer" instructions:

"THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL FOR THIS CORPORATION, IS AVAILABLE."

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE TERMS AND CONDITIONS CONTAINED IN A SECURITIES ESCROW AGREEMENT (THE "AGREEMENT") AND MAY NOT BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED DURING THE TERM OF THE ESCROW PERIOD (AS DEFINED IN THE AGREEMENT)."

- 4.2. <u>Subscriber's Compliance</u>. Nothing in this Section 4 shall affect in any way the Subscribers' obligations and agreements to comply with all applicable securities laws upon resale of the Securities.
- 4.3. <u>Company's Refusal to Register Transfer of the Securities</u>. The Company shall refuse to register any transfer of the Securities, if in the sole judgment of the Company such purported transfer would not be made (i) pursuant to an effective registration statement filed under the Securities Act, or (ii) pursuant to an available exemption from the registration requirements of the Securities Act.
- 5. Escrow. Upon consummation of the IPO, the holders of the Warrants shall enter into a securities escrow agreement (the "Escrow Agreement") with Continental Stock Transfer & Trust Company, whereby the Warrants shall be held in escrow until the earlier of (i) the 90th day following consummation of a Business Combination (as defined therein) or (ii) liquidation of the Company.

6. Securities Laws Restrictions.

In addition to the restrictions contained in the Escrow Agreement, subscriber agrees not to sell, transfer, pledge, hypothecate or otherwise dispose of all or any part of the Securities unless, prior thereto (a) a registration statement on the appropriate form under the Securities Act and applicable state securities laws with respect to the Securities proposed to be transferred shall then be effective or (b) the Company shall have received an opinion from counsel reasonably satisfactory to the Company, that such registration is not required because such transaction complies with the Securities Act and the rules promulgated by the Securities and Exchange Commission thereunder and with all applicable state securities laws.

7. Waiver of Liquidation Distributions.

In connection with the Securities purchased pursuant to this Agreement, and with respect to any Common Stock purchased by Subscriber prior to the private placement, Subscriber hereby waives any and all right, title, interest or claim of any kind in or to any liquidating distributions by the Company in the event of a liquidation of the Company upon the Company's failure to timely complete a Business Combination. For purposes of clarity, in the event Subscriber purchases shares of Common Stock in the IPO or in the aftermarket, any additional shares so purchased shall be eligible to receive any liquidating distributions by the Company. In no event will a Subscriber have the right to exercise any Warrants prior to the later of: (i) one year from the date of the prospectus relating to the Company's IPO and (ii) the consummation of a Business Combination.

- 8. Forfeiture of Warrants.
- 8.1. <u>Failure to Consummate Business Combination</u>. The Warrants shall be forfeited to the Company in the event that the Company does not consummate a Business Combination within 24 months from the consummation of the IPO.

- 8.2. <u>Termination of Rights as holder; Escrow</u>. If the Warrants are forfeited in accordance with this Section 8, then after such time the Subscriber (or successor in interest), shall no longer have any rights as a holder of such Warrants, and the Company shall take such action as is appropriate to cancel such Warrants. To effectuate the foregoing, all certificates representing the Warrants shall be held in escrow as provided in Section 5 hereof. In addition, Subscriber hereby irrevocably grants the Company a limited power of attorney for the purpose of effectuating the foregoing.
 - 9. Rescission Right Waiver and Indemnification.
- 9.1. Subscriber understands and acknowledges an exemption from the registration requirements of the Securities Act requires there be no general solicitation of purchasers of the Warrants. In this regard, if the IPO were deemed to be a general solicitation with respect to the Warrants, the offer and sale of such Warrants may not be exempt from registration and, if not, the Subscriber may have a right to rescind its purchase of the Warrants. In order to facilitate the completion of the Offering and in order to protect the Company, its stockholders and the trust account from claims that may adversely affect the Company or the interests of its stockholders, Subscriber hereby agrees to waive, to the maximum extent permitted by applicable law, any claims, right to sue or rights in law or arbitration, as the case may be, to seek rescission of its purchase of the Warrants. Subscriber acknowledges and agrees this waiver is being made in order to induce the Company to sell the Warrants to the Subscriber. Subscriber agrees the foregoing waiver of rescission rights shall apply to any and all known or unknown actions, causes of action, suits, claims or proceedings (collectively, "Claims") and related losses, costs, penalties, fees, liabilities and damages, whether compensatory, consequential or exemplary, and expenses in connection therewith, including reasonable attorneys' and expert witness fees and disbursements and all other expenses reasonably incurred in investigating, preparing or defending against any Claims, whether pending or threatened, in connection with any present or future actual or asserted right to rescind the purchase of the Warrants hereunder or relating to the purchase of the Warrants and the transactions contemplated hereby.
- 9.2. Subscriber agrees not to seek recourse against the trust account for any reason whatsoever in connection with its purchase of the Warrants or any Claim that may arise now or in the future.
- 9.3. Subscriber acknowledges and agrees the stockholders of the Company and Morgan Joseph & Co. Inc. are and shall be third-party beneficiaries of the foregoing provisions of this Agreement.
- 9.4. Subscriber agrees that to the extent any waiver of rights under this Section 9 is ineffective as a matter of law, Subscriber has offered such waiver for the benefit of the Company as an equitable right that shall survive any statutory disqualification or bar that applies to a legal right. Subscriber acknowledges the receipt and sufficiency of consideration received from the Company hereunder in this regard.
 - 10. Terms of the Warrant

The Warrants are substantially identical to the warrants included in the units offered in the IPO, except: (i) they are not being registered in the Registration Statement; (ii) they are not transferable until the consummation of a Business Combination; (iii) they are not redeemable so long as they are held by the initial holder thereof (or any of their permitted transferees) and (iv) they may be exercised on a "cashless" basis if held by the initial holder thereof or its permitted assigns. The Warrant Shares will be granted certain registration rights. In no event will the Company be required to net cash settle the Warrant exercise.

11. Voting of Shares.

Subscriber agrees to vote the shares of Common Stock owned by him immediately before this private placement in accordance with the majority of the shares of Common Stock voted by the public stockholders. Subscriber further agrees to vote the Common Stock acquired in the IPO or the aftermarket in favor of a Business Combination that the Company negotiates and presents for approval to the Company's stockholders.

12. Governing Law; Jurisdiction; Waiver of Jury Trial

This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware for agreements made and to be wholly performed within such state. The parties hereto hereby waive any right to a jury trial in connection with any litigation pursuant to this Agreement and the transactions contemplated hereby.

- 13. Assignment; Entire Agreement; Amendment
- 13.1. <u>Assignment</u>. Neither this Agreement nor any rights hereunder may be assigned by any party to any other person other than by Subscriber to a person agreeing to be bound by the terms hereof.
- 13.2. <u>Entire Agreement.</u> This Subscription Agreement sets forth the entire agreement and understanding between the parties as to the subject matter thereof and merges and supersedes all prior discussions, agreements and understandings of any and every nature among them.
- 13.3. <u>Amendment</u>. Except as expressly provided in this Agreement, neither this Agreement nor any term hereof may be amended, waived, discharged or terminated other than by a written instrument signed by the party against whom enforcement of any such amendment, waiver, discharge or termination is sought.
- 13.4. <u>Binding upon Successors</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and to their respective heirs, legal representatives, successors and permitted assigns.
 - 14. Notices; Indemnity

- 14.1 Notices. Unless otherwise provided herein, any notice or other communication to a party hereunder shall be sufficiently given if in writing and personally delivered or sent by facsimile or other electronic transmission with copy sent in another manner herein provided or sent by courier (which for all purposes of this Agreement shall include Federal Express or other recognized overnight courier) or mailed to said party by certified mail, return receipt requested, at its address provided for herein or such other address as either may designate for itself in such notice to the other. Communications shall be deemed to have been received when delivered personally, on the scheduled arrival date when sent by next day or 2-day courier service, or if sent by facsimile upon receipt of confirmation of transmittal or, if sent by mail, then three days after deposit in the mail. If given by electronic transmission, such notice shall be deemed to be delivered (a) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice; (b) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (1) such posting and (2) the giving of such separate notice; and (c) if by any other form of electronic transmission, when directed to the stockholder.
- 14.2 <u>Indemnification</u>. Each party shall indemnify the other against any loss, cost or damages (including reasonable attorney's fees and expenses) incurred as a result of such party's breach of any representation, warranty, covenant or agreement in this Agreement.

15. Counterparts

This Agreement may be executed in one or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission or by e-mail delivery of a ".pdf" format data file, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or ".pdf" signature page were an original thereof.

- 16. Survival; Severability
- 16.1. Survival. The representations, warranties, covenants and agreements of the parties hereto shall survive the Closing.
- 16.2. <u>Severability</u>. In the event that any provision of this Agreement becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, this Agreement shall continue in full force and effect without said provision; provided that no such severability shall be effective if it materially changes the economic benefit of this Agreement to any party.

17. Headings.

The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement.

This subscription is accepted by the Company on the 20th day of November, 2007.

CAMDEN LEARNING CORPORATION

By: /s/ David L. Warnock

David L. Warnock President

CAMDEN LEARNING, LLC

By: Camden Partners Strategic III, LLC

Its: Manager

By: Camden Partners Strategic Manager, LLC

Its: Managing Member

By: /s/ Donald W. Hughes

Donald W. Hughes Managing Member EX-23.1 17 v095534_ex23-1.htm **EXHIBIT 23.1**

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the reference to our firm under the caption "Experts" and to the inclusion of our report dated November 21, 2007 on our audit of the financial statements of Camden Learning Corporation as of September 30, 2007 and for the period from April 10, 2007 (inception) through September 30, 2007, which report includes an explanatory paragraph regarding the Company's ability to continue as a going concern, in Amendment No. 6 to the Registration Statement on Form S-1 and the related Prospectus of Camden Learning Corporation.

/s/ EISNER LLP EISNER LLP

New York, New York November 26, 2007